Investigating hyperheuristics for solving bi-objective simulation optimisation problems



Thesis presented in fulfilment of the requirements for the degree of Master of Engineering (Industrial Engineering) in the Faculty of Engineering at Stellenbosch University

> Supervisor: Prof JF Bekker Co-supervisor: Dr GS Nel

> > March 2023

Declaration

By submitting this thesis electronically, I declare that the entirety of the work contained therein is my own, original work, that I am the sole author thereof (save to the extent explicitly otherwise stated), that reproduction and publication thereof by Stellenbosch University will not infringe any third party rights and that I have not previously in its entirety or in part submitted it for obtaining any qualification.

March 2023

Copyright © 2023 Stellenbosch University

All rights reserved

Abstract

The *investigation* and *exploration* of search and optimisation methodologies are crucial research areas. Take for example the potential impact of an effective and computationally efficient decision support methodology, it could be the difference between life and death in healthcare scheduling. Schedule too few doctors and patients could die; schedule long work-hours and doctors could make fatal mistakes due to fatigue. *Simulation optimisation* is typically used to approximately solve large and complex problems that cannot be solved by exact methods. In addition, the need for better simulation optimisation approaches are further motivated by the combinatorial relationship that results in significant search spaces. One of the biggest problems that researchers face with metaheuristic approaches is the lack of *general applicability* and the high number of hyperparameter combinations that algorithms have and a lack of insight on how to choose them. This is due to the fact that the performance of metaheuristics greatly depends on the type of problem being optimised, as supported by the *no free lunch* (NFL) theorem. Accordingly, each optimisation algorithm has its strengths and weaknesses when it comes to exploring the search space.

Hyperheuristics propose to compensate, to some extent, for the weaknesses of the individual low-level heuristics (LLHs) by method of *algorithmic cooperation*, creating ensemble algorithms that are more generally applicable, *i.e.* can solve a larger range of problems than the individual LLHs are capable of solving. In this study, two hyperheuristic approaches are developed, one for population-based search and the second for single-solution based search, and are assessed using five discete-event dynamic stochastic bi-objective simulation optimisation problems. The hyperheuristics as well as their individual LLHs are implemented and assessed in Tecnomatix Plant Simulation. In addition, an algorithmic parameter study is presented for the respective LLHs to determine good hyperparameter combinations and possibly infer insights from the complex interaction.

Furthermore, due to the dynamic and stochastic nature of simulation models, there exists a sufficient number of observations per solution that need to be evaluated to be able to construct suitable narrow confidence intervals. This renders simulation optimisation computationally expensive and for that reason a pilot study is conducted to determine the feasibility of an ANN as metamodel to screen out solutions that are predicted to be of low-quality, thereby reducing the number of computationally expensive evaluations that need to be made by the simulation model.

The statement that hyperheuristics perform better (or at least similar) to its individual LLHs does not hold true for the population-based hyperheuristic. The statement, however, holds true for the single-solution based hyperheuristic. It can be concluded that both hyperheuristics failed to exhibit superior performance and did not indicate favourable performance improvements relative to *all* the individual applications of the LLHs. Furthermore, the novel pilot study provided valuable insights pertaining to the complex interaction within an ANN, however, the study could not conclude whether or not an ANN metamodel is a feasible solution to enhance the simulation optimisation process. The study does provide valuable insights which could inspire further research.

Opsomming

Die ondersoek en verkenning van optimeringmetodes word beskou as 'n kritieke navorsings gebied. Simulasie-optimeringsmetodes word dikwels gebruik om benaderde oplossings te vind vir komplekse probleme wat onoplosbaar is deur presiese metodes. Die behoefte vir beter optimeringsmetodes word verder gemotiveer deur kombinatoriese verwantskappe wat lei tot beduidende groot besluitnemingsruimtes. Een van die grootste tekortkominge in navorsing in hierdie gebied is die gebrek aan metaheuristieke wat oor die algemeen toepaslik is asook die groot aantal algoritme parameters wat kundigheid verg om van te kies. Dit is as gevolg van die metaheuristieke se sterk en swakpunte, in terme van die besluitnemingsruimte verken, wat lei tot die metaheuristiek se vermoë om sekere probleme goed te kan benader en ander nie.

Hiperheuristieke is voorgestel met die doel om vir die swakpunte van die enkele metaheuristieke of heuristieke komponente te kompenseer deur middel van algoritme samewerking om gevolglik 'n groter verskeidenheid van optimeringsprobleme te kan benader as voorheen. Twee hiperheuristieke was voorgestel en ontwikkel in hierdie studie. Die een volg 'n enkel-oplossing benadering, terwyl die ander 'n populasie-oplossing benadering volg. Die uitvoer van die voorgestelde hiperheuristieke was in Tecnomatix Plant Simulation verrig en geevalueer op vyf tweedoelige stogastiese en dinamies simulasie optimeringsprobleme. Daarby, was 'n omvattende parameterevaluarings studie uitgevoer met die doel om afleidings te kan maak oor die komplekse interaksies wat plaasvind tussen die parameters.

As gevolg van die stogastiese en dinamies aard van die simulasie-optimeringsprobleme moet daar 'n sekere aantal observasies per oplossing geevalueer word om sodoende nou vertrouensintervalle te bewerkstellig. Dit is as gevolg van die statistiese steekproefneming dat die produksielopies tydrowend is. Gevolglik is 'n ondersoek ingestel om te bepaal of 'n kunsmatige neurale netwerk as 'n metamodel die simulasie optimeringsoekproses kan ondersteun.

Die opvatting dat hiperheuristieke se prestasie beter of ten minste dieselfde is as sy individuele algoritmes, geld nie vir die populasie-oplossing hiperheuristiek nie. Die verklaring geld wel vir die enkel-oplossing hiperheuristiek wat ten minste so goed gedoen het of beter as al sy individuele heuristieke. Om af te sluit, die beslissing wat gevind was in die studie is dat die hiperheuristieke nie beter presteer het in vergelyking met al hulle individuele algoritmes nie. Die uiteinde van die ondersoek kon nie oortuigend tot 'n beslissing kom nie en daarom moet verdere ondersoek ingestel word. Waardevolle insigte is wel verkry uit die studie en kan gebruik work om verdere navorsing te inspireer.

Acknowledgements

The author wishes to acknowledge the following people and institutions for their various contributions towards the completion of this work: I am grateful to the following people who supported me toward completing this research project:

- Derrick van Schalkwyk, my husband, for his continuous love and support.
- Professor James Bekker, my supervisor and role model, for his guidance and wisdom throughout this study. Providing a strategic level guidance, always helping me see the bigger picture and helping me identify the intermediate steps that I needed to take to get where I wanted to go.
- Doctor Gerrit Stephanus Nel, my co-supervisor, for helping me dream big but also helping me scope my project, otherwise I would have never finished.
- My parents, for teaching me to always finish what I started.
- St Patrick Nice, for lending me three of his computers, which allowed me to run approximately 19 500 simulation runs and thereby complete my masters on time.

Table of Contents

A	bstra	act			iii	
U	Uittreksel					
A	cknov	owledgements			vii	
Li	ist of	f Acronyms			xv	
\mathbf{Li}	ist of	f Figures			xvii	
\mathbf{Li}	ist of	f Tables			xxv	
Li	ist of	f Algorithms		x	xxiii	
1	Intr	roduction			1	
	1.1	Research background			1	
	1.2	Problem description and research assignment			3	
	1.3	Research scope			4	
	1.4	Research objectives			6	
	1.5	Research methodology			7	
	1.6	Thesis organisation			8	
2	\mathbf{Lite}	erature Study			11	
	2.1	Simulation optimisation			11	
	2.2	Multi-objective optimisation preliminaries			14	
		2.2.1 The notion of solution dominance			15	
		2.2.2 The quality of MOO algorithm			17	
		2.2.3 Archiving			18	
		2.2.4 Performance assessment of MOO algorithms			19	
	2.3	Metaheuristics			20	
	2.4	Hyperheuristics			22	
	2.5	Simulation metamodeling			25	
		2.5.1 The fundamentals of artificial neural networks			26	
		2.5.2 The network architecture			28	
		2.5.3 The network training algorithms			31	

		2.5.4	Activation functions	33
	2.6	Summ	ary	36
3	AN	N as N	Aetamodel Pilot Study	37
	3.1	Machi	ne learning models: Regression	37
		3.1.1	Linear regression	37
		3.1.2	Polynomial regression	38
		3.1.3	Support vector regression	39
		3.1.4	Decision tree regression	39
		3.1.5	Random forest regression	40
	3.2	Perfor	mance measures for assessing regression models	41
	3.3	Hyper	parameter optimisation	42
		3.3.1	ANN hyperparameters	43
		3.3.2	An introduction to generalisation techniques	49
	3.4	Traini	ng, validating and testing the regression models	50
	3.5	Conclu	usion: Chapter 3	53
4	Sim	ulatior	n Models and Statistical Prerequisites	55
	4.1	Statist	ical analysis preliminaries	55
	1.0			
	4.2	The si	mulation model problems	57
	4.2	The si 4.2.1	mulation model problems	57 59
	4.2	The si 4.2.1 4.2.2	mulation model problemsThe Open mine problemThe (s, S) inventory problem	57 59 62
	4.2	The si 4.2.1 4.2.2 4.2.3	mulation model problemsThe Open mine problemThe (s, S) inventory problemThe buffer-allocation problem: five machines	57 59 62 63
	4.2	The si 4.2.1 4.2.2 4.2.3 4.2.4	mulation model problemsThe Open mine problemThe (s, S) inventory problemThe buffer-allocation problem: five machinesThe buffer-allocation problem: 10 machines	57 59 62 63 65
	4.2	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5	mulation model problemsThe Open mine problemThe (s, S) inventory problemThe buffer-allocation problem: five machinesThe buffer-allocation problem: 10 machinesThe non-linear buffer-allocation problem: 16 machines	57 59 62 63 65 67
	4.2	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu	mulation model problemsThe Open mine problemThe (s, S) inventory problemThe buffer-allocation problem: five machinesThe buffer-allocation problem: 10 machinesThe non-linear buffer-allocation problem: 16 machinesusion: Chapter 4	57 59 62 63 65 67 69
5	4.2 4.3 Met	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri	mulation model problemsThe Open mine problemThe (s, S) inventory problemThe buffer-allocation problem: five machinesThe buffer-allocation problem: 10 machinesThe non-linear buffer-allocation problem: 16 machinession: Chapter 4	 57 59 62 63 65 67 69 71
5	4.24.3Met5.1	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri The m	mulation model problemsThe Open mine problemThe (s, S) inventory problemThe buffer-allocation problem: five machinesThe buffer-allocation problem: 10 machinesThe non-linear buffer-allocation problem: 16 machinesIsion: Chapter 4istics and Hyperheuristicsmain concepts for metaheuristics	57 59 62 63 65 67 69 71 72
5	4.2 4.3 Met 5.1	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri The m 5.1.1	mulation model problems	57 59 62 63 65 67 69 71 72 72
5	4.2 4.3 Met 5.1	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri The m 5.1.1 5.1.2	mulation model problems	57 59 62 63 65 67 69 71 72 72 72
5	4.2 4.3 Met 5.1	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri 5.1.1 5.1.2 5.1.3	mulation model problems	57 59 62 63 65 67 69 71 72 72 72 72 74
5	 4.2 4.3 Met 5.1 	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri 5.1.1 5.1.2 5.1.3 Multi-	mulation model problems	57 59 62 63 65 67 69 71 72 72 72 72 74 74
5	 4.2 4.3 Met 5.1 5.2 	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri 5.1.1 5.1.2 5.1.3 Multi- 5.2.1	mulation model problems	57 59 62 63 65 67 69 71 72 72 72 72 72 74 74
5	 4.2 4.3 Met 5.1 5.2 	The si 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 Conclu taheuri 5.1.1 5.1.2 5.1.3 Multi- 5.2.1 5.2.2	mulation model problems	57 59 62 63 65 67 69 71 72 72 72 72 74 74 74 74

x_____

		5.3.1	The NSGA-II	84
		5.3.2	Selection operators	89
		5.3.3	Crossover operators	91
		5.3.4	Mutation operators	93
	5.4	Domin	ance-based bi-objective simulated annealing	95
		5.4.1	The DBMOSA algorithm	95
		5.4.2	The annealing schedule	99
		5.4.3	The initial temperature	101
		5.4.4	Neighbourhood move operators	102
	5.5	The B	OCEGAH algorithm	105
	5.6	The B	OSAH algorithm	109
	5.7	Summ	ary	116
6	Alg	orithm	ic Parameter Evaluation	117
	6.1	Introd	uction to multi-objective statistical analysis	117
	6.2	Deterr	nining the MOOCEM algorithm hyperparameters	120
		6.2.1	Open mine problem	120
		6.2.2	(s,S) Inventory problem	123
		6.2.3	Buffer allocation problem: five machines	125
		6.2.4	Buffer allocation problem: 10 machines	125
		6.2.5	Non-linear buffer allocation problem: 16 machines	127
	6.3	Deterr	nining the NSGA-II hyperparameters	130
		6.3.1	Open mine problem	131
		6.3.2	(s,S) Inventory problem	133
		6.3.3	Buffer allocation problem: five machines	134
		6.3.4	Buffer allocation problem: 10 machines	136
		6.3.5	Non-linear buffer allocation problem: 16 machines	136
	6.4	Deterr	nining the DBMOSA algorithm hyperparameters	141
	6.5	Conclu	usion: Chapter 6	149
7	Alg	orithm	Performance Assessment and Comparison	151
	7.1	BOCE	GAH versus MOOCEM and NSGA-II	151
	7.2	BOSA	H versus DBMOSAs move operators	154
	7.3	BOCE	GAH versus BOSAH	156
	7.4	Conclu	sion: Chapter 7	158

8	Conclusion					
	8.1	Thesis	summary	159		
	8.2	Appra	isal of thesis contributions	160		
	8.3	Sugges	stions for future work	161		
Re	efere	nces		165		
\mathbf{A}	Met	amode	el Pilot Study Results	181		
в	Alg	\mathbf{orithm}	ic Parameter Evaluation Results	193		
	B.1	MOOO	CEM	193		
		B.1.1	Open mine problem	193		
		B.1.2	(s,S) Inventory problem	195		
		B.1.3	Buffer allocation problem: five machines	197		
		B.1.4	Buffer allocation problem: 10 machines	199		
		B.1.5	Non-linear buffer allocation problem: 16 machines	201		
	B.2	NSGA	-II	214		
		B.2.1	Open mine problem	214		
		B.2.2	(s,S) Inventory problem	216		
		B.2.3	Buffer allocation problem: five machines	222		
		B.2.4	Buffer allocation problem: 10 machines	227		
		B.2.5	Non-linear buffer allocation problem: 16 machines	232		
	B.3	DBMO	DSA	259		
		B.3.1	Open mine problem	259		
		B.3.2	(s,S) Inventory problem	259		
		B.3.3	Buffer allocation problem: five machines	259		
		B.3.4	Buffer allocation problem: 10 machines	259		
		B.3.5	Non-linear buffer allocation problem: 16 machines	259		
С	The	Back	propagation Training Algorithm	271		
	C.1	Backp	ropagation	271		
D	Disc	crete-e	vent Simulation Optimisation Problem Definitions	275		
	D.1	The $(s$	(S,S) inventory problem $\ldots \ldots \ldots$	275		
	D.2	The bu	uffer-allocation problems	278		
		D.2.1	The buffer-allocation problem: five machines	279		
		D.2.2	The buffer-allocation problem: 10 machines	280		

xii

Ta	ble of Contents	xiii
	D.2.3 The non-linear buffer-allocation problem: 16 machines	281
	D.3 The Open mine problem	284
\mathbf{E}	Algorithmic Comparison Results	287

List of Acronyms

AI	Artificial Intelligence
AMALGAM	A Multi-algorithm, Genetically Adaptive Multi-objective
ANN	Artificial Neural Network
ANOVA	Analysis of Variance
BAP	Buffer-allocation Problem
BAP5	Buffer-allocation Problem: five machines
BAP10	Buffer-allocation Problem: 10 machines
BAP16	Non-linear Buffer-allocation Problem: 16 machines
COP	Combinatorial Optimisation Problem
CNN	Convolutional Neural Network
DBMOSA	Dominance-based Multi-objective Simulated Annealing
DES	Discrete-event Simulation
DNN	Deep Neural Network
EA	Evolutionary Algorithm
ELU	Exponential Linear Units
FNN	Feedforward Neural Network
\mathbf{GA}	Genetic Algorithm
GD	Great Deluge
GRASP	Greedy Randomised Adaptive Search Procedure
GLS	Guided Local Search
HA	Hyperarea
HC	Hill Climbing
ILS	Iterated Local Search
IP	(s, S) Inventory Problem
LLH	Low-level Heuristic
\mathbf{LReLU}	Leaky Rectified Linear Units
MAE	Mean Absolute Error
ML	Machine Learning

MOO	Multi-objective Optimisation
MOOCEM	Multi-objective Cross-entropy Method
MOOP	Multi-objective Optimisation Problem
MOP	Multi-objective Problem
MOSA	Multi-objective Simulated Annealing
MOSO	Multi-objective Simulation Optimisation
MSE	Mean Squared Error
MTTR	Mean Time to Repairs
NFL	No Free Lunch
NSGA-II	Non-dominated Sorting Genetic Algorithm II
OMP	Open Mine Problem
\mathbf{PReLU}	Parametric Rectified Linear Units
PSO	Particle Swarm Optimisation
RBM	Restricted Boltzman Machine
\mathbf{ReLU}	Rectified Linear Units
RMSE	Root Mean Squared Error
RNN	Recurrent Neural Network
RNS	Random Number Seed
\mathbf{SA}	Simulated Annealing
${f SeLU}$	Scaled Exponential Linear Units
SO	Simulation Optimisation
SOOP	Single Objective Optimisation Problem
SGD	Stochastic Gradient Descent
SS	Scatter Search
TA	Threshold Accepting
TPS	Tecnomatix Plant Simulation
TS	Tabu Search
TSP	Travelling Salesman Problem
WWO	Water Wave Optimisation

List of Figures

1.1	Illustration of the search space of hyperheuristics and metaheuristics	2
1.2	An illustration of the proposed framework of a hyperheuristic optimiser with a metamodel filter	4
2.1	A taxonomy of the simulation optimisation approaches in literature	12
2.2	Illustration of the black-box scenario used in simulation optimisation	13
2.3	Illustration of the relationships that exist among real-world systems, simulation models and simulation metamodels	14
2.4	An illustration of solution dominance	15
2.5	Illustration of the concept of Pareto optimality for deterministic output	16
2.6	Illustration of the computational burden of simulation optimisation for stochastic output	16
2.7	Examples of approximation sets displaying different convergence and diversity properties	18
2.8	An example illustrating the notion of archiving	18
2.9	An example illustrating the concept of hyperarea	20
2.10	Exploitation versus exploration	21
2.11	Classification of hyperheuristic approaches	23
2.12	Mathematical model of an artificial neuron and its processing capabilities	28
2.13	Illustration of a feedforward neural network with one hidden layer	29
2.14	Training, validation and test set split	32
2.15	Illustration of the two simplest activation functions, the identity and the heavside functions	33
2.16	Sigmoid activation functions	34
2.17	ReLU activation functions	35
3.1	An illustration of multiple linear regression	38
3.2	Illustration of polynomial regression fitting data	38
3.3	An illustration of support vector regression	39
3.4	An illustration to aid in explaining how decision tree regression works $\ldots \ldots$	40
3.5	An example of Bayesian optimisation	43
3.6	An illustration of the curve fitting phenomena	44
3.7	Online, mini-batch and batch learning	46
3.8	Gradient descent for different learning rates	47
3.9	The learning slowdown phenomenon	48

List of Figures

3.10	k-Fold cross validation	51
3.11	The proposed machine learning pipeline	51
4.1	Illustration of a simple deterministic and stochastic system	56
4.2	Illustration of the difference between cross-sample and in-sample variation	56
4.3	Upper bound for BAP5	59
4.4	The respective CI plots and <i>p</i> -value tables obtained for the work-in-progress and throughput objectives for the OMP	61
4.5	The true Pareto front of the OMP	61
4.6	The respective CI plots and <i>p</i> -value tables obtained for the total inventory cost and service level objectives for the IP	62
4.7	The true Pareto front of the IP \ldots	63
4.8	The respective CI plots and <i>p</i> -value tables obtained for the work-in-progress and throughput objectives for the BAP5	65
4.9	The true Pareto front of BAP5	65
4.10	The respective CI plots and <i>p</i> -value tables obtained for the work-in-progress and throughput objectives for the BAP10	66
4.11	The true Pareto front of BAP10	67
4.12	The respective CI plots and p -value tables obtained for the work-in-progress and	
	throughput objectives for the BAP16	68
4.13	The true Pareto front of BAP16	69
5.1	Solution representations for optimisation problems	73
5.2	The neighbourhoods of solution representations	73
5.3	Examples of truncated distributions	76
5.4	The MOOCEM tool user-interface	77
5.5	The input and output tables in Tecnomatix $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	77
5.6	Example of the histogram concept used in MOOCEM \ldots	80
5.7	An example of inverting histogram frequencies	81
5.8	The NSGA-II tool user-interface	84
5.9	The mutation operators	85
5.10	An illustration of candidate solutions partitioned into their respective sets and fronts	86
5.11	The fitness assignment procedure followed by NSGA-II	88
5.12	An illustration of a generation in EAs	89
5.13	Example of binary string representations	90
5.14	The DBMOSA tool user-interface	95
5.15	The decision variable table for DBMOSA in Tecnomatix	96

5.16	An example illustrating the energy measure of DBMOSA $\hfill \ldots \ldots \ldots \ldots$.	97
5.17	The probability of accepting non-improving moves vs . the temperature	101
5.18	The perturbation of solution x for a minimisation problem	102
5.19	The Hyperheuristic tool user-interface for population-based search $\ . \ . \ . \ .$	106
5.20	The heuristic selection procedure for the BOCEGAH	108
5.21	The Hyperheuristic tool user-interface for single-solution based search $\ . \ . \ .$.	110
5.22	The heuristic selection procedure for BOSAH	113
6.1	Non-parametric tests and <i>post hoc</i> procedures for multiple comparisons	119
6.2	Hyperarea phenomena for discontinuous approximation fronts	122
6.3	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A1.1.1–A1.1.9	122
6.4	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A1.2.1–A1.2.9	124
6.5	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A1.3.1–A1.3.9	125
6.6	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A1.4.1–A1.4.9	126
6.7	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A1.5.1–A1.5.9	127
6.8	The average hyperareas obtained for hyperparameter combinations A1.1–A1.9 for the respective simulation problems	129
6.9	An example illustrating the hyperparameter selection process followed for BAP10 based on average hyperareas	130
6.10	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A2.1.1–A2.1.24	132
6.11	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A2.2.1–A2.2.24	134
6.12	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A2.3.1–A2.3.24	135
6.13	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A2.4.1–A2.4.24	137
6.14	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A2.5.1–A2.5.24	138
6.15	The average hyperareas obtained for hyperparameter combinations A2.1–A2.24 for the respective simulation optimisation problems	139
6.16	An example illustrating the hyperparameter selection process followed for BAP16 based on average hyperareas	140
6.17	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A3.1.1–A3.1.4	142

6.18	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A3.2.1–A3.2.4	143
6.19	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A3.3.1–A3.3.4	144
6.20	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A3.4.1–A3.4.4	145
6.21	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A3.5.1–A3.5.4	147
6.22	The average hyperareas obtained for hyperparameter combinations A3.1–A3.4 for the respective simulation problems	148
6.23	An example illustrating the hyperparameter selection process followed for BAP16 based on average hyperareas	148
7.1	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A1.9, A2.3 and the BOCEGAH	152
7.2	The <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test and the averages for the hyperareas and number of non-dominated solutions for A1.9, A2.3 and the BO-CEGAH	153
7.3	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for A3.1, A3.3–A3.4 and the BOSAH	154
7.4	The <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test and the averages for the hyperareas and number of non-dominated solutions for A3.1, A3.3–A3.4 and the BOSAH	156
7.5	Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for the BOCEGAH and BOSAH	157
7.6	The averages of the hyperareas and number of non-dominated solutions for the BOSAH and the BOCEGAH	157
B.1	The best and worst best approximation fronts for hyperparameter combinations A1.1.1–A1.1.5	194
B.2	The best and worst best approximation fronts for hyperparameter combinations A1.1.6–A1.1.9	195
B.3	The best and worst best approximation fronts for hyperparameter combinations A1.2.1–A1.2.5	196
B.4	The best and worst best approximation fronts for hyperparameter combinations A1.2.6–A1.2.9	197
B.5	The best and worst best approximation fronts for hyperparameter combinations A1.3.1–A1.3.5	198
B.6	The best and worst best approximation fronts for hyperparameter combinations A1.3.6–A1.3.9	199
B.7	The best and worst best approximation fronts for hyperparameter combinations A1.4.1–A1.4.5	200

B.8	The best and worst be $A1.4.6-A1.4.9$	best	approximation	fronts f	for	hyperparameter combinations	201
B.9	The best and worst be $A1.5.1-A1.5.5$	best	$ \begin{array}{c} \text{approximation} \\ \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \end{array} $	fronts f	for	hyperparameter combinations	202
B.10	The best and worst be $A1.5.6-A1.5.9$	best	$ \begin{array}{c} \text{approximation} \\ \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \end{array} $	fronts f	for	hyperparameter combinations	203
B.11	The average number combinations A1.1–A	of 1.9	non-dominated for the respecti	solutic ve simu	ons ilat	obtained for hyperparameter ion problems	207
B.12	The best and worst b A2.1.1–A2.1.6, A2.1.8	best 8, Až	approximation 2.1.12–A2.1.14	fronts f	for	hyperparameter combinations	215
B.13	The best and worst b A2.1.7, A2.1.9–A2.1.1	oest 11.	$\begin{array}{c} \text{approximation} \\ \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \end{array}$	fronts f	for	hyperparameter combinations	216
B.14	The best and worst b $A2.2.16-A2.2.24$	best	approximation	fronts f	for	hyperparameter combinations	217
B.15	The best and worst b $A2.2.1-A2.2.5$	best	approximation	fronts f	for	hyperparameter combinations	218
B.16	The best and worst b $A2.2.6-A2.2.10$	best	approximation	fronts f	for	hyperparameter combinations	219
B.17	The best and worst b $A2.2.11-A2.2.15$	best	approximation	fronts f	for	hyperparameter combinations	220
B.18	The best and worst b $A2.2.16-A2.2.20$	best	$ \begin{array}{c} \text{approximation} \\ \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \end{array} $	fronts f	for	hyperparameter combinations	221
B.19	The best and worst b $A2.2.21-A2.2.24$	best	approximation	fronts f	for	hyperparameter combinations	222
B.20	The best and worst b $A2.3.1-A2.3.5$	best	$\begin{array}{c} \text{approximation} \\ \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \end{array}$	fronts f	for	hyperparameter combinations	223
B.21	The best and worst b $A2.3.6-A2.3.10$	best	approximation	fronts f	for	hyperparameter combinations	224
B.22	The best and worst b A2.3.11–A2.3.15	best	$\begin{array}{c} \text{approximation} \\ \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \end{array}$	fronts f	for	hyperparameter combinations	225
B.23	The best and worst b A2.3.16–A2.3.20	best	approximation	fronts f	for	hyperparameter combinations	226
B.24	The best and worst b A2.3.21–A2.3.24	best	approximation	fronts f	for	hyperparameter combinations	227
B.25	The best and worst b $A2.4.1-A2.4.5$	best	approximation	fronts f	for	hyperparameter combinations	228
B.26	The best and worst b $A2.4.6-A2.4.10$	best	$ \begin{array}{c} \text{approximation} \\ \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \end{array} $	fronts f	for	hyperparameter combinations	229
B.27	The best and worst b $A2.4.11-A2.4.15$	best	approximation	fronts f	for	hyperparameter combinations	230

List	of	Figures
-----------------------	----	---------

B.28	The best and worst best approximation fronts for hyperparameter combinations A2.4.16–A2.4.20	231
B.29	The best and worst best approximation fronts for hyperparameter combinations A2.4.21–A2.4.24	232
B.30	The best and worst best approximation fronts for hyperparameter combinations A2.5.1–A2.5.5	233
B.31	The best and worst best approximation fronts for hyperparameter combinations A2.5.6–A2.5.10	234
B.32	The best and worst best approximation fronts for hyperparameter combinations A2.5.11–A2.5.15	235
B.33	The best and worst best approximation fronts for hyperparameter combinations A2.5.16–A2.5.20	236
B.34	The best and worst best approximation fronts for hyperparameter combinations A2.5.21–A2.5.24	237
B.35	The average number of non-dominated solutions obtained for hyperparameter combinations A2.1–A2.24 for the respective simulation problems	258
B.36	The best and worst best approximation fronts for hyperparameter combinations A3.1.1–A3.1.4	261
B.37	The best and worst best approximation fronts for hyperparameter combinations A3.2.1–A3.2.4	263
B.38	The best and worst best approximation fronts for hyperparameter combinations A3.3.1–A3.3.4	265
B.39	The best and worst best approximation fronts for hyperparameter combinations A3.4.1–A3.4.4	267
B.40	The best and worst best approximation fronts for hyperparameter combinations A3.5.1–A3.5.4	269
B.41	The average number of non-dominated solutions obtained for hyperparameter combinations A3.1–A3.4 for the respective simulation problems	270
C.1	Backpropagation visual representation of the chain rule	271
D.1	Some characteristics of the generalised (s, S) inventory problem	276
D.2	A screenshot of the IP model in Tecnomatix	277
D.3	A series of five machines in a queuing network	279
D.4	A screenshot of the BAP5 model in Tecnomatix	279
D.5	A series of ten machines in a queuing network	281
D.6	A screenshot of the BAP10 model in Tecnomatix $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	281
D.7	The non-linear BAP with sixteen machines in a queuing network \hdots	282
D.8	A screenshot of the BAP16 model in Tecnomatix	283
D.9	Examples of a shovel and mine truck in an open mine	285
D.10	A screenshot of the OMP model in Tecnomatix	286

List of Figures		xxiii
E.1	The best and worst best approximation fronts in terms of hyperarea for the BOSAH and the BOCEGAH	288
E.2	The best and worst best approximation fronts in terms of the number of non- dominated solutions for the BOSAH and the BOCEGAH	289

List of Tables

2.1	Metaheuristic classification criteria	21
2.2	Move acceptance strategies in perturbation selection hyperheuristics \ldots	24
2.3	Heuristic selection strategies in perturbation selection hyperheuristics $\ldots \ldots$	25
3.1	Hyperparameter optimisation algorithms	42
3.2	An example of the data in the IP dataset	51
3.3	The performance of the respective regressors	52
3.4	The hyperparameters search space for the FNN	52
3.5	The best hyperparameters obtained during Bayesian optimisation	53
3.6	The average performance of the respective hyperparameter combinations	53
3.7	Comparison of the performances of the respective regression models on the test set	54
4.1	Summary of the experiments used to determine a sensible upper bound \ldots .	57
4.2	The results of the ANOVA test to determine the upper bound for BAP5	58
4.3	Table of the $p\mbox{-values}$ of the $t\mbox{-test}$ for the work-in-progress objective for BAP5	59
4.4	The results of the ANOVA test for the OMP	60
4.5	The experiments that are statistically significantly different for the OMP $\ . \ . \ .$	60
4.6	The results of the ANOVA test for the IP	62
4.7	The results of the ANOVA test for BAP5	64
4.8	The experiments that are statistically significantly different for BAP5 \ldots .	64
4.9	The results of the ANOVA test for BAP10	66
4.10	The experiments that are statistically significantly different for BAP10	66
4.11	The results of the ANOVA test for BAP16	68
4.12	The experiments that are statistically significantly different for BAP16	68
4.13	Summary of the sufficient number of observations per solution for the respective problems	70
5.1	An example illustrating the incrementing process followed for an epoch and a poorEpoch	99
5.2	The temperature ranges used to specify the <i>stepsize</i> of a move	103
5.3	The initial allocations for BOSAH	111
6.1	The hyperparameter search space considered for the MOOCEM	121
6.2	The number of runs that MOOCEM obtained the true Pareto set for the OMP $\ . \ .$	123
6.3	The Friedman and Iman-Davenport test results for A1.1.1–A1.1.9	123
6.4	The Friedman and Iman-Davenport test results for A1.2.1–A1.2.9	124

List	of	Tables

6.5	The adjusted p -values obtained by the Nemenyi <i>post hoc</i> test based on the number of non-dominated solutions found for the approximation fronts for A1.2.1–A1.2.9	124
6.6	The Friedman and Iman-Davenport test results for A1.3.1–A1.3.9	125
6.7	The adjusted p -values obtained by the Nemenyi <i>post hoc</i> test based on the number of non-dominated solutions found for the approximation fronts for A1.3.1–A1.3.9	126
6.8	The Friedman and Iman-Davenport test results for A1.4.1–A1.4.9	127
6.9	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test based on the hyper- areas and number of non-dominated solution found of the approximation fronts for A1.4.1–A1.4.9	127
6.10	The Friedman and Iman-Davenport test results for A1.5.1–A1.5.9	128
6.11	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test based on the hyperareas of the approximation fronts for A1.5.1–A1.5.9	128
6.12	The hyperparameters for the MOOCEM that are considered for implementation in BOCEGAH	130
6.13	The hyperparameter search space considered for the NSGA-II	131
6.14	The number of runs that the NSGA-II obtained the true Pareto set, for the respective hyperparameter combinations	133
6.15	The Friedman and Iman-Davenport test results for A2.1.1–A2.1.24 $\ .$	133
6.16	The Friedman and Iman-Davenport test results for A2.2.1–A2.2.24 $\ .$	134
6.17	The Friedman and Iman-Davenport test results for A2.3.1–A2.3.24 $\ .$	136
6.18	The Friedman and Iman-Davenport test results for A2.4.1–A2.4.24 \hdots	136
6.19	The Friedman and Iman-Davenport test results for A2.5.1–A2.5.24 \hdots	137
6.20	The hyperparameters combinations for the NSGA-II that are considered for implementation in BOCEGAH	140
6.21	The hyperparameters that are held constant during the hyperparameter study for DBMOSA	141
6.22	DBMOSA setup information	141
6.23	The hyperparameter search space considered for the DBMOSA $\ . \ . \ . \ .$.	142
6.24	The Friedman and Iman-Davenport test results for A3.1.1–A3.1.4	143
6.25	The Friedman and Iman-Davenport test results for A3.2.1–A3.2.4	144
6.26	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test for both the hyper- area and number of non-dominated solutions found for the approximation fronts for A3.2.1–A3.2.4	144
6.27	The Friedman and Iman-Davenport test results for A3.3.1–A3.3.4	145
6.28	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test for both the hyper-	
	area and number of non-dominated solutions found for the approximation fronts for A3.3.1–A3.3.4	145
6.29	The Friedman and Iman-Davenport test results for A3.4.1–A3.4.4	146

xxvi

List	of	Tab	les

	٠	٠
XXV	1	1

6.30	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test for both the hyper- area and number of non-dominated solutions found for the approximation fronts for A3.4.1–A3.4.4
6.31	The Friedman and Iman-Davenport test results for A3.5.1–A3.5.4
6.32	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test for both the hyper- area and number of non-dominated solutions found for the approximation fronts for A3.5.1–A3.5.4
6.33	The move operators that are considered for implementation in BOSAH 148
6.34	The final hyperparameter combinations
7.1	The critical values for the BOCEGAH comparison
7.2	The Friedman and Iman-Davenport test results for A1.9, A2.3 and the BOCEGAH152 $$
7.3	A summary indicating the statistically significant differences for A1.9, A2.3 and the BOCEGAH
7.4	The critical values for the BOSAH comparison $\ldots \ldots 154$
7.5	The Friedman and Iman-Davenport test results for A3.1, A3.3–A3.4 and the BOSAH155 $$
7.6	A summary indicating the statistically significant differences for A3.1, A3.3–A3.4 and the BOSAH
7.7	The critical values for the BOSAH and the BOCEGAH comparison 156
7.8	The Friedman and Iman-Davenport test results for the BOSAH and the BOCEGAH158 $$
7.9	A comparison between the BOSAH and the BOCEGAH \ldots
A.1	Summary of the performance of MLR in terms of MSE and MAE for the 30 trails on the IP dataset
A.2	Summary of the performance of SVM in terms of MSE and MAE for the 30 trails on the IP dataset
A.3	Summary of the performance of DTR in terms of MSE and MAE for the 30 trails on the IP dataset
A.4	Summary of the performance of RFR in terms of MSE and MAE for the 30 trails on the IP dataset
A.5	Summary of the performance of XGBoost in terms of MSE and MAE for the 30 trails on the IP dataset
A.6	Summary of the performance of the ANN with hyperparameter combination A in terms of MSE and MAE for the 30 trails on the IP dataset $\dots \dots \dots$
A.7	Summary of the performance of the ANN with hyperparameter combination B in terms of MSE and MAE for the 30 trails on the IP dataset
A.8	Summary of the performance of the ANN with hyperparameter combination C in terms of MSE and MAE for the 30 trails on the IP dataset
A.9	Summary of the performance of the ANN with hyperparameter combination D in terms of MSE and MAE for the 30 trails on the IP dataset

xxviii	List of Ta	bles
B.1	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A1.1.1–A1.1.9	204
B.2	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A1.1.1–A1.1.9	204
B.3	The worst and best runs for A1.1.1–A1.1.9	205
B.4	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test based on the adjusted hyperareas of the approximation fronts for A1.1.1–A1.1.9	205
B.5	The adjusted <i>p</i> -values obtained by the Nemenyi <i>post hoc</i> test based on the number of non-dominated solutions found for the approximation fronts for A1.1.1–A1.1.9	205
B.6	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A1.2.1–A1.2.9	206
B.7	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A1.2.1–A1.2.9	206
B.8	The worst and best runs for A1.2.1–A1.2.9	207
B.9	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A1.3.1–A1.3.9	208
B.10	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A1.3.1–A1.3.9	208
B.11	The worst and best runs for A1.3.1–A1.3.9	209
B.12	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A1.4.1–A1.4.9	210
B.13	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A1.4.1–A1.4.9	210
B.14	The worst and best runs for A1.4.1–A1.4.9	211
B.15	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A1.5.1–A1.5.9	212
B.16	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A1.5.1–A1.5.9	212
B.17	The worst and best runs for A1.5.1–A1.5.9	213
B.18	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A2.1.1–A2.1.24	238
B.19	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A2.1.1–A2.1.24	238
B.20	The worst and best runs for A2.1.1–A2.1.24 \ldots	239
B.21	The adjusted <i>p</i> -values obtained by the Nemenyi post hoc test based on the hyperareas of the approximation fronts for A2.1.1–A2.1.24 \ldots	240
B.22	The adjusted p -values obtained by the Nemenyi post hoc test based on the number of non-dominated solutions found for the approximation fronts for A2.1.1–A2.1.24	241
B.23	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A2.2.1–A2.2.24	242

List	of	Tab	les
------	----	-----	-----

B.24	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A2.2.1–A2.2.24	242
B.25	The worst and best runs for A2.2.1–A2.2.24 \ldots	243
B.26	The adjusted <i>p</i> -values obtained by the Nemenyi post hoc test based on the hyperareas of the approximation fronts for A2.2.1–A2.2.24 \ldots	244
B.27	The adjusted p -values obtained by the Nemenyi post hoc test based on the number of non-dominated solutions found for the approximation fronts for A2.2.1–A2.2.24	245
B.28	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A2.3.1–A2.3.24	246
B.29	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A2.3.1–A2.3.24	246
B.30	The worst and best runs for A2.3.1–A2.3.24 \ldots	247
B.31	The adjusted <i>p</i> -values obtained by the Nemenyi post hoc test based on the hyperareas of the approximation fronts for A2.3.1–A2.3.24 \ldots	248
B.32	The adjusted p -values obtained by the Nemenyi post hoc test based on the number of non-dominated solutions found for the approximation fronts for A2.3.1–A2.3.24	249
B.33	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A2.4.1–A2.4.24	250
B.34	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A2.4.1–A2.4.24	250
B.35	The worst and best runs for A2.4.1–A2.4.24 \ldots	251
B.36	The adjusted <i>p</i> -values obtained by the Nemenyi post hoc test based on the hyperareas of the approximation fronts for A2.4.1–A2.4.24 \ldots	252
B.37	The adjusted p -values obtained by the Nemenyi post hoc test based on the number of non-dominated solutions found for the approximation fronts for A2.4.1–A2.4.24	253
B.38	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A2.5.1–A2.5.24	254
B.39	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A2.5.1–A2.5.24	254
B.40	The worst and best runs for A2.5.1–A2.5.24 \ldots	255
B.41	The adjusted <i>p</i> -values obtained by the Nemenyi post hoc test based on the hyperareas of the approximation fronts for A2.5.1–A2.5.24 \ldots	256
B.42	The adjusted p -values obtained by the Nemenyi post hoc test based on the number of non-dominated solutions found for the approximation fronts for A2.5.1–A2.5.24	257
B.43	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A3.1.1–A3.1.4	260
B.44	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A3.1.1–A3.1.4	260
B.45	The worst and best runs for A3.1.1–A3.1.4	261

List	of	Tables
------	----	--------

B.46	The hyperareas and corresponding ranks obtained for hyperparameter combinations A3.2.1–A3.2.4	262
B.47	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A3.2.1–A3.2.4	262
B.48	B The worst and best runs for A3.2.1–A3.2.4	263
B.49	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A3.3.1–A3.3.4	264
B.50	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A3.3.1–A3.3.4	264
B.51	The worst and best runs for A3.3.1–A3.3.4	265
B.52	2 The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A3.4.1–A3.4.4	266
B.53	³ The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A3.4.1–A3.4.4	266
B.54	The worst and best runs for A3.4.1–A3.4.4	267
B.55	The hyperareas and corresponding ranks obtained for hyperparameter combina- tions A3.5.1–A3.5.4	268
B.56	The number of non-dominated solutions and their corresponding ranks obtained for hyperparameter combinations A3.5.1–A3.5.4	268
B.57	The worst and best runs for A3.5.1–A3.5.4	269
D.1	Experiments used to determine a sufficient number of observations per solution for the IP	277
D.1 D.2	Experiments used to determine a sufficient number of observations per solution for the IP	277 277
D.1 D.2 D.3	Experiments used to determine a sufficient number of observations per solution for the IP	277 277 279
D.1 D.2 D.3 D.4	Experiments used to determine a sufficient number of observations per solution for the IP	277 277 279 280
D.1 D.2 D.3 D.4 D.5	Experiments used to determine a sufficient number of observations per solution for the IP	277 277 279 280 280
D.1 D.2 D.3 D.4 D.5 D.6	Experiments used to determine a sufficient number of observations per solution for the IPSummary of the experiments used for the IPProcessing times for the machines in BAP5The experiments used to determine a suiteable upper bound for BAP5Experiments used to determine a sufficient number of observations per solution for BAP5Summary of the experiments used for BAP5	 277 277 279 280 280 280
 D.1 D.2 D.3 D.4 D.5 D.6 D.7 	Experiments used to determine a sufficient number of observations per solution for the IP	 277 277 279 280 280 280 280 281
 D.1 D.2 D.3 D.4 D.5 D.6 D.7 D.8 	Experiments used to determine a sufficient number of observations per solution for the IP	277 277 279 280 280 280 281 281
 D.1 D.2 D.3 D.4 D.5 D.6 D.7 D.8 D.9 	Experiments used to determine a sufficient number of observations per solution for the IPSummary of the experiments used for the IPProcessing times for the machines in BAP5The experiments used to determine a suiteable upper bound for BAP5Experiments used to determine a sufficient number of observations per solution for BAP5Summary of the experiments used for BAP5Processing times for the machines in BAP10Summary of the experiments used for BAP5Summary of the experiments used for BAP10Summary of the experiments used to determine a sufficient number of observations per solution	277 277 279 280 280 280 281 281 281 282
D.1 D.2 D.3 D.4 D.5 D.6 D.7 D.8 D.9 D.10	Experiments used to determine a sufficient number of observations per solution for the IP	277 277 279 280 280 280 281 281 281 282 282
 D.1 D.2 D.3 D.4 D.5 D.6 D.7 D.8 D.9 D.10 D.11 	Experiments used to determine a sufficient number of observations per solution for the IP	277 279 280 280 280 281 281 282 282 282 283
 D.1 D.2 D.3 D.4 D.5 D.6 D.7 D.8 D.9 D.10 D.11 D.12 	Experiments used to determine a sufficient number of observations per solution for the IPSummary of the experiments used for the IPProcessing times for the machines in BAP5The experiments used to determine a suiteable upper bound for BAP5Experiments used to determine a sufficient number of observations per solution for BAP5Summary of the experiments used for BAP5Summary of the experiments used for BAP5Summary of the experiments used for BAP10Processing times for the machines in BAP10Summary of the experiments used for BAP10Experiments used to determine a sufficient number of observations per solution for BAP10Summary of the experiments used for BAP16Summary of the experiments used for BAP16Summary of the experiments used for BAP16	 277 277 279 280 280 281 281 282 282 283 283
 D.1 D.2 D.3 D.4 D.5 D.6 D.7 D.8 D.9 D.10 D.11 D.12 D.14 	Experiments used to determine a sufficient number of observations per solution for the IPSummary of the experiments used for the IPProcessing times for the machines in BAP5The experiments used to determine a suiteable upper bound for BAP5Experiments used to determine a sufficient number of observations per solution for BAP5Summary of the experiments used for BAP5Processing times for the machines in BAP10Summary of the experiments used for BAP5Processing times for the machines in BAP10Summary of the experiments used for BAP10Experiments used to determine a sufficient number of observations per solution for BAP10Summary of the experiments used for BAP10Processing times for the machines in BAP16Experiments used to determine a sufficient number of observations per solution for BAP16Processing times for the machines in BAP16Experiments used to determine a sufficient number of observations per solution for BAP16Processing times for the machines in BAP16	 277 277 279 280 280 281 281 282 282 283 283 286

List of	Tables	xxxi
E.1	The worst and best runs for the BOCEGAH	287
E.2	The worst and best runs for the BOSAH	287

List of Algorithms

2.1	Pareto ranking algorithm	17
3.1	Random Forest Regression	40
5.1	Truncate a probability distribution	75
5.2	Sample from a truncated probability distribution	76
5.3	Ranking threshold adaptation of the Pareto ranking algorithm $\ldots \ldots \ldots \ldots$	79
5.4	The MOOCEM algorithm in the context of simulation optimisation $\ldots \ldots \ldots$	83
5.5	Fast non-dominated sorting algorithm	86
5.6	Crowding distance sorting and assignment algorithm	87
5.7	The NSGA-II in the context of simulation optimisation $\ldots \ldots \ldots \ldots \ldots \ldots$	89
5.8	Simulated binary crossover	92
5.9	Laplace crossover	92
5.10	Blend crossover	93
5.11	Polynomial mutation	94
5.12	Dynamic mutation	94
5.13	The DBMOSA algorithm in the context of simulation optimisation	100
5.14	Neighbouring move 1	103
5.15	Neighbouring move 3	104
5.16	The BOCEGAH algorithm for population-based search	109
5.17	The BOSAH method for single-solution based search	115
CHAPTER 1

Introduction

1.1	Research background	1
1.2	Problem description and research assignment	3
1.3	Research scope	4
1.4	Research objectives	6
1.5	Research methodology	7
1.6	Thesis organisation	8

This chapter serves as an introduction to the research assignment and is formalised by research objectives, a proposed research methodology and scope. Finally, the structure of the document is discussed.

1.1 Research background

The *investigation* and *exploration* of search and optimisation methodologies are crucial research areas of decision support systems, which can be applied across industry, commerce, science, healthcare and military. Take for example the potential impact of an effective and computationally efficient decision support methodology, it could be the difference between life and death in healthcare scheduling. Schedule too few doctors and patients could die; schedule too long working hours and doctors could make fatal mistakes due to fatigue [80].

Many real-world problems are too complex to be modelled and solved by exact methods. These complex real-world problems often have no convenient mathematical representation, linear or non-linear. A subset of these problems consists of *dynamic*, *discrete*, *stochastic* problems, representing systems that evolve over time, for example a production line or a traffic intersection. Simulation is a markedly powerful and widely adopted technique for analysing and studying these complex systems [20, 256].

Attempts to use exact methods such as linear programming, non-linear programming or integer programming require several *simplifications* of assumptions and abstractions of the problem under study, resulting in two inadequate outcomes. The *first* outcome is where an optimal solution is found, but for a model that does do not adequately represent the real system, or the *second* outcome according to which the developed model is an adequate representation of the real system, but for which only low-quality solutions can be obtained [156]. Thus creating the need for *approximate* methods.

Simulation optimisation (SO) can be used to approximately solve these complex problems without the need for simplifying assumptions or abstractions. In the context of *simulation*, it is not necessary to formulate the objective function mathematically, since the simulation model (or black-box¹) becomes the objective function, subjected to some optimisation approach [80]. Although simulation is a powerful alternative to solving complex problems, it must be noted that there is an embedded abstraction error when approximating real-world problems.

¹A system where the inputs and outputs are known, without any knowledge of the systems inner workings.

Furthermore, complex optimisation problems, typically have significantly large search spaces, rendering them computationally intractable for most optimisation procedures. These problems are classified as *combinatorial optimisation problems* (COPs), *i.e.* the decision variables can assume a large number of values, subject to numerous and intricate constraints, which, taken together, can result in large, complex *search spaces* that are difficult to traverse by optimisation procedures. The combinatorial nature of large-scale problems serve as motivation for the use of *metaheuristics* and is considered a widely adopted simulation optimisation approach based on treating the simulation model as a black-box function evaluator [5].

One of the biggest problems that researchers face with metaheuristic approaches is the lack of *general applicability* and *metaheuristic selection*. This is due to the fact that the performance of metaheuristics greatly depends on the type of problem being optimised. This is where the *No Free Lunch* (NFL) theorem originated as each metaheuristic has its strengths and weaknesses when it comes to searching the search space [80],

The NFL theorem states that, no search algorithm, irrespective of the performance measures used, performs better than (or worse) than another based on its average performance over all optimisation problems [257].

Researchers therefore tend to search for algorithms that are considered to be generally applicable, *i.e.* algorithms that can perform satisfactory on a subset of optimisation problems or specific problem domain. The problem, however, is that a vast majority of problems do not have an *exploitable* structure that can be used to characterise them into a subset of properties, and consequently one cannot design an effective algorithm for a subset of problems with unknown properties. The application of *hyperheuristics* to the field of optimisation, specifically simulation optimisation, is motivated by the NFL theorem, as they circumvent and mitigate (to an extent) the issue of characterising and designing problem-specific algorithms, by instead operating in the algorithm search space (first) before operating on the solution space, as illustrated in Figure 1.1 [80, 213].



Figure 1.1: Illustration of the search space of (a) hyperheuristics, and (b) metaheuristics, adapted from [93].

Hyperheuristics can be used to construct an ensemble of algorithms by combining known heuristics in ways that enable each to compensate, to some degree, for the weaknesses of others by method of *algorithmic cooperation*, thereby solving a larger range of problems than their individual counterparts are capable of solving. Hyperheuristics are commonly referred to as "*heuristics* to choose heuristics" in the context of combinatorial optimisation [36], importantly, unlike most metaheuristic applications, hyperheuristics work with the search spaces of heuristics and not the search spaces of solutions and can therefore be used to construct a more general-purpose optimisation procedure. The NFL theorem still applies, the only difference is that a larger subset of problems can be solved using hyperheuristics [80].

It is important to note that optimisation procedures can be classified as either *single solution-based* search, which operate iteratively on a single candidate solution, or *population-based* search,

 $\mathbf{2}$

which operate iteratively on a population of candidate solutions [93]. In this study both metaheuristics and hyperheuristics are classified as such. An example of a single-solution based metaheuristic (or S-metaheuristic) is the *simulated annealing* (SA) [143] algorithm and a populationbased metaheuristic (or P-metaheuristic) is the *genetic algorithm* (GA) [120].

Due to the dynamic and stochastic nature of simulation models, it is required that a sufficient number of observations per solution be evaluated. Consequently, the simulation optimisation process should intelligently search the search space as to minimise its computationally expensive nature [20]. Metaheuristic solution methodologies often use *metamodels* to screen out solutions that are predicted to be of low-quality when compared with the incumbent solution, *i.e.* they reduce the number of solutions to be evaluated by filtering out low-quality solutions. The simulation optimisation process can therefore be enhanced by combining heuristic procedures with metamodels to search the solution space less *exhaustively* and more *intelligently* by eliminating low-quality solutions from the search space. Note that quality is based on the performance measure being optimised, for example solution dominance [5].

Some metamodel procedures use *artificial neural networks* (ANNs), to build metamodels that aid the metaheuristic search procedure. The ANN uses predefined rules to filter out the solutions that are potentially low-quality solutions, thereby mitigating computationally expensive evaluations of the black box evaluator [5, 156]. The ANN is trained using the decision variable values together with their corresponding objective function values obtained during the search, over many iterations, as datapoints. The trained ANN then becomes the metamodel that evaluates solutions, if the solutions are not considered to be low-quality solutions, then the simulation model may evaluate the solution. The metamodel is less computationally expensive to evaluate than the simulation model.

The hyperheuristic approach to simulation optimisation is an attempt to create a more generally applicable optimisation tool, whereby decreasing the effects of the NFL theorem. Additionally, an ANN metamodel filter attempts to accelerate the simulation optimisation process, thereby decreasing the effects of computationally expensive problems.

1.2 Problem description and research assignment

From the research background given, it should be apparent that there is a lack of general-purpose (context-independent) optimisation tools, specifically within the field of simulation optimisation. Therefore, the aim of this thesis is twofold. The first aim is to create a general-purpose bi-objective simulation optimisation procedure using a hyperheuristic framework, to successfully explore and approximately solve a subset of five simulation optimisation problems. The second is to conduct a pilot study to determine whether or not an ANN metamodel is a feasible solution to enhance the simulation optimisation process. The procedure is to be implemented in Tecnomatix, thereby extending its simulation optimisation capabilities, which in turn may be beneficial to industries using Tecnomatix as their decision support system.

A high-level overview of the proposed general-purpose optimisation framework is presented in Figure 1.2. Assume that the metamodel has been trained and that the initial solution(s) have been generated. The proposed simulation optimisation process *starts* by evaluating the initial solution(s), the hyperheuristic is *run* on the current solution, from which the hyperheuristic generates the neighbouring solution(s) \boldsymbol{x} . Thereafter, the metamodel predicts the objective function value(s), *i.e.* $F(\boldsymbol{x})$ of \boldsymbol{x} . If the predicted objective function value(s) is *good*, then the simulation model may *evaluate* \boldsymbol{x} , resulting in $f(\boldsymbol{x})$, otherwise \boldsymbol{x} is discarded. This process repeats until the *stopping condition* is reached, after which the simulation optimisation process

4

terminates.

Note that for single-solution based search framework the simulation model takes as input a single solution, *i.e.* one value for each decision variable, to evaluate which is then transformed by the hyperheuristic. For a population-based search framework the simulation model takes as input a population of solutions, *i.e.* each decision variable has several values to be evaluated (simultaneously) and then transformed by the hyperheuristic. In this study both search frameworks are considered, namely a single-solution based hyperheuristic and a population-based hyperheuristic framework.



FIGURE 1.2: An illustration of the proposed framework of a hyperheuristic optimiser with a metamodel filter [5].

To summarise, the research assignment can be formulated as,

- 1. *Design* two hyperheuristics based on the two classifications mentioned previously, namely a single-solution based hyperheuristic and a population-based hyperheuristic to solve for *bi-objective simulation optimisation* of discrete-event *dynamic* and *stochastic* problems.
- 2. *Determine* whether an ANN metamodel is a feasible solution to enhance the simulation optimisation process and if so, *construct* the metamodel with the appropriate *architecture* that can be applied to a large subset of problems.
- 3. Design, develop and implement a computerised user-interface that will enable the simulation analyst using Tecnomatix to solve bi-objective simulation optimisation of discreteevent dynamic, stochastic problems with the hyperheuristics and possible ANN metamodel filter, thereby extending the bi-objective optimisation functionality of Tecnomatix.

In conclusion, the novelty of the research problem under investigation necessitates the design of a hyperheuristic optimiser with an ANN metamodel filter feature within Tecnomatix that may be used to solve bi-objective simulation optimisation problems. The research assignment warrants an in-depth exploration into the workings of metaheuristics and hyperheuristics as well as investigation into the different ANN architectures that can generalise well to unseen datapoints and consequently approximate various problems with sufficient accuracy.

1.3 Research scope

The following delimitations are adopted to narrow down the scope of the research problem considered in this thesis:

1.3. Research scope

- system that they were modelled for. A limited number of discrete-event, dynamic and stochastic simulation optimisation problems form the testbed so as to validate the optimisation tools. The problems include the (s, S) inventory problem (IP), three instances of the buffer-allocation problem (BAP), namely the BAP with five machines (BAP5), the BAP with 10 machines (BAP10) and the non-linear BAP with 16 machines (BAP16), and lastly the Open mine problem (OMP).
- **Bi-objective optimisation** of *two* objective functions simultaneously. This may involve the following combinations of minimisation (min) or maximisation (max); *min-min*, *min-max* and *max-max*.
- **Hyperheuristic** designs that result in more general-purpose, robust optimisation procedures that are novel in its contribution due to the given optimisation context, *i.e.* simulation optimisation. Hyperheuristics are considered more appropriate when solving different problems or even instances of the same problem and do so by managing a set of low-level (meta)heuristics (LLHs), *i.e.* LLHs are either complete metaheuristics or the heuristic components of metaheuristics.

Accordingly, the LLHs (heuristic components) considered for the inclusion in the singlesolution based hyperheuristic (SBH) are *move operators* of the multi-objective variant of simulated annealing, namely *dominance-based multi-objective simulated annealing* (DB-MOSA) [233]. The LLHs (metaheuristics) considered for the inclusion in the populationbased hyperheuristic (PBH) are the multi-objective cross-entropy method (MOOCEM) [21] and the non-dominated sorting genetic algorithm (NSGA-II) [67].

- Metaheuristics have a plethora of hyperparameters, each influencing the performance of the metaheuristic. For this reason, this study includes an empirical study conducted to determine common hyperparameters for the MOOCEM, the NSGA-II and DBMOSA, respectively.
- Artificial neural networks are considered to form part of the feasibility analysis for metamodeling purposes. According to Hornik *et al.* [125] and supported by Faucett [84] an ANN with one hidden layer that feeds datapoints in a so-called forward fashion, referred to as a feedforward neural network (FNN), are considered universal function approximators. The fundamental premise of training ANNs are more or less the same however their applications are different. For that reason, ANNs such as convolutional neural networks (CNNs) and recurrent neural networks (RNNs) are excluded from consideration in this study. Furthermore, hardware constraints provide further motivation for this delimitation. The interest is simulation metamodeling for regression problems.

Similar to simulation models, ANN as well as many machine learning models are essentially black box optimisers and as a result, building and gaining insight into how the ANN solves a problem is not straightforward. Interestingly, despite the possibilities of ANN, they are rarely part of introductory optimisation classes [237]. Consequently, hyperparameter optimisation of a FNN with one hidden layer is of great importance.

- **Supervised learning** is the machine learning paradigm used to facilitate function approximation as the ANN learns the relationship between the input values(s) and the corresponding output values(s) to ultimately make predictions based on unseen inputs [81].
- **Regression models** are used for comparison purposes to determine whether or not the FNN as simulation metamodel is feasible. This is done by comparing the performance of the

FNN to that of other regression models, namely linear regression [2], polynomial regression [113], support vector regression [234], decision tree regression [205] and random forest regression [168, 55].

- **Dataset** is obtained, for the sake of relevancy, by running exhaustive enumeration on the (s, S) inventory problem (IP). This dataset is used to train, validate and test the network generalisation capabilities as well as to facilitate the hyperparameter optimisation.
- Solution representation considered in this study includes; discrete and binary variables. Continuous variables, permutations and mixed solution representations are mentioned– however they are excluded from the functionality of the optimisation tool.
- **Development** of the optimisation tools are built within Tecnomatix using the objects and methods to create an interactive user-interface facilitating the simulation optimisation process. The programming language used within Tecnomatix is *SimTalk*, and will be used to program the various algorithms.

Note that the DBMOSA, the MOOCEM and NSGA-II are built individually as well as part of their hyperheuristics to facilitate a comparison between the individual algorithms and its ensemble.

Algorithmic comparisons are limited to the BOCEGAH and BOSAH and their constituent LLHs and is facilitated using *non-parametric statistical testing*. The principal aim of this study as per the problem statement in §1.2, is twofold. The first aim is to build an optimisation tool (hyperheuristic) that performs better (measured accross five problems) than its constituent LLHs, and the second is to determine the feasibility of an ANN as metamodel to enhance the simulation optimisation process.

It is anticipated that further assumptions may be necessary as the project progresses, and these will be reported in detail at the appropriate time.

1.4 Research objectives

The following *eight* objectives are pursued in this thesis:

- 1. *Conduct* a review of the academic literature related to simulation optimisation, more precisely the application of hyperheuristics as well as literature related to applying FNNs as simulation metamodels.
- 2. *Perform* a preliminary study to determine the feasibility of a FNN with one-hidden layer as metamodel, to determine its capabilities as function approximator, before implementation.
- 3. *Determine* the specific parameters related to each simulation problem studied, parameters that are necessary to conduct a statistically sound simulation optimisation study, for example a sensible upper bound.
- 4. *Review* the literature covering the LLHs and *identify* based on the review, which hyperparameters to include in the hyperparameter tuning search space. Also, *build* a computerised user-interface within Tecnomatix to facilitate the simulation optimisation process for the MOOCEM, the NSGA-II and the DBMOSA algorithm.

 $\mathbf{6}$

- 5. *Design* the population-based and single-solution based hyperheuristic based using the chosen LLHs, and build a computerised user-interface within Tecnomatix to facilitate the simulation optimisation process.
- 6. Perform hyperparameter tuning for the LLHs, respectively, and report the results.
- 7. *Employ* the common best hyperparameters and *evaluate* the hyperheuristic optimisation tools within Tecnomatix on all five discrete-event dynamic stochastic simulation optimisation problems and *interpret* the results obtained from the various tools and report valuable insights realised.
- 8. *Recommend* possible follow-up work and further research avenues that may be pursued to continue the work done in this thesis.

1.5 Research methodology

The following methodological procedure is adopted to achieve the objectives set out for this thesis:

- 1. Conduct an extensive literature review on the following topics relating to this thesis and the completion thereof. The topics aim to assist the researcher in gaining the required knowledge and insight necessary to fulfil the research objectives 1–7.
 - (a) Multi-objective simulation optimisation,
 - (b) multi-objective optimisation preliminaries,
 - (c) metaheuristics,
 - (d) hyperheuristic design, and
 - (e) simulation metamodeling by method of ANNs.
- 2. Perform a preliminary study on FNN metamodel feasibility, in fulfilment of Objective 2.
 - (a) Perform hyperparameter optimisation on a FNN with one hidden layer,
 - (b) evaluate the performance of the optimised FNN on the IP dataset,
 - (c) evaluate the regression models on the IP dataset, and
 - (d) compare their performances, *i.e.* FNN versus regressors.
- 3. *Define* the statistical prerequisites necessary to conduct a simulation study and *determine* the following specific to each simulation problem, in fulfilment of Objective 3.
 - (a) The upper bounds,
 - (b) the sufficient number of observations per solution.
- 4. Based on relevant literature, *report* the hyperparameters to be included in the hyperparameter tuning search space, in fulfilment of Objective 4.
- 5. *Design* the population-based hyperheuristic using the MOOCEM and the NSGA-II, called the **bi-o**bjective **c**ross-**e**ntropy and **g**enetic **a**lgorithm **h**yperheuristic (BOCEGAH). Also, *design* the single-solution based hyperheuristics, called the **bi-o**bjective **s**imulated **a**nnealing **h**yperheuristic (BOSAH) in fulfilment of Objective 5.

- 6. *Design* and *build* a computerised user-interface within Tecnomatix to facilitate the simulation optimisation process for the DBMOSA, MOOCEM, NSGA-II, BOCEGAH and BOSAH, also in fulfilment of Objectives 4 and 5.
 - (a) Test and validate each metaheuristic and hyperheuristic in terms of logic and output using structured walkthroughs,
 - (b) test the user-interface to ensure that it meets the conceptual and operational validity requirements and is credible,
 - (c) the functionality must exhibit reasonableness in continuity, consistency, and produce errors based on absurd conditions,
 - (d) reiterate the optimisation tools where necessary.
- 7. *Perform* hyperparameter tuning and extensive algorithmic parameter evaluations in a structured and statistically sound manner in order to determine the best hyperparameter combinations for BOCEGAH and BOSAH and its constituent LLHs, considering all five simulation problems. This is done in fulfilment of Objective 6.
- 8. *Evaluate* the optimisation tools within Tecnomatix for each simulation optimisation problem in a structured and statistically sound manner, in fulfilment of Objective 7.
 - (a) Compare the performance of each hyperheuristic with it constituent LLHs as well as with each other,
 - (b) evaluate the general application capabilities of the optimisation tools developed, and
 - (c) report valuable insights realised.
- 9. *Discuss* the contribution of the addition of a general-purpose simulation optimisation tool within Tecnomatix and industries using Siemens and *recommend* future-work to further the work done in this thesis, based on the research findings, in fulfilment of Objective 8.

1.6 Thesis organisation

This thesis comprises a total of eight chapters, including this introductory chapter, a bibliography, and six appendices. Chapter 2 comprise a literature study of the work related to this thesis. Thereafter, the detailed documentation of the preliminary study conducted to determine the feasibility of an FNN as metamodel is delineated in Chapter 3. Chapter 4 reports the statistical prerequisites required to conduct a simulation study and introduced the simulation models studied. Chapter 5 comprises a review of the LLHs and their algorithm specific parameters used, as well as the population-based and single-solutions based search hyperheuristics proposed in this study. Chapter 6 reports the empirical study conducted to determine suitable hyperparameter combinations for each LLH, followed by the results chapter, Chapter 7, where both BOCEGAH and BOSAH are compared with their constituent algorithms and later compared with each other. Finally, Chapter 8 comprises a summary of the work conducted in this thesis and a documentation of the research contributions and concludes with recommended future work to be done.

This chapter contains a contextual description of time-consuming nature of simulation optimisation and the need for enhancements. This led to the formulation of a research assignment and research objectives constrained by the scope. In Chapter 2, a literature study is conducted on simulation optimisation and metamodeling. More specifically simulation optimisation approaches used in the literature are briefly discussed after which the multi-objective optimisation preliminaries are documented. Thereafter, literature pertaining to metaheuristics and hyperheuristics are discussed. Lastly, metamodels form the basis of discussion, specifically FNN.

Chapter 3 reviews the regression models that are used for comparison purposes. Next, the hyperparameter optimisation approach followed for the FNN is documented. Some performance metrics are introduced, some of which are used to compare the regressors. Lastly, the performances of the FNN is compared to that of the other regression models.

Chapter 4 briefly introduces some statistical prerequisites required to conduct a simulation study, thereafter the dynamic, stochastic simulation problems are discussed. For each problem a sensible upper bound and a sufficient number of observations per solution is determined. This chapter also documents the decision variables and objectives for each problem as well as the true Pareto fronts found by method of exhaustive enumeration and lastly, describes each simulation model in terms of its complexity. The simulation problems considered in this study include the (s, S) inventory problem, the buffer-allocation problem with five machines, the buffer-allocation problem with ten machines, the non-linear buffer-allocation problem with sixteen machines, and the Open mine problem. See Appendix D for the respective problem descriptions.

Chapter 5, presents the theoretical foundation of the LLHs, as well as its formulation and application to simulation optimisation specifically. Furthermore, the chapter discusses each LLHs algorithm specific parameters that will be included in the hyperparameter tuning search space. Thereafter, the theoretical foundation of the hyperheuristics proposed are presented, including its formulation and application to simulation optimisation context specifically.

Chapter 6 documents the empirical study conducted to determine common hyperparameter combinations for each LLH across all five simulation problems by method of statistical inferences. Penultimately, Chapter 7 documents the performances of the hyperheuristics as well as compares the hyperheuristics with its constituent algorithms and later with each other. The comparisons are performed using non-parametric statistical tests with relevant *post hoc* analysis, where applicable.

Finally, Chapter 8 summarises the work done, documents avenues for improvements and future work contributions as well as formally documents the research contributions made by the completion of this thesis.

CHAPTER 2

Literature Study

2.1	Simulation optimisation							
2.2	2 Multi-objective optimisation preliminaries							
	2.2.1	2.2.1 The notion of solution dominance						
	2.2.2 The quality of MOO algorithm							
	2.2.3 Archiving							
	2.2.4	Performance assessment of MOO algorithms	19					
2.3	Metah	neuristics	20					
2.4	Hyperheuristics							
2.5	5 Simulation metamodeling							
	2.5.1	The fundamentals of artificial neural networks	26					
	2.5.2	The network architecture	28					
	2.5.3	The network training algorithms	31					
	2.5.4	Activation functions	33					
2.6	6 Summary							

This chapter serves to fulfil Objective 1 as stated in Chapter 1. In this chapter, the reader is introduced to some of the fundamental concepts of multi-objective simulation optimisation (MOSO). First, a short overview is given of the simulation optimisation (SO) approaches in the literature. Next, an introduction to multi-objective optimisation (MOO) preliminaries pertaining to the notion of Pareto dominance and methods for determining the non-dominated solution sets, the quality of the Pareto sets are then discussed in terms of the convergence and diversity of the solution set. Thereafter an introduction to metaheuristic and hyperheuristic methods are given. Penultimately, an introduction to artificial neural networks (ANNs) as a metamodel is proposed and discussed. Finally, the chapter closes in with a summary of its contents.

2.1 Simulation optimisation

The following example demonstrates the need for simulation as a research field: A nuclear reactor operates within safe operating zones, according to which abnormal operating conditions could lead to events similar to that of the Chernobyl disaster in 1986. By method of simulation, however, abnormal operating conditions can be reproduced in theory and then avoided in practice. Simulation enables the *trail-and-error* process that real-world implementations do not, whereby reducing the cost, time and potential risk associated with the implementation of new designs [219].

Simulation can be characterised by the *time increment, variable types* and *time dependency* of the problem and whether or not the problem is *terminating*. In this study, the nature of the problems is terminating discrete-event simulation (DES) problems. To elaborate, problems are *discrete* (time increment), *stochastic* (variable type), *dynamic* (time dependency) and are evaluated for a specified amount of simulated time (terminating) [20].

In DES models, the state of the system changes at *discrete* and usually *random* points in time [224]. Take for example a single-server queueing system – customers arrive from an unknown population at an arrival rate lambda; the customer can go to the server if the server is idle, or join the queue and wait until the server becomes idle [20]. To summarise: the single-server queueing system is discrete (unique events can be identified), dynamic (the state of the system evolves over time), and stochastic (the arrival rate of customers is determined by the Poisson distribution) and starts when the store opens and terminates when the store closes.

Simulation optimisation provides a way to overcome the limitations of traditional optimisation, attributable to its ability to model uncertainty and complex interactions. The main drawback of simulation optimisation is the *computational cost* required for the trail-and-error (simulated) process [99]. Figure 2.1 shows the optimisation approaches used in literature in combination with simulation to enhance the simulation optimisation process and reduce the associated computational cost [3, 4, 18, 90, 258].



Figure 2.1: A taxonomy of the simulation optimisation approaches in literature [18].

Random Search & Metaheuristics are used in combination with simulation to search the decision space and find good solutions to large and typically complex problems intelligently. These methods provide the simulation model with inputs (or decision variable combinations) to evaluate and thereby reduce the search space by intelligently searching the decision space. The result is a smaller part of the total decision space that needs to be explored and typically less computational power is required to find good objective function values [20].

Ranking & Selection (R&S) procedures are statistical methods developed to compare a finite number of simulation alternatives and select the best system from a set of competing alternatives [105]. R&S determines the minimum number of observations per solution required, with a certain probability, to ensure that the optimal input parameter values over a finite set k, where k is 'small', *i.e.* 2–200, is found to be different at some level of statistical significance [3]. Statistical methods, specifically R&S, are recommended in case of a discrete search space with limited number of solutions, whereas random search and metaheuristics are preferred when there are a large number of alternatives, *i.e.* combinatorial nature [20]. For an overview of R&S methods, the reader is referred to [260].

Direct gradient methods are considered efficient estimators as they derive the gradient of the objective function(s) and determines the search direction used in simulation optimisation algorithms and include perturbation analysis [27, 94, 117], likelihood ratios or score functions [102, 215, 216]. Metamodel methods are used to find the functional relationship between inputs and their corresponding outputs. In the context of this study, metamodels are used to enhance the simulation optimisation process by filtering and screening out solutions that are predicted to be of *low-quality* when compared with the incumbent solution(s) [18], see §2.5 for more detail.

The use of these approaches is motivated by the sheer size of most real-world problems, specifically *combinatorial optimisation problems* (COPs). COPs are a class of optimisation problems comprising multiple discrete decision variables, as opposed to continuous decision variables. Even though COPs have a finite search space they can often be too large for exhaustive enumeration to be considered a feasible option [150].

In support of this statement, the so-called combinatorial relationship is explained. As the number of decision variables increase the number of combinations proverbially explode, which is often referred to as the *combinatorial explosion problem* [150]. Consider the buffer-allocation problem (BAP) with 16 machines as an example: there are 15 buffers and suppose that each buffer has 10 buffer spaces. This example results in a search space cardinality of 1×10^{15} solutions. Now imagine that each solution evaluation takes 0.5 seconds and because of the stochastic nature of simulation models, several observations per solution evaluation is required, say 100. This results in an exhaustive search that will take 1585489599 years to complete. Also, to further this explanation, it is said that there are approximately 1×10^{25} grains of sand on earth.

Some examples of real-world COPs are vehicle routing and scheduling problems. Most COPs are considered NP-Hard, which requires exponential time to be solved to optimality [93]. For an exhaustive list of COPs being studied in literature, the reader is referred to [134].

There are two main approaches towards optimisation, namely *exact* and *approximate* (or heuristic) methods. Even though exact methods guarantee optimal solutions within a finite amount of time, they are not considered suitable for solving many real-world problems [198]. On the other hand, approximation methods, such as metaheuristics, do not guarantee optimality but are widely used for solving complex COPs [198]. Simulation used in combination with metaheuristics can guide the search, evaluated by the simulation model, in the most effective way, instead of blindly itemising scenarios or performing exhaustive enumeration [99].

The simulation model represents the objective function that needs to be optimised, representing the complex but unknown relationship between the decision variables, termed black-box optimisation [80]. The simulation optimisation process is illustrated in Figure 2.2. First, the number of observations per solution, the confidence interval and the simulation time specific to the simulation problem are to be defined as it influences the simulation output as well as the analysis of the output. This matter is discussed in detail in Chapter 4.



Figure 2.2: Illustration of the black-box scenario used in simulation optimisation, adapted from [93].

Next, values for the decision variables are used as input x which are then evaluated by the simulation model (or black-box or objective function) to generate the output f(x). The output is then subjected to some optimisation approach (metaheuristic), which determines the next input value(s) of the decision variables to be evaluated by the simulation model. This procedure continues until the optimisation approach terminates due to a predefined stopping condition. Note that many decision variables may be evaluated, and as such x is a vector of n values, where n is the number of decision variables. Similarly many objective(s) may be of interest, and as such f(x) corresponds to a vector of m values.

There is a trade-off relationship between *time*, *cost*, *accuracy* and *risk* for the real-world system, a simulation model and simulation metamodel. This relationship is depicted in Figure 2.3, where the inputs $\{x_1, \ldots, x_i, \ldots, x_n\}$ for the real-world system, the simulation model and metamodel are the same, however, the outputs are different and distinguishes each in terms of *time*, *cost*, *accuracy* and *risk* associated. In Figure 2.3 there are *n* inputs and *m* outputs. The inputs are natural numbers $(\mathcal{N} \in \{1, 2, \ldots, \infty\})$ since the inputs considered in this study are integers, while the outputs are real numbers (\mathcal{R}) , *i.e.* integer or non-integer values.



Figure 2.3: Illustration of the relationships that exist among real-world systems, simulation models and simulation metamodels, adapted from [165].

The real-world implementation is the most time-consuming, costly and embodies the most risk, but is the most accurate. Simulation metamodels require the least amount of time, are the least costly and embody the least risk, but is also the least accurate. Simulation represents a good trade-off between each of the aforementioned criteria and is therefore the preferred choice by many. Simulation model can be used in combination with a metamodel, to benefit from the accuracy of the simulation model and the time, cost and risk of metamodels [165].

Each of the simulation problems considered in this study are combinatorial in nature and therefore simulation in combination with some optimisation approach is the preferred choice. Furthermore, an optimisation problem can be classified as *single-objective* or *multi-objective*, in this study bi-objective simulation optimisation problems are considered. The next section addresses the preliminaries for solving multi-objective COPs.

2.2 Multi-objective optimisation preliminaries

Multi-objective optimisation problems (MOOPs) have multiple, usually *conflicting* and often *non-commensurate* objectives that need to be optimised simultaneously. For this reason, MOO searches for a set of *tradeoff* solutions, termed the *Pareto set* [186].

Formally, a MOOP has a set of m objective functions, n decision variables with J equality and K inequality constraints. As mentioned previously, in multi-objective simulation optimisation (MOSO) there is no formulation of an objective function, the simulation model is the objective function. However, for the sake of completeness, the mathematical formulation of a MOOP can be stated as follows, where the minimisation of all the objectives is assumed without loss of generality,

minimise
$$f(\boldsymbol{x}) = \{f_1(\boldsymbol{x}), \dots, f_m(\boldsymbol{x})\},$$
 (2.1)

subject to
$$g_k(\boldsymbol{x}) \ge 0, \qquad k \in \{1, \dots, K\},$$
 (2.2)

$$h_j(\boldsymbol{x}) = 0, \qquad j \in \{1, \dots, J\},$$
(2.3)

$$x_i^{(L)} \le x_i \le x_i^{(U)}, \quad i \in \{1, \dots, n\},$$
(2.4)

where
$$\boldsymbol{x} = \{x_1, \dots, x_n\} \in X,$$
 (2.5)

$$\boldsymbol{f} = \{f_1, \dots, f_m\} \in F,\tag{2.6}$$

where X denotes the decision space $x = (x_1, x_2)$, where x_1 and x_2 have discrete values, and F the objective space $y = (y_1, y_2)$. Decision variables are bounded by an upper and lower bound

as stated in (2.4). If solution x satisfies the constraints (2.2)–(2.4), it is considered a feasible solution [46, 169, 266].

For any MOOP, there exists a *true* but *unknown* set of non-dominated (or Pareto optimal) solutions, when considering the entire feasible decision space. These solutions form the true Pareto set (\mathcal{P}_T) and visually they represent the boundary called the true Pareto front. Solutions are guided towards the Pareto front by method of *fitness assignment*. The fitness assignment strategy used in this paper is dominance-based [266]. Approximate methods search the decision space and identify a set of *high-quality* non-dominated solutions that form the approximate Pareto set, for a given MOOP, by means of the notion of *solution dominance* (or *Pareto dominance*) [186]. The notion of solution dominance is discussed next.

2.2.1 The notion of solution dominance

Solutions are considered non-dominated if no other solutions in the search space are superior to them, when *all* objectives are considered [266]. Formally, a solution x_1 dominates another solution x_2 , if the following criteria are true:

- 1. Solution x_1 is no worse than solution x_2 in respect of all objectives, and
- 2. solution x_1 is better than solution x_2 in at least one objective, f_1, \ldots, f_m .

If criteria <u>1</u> and <u>2</u> are *true*, it can be said that x_1 dominates x_2 , denoted by $x_1 \prec x_2$. This is elucidated in the *max-min* bi-objective optimisation problem (BOOP) given in Figure 2.4. The members $\{x_3, x_5\}$ form approximate Pareto Front 1, *i.e.* $\mathcal{F}_1 = \{3, 5\}$ and are considered nondominated solutions [68, 266]. For the sake of brevity, when referring to true Pareto solutions, the solutions are explicitly referred to as *true*, otherwise it is regarded as approximate.



Figure 2.4: An illustration of solution dominance using a population of five solutions, adapted from [60].

Depending on the space considered, the number of Pareto solutions may differ. In Figure 2.5, a solution in the objective space F is considered as a single point $\{f_1, f_2\}$ (\bigcirc), however it is represented by two different solutions x_1 and x_2 in the decision space (\bigcirc). The convention used in this paper is that Pareto set refers to the objective vector, while Pareto optimal solutions refer to the corresponding decision variable combinations. If the system is operated using any of the Pareto optimal solutions (\bigcirc), then the system is considered optimal. Mathematically, non-dominated solutions are regarded as equally desirable. However, in the real world, a final solution is selected for implementation and is based on some preference information.

This selection process is typically the responsibility of the decision maker [181, 266], but is not part of the scope of this study.



Figure 2.5: Illustration of the concept of Pareto optimality for two decision variables $\{x_1, x_2\} \in X$ and two objectives $\{f_1, f_2\} \in F$ for a *min-min* and deterministic output [20].

MOO of *stochastic output* is very complex and computationally intensive and is elucidated using Figure 2.6. For each decision variable combination $\{x_1, x_2\}$, the simulation model must execute a number of observations per solution to be confident, at a certain level of significance, that the point estimator value is a *good approximation* of the true but unknown values for $\{x_1, x_2\}$ [15]. This matter is discussed in detail in Chapter 4.



Figure 2.6: Illustration of the computational burden of simulation optimisation for two decision variables $\{x_1, x_2\} \in X$ and two objectives $\{f_1, f_2\}$ in the simulated objective space (\hat{F}) for stochastic output, adapted from [20].

Many solutions are evaluated during multi-objective simulation optimisation (MOSO), especially if metaheuristics are used to guide the search. This requires some form of ranking to determine which solutions form part of the approximate Pareto set. A popular ranking method is described in Algorithm 2.1, however, this method only applies to *deterministic results* [20, 104]. Recall that, the results generated by a simulation model are *stochastic*, therefore the more appropriate ranking method is multi-objective ranking and selection (MORS), as it ensures that the solutions that constitute the Pareto set are statistically significantly different [72, 160]. However, MORS falls outside of the scope of this project. Consequently, some work has to be done to ensure that the solutions in the Pareto set are *statistically significantly different* to enable the use of Algorithm 2.1. MORS is replaced by determining a sufficient number of observations per solution that results in statistically significant solutions. This is discussed in more detail in Chapter 4.

Consider Algorithm 2.1 for a min-min BOOP. The algorithm takes as input the approximate Pareto set S and returns the non-dominated fronts and corresponding ranks. The algorithm starts by sorting the first objective (f_1) in descending order. Thereafter, the values of the second objective (f_2) are considered and determine its rank. Remember that both objectives are to be minimised, therefore if the value of f_2 in row *i* is larger than the value of f_2 in row *j*, the rank of row *i* is incremented since it is dominated. Initially each row has a rank of zero.

Algorithm 2.1: Pareto ranking algorithm, for a min-max bi-objective problem [104]

	Input : A set of solutions S containing the value	les for two objective functions.				
	Output : The approximate Pareto set \mathcal{P}_S which	contain all the solutions with $\rho = 1$				
1	$\mathcal{S} \leftarrow sort(\mathcal{S}, 1);$ // Sor	t the first objective in descending order				
2	for $i \leftarrow 1$ to $ \mathcal{S} $ do					
3	$penalty \leftarrow 0;$	<pre>// Initialise Pareto rank to zero</pre>				
4	for $j \leftarrow (i+1)$ to $ \mathcal{S} $ do					
5	if $f_j^2 \prec f_i^2$ then // <i>i.e.</i> if $f_j^2 \ge f_i^2$					
6		<pre>// Increment the penalty</pre>				
7	$\begin{bmatrix} \rho_i \\ \leftarrow penalty; \end{bmatrix}$	// Rank of solution i				
8 for $i \leftarrow 1$ to $ \mathcal{S} $ do						
9	if $\rho_i = 0$ then					
10	$ F_1 \leftarrow F_1 \cup \{i\}$					
11	$\mathcal{P}_S \leftarrow \mathcal{P}_S \cup \{F_1\};$	<pre>// Approximate Pareto set</pre>				

Specifically, the rank of Row 1 is incremented for every other row it is dominated by. Similarly, the rank of Row 2 gets incremented for every other row it is dominated by, except itself and the rows before it. This is repeated until the last row, which is given a default rank of zero. All rows that have a rank of zero represent non-dominated solutions (or the Pareto set) and subsequently \mathcal{F}_1 and Pareto rank $\rho_1 = 1$. The rank value of zero indicates solution dominance, *i.e.* no solution pair $\{f_1, f_2\}$ dominates any other pair in the Pareto set (\mathcal{P}_S) [20].

In the next section the properties that determine the quality of an approximated Pareto front are discussed.

2.2.2 The quality of MOO algorithm

The quality of a MOO algorithm depends on two properties that describe the approximated Pareto front, namely *convergence* and *diversity*. The goal of a MOO algorithm is to converge toward the true Pareto front, *i.e.* generate solutions near the true Pareto solutions while maintaining a diverse set of solutions, *i.e.* generate solutions that represent the entire range of the Pareto front [61, 266]. Figure 2.7a shows an approximation front having a diverse set of solutions, but does not converge to the true Pareto front. Next, Figure 2.7b shows an approximation front that is close to the true Pareto front, *i.e.* converges, however some regions are unsearched resulting in poor diversity. Lastly, Figure 2.7c shows an approximation with both desirable properties, *i.e.* good convergence and diversity. It can be seen that too much diversity disrupts the search process and too little could lead to premature convergence [142, 266].



Figure 2.7: Examples of approximation sets illustrating (a) bad convergence and good diversity, (b) good convergence and bad diversity and (c) good convergence and diversity.

An algorithm's ability to generate good quality approximations, as depicted in Figure 2.7c, is governed by its ability to efficiently *explore* the search space and effectively *exploit* prospective regions [77], *i.e.* to generate a diverse set of non-dominated solutions near the true Pareto front. Therefore, a balance between exploration and exploitation needs to be maintained [107]. Many MOO algorithms use an archive to prevent non-dominated solutions from being lost or to guide the search based on previously archived solutions [266]. Note that archived solutions are not necessarily only non-dominated solutions, for example simulated annealing (SA) [143] accepts dominated solutions aids the search algorithm in escaping local optima. The notion of archiving is discussed next.

2.2.3 Archiving

For the sake of simplicity Figure 2.8 represents a max-max BOOP in which the contents of the archive consist only of non-dominated solutions, where the Pareto set will become clear by notion of archiving. The non-dominated solutions generated are represented by \bullet and form the archive \mathcal{A} .



Figure 2.8: An example illustrating the notion of archiving for a max-max bi-objective optimisation problem, adapted from [68].

19

Solution \bullet was not dominated until solution \bullet was generated, replacing \bullet by \bullet in \mathcal{A} . Solution \bullet represent the dominated solutions that are not archived [68]. For comparison purposes, it is necessary to assess the performance of a search algorithm, a matter discussed in the next section.

2.2.4 Performance assessment of MOO algorithms

Attention is afforded to quality performance indicators because performance indicators refer to both time and quality. Simulation time is a function of the computer's processing power, *i.e.* the computer used, and the number of simulations optimised simultaneously. To elaborate, the amount of time it takes to run a single algorithm on a simulation problem differs from machine to machine, also, the simulation time increases when the number of simulation models being optimised simultaneously increases, however not linearly. For that reason time cannot be used to assess the performance of a MOSO algorithm. As a result, quality performance indicators are considered to assess the performance of a MOO algorithm in the objective space, *i.e.* in terms of the approximate Pareto set obtained [222].

Indicator approaches measure quantify the performance of an approximation set as a numerical value. Quality performance indicators can be classified as *unary* or *binary*. Unary indicators take one approximation set as input and returns a numerical value (that describes its cardinality, convergence or diversity) as output. Whereas binary indicators, take as input two approximation sets and considers the relationship between them (in terms of dominance) and returns a numerical value (that dictates which set is better) as output [147]. In the absence of a true Pareto front, a unary performance assessment methodology is required to enable the comparative analysis between approximation fronts returned by each algorithm (for a specific problem), relative to one another.

An extensive review of performance indicators conducted by Riquelme *et al.* [210] concluded that the hypervolume indicator is the most widely used quality performance indicator, and is attributable to being *Pareto compliant*, evaluation properties and ease of implementation since it does not require a true Pareto set as reference [162]. The hypervolume¹ [267, 269] is a unary quality indicator that measures both the *convergence* and *diversity* of an approximation set [166], denoted by I_H . In this paper, I_H is referred to as a *hyperarea* since the simulation problems being optimised are bi-objective. The hyperarea indicator essentially measures the portion of the objective space that is dominated by the approximation front, relative to prespecified reference point. Note that the reference point has to be dominated by the entire approximation front, *i.e.* by both objectives. Formally, the hyperarea can be defined as

$$I_H(Z_{\text{ref}}, \boldsymbol{X}) = \Lambda \left(\bigcup_{X_n \in X} \left[f_1(X_n), f_1^{\text{ref}} \right] \times \left[f_2(X_n), f_2^{\text{ref}} \right] \right),$$
(2.7)

where $I_H(Z_{\text{ref}}, X)$ represents the area covered by the approximation set X, $Z_{\text{ref}} = \{f_1^{\text{ref}}, f_2^{\text{ref}}\}$ refers to the problem specific reference point and $\Lambda(\cdot)$ refers to the Lebesque measure. Note that when conducting multiple comparisons of optimisation algorithms, the reference point must be the same, otherwise the resulting I_H indicator values are not comparable [214, 268]. A simple example is shown in Figure 2.9 with the min-max bi-objective approximation set containing seven solutions, $\mathbf{X} = \{(2, 1), (3, 3), (4, 5), (6, 5.6), (7, 6), (8, 7.5), (9.5, 8)\}$ and the reference point $Z_{\text{ref}} = (10.5, 0.5)$. The hyperarea is represented by the shaded (blue) area and corresponds to a hyperarea value $I_H = 44.53$, which is determined by summing the individual hyperareas as shown in Figure 2.9.

¹Also termed the S-metric [269], hyperarea in the case of two objectives or Lebesgue measure [85]



Figure 2.9: An example illustrating the concept of hyperarea for a *min-max* bi-objective approximation front. The blue shaded area corresponds to the value of the hyperarea indicator for this approximation front, relative to the reference point.

Notice that $A_1 - A_6$ are trapezoids and are calculated as such, whereas A_7 is simply a rectangle. A larger hyperarea value is preferred, and because the hyperarea indicator is Pareto compliant, it can be that that if the hyperarea of approximation set A is larger than approximation set B $(I_A > I_B)$, then that approximation set A is preferable to B. However, the indicator value does not provide any information with regards to the extent according to which A outperforms B, and whether or not it is significant. This matter is discussed in detail in Chapter 6. For more on quality performance indicators refer to [267, 270].

This concludes the discussion of MOO preliminaries for solving COPs, hereafter metaheuristics for solving MOOPs are discussed.

2.3 Metaheuristics

It has been shown that metaheuristics provide acceptable solutions in a reasonable amount of computational time, and are therefore considered good substitutes for exact algorithms [93, 114, 198]. Metaheuristics are classified using many criteria, as seen in Table 2.1, namely *nature* vs. *non-nature* inspired, *memory* vs. *memoryless*, *deterministic* vs. *stochastic*, *population-based* vs. *single-solution based* search and *iterative* vs. *greedy* [93]. Metaheuristics that are considered memoryless, do not use the historical information during the search process, whereas metaheuristics that use memory, such as TS, use a specific structure to take historical information into account. Moreover, deterministic algorithms generate the same final solution(s) for the same initial solution(s). Note that this characteristic influences how the performance of the metaheuristic is evaluated [93], refer to Chapter 6.

Most of the metaheuristics are iterative algorithms, in that they start with a complete solution and transform it using a set of search operators; this is performed at each iteration [93, 134]. Notice that the algorithms belonging to each family of metaheuristics, as discussed, share many search mechanisms. However, the most used classification of a metaheuristic takes into account the number of solutions transformed in each step of the iterative algorithm. For that reason, metaheuristics will be classified as either single-solution based which transform a single solution or population-based which transform a population of solutions [134]. Figure 2.10 demonstrates the complementary strengths and weaknesses of S-metaheuristics and P-metaheuristics.

2.3. Metaheuristics

Single-solution based	Inspiration	Memory	Stochastic	Iterative
Simulated Annealing (SA) [143]	Physics		\checkmark	\checkmark
Tabu Search (TS) [100]	AI	\checkmark		\checkmark
Threshold Accepting (TA)	-			\checkmark
Iterated Local Search (ILS) [170]	-		\checkmark	\checkmark
Breakout Local Search (BLS) [23]	-	\checkmark	\checkmark	
Descent-based Local Search (DLS) [265]	-		\checkmark	
Guided Local Search (GLS) [253]	-	\checkmark	\checkmark	\checkmark
Variable Neighborhood Search (VNS) [185]	-		\checkmark	\checkmark
Hill Climbing (HC) [131]	-			\checkmark
Large Neighborhood Search (LNS) [228]	-	\checkmark	\checkmark	\checkmark
Great Deluge (GD) [79]	-	\checkmark	\checkmark	\checkmark
Greedy randomised adaptive search procedure (GRASP)	-		\checkmark	
Population-based	Inspiration	Memory	Stochastic	Iterative
Genetic Algorithm (GA) [120]	Biology		\checkmark	\checkmark
Memetic Algorithm (MA) [187]	Biology	\checkmark	\checkmark	\checkmark
Differential Evolution (DE) [240]	Biology	\checkmark	\checkmark	\checkmark
Artificial Bee Colony (ABC) [133]	Bees	\checkmark	\checkmark	\checkmark
Ant Colony Optimisation (ACO) [73]	Ants	\checkmark	\checkmark	\checkmark
Particle Swarm Optimisation (PSO) [139]	Flocking birds	\checkmark	\checkmark	\checkmark
Harmony Search (HS) [91]	Music	\checkmark	\checkmark	\checkmark
Scatter Search (SS) [98, 101]	Evolution	\checkmark	\checkmark	\checkmark
Water Wave Optimisation (WWO) [264]	Wave theory		\checkmark	\checkmark
Imperialist Competitive Algorithm (ICA) [7]	Society		\checkmark	\checkmark

Table 2.1: Criteria used for the classification of metaheuristics.

P-metaheuristics are considered exploration (diversification) oriented as they are powerful in the approximation of the whole Pareto set while S-metaheuristics are efficient in exploiting (intensify) the search around the obtained approximations [196, 197, 204]. Hybrid metaheuristics combine S- and P-metaheuristics to work together [134].



Figure 2.10: The two conflicting criteria in the design of a metaheuristic, namely exploitation and exploration, adapted from [93].

Random search only *explores* the search space which means that non-improving solutions are also accepted and because promising regions are not exploited it could result in suboptimal solutions. Local search on the other hand, only *exploits* the search space which means that only improving solutions are accepted and because of the lack of exploration, promising regions may not be found and could result in premature convergence and consquently suboptimal solutions. There is a balance that needs to be maintained, called the *exploration-exploitation* tradeoff. Exploration of the search space and exploitation of promising regions which are determined by good solutions [93]. Some of the main classes of metaheuristics are listed below, with examples, some of which are regularly used or have been adapted to solve large COPs.

- Local Search Methods start with an initial solution and seeks to improve the solution by searching in the neighbourhood (or locally). When an improvement is found, the process seeks to improve the solutions, *etc.* Examples are *variable neighbourhood search* [111, 185] and *guided local search* [246, 253].
- **Simulated Annealing** resembles local search, but accepts non-improving solutions with some probability, whereby escaping the local optimas [143].
- **Tabu Search** resembles local search, but uses some memory structure to prohibit recent solutions to be revisited, TS also escapes from local optima by allowing non-improving moves [97, 100, 115, 167].
- **Evolutionary Algorithms** apply the theory of evolution to a population of solutions. Examples include genetic algorithms [120], evolutionary strategies [93], scatter search [101], genetic programming [17].
- Swarm intelligence inspired by the collective behaviour of swarms and social insect colonies [80, 77]. Examples include artificial bee colony [133], ant-colony optimisation [73] and particle swarm optimisation [139].
- Mememic Algorithms are population-based search techniques in combination with local search techniques, where information is transferred by imitation, rather than genetically [187].

Despite the individual success of metaheuristics and their widespread adoption, there is some reluctance to use them. This is attributable to each having several parameter or operator choices which directly influence their performance, but to which choice of parameters of operators to use are unclear to non-expert users [213].

Consequently, the application of *hyperheuristics* to the field of optimisation, spesifically simulation optimisation, is motivated by the above-mentioned drawbacks of metaheuristic search methodologies. Hyperheuristics mitigate, to some extent, the difficulty of characterising and designing problem-specific algorithms, by operating in the search space of metaheuristics (or heuristic components) before operating on the search space of solutions, whereas metaheuristics operate directly on the search space of solutions.

2.4 Hyperheuristics

The main motivations for research on hyperheuristics are to attempt to respond to the legitimate limitations of existing search methods, some of these limitations include [213]:

- 1. Metaheuristics cannot be applied to new optimisation problems from diffrent domains or even instances of the same problem without some changes to the algorithm, and
- 2. usually there is a large number of parameters or algorithm choices that must be defined, without guidelines for how to select them [36, 39].

Consequently, hyperheuristics have been developed to represent a class of high-level adaptive search techniques to raise the level of generality of search algorithms [36, 39]. Accordingly, the application of hyperheuristics is more appropriate to solve different problems or even instances of the same problem and does so by managing a set of low-level (meta)heuristics (LLHs).

By combining LLHs, ensemble algorithms are constructed that enable each heuristic to compensate, to some degree, for the weaknesses of others by method of *algorithmic cooperation* [142]. The result is a more effective search method that leads to a better overall performance when compared with the performance of the individual heuristics, *i.e.* hyperheuristics are adept at solving a larger range of problems [80, 36, 137]. Hyperheuristics can be classified in terms of (i) the different sources of feedback information used and (ii) the nature of the heuristic search space, as shown in Figure 2.11. Firstly, a hyperheuristic can be considered a *learning algorithm* when it uses some form of feedback *during* the search process. The use of information *during* the search process is termed *online learning*, used *after* the search process is termed *offline learning* [38]. Some studies incorporate both, called *mixed learning*, see [6]. Whenever *no learning* takes place, a random or exhaustive process dictates the selection strategy [36].

Secondly, the nature of the heuristic search space, classified as either *heuristic selection* where existing LLHs are selected or *heuristic generation* where new heuristics are generated from the components of existing LLHs [38]. The second level corresponds to the *nature* of the search structure, delineated into two types of LLHs, either *constructive* or *perturbative* [38]. Constructive heuristics incrementally build a complete solution from nothing by applying a selected LLH to partial solutions at each step. Whereas, perturbation heuristics operate on complete solutions, iteratively performing local search operations until some stopping (or termination) criterion is reached [76, 123].



Figure 2.11: Classification of hyperheuristic approaches according to two dimensions: (i) the source of feedback during learning, and (ii) the nature of a heuristic search space, adapted from [38].

MO selection hyperheuristics can either manage multiple MO metaheuristics [163, 164, 173] within a single execution, for example MOEAs, or manage the components of a single MO metaheuristic such as the *crossover* or *mutation operators* [108]. To conclude, a set of LLHs can either be a set of metaheuristics or a set of metaheuristic components. A selection hyperheuristic consists of two components: *heuristic selection* and *move acceptance* [29, 199]. In heuristic selection, given a set of perturbative LLHs, a strategy is used to *select* the most appropriate LLH to run at a certain point in the search. Thereafter, the perturbed solution(s) is either accepted or rejected based on the move acceptance strategy.

The nature of the move acceptance strategy for selection hyperheuristics is classified as *stochastic* if the accept (or reject) decision is determined with some probability, for example SA, or *deterministic*, otherwise [38]. It is important to note that improving moves are always accepted, the move acceptance strategy determines whether or not *worsening* moves are accepted [142]. Tables 2.3 and 2.2 shows some of the different strategies for move acceptance and heuristic selection, respectively, note that this is not a taxonomy of all the strategies. When applying selection hyperheuristics to population-based methods, it is worth noting that the move acceptance strategy often becomes a replacement strategy [37, 38].

Note that it is possible to apply more than one heuristic selection and or move acceptance strategy, however only one Heuristic Selection–Move Acceptance pair is considered in this paper. Selection hyperheuristics are described in terms of these two components, allowing for a high level of modularity, *i.e.* when either component is replaced by another, a new hyperheuristic is created. An instance of this will be denoted as Heuristic Selection–Move Acceptance from this point forward. A plethora of different *Heuristic Selection–Move Acceptance* combinations have been explored within the context of perturbative selection hyperheuristics, refer to [36].

It is advised to introduce some learning into the heuristic selection process, to select the next LLH intelligently and thereby aid the decision-making process [36]. Note that two approaches are available when employing perturbative selection hyperheuristics, namely single-solution based search or population-based search. The majority of perturbative selection hyperheuristic research covers single-solution based search methods [51, 108, 201]. However, research is growing for hyperheuristics using population-based search methods [76] and mixed approaches, *i.e.* using single-solution and population-based search methods [126, 161].

Deterministic methods	Description	References
All moves	Accepts all moves.	[50, 51]
Only improving	Only accepts improving moves.	[50, 51]
Improving or equal	Accepts improving or equal moves.	[50, 51]
Threshold acceptance	Accepts non-improving moves that are less than a prefixed threshold, in terms of solution quality.	[29, 138, 182]
Late acceptance	The new solution is compared to the L^{th} solution step and the decision to accept is made accordingly.	[35, 200]
Stochostic mothodo		
Stochastic methods	Description	References
Monte Carlo	Description Accepts non-improving moves with a probability that decreases in relation to the decrease in objective function value.	References [9]
Monte Carlo Great Deluge	Description Accepts non-improving moves with a probability that decreases in relation to the decrease in objective function value. Accepts a move as long as it is no worse than an expected objective value that changes linearly at each step.	References [9] [29, 138]
Monte Carlo Great Deluge Tabu search	Description Accepts non-improving moves with a probability that decreases in relation to the decrease in objective function value. Accepts a move as long as it is no worse than an expected objective value that changes linearly at each step. Accepts the move if it is not on the tabu list.	References [9] [29, 138] [42]

 Table 2.2:
 Move acceptance strategies in perturbation selection hyperheuristics.

The majority selection hyperheuristics generate an online score based on each LLHs performance, and based on the selection strategy chosen, the scores are processed and/or combined in a systematic manner to select the heuristic to be applied to the candidate solution or population of solutions at each step. A *score-based* hyperheuristic framework comprises of the following components: (i) an initial score, (ii) memory length adjustment, (iii) heuristic selection strategy based on the scores, (iv) score update rules in case of improvement of solution quality and (v) score update rules in case of degradation of solution quality, respectively [37].

The first component, the *initial score* is assigned to all the LLHs, the initial score is typically zero. The *memory length adjustment* controls how the previous LLHs performance influences the selection strategy. Note that the choice of heuristic selection strategy may directly affect the performance of the hyperheuristic [37]. For example, the *max* heuristic selection strategy selects the heuristic with the maximum score, the *Roulette wheel* selects an LLH based on its probability which corresponds to its performance [36]. The *choice function* (CF) is an example of the max strategy and maintains a utility score for each LLH, the heuristic with the maximum score is selected at each iteration. Another example and commonly used method is *reinforcement learning* (RL), for more detail, refer to [132, 242].

Max strategy	Description	References
Greedy	Exhaustively applies all LLHs and selects the LLH that produced the best solution.	[50, 51]
Choice function	Adaptively ranks the LLHs with respect to a combined score on its indi- vidual performance and selects the LLH with the best score.	[50, 51]
Roulette-wheel strategy	Description	References
Simple random	Randomly selects a LLH at each step.	[50, 51]
Random descent	A LLH is selected at random and applied until no improvement is made.	[50, 51]
Random permutation	Generates a random order in which a LLH is selected at each step.	[50, 51]
Random permutation descent	Generates a random order in which a LLH is selected but does so when no improvement is made.	[50, 51]
Reinforcement learning	Tries to learn which LLH to select at which step and is based on scores or cumulative reward.	[40, 74, 183]

Table 2.3: Heuristic selection strategies in perturbation selection hyperheuristics.

RL heuristic selection selects an LLH by method of trial-and-error, the system attempts to learn which actions result in reward and which actions result in penalty. To elaborate, if an LLH improves a non-dominated solution, then it is rewarded and its score increases, otherwise it is punished and its score decreases [36]. Empirical evidence shows that different combinations of Heuristic Selection–Move Acceptance components yield different performances [29, 199]. The results show that All Moves (AM) perform the worst, regardless of the selection strategy. Experiments have shown that the move acceptance strategy significantly affects the performance of the hyperheuristic as compared with heuristic selection. The most used and successfully implemented heuristic selection methods include the choice function and the reinforcement learning variants. For move acceptance methods, the best is SA and great deluge variants. Experiments have shown that the choice of move acceptance methods significantly affects the performance of the hyperheuristics, more so than the choice of heuristic selection methods [142].

Different frameworks for population-based searches have also been proposed, for example *a* multi-algorithm genetically adaptive multi-objective (AMALGAM) approach [254]. AMALGAM applies a set of LLHs simultaneously, where the number of solutions that each LLH may produce is proportional to its individual past performance, *i.e.* the percentage of previously created solutions that are non-dominated [38, 76]. An enhanced AMALGAM method was proposed for population evolution during the optimisation process [140]. Later AMALGAM was applied to single-solution based searches, proposed in [128]. So far, this chapter has discussed simulation and its optimisation approaches, MOO preliminaries, metaheuristics and hyperheuristics. The next section discusses artificial neural networks for the purpose of simulation metamodeling.

2.5 Simulation metamodeling

The DES models studied in this paper are complex, *i.e.* they have multiple sources of *randomness* that require substantial computational effort to simulate. The computational inefficiency of simulation motivated the development of *metamodels* in order to enhance the overall effectiveness of the simulation optimisation process [5]. Using metamodels in combination with simulation reduces the computationally expensive (time-intensive) nature of simulation by searching the solution space more *intelligently* and less *exhaustively* [5]. The computational time (or simulation time) required by a simulation model depends greatly on the complexity of the model and may range from minutes to hours and even weeks.

The more complex the system, the harder it is to model, resulting in lengthy computational requirements [248], which is particularly true for complex COPs [18, 248, 258]. Despite the potential of SO, the excessive computational time required is a limiting factor, and reducing the model complexity is not an option. Because even if an optimal solution is found, it is for a model that does do not adequately represent the real system [156]. To this end, metamodels are proposed to aid in the simulation optimisation procedure and reduce the computational time required, by acting as a filter for potentially *low-quality* solutions, thereby mitigating computationally expensive evaluations of the simulation model [5, 156].

As discussed previously, and as seen in Figure 2.3, metamodels require less computational time than simulations because they are deterministic rather than stochastic and can be used in combination with simulation so as to leverage the trade-off between the time, cost, risk and quality [109], thereby simplifying the simulation optimisation process. Simulation metamodels offer a trade-off between *accuracy* and *efficiency*, by trying to predict the *input-output* relationships inherent to the simulation model, thereby offering significant advantages regarding computational efficiency [87]. For an example, see the development of the OptQuest tool [99].

Metamodels can provide a *fast* decision support that aids the overall effectiveness of the simulation optimisation process [109, 219]. Note that the inputs (or decision variables) of the simulation model are also the inputs of the metamodel, and the outputs (or objective function values) of the simulation model are used to train the metamodel [109]. There are several techniques for simulation metamodelling, each having advantages, disadvantages and applicable domains [18, 248]. The most frequently used techniques in recent literatures include *artificial neural networks* (ANNs) [86, 87], *genetic programming* (GP), *multivariate adaptive regression splines* (MARS) [88], *spatial correlation* or *Kriging* (KG) [144, 145, 146, 220], *radial basis functions* (RBF) [178, 190, 230] and *support vector regression* (SVR) [44].

Publications combining ANNs and simulation are on the rise, the reader is referred to [41, 86, 155, 191, 248], either for simulation metamodeling purposes [248] or as part of the simulation optimisation process which is part of simulation software such as *OptQuest*, *Optimiz* and *Simul8* [3, 5], however not yet in *Tecnomatix* [16]. The next section discusses the pertinent literature related to the fundamentals of ANNs, mainly to introduce concepts and the notations used throughout this paper.

2.5.1 The fundamentals of artificial neural networks

ANNs are mathematical models inspired by the generally accepted notion of how the brain *learns from experience* by *processing information* enabling machines to learn in an automated fashion and is the fundamental underpinning of the domain of *machine learning* (ML) [106]. To elaborate, ANNs learn from training data which consist of a set of input values and, in the case of supervised learning, its corresponding output values. The network is able to map the input-output relationship intrinsic to the data and does so in an iterated manner. Mitchell *et al.* [184] provides a succinct definition:

"A computer program is said to learn from experience E with respect to some class of tasks T and performance measure P if its performance at tasks in T, as measured by P, improves with experience E."

Learning from experience is defined as the method of attaining the ability to perform a task, and can be accomplished in a supervised, semi-supervised, unsupervised or reinforced manner, depending on the experience E.

Unsupervised learning requires a model to learn from inputs what the underlying function of the dataset is without having the corresponding outputs to compare to (or learn from). Semisupervised learning combines supervised and unsupervised learning, *i.e.* some of the inputs include a corresponding output to learn from while others do not. Reinforcement learning (RL) is different to the other machine learning paradigms as it follows a trail-and-error search approach and delays reward. RL attempts to learn what the best action is that will maximise the cumulative reward, the reader is referred to [242].

The two main supervised learning tasks that ANNs can perform are classification and regression and some applications include clustering, function approximation and optimisation [179]. Given some input, a classification task would predict the category to which some input belongs, whereas a regression task would predict some numerical value [106]. To evaluate the performance of an ML algorithm, a quantitative measure of its performance P is required and is dependent on the task T being performed and is discussed later. This study focusses on function approximation for regression tasks. The focus of this study is on regression tasks T and supervised learning experiences E using ANNs.

ANNs are generally classified into three main categories, exhibiting noteworthy differences, recurrent neural networks (RNNs), convolutional neural networks (CNNs) and Feedforward neural networks (FNNs). RNNs use memory in the form of feedback connections, where information is transmitted to and from preceding layers, that enables the network to process and analyse sequential data and perform tasks such as speech recognition with exceptional performance [262]. CNNs perform tasks such as pattern-and-image classification by taking as input an image and encoding specific features into the network architecture that allows the network to learn from the pixels and make predictions [81, 106]. For a systematic introduction see [211]. FNNs (or multi-layer perceptrons (MLPs)) allow data to be transmited thought the network, strictly in a forward direction, *i.e.* no feedback connections exist [106].

The three basic elements of an ANN are illustrated by the perceptron in Figure 2.12 [153]:

- 1. A set of weighted connections, also called synapses. Specifically, when referring to Figure 2.12, input signal x_i is connected (or multiplied) by weighted connection w_i .
- 2. The weighted sum z, sums the input signals multiplied by their respective weighted connections and adding a bias, denoted by

$$z = \sum_{i=1}^{n} x_i w_i + b,$$
 (2.8)

where b is the bias associated with each neuron and n is the number of inputs. The purpose of incorporating this bias value is to shift the activation function either to the left or right, adjusting the neuron's inherent threshold.

3. An activation function, denoted by $\sigma(z)$, mathematically models the firing process of a biological neuron, *i.e.* a node only relays information if the weighted sum of that node exceeds some inherent threshold, called the activation of the node.

A biological brain comprises of a network of *biological neurons*, therefore neurons are the core building blocks of ANNs as they try to mimic how the brain functions by controlling a network of artificial neurons, referred to as a node from this point on [153]. As shown in Figure 2.12, a neuron receives inputs $\{x_1, \ldots, x_i, \ldots, x_n\}$, weighted with $\{w_1, \ldots, w_i, \ldots, w_n\}$ and a bias b, respectively. The sum of the weighted connections $z = \sum_{i=1}^n x_i w_i + b$ then produces the predicted output $\hat{y} = \sigma(z)$ of the neuron, according to the activation function σ .



Figure 2.12: Mathematical model of an artificial neuron and its processing capabilities.

The output represents the transmitted signal strength of the neuron which is influenced by the weighted connections and bias. Unfortunately, a perceptron is unable to learn from data that are not linearly separable, which is the case for the logical operation exclusive or (XOR). However, two perceptrons combined could. ANNs may be defined by the topology in which the neurons are connected, termed the *network architecture*, the training algorithm used and activation function applied [84]. These components and other relevant information are discussed in the following sections.

2.5.2 The network architecture

In terms of network architecture, an ANNs can be classified as a *single-layer* or *multi-layer network*. A neural network may be described in terms of layers, where the layers consist of neurons, *i.e.* neurons are partitioned into different layers as depicted in Figure 2.13, the first layer is called the *input layer* and the neurons in that layer input neurons, the last layer is called the *output layer* and the neurons. The layers in between are called *hidden layers* and the neurons hidden neurons. Typically, the same activation function is applied to neurons in the same layer.

Single-layer networks, similar to the network presented in Figure 2.12, do not have hidden layers. Multi-layer networks consist of an input layer, at least one hidden layer and an output layer and is capable of learning in respect of linearly separable data [84]. An ANN with more than one hidden layer is referred to as a *deep neural network* (DNN) [192]. Note that there are other conventions being followed, as seen in [192].

In FNNs, signals (or information) are propagated from one neuron to the next in a forward direction through the network layers, starting at the input layer moving through the hidden layers towards the output layer [84]. Consequently, the architecture of a *fully-connected* (or interconnected) ANN is explained using an FNN with one hidden layer, as depicted in Figure 2.13. The network has n input neurons each representing an input, h hidden neurons with a bias term in each layer and m output neurons each representing an output. The bracketed superscripts denote the relevant layers, adopting the convention that the input layer is layer one, hidden layer is layer two and output layer is layer 3. It should be noted that the neurons from one layer are connected to the neurons in the subsequent layer, however, neurons in the same layer are not connected to one another.



Figure 2.13: Illustration of the notation used for a FNN with one hidden layer.

Furthermore, Figure 2.13 demonstrates the notation adopted throughout this study. The weights are numbered sequentially, starting at the first neuron in the input layer to the first hidden neuron in the hidden layer. Specifically, let $w_{ji}^{(1)}$ denote the weighted connection associated with the input neuron *i* and the hidden neuron *j*, and let $b_j^{(1)}$ denote the bias for hidden neuron *j* and let $w_{kj}^{(2)}$ denote the weighted connection associated with the hidden neuron *k*, and let $b_k^{(2)}$ denote the bias for output neuron *k*. The biases are represented as adjustable weights from additional inputs with a value of one [30].

The main components of the FNN with one hidden layer presented in Figure 2.13 can be summarised as:

- Input layer activations, denoted by $\boldsymbol{a}^{(1)} = [a_1^{(1)}, \dots, a_i^{(1)}, \dots, a_n^{(1)}]^T$,
- hidden layer activations, denoted by $\boldsymbol{a}^{(2)} = [a_1^{(2)}, \dots, a_j^{(2)}, \dots, a_h^{(2)}]^T$,
- output layer activations, denoted by $\boldsymbol{a}^{(3)} = [a_1^{(3)}, \dots, a_k^{(3)}, \dots, a_m^{(3)}]^T$,
- weights corresponding to the connection between input neuron $i \in \{1, \ldots, n\}$ and hidden neuron $j \in \{1, \ldots, h\}$, denoted by $w_{ji}^{(1)}$ and contained within a weight-matrix $\boldsymbol{W}^{(1)}$ and bias denoted by $b_{j}^{(1)}$,
- weights corresponding to the connection between hidden neuron $j \in \{1, \ldots, h\}$ and output neuron $k \in \{1, \ldots, m\}$, denoted by $w_{kj}^{(2)}$ and contained within a weight-matrix $\boldsymbol{W}^{(2)}$ and bias denoted by $b_k^{(2)}$,
- input layer activation functions are excluded as the identity function is used, the hidden layer activation functions are denoted by $\sigma^{(1)}(\cdot)$ and the output layer activation functions are denoted by $\sigma^{(2)}(\cdot)$.

The process followed to determine the predicted output \hat{y}_k is termed forward propagation. Given the aforementioned information, the FNN with one hidden layer can be expressed analytically in terms of a mathematical function. Recall that the activation function used in the input layer is the identity function, therefore $\sigma(x_i) = a_i^{(1)} = 1(x_i)$, *i.e.* the input signal is simply transmitted onwards, as is. The input layer activations are subsequently used to calculate the net input (or weighted sum) to hidden neuron j, denoted by $z(j)^{(2)}$ for $j \in \{1, \ldots, h\}$.

The net input is given by

$$z_j^{(2)} = \sum_{i=1}^n w_{ji}^{(1)} a_i^{(1)} + bj^{(1)}.$$
(2.9)

The matrix expression that defines the net input for all the hidden nodes may be written as

$\begin{bmatrix} w_{11}^{(1)} \\ w_{j1}^{(1)} \end{bmatrix}$	$w_{1i}^{(1)} \\ w_{ji}^{(1)}$	· · · ·	$\begin{bmatrix} w_{1n}^{(1)} \\ w_{hi}^{(1)} \end{bmatrix}$	$\begin{bmatrix} a_1^{(1)} \\ a_i^{(1)} \end{bmatrix}$	$\begin{bmatrix} b_1^{(1)} \\ b_j^{(1)} \end{bmatrix}_{-}$	$\begin{bmatrix} z_1^{(2)} \\ z_j^{(2)} \end{bmatrix}$
$\begin{bmatrix} \vdots \\ w_{h1}^{(1)} \end{bmatrix}$	$\vdots \\ w_{hi}^{(1)}$	۰۰.	$\begin{bmatrix} \vdots \\ w_{hn}^{(1)} \end{bmatrix}$	$\begin{bmatrix} \vdots \\ a_n^{(1)} \end{bmatrix}$	$\begin{bmatrix} \vdots \\ b_h^{(1)} \end{bmatrix} =$	$\begin{bmatrix} \vdots \\ z_h^{(2)} \end{bmatrix},$

where \odot is the *Hadamard product* (or element-wise multiplication). Next, the activation of hidden neuron j can be calculated by applying the activation function $\sigma^{(1)}(\cdot)$ to the weighted sum calculated in (2.9), such that

$$a_j^{(2)} = a\left(z_j^{(2)}\right) = \sigma^{(1)}\left(z_j^{(2)}\right).$$
(2.10)

The matrix expression that defines the activation for all the hidden nodes may be written as

$$\sigma^{(1)} \begin{pmatrix} z_1^{(2)} \\ z_j^{(2)} \\ \vdots \\ z_h^{(2)} \end{pmatrix} = \begin{bmatrix} a_1^{(2)} \\ a_j^{(2)} \\ \vdots \\ a_h^{(2)} \end{bmatrix}.$$

The calculations for calculating the weighted sum and activation of output node k is similar to that of hidden neuron j, and is given as

$$z_k^{(3)} = \sum_{j=1}^h w_{kj}^{(2)} a_j^{(2)} + b_k^{(2)}, \qquad (2.11)$$

where h is the number of hidden nodes. The matrix expression that defines the weighted sum for all the output nodes may be written as

$$\begin{bmatrix} w_{11}^{(2)} & w_{1j}^{(2)} & \dots & w_{1n}^{(2)} \\ w_{k1}^{(2)} & w_{kj}^{(2)} & \dots & w_{kn}^{(2)} \\ \vdots & \vdots & \ddots & \vdots \\ w_{m1}^{(2)} & w_{mj}^{(2)} & \dots & w_{mn}^{(2)} \end{bmatrix} \odot \begin{bmatrix} a_1^{(2)} \\ a_j^{(2)} \\ \vdots \\ a_k^{(2)} \end{bmatrix} + \begin{bmatrix} b_1^{(2)} \\ b_k^{(2)} \\ \vdots \\ b_m^{(2)} \end{bmatrix} = \begin{bmatrix} z_1^{(3)} \\ z_k^{(3)} \\ \vdots \\ z_m^{(3)} \end{bmatrix} .$$

Next, the activation of output neuron k can be calculated by applying the activation function $\sigma^{(2)}(\cdot)$ to the weighted sum calculated in (2.11), such that

$$a_k^{(3)} = a\left(z_k^{(3)}\right) = \sigma^{(2)}\left(z_k^{(3)}\right), \qquad (2.12)$$

where $a_k^{(3)}$ represents the output \hat{y}_k . The matrix expression that defines the activation for all the output nodes may be written as

$$\sigma^{(2)} \begin{pmatrix} z_1^{(3)} \\ z_k^{(3)} \\ \vdots \\ z_m^{(3)} \end{pmatrix} = \begin{bmatrix} a_1^{(3)} \\ a_k^{(3)} \\ \vdots \\ a_m^{(3)} \end{bmatrix}.$$

The FNN is simply a non-linear function which is controlled by adjusting the network weights and biases [30]. The weights and biases that represent the underlying input-output relationship of the data are determined by a *training* (or *learning algorithm*) according to which network weights and biases are updated in an iterative and algorithmic fashion.

2.5.3 The network training algorithms

Recall that supervised learning is adopted in this study, *i.e.* the ANN is presented with input– output pairs from which the network adjusts its weights and biases so as to best approximate the functional mapping from inputs to outputs. There are three approaches that may be adopted to train an ANN, namely *first-order optimisation algorithms*, *second-order optimisation algorithms* and *metaheuristics* [106]. This section focuses on first-order algorithmic approaches as second-order approaches are rendered intractable as the problem complexity and cardinality increases [106]. More on this in Chapter 3. The backpropagation training algorithm introduced a computationally efficient method for minimising the the error function that measures the prediction error of the network. The derivatives of the error function with respect to the network weights and biases are used to determine what changes to the weights and biases would result in the largest decrease in network error [30, 83, 194]. There are four steps involved during backpropagation training of neural networks [83, 194]:

- 1. Forward propagate an input training vector (or training example) through the network, as shown in the previous section using (2.9)-(2.12).
- 2. Calculate the network errors using

$$E = \frac{1}{2} \sum_{k=1}^{m} (a_k - y_k)^2, \qquad (2.13)$$

where *m* is the number of output nodes, and the network error is the sum of the individual output errors denoted by $E = \frac{1}{2} \sum_{k=1}^{m} E_k$. Recall that $a_k = \hat{y}_k$, *i.e.* the predicted output.

- 3. Determine the derivatives of the error function with respect to the network weights and biases.
- 4. Adjust the network weights and biases according to the chosen learning $rate^2$.

 $^{^{2}}$ A hyperparameter which controls the *step size* used to minimise the error function, *i.e.* speed of learning.

During forward propagation, a training example is transmitted through the network. The activation of the output layer is the predicted output for the given training example. Next, the network error may be calculated by comparing the predicted output (or response) a_k with the actual output (or target value) y_k using (2.13). Now the derivatives of the error function with respect to the network weights and biases can be calculated and the necessary adjustments to the weights and biases can be made. The derivation for the backpropagation algorithm is presented within the context of the FNN with one hidden layer, as depicted in Figure 2.13 in Appendix C.

The adjustments to the weights and biases can be performed by stochastic or batch update. Stochastic (or online) updates occur for each training example, whereas when batch updates are performed, the error accumulates and one adjustment is made after n training examples, where n represents the batch size. Stochastic updates have the advantage of escaping local minima, however batch updates enable the network to learn faster because the adjustments can typically be more considerable. The weights regulate how much an input influences the predicted output and the biases can have an influence on feature importance. The goal of training the network is to find the weights and biases that result in the predicted output matching the actual output [134, 188].

When training a neural network, only a percentage of the entire dataset is used, typically 80% of the dataset which is commonly referred to as the *training set*. After training the network, the remaining 20% of the dataset is used to measure the network performance, which is called the *test set*. When the network exhibits good performance when measured on the training set, but poor performance when measured on the testing set, the network is deemed to overfit the data, *i.e.* it memorises well but generalises poorly [30]. Consequently, performance with respect to the test set is of paramount importance, *i.e.* how well the network was able to generalise to new (or unseen) data. Note that the performance measure used depends on the task, in this case regression. This matter is discussed in detail in Chapter 3.

It is also considered good practice to have a validation set which is used for regularisation purposes and to adjust certain parameters of the network, called the hyperparameters (e.g. learning rate). Accordingly, the dataset is partitioned into the training set, validation set and test set, typically with a proportion of 60%:20%:20%, as seen in Figure 2.14. Note that it is also common practice to partition the dataset into only a training and test set and then partition the training dataset into a training and validation set, *i.e.* a 80%:20% partition and then the 80% is split again to form 60%:20%, *i.e.* a quarter of the training dataset becomes the validation set [34, 154, 212, 218, 227].



Figure 2.14: Illustration of a dataset partitioned into a training, validation and test set respectively, adapted from [227].

It should be noted that during training, only the parameters, *i.e.* the weights and biases, are typically optimised, not the hyperparameters. Hyperparameters may be adjusted and has been reported to considerably influence the performance of the network [30].

The hyperparameters typically include the number of hidden neurons per hidden layer, the number of hidden layers, the activation function used in the hidden layers and output layer, the training algorithm and its learning rate, see Chapter 3. According to Fausett [83], when using a gradient-based training algorithm, the activation function should be *continuous* and *differentiable*, this is because the derivative of the activation function is required to determine the partial derivatives of the activation of a neuron with respect to the weighted sum of the neuron which in turn is required to determine the derivative of the error function with respect to the weights and biases. Some of the activation functions used in literature are discussed next.

2.5.4 Activation functions

The choice of activation function (or *transfer function*) is a critical part of the network architecture design, this is because the activation function can greatly affect the performance of the network. Different activation functions can be used in different layers. A caveat when choosing an activation function for both the hidden and output layers. When choosing an activation function for the hidden layer, the nature of the input variables has to be considered, since it controls the networks ability to learn from the inputs. The choice of activation function for the nature of the output variables, since it determines what predictions (of the outputs) the network can make [83].

Activation functions essentially model the firing process of biological neurons mathematically, which can be a *linear* or *non-linear* function. However, non-linear functions exhibit the same level of functionality in a network with a single hidden layer as linear activation functions do in a network with multiple hidden layers [192]. Moreover, the non-linear nature of the activation functions is what empowers neural networks with respect to their nonlinear capabilities [159]. As such, non-linear activation functions are discussed in this section and will include the form, mathematical expression, the advantages and disadvantages, as well as suitable applications of each. The two simplest activation functions are the *identity* and the *heavside* (or *binary step*) function, as shown in Figure 2.15a and 2.15b, respectively. The mathematical expression for the identity function is given as $\sigma(z) = 1(z)$, for all weighted sums z. The heaveside function is given by

$$\sigma(z) = \begin{cases} 0, & z < 0, \\ 1, & z \ge 0. \end{cases}$$
(2.14)



Figure 2.15: Illustration of the two simplest activation functions (a) the identity function and (b) the heavside function, as given in (2.14).

Recall, that the activation used in the output layer is dependent on the data type of the output variable that the model must predict. Similar to the activation function chosen for the output layer being dependent on the data type of the output variable, so is the activation function chosen for the hidden layers [195]. The nature of the inputs and outputs must be considered. For example, for binary inputs, an appropriate activation function could the *logistic sigmoid* or the *hyperbolic tangent* function [83].

The sigmoid function can be seen in Figure 2.16a, where the logistic sigmoid function has a slope of $\alpha = 1$. These functions take any real number and map it to a probability between 0 and 1 depending on the threshold chosen. Suppose the threshold is set to 0.5, then a probability smaller than 0.5 would be mapped to the outcome 0 (*i.e.* false), otherwise the outcome would be 1 (*i.e.* true). The sigmoid function is one of the most common non-linear activation functions [81, 106], given by (2.15). The slope parameter is a *hyperparameter*, and may be adjusted to influence the performance of the network, or may be learnt in a trail-and-error manner. Figure 2.16 illustrates the sigmoid function for $\alpha = \{1, 1.5, 3\}$,

$$\sigma(z) = \frac{1}{1 + e^{-\alpha z}},$$
(2.15)

and its derivative is given by

$$\sigma'(z) = \alpha \left(\frac{1}{1+e^{-\alpha z}}\right) \left(1 - \frac{1}{1+e^{-\alpha z}}\right), \qquad (2.16)$$

which can be written as $\sigma'(z) = \alpha \ (\sigma(z)(1 - \sigma(z)))$ for the sake of brevity. It is clear by the 'S' shape of the hyperbolic tangent function that it is a member of the class of sigmoid functions, illustrated in Figure 2.16b. Unfortunately, both functions in Figure 2.16 are insensitive to small changes in the inputs and saturate when the network weights are either too large or too small. The result is slow gradient descent, *i.e.* slow learning [116].



Figure 2.16: The class of sigmoid activation functions.

The mathematical expression of the hyperbolic tangent function is given by

$$\sigma(z) = \tanh(z), \tag{2.17}$$

and its derivative is given by

$$\sigma'(z) = (1 + \tanh(z))(1 - \tanh(z)), \qquad (2.18)$$

which can be written as $\sigma'(z) = (1 + \sigma(z))(1 - \sigma(z))$, for the sake of brevity.

It has been reported that the performance of FNNs can be improved by employing piecewise linear activation functions, such as rectified linear units (ReLUs), in the hidden layers rather than sigmoid or hyperbolic tangent functions [106]. ReLU activation function was first proposed for *restricted Boltzmann machines* (RBMs) [81, 189] and then successfully applied to neural networks in [96]. It can be seen that the ReLU activation function is a combination of Figure 2.15b (for negative values) and Figure 2.15a (for positive values) and is reported to perform better than the sigmoid and hyperbolic tangent functions [151].

The application of the ReLUs in DNNs have been reported to deliver promising results [96], however because all negative values are zero, neurons may be made inactive prematurely. This phenomenon is referred to as the 'dying ReLU' problem. The parametric ReLU (PReLU), leaky ReLU (LReLU) and the exponential linear unit (ELU) or Scaled ELU (SELU) are alternatives that permit small negative values by having some slope in the negative region, thereby preventing the issue [172]. Figure 2.17 is the graphical representation of the different activation functions, and are discussed accordingly.

PReLU has a hyperparameter *alpha* (α) that controls the shape of the negative region of the function and can be trained and improve the networks performance. PReLU function achieves a notable performance advantage in terms of speed of convergence of the gradient-descent training algorithm when compared with the sigmoid, hyperbolic tangent functions[135], and with negligible additional computational cost [112]. Note that ReLU is the PReLU function, where $\alpha = 0$. The mathematical expression of PReLU is given by

$$\sigma(z) = \begin{cases} \alpha z, & z < 0, \\ z, & z \ge 0. \end{cases}$$
(2.19)

LReLU is the PReLU function, where $\alpha = 0.01$, *i.e.* LReLU has a predetermined slope whereas PReLU learns the slope. LReLU replaces the negative region of the ReLU function with a linear function and has exhibited superior performance with respect to to ReLU [172]. ELU was first proposed by Clevert *et al.* [45], designed to combine the advantages of ReLU and LReLU — to avoid the so-called dying ReLU problem. ELU is similar to the LReLU, having a small slope for negative values, however an exponential function controls the negative region as opposed to a straight line. The mathematical expression of the ELU (or SELU) function is given by



Figure 2.17: The different ReLU activation functions, for $\alpha \in \{0, 0.01, 0.25\}$, and ELU.

Given a sufficient number of hidden neurons in the hidden layer(s), MFNNs are considered universal function approximators, capable of approximating function arbitrarily [125].

Note that more hidden layers and/or hidden neurons do not directly translate to a better function fit. It does, however, result in a more complex network, a matter discussed in greater detail later. Supporting this, Fausett [84] suggests that one hidden layer is sufficient for most function approximations. Moreover, the universal approximation theorem [56, 124] support the use of FNNs as simulation metamodels, as it states that an FNN with a single hidden layer containing a finite number of neurons is sufficient to approximate any measurable function. However, the theorem does not give any information about the network parameterisation and requires a preliminary study to determine its feasibility. Intuitively, more training examples coincide with improved network performance, here overfitting is not a consideration since the motivation behind a metamodel is to reduce the computational time and therefore the goal is to build a metamodel with limited training and be able to approximate the simulation models with a certain level of accuracy [248].

2.6 Summary

This chapter contained a review of the most relevant and pertinent literature pertaining to simulation optimisation methods for MOO as well as ANNs as metamodels. The reader was presented with the necessary information to facilitate an understanding of the remainder of the research reported in this study.

First, the need for simulation optimisation was discussed as well as methods for doing so. Next, the MOO preliminaries required for solving MOOPs were documented. Metaheuristics and hyperheuristics were discussed next. Lastly, ANNs as metamodels were discussed as they are considered good candidates for metamodelling purposes, specifically FNN with one hidden layer. The next chapter reports the preliminary study conducted to determine the feasibility of an FNN with one hidden layer as a metamodel.
CHAPTER 3

ANN as Metamodel Pilot Study

3.1	Machin	ne learning models: Regression	7
	3.1.1	Linear regression	7
	3.1.2	Polynomial regression	8
	3.1.3	Support vector regression	9
	3.1.4	Decision tree regression	9
	3.1.5	Random forest regression	0
3.2	Perform	mance measures for assessing regression models	1
3.3	Hyper	parameter optimisation $\ldots \ldots 42$	2
	3.3.1	ANN hyperparameters	3
	3.3.2	An introduction to generalisation techniques 48	9
3.4	Traini	ng, validating and testing the regression models	0
3.5	Conclu	ısion: Chapter 3	3

The previous chapter introduced the fundamental concepts that will be used and build-upon throughout this study.

The main objective of this chapter is to determine whether or not an artificial neural network (ANN) as metamodel is a feasible solution to enhance the simulation optimisation process, in fulfilment of Objective 2 as stated in Chapter 1. The first section introduces the supervised learning regression models that are used as a benchmark for comparison. Next, the performance metrics used to train the regression models as well as to quantify their performances are discussed. Thereafter, the hyperparameters considered for the hyperparameter search space are discussed. The performance of the respective regression models are presented, followed by the performance results of the best hyperparameters for the ANN. Penultimately, the regression models (including the ANN) are compared and a conclusion is made whether or not an ANN as metamodel is feasible.

3.1 Machine learning models: Regression

It is necessary to compare the ANN with other regression models, specifically other supervised learning regression models, which include *linear regression* (LR) [2], *polynomial linear regression* (PLR) [113], *support vector regression* (SVR) [234], *decision tree regression* (DTR) [205], and *random forest regression* (RFR) [55, 168]. The following sections provide a high-level overview of the theory of each regressor.

3.1.1 Linear regression

Linear regression approximates the linear relationship between the independent variable (or decision variable) and the dependent variable (or objective variable). However, more than one

independent variable is of interest, *i.e.* multiple linear regression (MLR) and is formulated as [2]

$$\hat{y} = \beta_0 + \beta_1 x_1 + \ldots + \beta_n x_n, \tag{3.1}$$

where \hat{y} is the dependent variable (to be predicted), β_0 is the *y*-intercept and x_1, \ldots, x_n are the independent variables with their corresponding coefficients β_1, \ldots, β_n that the regressor is solving for. Moreover, the coefficients (β_n) is the effect that x_n has on the predicted value *y*. For the sake of completeness, in this study, \hat{y} is a vector of two elements, *i.e.* two objective variables each having their own linear relationship with the decision variables.

The goal of MLR, as visually presented in Figure 3.1, is to find the line that minimises the distance between the actual value y and the predicted value \hat{y} [81], given by $\sum (y - \hat{y})^2$.



Figure 3.1: An illustration of multiple linear regression.

A caveat, however, is that the assumptions of linear regression which include linearity, homoscedasticity, multivariate normality, independence of errors and lack of multi-collinearity [2] are not guaranteed, the model will simply perform poorly if the relationships embedded within the data are non-linear [81]. Polynomial regression is discussed next.

3.1.2 Polynomial regression

Polynomial regression, a special case of linear regression, however, can approximate non-linear relationships by method of an n^{th} degree polynomial. The function \hat{y} can be expressed as a linear combination of the coefficients. The n^{th} degree polynomial for one decision variable is given by [113]

$$\hat{y} = \beta_0 + \beta_1 x + \beta_2 x^2 + \ldots + \beta_n x^n,$$
 (3.2)

where β_0, \ldots, β_n represents the coefficients of x, not the x variable, that the regressor is solving for. Figure 3.2 illustrates how polynomial regression works by fitting a 1^{st} , 2^{nd} and 3^{rd} degree polynomial to the data.



Figure 3.2: Illustration of polynomial regression fitting data.

3.1.3 Support vector regression

Support vector regression is based on statistical learning theory [251] used to find the decision boundary between groups of data. Future data is predicted based on where they are in terms of the boundary [234]. For a more in-depth overview of SVR, the reader is referred to [176, 223, 8]. The formulation for SVR is given by [49]

$$\frac{1}{2}||w||^2 + C\sum_{i=1}^m (\xi_i + \xi_i^*) \to \min,$$
(3.3)

where $||w||^2$ is the l2-norm of the coefficient vector and C is a hyperparameter that determines the trade-off between the flatness of the function f and the tolerance for solutions outside ε that are allowed. The slack variables (ξ) contribute to the cost, if the solution is above the ε -insensitive tube, the solution is ξ otherwise, it is ξ^* .

Figure 3.3 depicts this graphically, only the points outside the tube, *i.e.* support vectors ξ_1^* , ξ_2 , ξ_3^* , ξ_4^* , ξ_5 , ξ_6^* , ξ_7 , ξ_8^* , ξ_9 , contribute to the cost and therefore determine how the tube is created and consequently the flexibility of the model.



Figure 3.3: An illustration of support vector regression, adapted from [223].

For the sake of completeness, note that the boundary can only be found for linearly separable data, otherwise SVR has to be extended by either mapping the data to a higher dimension, thereby rendering the data linearly separable, or by applying a kernel to the data [81, 234].

3.1.4 Decision tree regression

Decision tree regression performs predictions by learning decision rules inferred from the data. For example, in Figure 3.4, a dataset is divided into smaller subsets, which create the associated decision tree. The subsets are determined by splitting the data. The splits are determined by the information entropy, *i.e.* a split is made if it results in an increase in the amount of information about the dataset [205].

Figure 3.4a is used to aid in illustrating the working of decision tree regression. How and where the splits are made is determined by the algorithm, but for demonstration purposes assume the algorithm determined the splits as shown in Figure 3.4a.

Split 1 is x_1 at 20, splitting the data for the first time. Therefore, the first decision is $x_1 < 20$ as seen in Figure 3.4b. Next, the algorithm makes a second split, Split 2 x_2 at 170, creating the second decision, where $x_2 < 170$, and so on, until Split 4. In this example, Split 4 was the last split which means no more information could be added (or gained) by splitting a leaf again.

The final split resulted in five terminal leaves. The associated decision tree based on the splits is given in Figure 3.4b.



Figure 3.4: An illustration to aid in explaining how decision tree regression works, adapted from [81].

For example, the predicted value of y where $x_1 = 30$ and $x_2 = 50$ is determined as follows. It can be seen that the point $\{30, 50\}$ falls within the red terminal leaf depicted by the black data point. The predicted value for y is the average of the red terminal leaf, *i.e.* y = -64.1.

3.1.5 Random forest regression

Random forest regression applies ensemble learning¹ to regression trees, *i.e.* it creates a number of regression trees (N), and averages the prediction of all the trees. The y prediction is based on N trees (or a whole forest of trees), whereby improving the accuracy of the predictions because an average is taken of N trees (or predictions) [81]. The random forest regression algorithm is given in Algorithm 3.1.

Algorithm 3.1: Random Forest Regression [81]						
Input : Training set and testing set, the number of trees you want to build, denoted by N_{tree} .						
Output : Predicts the output y based on the input x						
1 repeat						
2 Pick at random K data points from the training set;						
3 Build the Decision Tree associated to these K data points;						
4 until Number of trees built = N_{tree} ;						
5 for New data point x do						
6 foreach Tree N do						
7 Predict y_N for tree N						
$x \leftarrow \frac{\sum_{i=1}^{N} y_i}{N}.$						

This concludes the high-level overview of the regression models used in this study. It is sufficient to know how to use the algorithms at a high-level of abstraction, since each regressor was implemented in Python using existing libraries [81].

¹When multiple algorithms or a single algorithm is used multiple times, to create a more powerful algorithm.

Additionally, because of its popularity, extreme gradient boosting (XGBoost) is also used for comparison purposes. XGBoost is an open-source implementation of the gradient boosted trees algorithm. For more detail the reader is referred to [43]. The next section discusses the performance measures that are used to train the regression models as well as quantify their performances.

3.2 Performance measures for assessing regression models

There are many performance metrics (within the context of regression problems) that describe a model's capability to learn from data (or so-called goodness of fit), some of which are discussed here [239].

Mean Squared Error (MSE) takes the square difference between the actual value and the predicted value. Therefore, when the difference is great, the squared difference is even greater, emphasising the error and can be expressed as [188]

$$MSE = \frac{1}{Q} \sum_{k=1}^{m} (y_k - \hat{y}_k)^2, \qquad (3.4)$$

where Q is the number of training examples, y_k is the target output value, \hat{y}_k the predicted value and m the number of outputs. MSE has been extended to root Mean Square Error (RMSE), which measures the standard deviation of the error between the predicted and the actual value, indicating the overall approximation capability of the regressor and can be expressed as [188]

RMSE =
$$\sqrt{\frac{1}{Q} \sum_{k=1}^{m} (y_k - \hat{y}_k)^2}$$
. (3.5)

The smaller the RMSE, the better the approximation.

Mean Absolute Error (MAE) represents the average of the absolute differences between the predicted and actual value and can be expressed as [188]

MAE =
$$\frac{1}{Q} \sum_{k=1}^{m} |y_k - \hat{y}_k|.$$
 (3.6)

Coefficient of Determination (R^2) is a statistical measure used to explain the correlation between variables and can be expressed as

$$R^{2} = 1 - \frac{\sum_{k=1}^{m} (\hat{y}_{k} - y_{k})^{2}}{\sum_{k=1}^{m} (\bar{y}_{k} - y_{k})^{2}},$$
(3.7)

where \bar{y}_k is the average. Note that R^2 will continue to increase as the number of parameters increase, therefore an adjusted R^2 is proposed which adds a penalty term to penalise complex models, because greater complexity does not necessarily translate to improved performance [188]. Adjusted R^2 is given by

$$R_{\rm adj}^2 = 1 - (1 - R^2) \frac{Q - 1}{Q - n - 1},$$
(3.8)

where n is the number of decision variables. If $R^2 = 1$, the variables are perfectly correlated. Each measure provides some indication of the prediction error made by the model and may be interpreted as the cost of predicting. If a batch learning approach is adopted, then the error is calculated across the Q input-output training examples that constitute the batch. If an online learning approach is followed, on the other hand, then the summation is omitted [188].

The next section discusses the hyperparameter optimisation performed to determine the best hyperparameter combination for the FNN with one hidden layer.

3.3 Hyperparameter optimisation

In the context of ANNs, hyperparameters are parameters of the neural network that are set prior to the training process, that is, prior to determining the weights and biases [33]. Determining the best hyperparameter combination, however, is not a trivial task and requires some form of optimisation. There are several hyperparameter optimisation techniques, Table 3.1 shows the high-level overview of the different techniques for sequential and parallel computing. Parallel algorithms are listed for the sake of completeness, however only sequential algorithms are considered. For in-depth discussions the reader is referred to [25, 26, 235]. Grid search exhaustively

Table 3.1:	A	summary	of	hyperparamet	ter	optimisation	algorithms.
------------	---	---------	----	--------------	-----	--------------	-------------

	Random	Adaptive	Evolutionary
Sequential Parallel	Grid search or random search Asynchronous successive halv-	Bayesian optimisation Bayesian optimisation with	Genetic algorithm Population-based
	ing algorithm (ASHA)	hyperband (BOHB)	training

evaluates a model for each hyperparameter combination specified, whereas random search samples each hyperparameter combination from a distribution, over all possible parameter values, and is likely to discover better combinations (faster) than grid search due to its exploration of the parameter space [25]. Bayesian optimisation samples regions that result in a better network performance with a higher likelihood. Initially, Bayesian optimisation resembles random search, but as the search progresses the algorithm uses information inferred from past trials to guide the search. It has been shown that Bayesian optimisation is able to find better hyperparameter combinations than random search [26].

Consequently, Bayesian optimisation is used to determine the best hyperparameter combination to use for the FNN with one hidden layer. Figure 3.5a shows one function evaluation of what the actual function g is at that point, next a Gaussian process is fit to this one training example and the grey shaded area represents the approximate function of g for one observations, it should be clear that the more function evaluations are performed the more the approximated functions mirrors g. Next, an acquisition function is used to determine the next point to evaluate g, with a high probability of getting a lower function value (for a minimisation problem). It can be seen in Figure 3.5h that after only eight function evaluations, the Gaussian process is a relatively good approximation of g.

Due to the time-consuming process of selecting and evaluating hyperparameter combinations, with the goal of finding the best combination [33], the next section discusses the most appropriate design choices that are considered in the hyperparameter optimisation search space.

42



(g) Seven function evaluations (h) Eight function evaluations

Figure 3.5: An example illustrating how Bayesian optimisation works, by fitting a Gaussian process for each training example.

3.3.1 ANN hyperparameters

The Nobel prize winning physicist *Enrico Fermi* said:

"I remember my friend Johnny von Neumann used to say, with four parameters I can fit an elephant, and with five I can make him wiggle his trunk."

The point is that models with many parameters might be able to fit to the training data, but overfitting does not make it a good model. For example, a 3^{rd} degree polynomial overfits to the data presented in Figure 3.2. Formally, the capability of the network to fit the training data is known as *memorisation*, which means that the model will be able to predict values for the existing data, but will fail to generalise to unseen data, known as *generalisation* [195].

Therefore, the main goal of an ANN is to be able to generalise to unseen data, *i.e.* make predictions based on data it has not been trained on. Consequently, a trade-off exists between the competing abilities of memorisation and generalisation [153].

Network architecture and training epochs

In Figure 3.6d and 3.6e, it can be seen that the network's generalisation error is the gap between the training error and validation error. The generalisation error is characterised into two terms: bias and variance. Bias measures the expected deviation between the predicted and the actual value, while variance measures how much the network predicted output varies between datasets. To elaborate, when a model performs poorly on the training set but well on the validation set (or test set), the model is biased. Alternatively, when a model fits the training set well, but not the validation set (or test set), the model is said to have high variance. Both cases are undesirable, because the result is the same: poor predictions in respect of out-of-sample data [30].



(d) An illustration of how the model's complexity influence the models generalisation capabilities



(e) An illustration of how the number of epochs influence the models generalisation capabilities

Figure 3.6: An illustration of the curve fitting phenomena that occurs when the ANN is too complex, or that is trained for too long. These may result in a learning model that either underfits or overfits the training set, resulting in large prediction errors, *i.e.* lack of generalisation in respect of the validation set, adapted from [222].

The bias-variance tradeoff is strongly associated with the concepts of *capacity* (or complexity), *underfitting* and *overfitting*. Informally, models with low capacity (for example an ANN with a small number of hidden neurons) may struggle to fit to the training set, while models with high capacity (for example an ANN with a large number of hidden neurons) can overfit by memorising properties of the training set. Generally, increasing capacity and number of epochs that the network is trained for tends to increase variance and decrease bias [30]. It can be shown that the minimum network error is when the sum of the bias and variance are minimal, illustrated by the red dashed lines in Figure 3.6d and 3.6e for the network capacity and number of epochs trained respectively.

To elucidate this further, refer to Figure 3.6d. On the left side of the *optimal capacity* (red dashed line), the training and validation errors are high and the model is underfit (or biased). This is represented by the underfitting zone (a) in Figure 3.6d and the result is a poor function fit as seen in Figure 3.6a. However, as the capacity increases, the training error decreases, but the validation error increases and as a result, the generalisation error increases. On the right of the optimal capacity, the model is considered overfit (or has high variance). This is represented by the overfitting zone (c) in Figure 3.6d, also resulting in a poor function fit as seen in Figure 3.6c. The optimal capacity is where the smallest generalisation error exists, *i.e.* the point where the validation error starts increasing while the training error gradually decreases [30, 179].

Early on in the training process, the networks predictions are far from the actual output and therefore the bias is large, this can be seen in zone (a) of Figure 3.6e, the underfitting zone. As training progresses, the bias decreases as the network learns the underlying function, however, when the network is trained for too many epochs, the network may learn the noise present in the dataset it was trained on causing the variance to increase, represented by zone (c) of Figure 3.6e the overfitting zone [151]. From this, two important questions arise: During the construction of an FNN (with one hidden layer), how many neurons should be used and for how many epochs should the model be trained for? Where the goal is to create a model that can generalise well to out-of-sample data. It is suggested that the required number of neurons depends on many factors, namely, training set size, problem complexity and the generalisation techniques used to name but a few [244].

Deciding on the *right* number of hidden neurons is clearly problem dependent and requires expert knowledge to infer. Essentially, it must be determined empirically, serving as motivation for its inclusion in the ANNs hyperparameter optimisation. Next, the learning in literature and suggested rates are discussed.

Learning algorithms and learning rates

The ideal learning algorithm has low bias and can accurately model the underlying input-tooutput relationship found by finding the trade-off between underfitting and overfitting. Gradient descent is the standard technique used for training ANNs. There are three variants of gradient descent, which are defined based on the amount of data used to compute the gradient of the error function and accordingly the frequency with which the weights and biases are updated. Again, a trade-off exists, between the accuracy of the weight and bias updates and the computational time required to perform the updates [194].

Figure 3.7 is used to elucidate this behaviour. Assume that there are 10 training examples in the training set. *Batch learning* is where the weights and biases are only updated after *all* or a *subset* of the training examples in the training set have been propagated through to the network [194]. When the whole training set is used, it is termed *batch gradient descent*, whereas, when a subset of training set is used, it is termed *mini-batch gradient descent*.

In Figure 3.7, the mini-batch size is two. During *online learning* the weights and biases are updated after each training example, termed *stochastic gradient descent* (SGD). It is termed stochastic since the examples are selected at random, rather than using the whole dataset [194].



Figure 3.7: An illustration of online, mini-batch and batch learning.

Whenever the weights and biases are updated, an iteration is incremented, whether it was updated after a single training example (online learning) or a batch of training examples (batch learning). Moreover, when the entire training set has been evaluated by the training algorithm, an *epoch* is incremented. For example, to elucidate these two definitions further, take the previous example of 10 training examples. If online learning is employed, then when 10 iterations pass an epoch is completed. On the other hand, when batch learning is employed with a batch size of 2, then an epoch will be incremented after 5 iterations have been completed. When batch learning is employed, where the batch size is equal to the number of training examples, then one iteration is equal to one epoch.

Stochastic gradient descent maintains a single learning rate for all weight updates and the learning rate does not change during training, *i.e.* adaptively. For this reason, some gradient descent optimisation algorithms have been proposed, where the learning rates are adapted during the search. The algorithms include:

- 1. Nesterov momentum [208, 241], was developed to help accelerate SGD in the relevant direction.
- 2. Nesterov accelerated gradient (NAG) [193], was developed to improve upon the momentum term, instead of blindly following the slope, Nesterov evaluates gradients using the previous velocity.
- 3. Adaptive gradient algorithm (AdaGrad) [78] improves upon NAG and is an algorithm that adapts the learning rate to the parameters.
- 4. Adadelta [263] is an extension of Adagrad that seeks to reduce its radically decreasing learning rates, by adapting the updates for each individual parameter.
- 5. Root mean square propagation (RMSprop), proposed by Geoffrey Hinton [116, 245], similar to Adadelta was developed to resolve Adagrad's radically diminishing learning rates.
- 6. Adaptive moment estimation (Adam) [141] is a combination of RMSprop and momentum, and compares favourably to other adaptive learning-method algorithms.
- 7. AdaMax [141] adapts the update rule in Adam.
- 8. Nesterov-accelerated Adaptive Moment Estimation (Nadam) [75], is a combination of Adam and NAG, Adam is essentially RMSprop with momentum, Nadam is Adam with Nesterov momentum.

SGD is included in the hyperparameter search space of the ANN, and because of their simplicity, computational efficiency, relatively low memory requirements and combined advantages [106], Adam and Nadam are also included in the ANN's hyperparameter search space. The learning rate determines how fast the model learns during training or the amount that the weights are updated during training, also referred as the *step size*. Therefore, the learning rate affects both the learning speed as well as the convergence of the model. In Figure 3.8, an illustration is presented of the effect of the learning rate on the networks capability to learn. Note that smaller learning rates require more training epochs as the 'steps' are smaller, whereas larger learning rates require fewer training epochs as a result of the large 'steps' taken to update the weights [106].

Figure 3.8a illustrates the effect of employing a small learning rate, training takes long and it is possible that the algorithm can get stuck at a local optimum because the 'step' is too small. Figure 3.8c illustrates the effect of employing a high learning rate, which can cause premature convergence to suboptimal values for the weights. The learning rate that results in the fastest convergence is η_{opt} , as seen in Figure 3.8b. The largest learning rate that can be used without causing divergence is $\eta_{max} = 2\eta_{opt}$ [159], if $\eta > 2\eta_{opt}$ the algorithm will diverge and high-quality weights will not be found, as illustrated in Figure 3.8d.

According to Goodfellow *et al.* [106], the learning rate might be the most important hyperparameter, serving as motivation for its inclusion in the hyperparameter search space. Note that a single learning rate is considered for the weights.



Figure 3.8: Gradient descent for different learning rates, adapted from [159].

Next, methods for network weight initialisation are discussed.

48

Network weight initialisation

Humans tend to learn faster when they make mistakes, however, the same does not hold true for neural networks. For example: Take a network with one input to a node. Suppose the goal was to train the single input neuron to take the input 1 and change it to the output 0. However, trivial this task may seem, it will help to explain the *learning slowdown phenomenon* that occurs when the initial weight value is too large, and is therefore worth computing the weight w and bias b using gradient descent.

- **Experiment 1** The first experiment uses the parameters and hyperparameter (the learning rate) as specified in Figure 3.9a, which leads to an initial output of 0.82 which is far from the desired 0.0. Figure 3.9a shows how the neuron rapidly learns the weight and bias that minimises the error and produces an output of 0.09 which is much closer to 0.0.
- **Experiment 2** The second experiment uses the parameters and hyperparameter as specified in Figure 3.9b, which leads to an initial output of 0.98 which is even further from the desired 0.0, although they use the same learning rates. Figure 3.9b shows how the neuron learns markedly slower, in fact, at the beginning the weight and bias barely change. Then, as learning kicks in the neuron's output rapidly moves closer to 0.0. The learned weight and bias minimises the error and gives an output of 0.20.

To summarise, Experiment 1 performs better and yields a better prediction for the output than Experiment 2 and is attributable to the chosen initial weight and bias. Based on the experiments and the results it is clear that the artificial neuron encounters considerable difficulty learning when the initial weight and bias is large, *i.e.* either too large or too small. Therefore, it can be concluded that smaller initialised values for w and b are preferred. This learning slowdown phenomenon occurs in larger networks too [195]. Consequently, weight and bias initialisation is included in the hyperparameter search space.



Figure 3.9: The learning slowdown phenomenon, illustrating the effect of weight initialisation and the chosen activation function.

The training algorithm is essentially an optimisation algorithm which requires a starting point in the space of possible network weight values from which the optimisation (learning or training) of the neural network can begin. Consequently, before training (or learning) can transpire, the network weights must be initialised. This is an important design choice, since the initial weight values have a substantial effect on the *convergence speed* and *quality of solutions* obtained during the training process [159, 106], as discussed previously.

3.3. Hyperparameter optimisation

A too-large initialisation leads to exploding gradients, resulting in the error to oscillate around the minimum value (or even diverge). Conversely, too-small initialisations lead to vanishing gradients, resulting in convergence of the error before it has reached the minimum value [106]. The basic convention is to initialise all the weights to values randomly drawn from a *uniform* or *Gaussian probability distribution*. Weight initialisation methods use information about the network, such as the activation function (and its associated derivative) and the number of weighted connections. The heuristic initialisation methods provide the prospects of a more effective optimisation process and became the *de facto* standard [106].

The following weight initialisation procedures are suggested based on their prolific use in literature.

- 1. The Sigmoid activation function requires a modification, since the function saturates for markedly small/large values of s, as shown previously in Figure 2.16(a), given by (2.15). The modification enables the linear region to dictate the range of values from which the weights can be drawn.
 - LeCun *et al.* [159] suggested that the weights be drawn from some probability distribution, with a mean of zero and a standard deviation of $\sqrt{\frac{1}{m}}$, where *m* is the number of weighted connections entering the neuron (or fan-in number).
 - Xavier Glorot and Yoshua Bengio [95] proposed the normalised 'xavier' or 'glorot' initialisation method, where the weights are drawn from the uniform probability distribution referred to as GlorotUniform, given by

$$w = \mathcal{U}\left[-\frac{\sqrt{6}}{\sqrt{n+m}}, \frac{\sqrt{6}}{\sqrt{n+m}}\right],\tag{3.9}$$

where n is the number of outgoing weighted connections (or fan-out number). Another initialisation method was proposed using the normal distribution, referred to as GlorotNormal.

2. When employing the PReLU activation function, as previously shown in Figure 2.17, He *et al.* [112] recommended the '*he*' initialisation method. Where the weight *w* is calculated as a random number from the *Gaussian probability distribution*, referred to as HeNormal, where the weights have a mean of zero and a standard deviation of $\sqrt{\frac{2}{m}}$. Another initialisation method was proposed using the uniform distribution, referred to as HeUniform.

Other than the number of hidden neurons and the number of training epochs, there are some other techniques that can be used to improve the generalisation capabilities of the regression model, some of which are discussed next.

3.3.2 An introduction to generalisation techniques

Network generalisation is achieved by controlling the capacity of the network. Goodfellow *et al.* [106] defines regularisation as any improvement to the learning algorithm to 'reduce its generalisation error but not its training error.' The most relevant generalisation strategies found in literature are *parameter norm penalisation*, known as L1, L2 and *elastic-net regularisation*, *batch normalisation*, *early stopping*, *dropout* and *cross-validation* [30]. Essentially, these techniques control the complexity of the network in one way or another to avoid the behaviour observed in the overfitting zone (c), depicted in Figures 3.6d and 3.6e [30].

L1, and L2 attempt to control model complexity by the addition of a *penalty term* to the error function. By adding a small amount of bias, a smoother input-to-output mapping is encouraged [30, 106, 152, 243]. Elastic-net combines L1 and L2 regularisers. According to Goodfellow *et al.* [106], early stopping is commonly used attributable to its effectiveness and simplicity. The premise of this strategy is as follows: During training, the validation error is monitored which can provide an indication for when the network is overfitting, as seen in Figure 3.6d and 3.6e. Initially the training and validation error decreases, but at some point, at the optimal number of epochs, the validation error starts to increase. The standard procedure is to stop the training process (with some patience) if the validation error does not improve. Patience is the number of epochs that the validation error is permitted to increase before training is stopped and is set to 15 in this study [106].

The dropout regularisation technique involves randomly shutting down (or dropping) hidden neurons in each iteration, however, is not considered in this study. Batch normalisation recentres and re-scales the activations of each input variable, for each mini-batch, by normalising them to have a mean of zero and a standard deviation of one [31]. Batch normalisation is said to stabilise the learning process and dramatically reduce the amount of training required [106]. It is important to note that not all the hyperparameters previously discussed form part of the hyperparameter search space. This is because hyperparameter optimisation is a time-consuming process due to the combinatorial relationship that exists.

The next section presents the results obtained during model training and testing for the respective regression models as well as delineates the methodology followed to determine the best hyperparameters for the FNN with one hidden layer.

3.4 Training, validating and testing the regression models

If a model is trained and evaluated on the same dataset then the evaluation metric produced is known as *training loss* (or error). Unfortunately, this leads to a model that has overfit to the data and is unlikely to generalise to out-of-sample data. Alternatively, the train-test split can be used, where the dataset is split into two sets, the training and test set. The model is trained on the training set and evaluated on the test set and results in a performance metric known as the *testing loss*. However, the testing loss is a high variance estimate of the out-of-sample data and is markedly sensitive to how the data is split, *i.e.* if you split the data with a different random number seed you are likely to obtain a different testing loss than previously obtained.

As mentioned, cross-validation can be used to train a machine learning model and then estimate how well a model is likely to perform on out-of-sample data. There are various ways to perform cross-validation, however, the method employed in this study is k-fold cross-validation, where k refers to the number of groups that the data set is split into. The training set is partitioned into k groups, as illustrated in Figure 3.10, for k = 10, *i.e. 10-fold CV*. The test set is held separate for the final evaluation of the ML model, while the remaining data forms the training set and is split into k groups (or folds). At each iteration, one of the k folds are used as the validation set, while the other k - 1 remaining folds are used for training.

By training and testing the ML model k times on different subsets of the training data and averaging the scores of each iteration allows for a more accurate representation of the model's generalisation capability, compared to one training and one validation set, a sort of ensemble method. The regression models were built following the proposed machine learning pipeline presented in Figure 3.11.



Figure 3.10: An illustration of how the dataset is split for k-fold cross validation.

The data collection phase consisted of performing exhaustive enumeration for the (s, S) inventory problem (IP) and store the data as inputs and corresponding outputs as illustrated in Table 3.2, which provides a snapshot of the first five examples from 250 000 training examples. The reorder point (ROP) and reorder quantities (ROQ) represent the decision variable values (or inputs) and the total inventory cost (TIC) and service level (SL) the objectives (or outputs).



Figure 3.11: The proposed machine learning pipeline.

Table 3.2: An example of the data in the IP dataset.

	ROP	ROQ	TIC	SL
1	20	20	7686.53	16.80
2	20	30	7394.59	18.79
3	20	40	7261.08	20.80
4	20	50	7154.16	22.66
5	20	60	7074.46	24.55

Assume the necessary libraries have been imported. In the data preparation phase, the dataset is imported, then partitioned into the training and test set. Accordingly, the training set consists of 200 000 training examples and the test set of 50 000. The validation set is then 25% of the training set, *i.e.* 12 500 training examples. After the data is split, feature scaling is applied, *i.e.* normalisation of the data to ensure all values are on the same scale. This prevents one feature (or variable) from dominating another [81]. Next, build the regressors (as described in the Python script presented in Listing A.1), then train the regressor on the training set using 10-fold CV, measuring the MSE, MAE, as discussed previously, on the training, validation and test set. The results for the 30 trails are presented in Tables A.1–A.5 for each regression model.

The stochastic nature of the regression models necessitates a sample of algorithmic performances, therefore the performance measure scores presented in Table 3.3 were calculated by performing 30 trials (using 30 different random number seeds to produce 30 different dataset splits) of the learning model and then averaging the individual scores for each performance measure.

The Python script demonstrating this for MLR is given in Listing A.2. The best performing algorithm is RFR, which achieved the lowest training, validation and test scores in terms of MSE and MAE, and the worst performing algorithm is MLR.

 Table 3.3: The performance of the respective regressors in terms of MSE and MAE for the IP dataset.

	Training		Valid	ation	Te	Test		
MSE		MAE	MSE	MAE	MSE	MAE		
MLR	0.148	0.311	0.148	0.311	0.148	0.311		
SVM	0.003	0.044	0.003	0.044	0.003	0.044		
DTR	0	0	0	0.003	0	0.002		
RFR	0	0.001	0	0.002	0	0.002		
XGBoost	0	0.007	0	0.007	0	0.007		

As discussed previously, ANNs have many hyperparameters that can be adjusted, however, it requires expert knowledge because the hyperparameter combination is dataset specific (problem specific). The hyperparameter search space considered for optimisation is presented in Table 3.4, resulting in 12 960 possible combinations, further motivating the use of Bayesian optimisation. Recall that only one hidden layer is considered, as supported by the literature presented in Chapter 2.

 Table 3.4:
 The proposed hyperparameters search space for the FNN with one hidden layer.

(a) Network h	yperparameters		(b) Algorithm	m hyperparameters
Neural netwo	rk search space	_	Algorith	m search space
Number of neurons Activation function Weight initialisation	2, 4,, 20 Sigmoid, ReLU, ELU HeNormal, HeUni- form, GlorotNormal, GlorotUniform	_	Learning rate Batch size	0.01, 0.001, 0.0001 512, 1028
Batch normalisationYes, NoTraining algorithmSGD, Adam, NadarKernal regulariserL1, L2, L1L2				

The results obtained by employing Bayesian optimisation (by running the Python code in Listing A.3) is presented in Table 3.5, for a batch size of 512 and 1028 and with or without batch normalisation. For example, for a batch size of 512 training examples and employing batch normalisation, the optimal hyperparameters, for the FNN with one hidden layer, are Nadam with a learning rate of 0.01, L_2 regularisation, the ELU activation function, HeUniform weight initialisation and 12 hidden neurons, which result in a MSE of 0.052.

The network has to be trained again, this time with the hyperparameter combinations found during Bayesian optimisation, as listed in Table 3.5. Once the model training is completed the generalisation capabilities of the network can be evaluated on the test set. Again, the stochastic nature of the regression model necessitates a sample of algorithmic performances, therefore the FNN is run for 30 independent trials, presented in Tables A.6–A.9 for the hyperparameter combinations A-D, as specified in Table 3.5. Note that each trial is run with a different random number seed in order to generate 30 different dataset splits. The performance of the model is then represented by the average of the 30 trials, as described by the Python code in Listing A.4. The performance measure scores are tabulated in Table 3.6, where the metrics represent the average for both objectives for all 30 trials.

Convention	A	В	C	D
Batch size	512	512	1028	1028
Batch normalisation	\checkmark	×	\checkmark	×
Hyperparameters				
Training algorithm	Nadam	Adam	Nadam	Nadam
Learning rate	0.01	0.001	0.001	0.0001
Regulariser	L2	L2	L2	L2
Activation	ELU	ELU	ELU	ELU
Initialiser	HeUniform	HeNormal	HeNormal	HeUniform
Units	12	14	14	20
Score:	0.052	0.052	0.052	0.055

Table 3.5: The best hyperparameters obtained for the FNN, with one hidden layer, during Bayesian optimisation.

According to Table 3.6, the best performance (in terms of the smallest MSE and MAE for the training, validation and test set) corresponds to hyperparameter combination B, *i.e.* a batch size of 512 without employing batch normalisation, employing the Adam optimiser with a learning rate of 0.001, L_2 regularisation, ELU as activation function, HeNormal as weight initialisation method and with 14 hidden neurons.

Table 3.6: The average performance of the respective hyperparameter combinations, as shown in Table 3.5, in terms of MSE and MAE for the IP dataset.

	Batch	Batch	Training set		Validation set			Test set	
_	size	normalisation	MSE	MAE	MSE	MAE	_	MSE	MAE
А	512	\checkmark	0.0532	0.1466	0.0532	0.1466		0.0669	0.1474
В	512	×	0.0492	0.1319	0.0492	0.1319		0.0494	0.132
\mathbf{C}	1028	\checkmark	0.0511	0.1391	0.0511	0.1391		0.0508	0.1376
D	1028	×	0.0599	0.1506	0.0599	0.0151		0.0601	0.1508

3.5 Conclusion: Chapter 3

The main objective of this chapter was to demonstrate whether or not an ANN as metamodel would be a feasible solution to enhance the simulation optimisation process. Table 3.7 summarises the performances of each regression model in terms of MSE and MAE. Overall, RFR performed the best and MLR the worst. The FNN with one hidden layer (with a batch size of 512 without employing batch normalisation) performed better than MLR. This chapter provided an in-depth inverstigation into the workings of an ANN, considering the network structures, network initialisation methods, learning algorithms and generalisation techniques, providing valuable insight into the complex interactions within an ANN. Therefore, this pilot study is considered *novel* and can be used as a starting point for further experimentation. This pilot study proposed to demonstrate the feasibility of an ANN metamodel, however, from the results in Table 3.7 it is not clear whether or not an ANN metamodel adds sufficient value.

Further experimentation is required regarding the following:

1. At what stage during training can the network start predicting whether or not solutions should be evaluated?

	MSE	MAE
MLR	0.1471	0.3101
SVM	0.0026	0.0438
DTR	0	0.002
RFR	0	0.0015
XGBoost	0.0001	0.0071
ANN	0.0494	0.132

Table 3.7: Comparison of the performances of the respective regression models, in terms of MSE and MAE for the test set for the IP dataset.

- 2. What is the optimal batch size? Consider for example a mini-batch size of 16, 32 and 64.
- 3. Consider the regularisation parameter λ for L1, L2 and L1L2 regularisation which controls the trade-off between minimising the error function and keeping the network parameters (*i.e.* weights and biases) small [195].
- 4. Consider dropout regularlisation and the dropout rate.

However, this is out of the scope of this study but should be considered for future work. The next chapter presents the simulation models studied as well as some statistical prerequisites required for simulation output analysis.

CHAPTER 4

Simulation Models and Statistical Prerequisites

4.1	Statist	ical analysis preliminaries	55
4.2	The si	mulation model problems	57
	4.2.1	The Open mine problem	59
	4.2.2	The (s, S) inventory problem $\ldots \ldots \ldots$	62
	4.2.3	The buffer-allocation problem: five machines $\ldots \ldots \ldots \ldots \ldots \ldots$	63
	4.2.4	The buffer-allocation problem: 10 machines	65
	4.2.5	The non-linear buffer-allocation problem: 16 machines	67
4.3	Conclu	sion: Chapter 4	69

Previously, Chapter 3 reported on the preliminary analysis conducted to determine whether or not the application of an ANN as a metamodel is a feasible solution to enhance the simulation optimisation process.

This chapter serves to fulfil Objective 3, as stated in Chapter 1. This chapter introduces some of the statistical aspects with regards to simulation output and each reports the problem-specific simulation parameters studied. Firstly, the statistical prerequisites necessary for simulation output analysis are discussed. Thereafter, the respective simulation models are discussed in terms of their decision variables (inputs) and objectives (or outputs) as well as their boundaries. Next, the sufficient number of observations per solution is determined scientifically and because R&S is out of the scope of this study, analysis-of-variance (ANOVA) is used to determine whether 100 observation per solutions are sufficient for each problem. Penultimately, a description of the complexity of each simulation model is presented as well as its true Pareto front. Finally, the chapter closes with a summary of its contents.

4.1 Statistical analysis preliminaries

The simulation of stochastic systems is simply *statistical sampling* and therefore relies on statistical inferences to be able to analyse and draw conclusions from the results. Stochastic systems or systems containing stochastic elements (or randomness) respond stochastically, *i.e.* the output(s) obtained for the same set of input(s) will differ for different runs, the reader is referred to [136] for further reading.

Consequently, several observations per solution is required to describe the output variable in terms of its point and interval estimators to then analyse and draw conclusions from. The analysis of the output depends on the type of system (terminating) analysed. For terminating systems, every termination (or simulation time) results in one observation for the output variable(s) being studied. Each observation is assumed to be statistically independent from other observations during the same simulation run [20, 157].

CHAPTER 4. SIMULATION MODELS AND STATISTICAL PREREQUISITES

For example, a simple deterministic system is given in Figure 4.1a. A customer arrives and leaves every minute, with a constant service time of 30s. If the system is changed to stochastic, as shown in Figure 4.1b, it is no longer possible to predict exit time. Assume the simulation time is from 07:00–17:00, where the mean of all the exit times observed, from 07:00–17:00, represent one observation. Suppose 100 days are observed, then the mean of the mean observed exit times are taken and represent the point estimator (x) for the output, *i.e.* 100 observations per solution was observed. A sufficient number of arrivals are required, to draw a conclusion from the unknown exit parameter x.



Figure 4.1: Illustration of (a) a simple deterministic and (b) a simple stochastic system, adapted from [20].

Formally, it is necessary to determine the number of observations per solution required to be 95% confident that the mean value x, is equal to the true but unknown population mean, where the confidence interval (CI) specifies the range in which it is to be expected. The output (or estimated value) generated by a stochastic system may be observed as numerically different, however they may not be *statistically significantly different* [20].

For example, $\bar{X}_1 = 22.4$ and $\bar{X}_2 = 23.5$ differ numerically, however, it is necessary to determine statistically whether they differ statistically significantly or not. To elaborate, Figures 4.2a and 4.2b are used. Figure 4.2a shows samples without a common mean and cross-sample variation, *i.e.* X, Y and Z are different and Figure 4.2b shows samples with a common mean and in-sample variation, *i.e.* X, Y and Z are considered similar.



 $\begin{array}{c|c} & & & \times Z \\ \hline 2 & 3 & 4 & 5 & 6 \\ \hline \text{(b) Samples with a common mean and high} \end{array}$

 μ_z

 $\bullet X$ $\diamond Y$

 μv

 μ_X

in-sample variability

(a) Samples without a common mean and high cross-sample variability



(c) Samples with different distribution characteristics

Figure 4.2: Illustration of the difference between (a) cross-sample, (b) in-sample variation and (c) distributions having these characteristics, adapted from [20].

Figures 4.2a and 4.2b can also be explained considering Figure 4.2c, where distribution X and Y are from the same population that exhibit in-sample variability whereas distribution Z is from a different population that exhibits cross-sample variation between X, Y versus Z. As stated in the project scope, it is assumed that all models have been built correctly for the purpose of validating the proposed solutions. Also, unless stated otherwise the simulation time for each model is *five simulation days*, denoted by 05:00:00:00. The simulation problems studied as well as other relevant information are discussed next. For a formal description of each, refer to Appendix D.

4.2 The simulation model problems

To reiterate, due to the stochastic nature inherent to simulation models, small-sample theory must be applied to deal with the observations generated by the simulation model. Typically, the expected values for the objectives are described by *point* and *interval estimators* which require, as mentioned, several observations per solution [157]. The more observations per solution, the smaller the CI becomes, *i.e.* less variance of the output distribution, which leads to a *good* approximation (or point estimator) of the output parameter(s) being studied [20].

However, this requires more simulation time to complete, *i.e.* the computational time may become considerable [28, 82]. For example, assume one observation per solution takes 0.1 seconds to evaluate and 100 observations per solution are required, then 10 seconds are required to evaluate one decision variable combination. Recall that earch algorithms evaluate many solutions during the search process, assuming 1 000 decision variable combinations are evaluated then the simulation optimisation process would take 2.77 hours.

Upon considering the above-mentioned, it can be derived that a good search algorithm should find good quality approximation sets and do so by the least number of function evaluations, *i.e.* find good solutions fast. The amount of time it takes to evaluate a single solution varies from simulation model to simulation model and is attributable to the complexity of the model. There are four elements that influence the complexity of a simulation model, namely the number of objectives, number of decision variables, number of stochastic elements, and size of the search space (or combinatorial nature).

Accordingly, all the problems studied differ in terms of their complexities and are reported in the following sections, respectively. In this study, the complexity of the simulation problems differ by the number of decision variables, stochastic elements and the search space cardinality, not influenced by the number of objectives since all the problems in this study are bi-objective simulation optimisation problems.

After careful consideration of the problem context, a sensible upper bound for each problem was chosen. The upper bounds were introduced, because when going beyond a certain value, a point of diminishing return is reached, also it is used as a method to contain the problem. The BAP with five machines (BAP5) is used to illustrate this. Table D.4 lists the experiments used for the hypothesis test to determine the upper bound for BAP5, and is summarised in Table 4.1.

Table 4.1: Summary of the experiments (in the columns) used to determine a sensible upper bound as given in Table D.4.

Groups	Count	Sum	Average	Variance
Work-in-progress	9	21.72	2.41	0.02
Throughput	9	830.09	92.23	0.03

CHAPTER 4. SIMULATION MODELS AND STATISTICAL PREREQUISITES

The possible outcomes of the hypothesis test is, either the data supports the research question stated as the *null hypothesis* H_0 (which is assumed to be true) or disproves the research prediction stated as the *alternate hypothesis* H_1 [71]. Specifically, the null hypothesis states that the means do not differ and consequently there is no statistically significant difference between the workin-progress (WIP) or throughput objective values for different buffer sizes, formally stated as (4.1). Alternatively, (4.2) the means differ significantly and therefore the buffer sizes influence the objectives significantly.

Note that one-way ANOVA is used, *i.e.* the test is conducted for each objective respectively. If H_0 is rejected for one objective then that objective becomes the deciding factor, be it for the upper bound chosen or the number of observations per solution. The following hypothesis is used for both the work-in-progress and throughput objectives for determining a sensible upper bound for BAP5.

$$H_0 : \mu_1 = \mu_2 = \mu_3 = \mu_4 = \mu_5 = \mu_6 = \mu_7 = \mu_8 = \mu_9$$
(4.1)

$$H_1 : \mu_i \neq \mu_j, \quad i \neq j, \quad i, \ j \in \{1, \dots, 9\}.$$

$$(4.2)$$

The values for each objective in Table D.4 may differ numerically, however ANOVA is conducted to determine whether they differ statistically. The ANOVA test results are summarised in Table 4.2 and describe the sum of squares (SS), degrees of freedom (df), mean squared (MS), the *F*statistic (*F*) the corresponding critical value (*F*crit) and *p*-value. Since *F*crit < F and p < 0.05the outcome of the ANOVA test is to reject H_0 , which indicates that the mean of at least one pair is statistically different and therefore a *post hoc* test is required to determine which experiments differ significantly. To identify the significant pair(s) *post hoc* analysis is required,. The multiple comparisons *t*-test is used for *post hoc* analysis.

Table 4.2: The results of the ANOVA test to determine the upper bound for BAP5.

Source of variation	SS	df	MS	F	Fcrit	<i>p</i> -value	Outcome
Between groups	36303.27	1	36 303.27	1599757.20	4.49	0	Reject H_0
Within groups	0.36	16	0.02				
Total	36303.63	17					

Figures 4.3a and 4.3b illustrate the CI plot per experiment for objectives work-in-progress and throughput and also presents the pair-wise comparisons. Note that the dot represents the expected mean and the line indicates the range of 95% of the observations. From the visual inspection of Figure 4.3b, it can be seen that the throughput objective values do not differ statistically and consequently has no influence on the decision regarding the problem upper bound. Considering the work-in-progress objective a sensible upper bound is chosen. The larger the upper bound, the larger the combinatorial nature of the problem, therefore a lower upper bound is preferred.

From the *p*-values in Table 4.3 obtained from the *post hoc t*-test, regarding the work-in-progress objective, it can be said that Exp~5 is not statistically significantly different to Exp 6–9. Consequently, implementing more than 17 buffers per machine does not yield a work-in-progress value that is significantly better, *i.e.* the threshold is reached and the work-in-progress becomes dependent on the arrival rate of the work. This means that even if 100 buffer spaces are allocated, the point of diminishing return is met at 17 buffers. Logically, further analysis should be limited to 17 buffer spaces per machine. A similar process was followed to determine sensible upper bounds for the *Open mine problem* (OMP), the (s, S) inventory problem (IP), the BAP with 10 machines (BAP10) and the non-linear BAP with 16 machines (BAP16). The resulting upper bounds are reported in the following respective sections.



Figure 4.3: Illustration of the meticulous consideration taken with the choice of upper bound for BAP5.

	Exp 2	Exp 3	Exp 4	Exp 5	Exp 6	Exp 7	Exp 8	Exp 9
Exp 1	0	0	0	0	0	0	0	0
Exp 2		0.01	0	0	0	0	0	0
Exp 3			0.16	0.04	0.01	0.01	0	0
Exp 4				0.54	0.29	0.17	0.02	0.02
Exp 5					0.66	0.44	0.08	0.08
Exp 6						0.74	0.18	0.18
Exp 7							0.31	0.31
Exp 8								1

Table 4.3: Table of the p-values of the t-test for the work-in-progress objective for BAP5.

The next sections set out to describe each simulation problem in terms of their inputs and outputs, upper bounds and solution cardinality. Next, hypothesis testing is conducted to determine whether or not 100 observations per solution are sufficient and lastly, a description of the complexity of each problem is given as well as its true Pareto front.

4.2.1 The Open mine problem

There are two conflicting objectives that need to be optimised, namely minimise the *total cost* while maximising the *number of trains served*. The decision variables are the number of shovels, crushers and trucks. The sensible upper bound was found at 15 for all decision variables, resulting in a discrete search space cardinality of 3 375 solutions.

The MOO question can be formulated as: what values of *shovels*, *crushers* and *trucks* that result in a minimum total cost and maximum number of trains served? In the context of simulation optimisation, a search algorithm samples values for shovels, crushers and trucks within the specified range, the values are then evaluated by the simulation model and point estimators for the objectives are returned. The evaluation of a decision variable combination $\{shovels, crushers, trucks\}$ results in a corresponding value for total cost and number of trains served.

A wide spectrum of problems sizes are studied, from small (the OMP) to very large (BAP16). The reason for including a smaller problem such as the OMP is to validate that the algorithms can solve the problems studied. The reasoning is that if the algorithms are able to find the true Pareto set (or markedly good approximations of the true Pareto set) for a reasonably small problem then they should be able to solve the larger problems.

Chapter 4. Simulation Models and Statistical Prerequisites

Observations per solution

The proposed number of observations per solution is 100 due to the limited timeframe in which the study must be completed. To motivate this decision further, the respective problems are tested for 10, 100 and 1000 observations per solution for 10 different experiments (or decision variable combinations), respectively, to test whether there is a statistically significant difference in the number of observations chosen and whether or not 100 observations are sufficient.

The following procedure was followed for all the problems. Note that the decision variable combinations are chosen arbitrarily and has no effect on the outcome of the hypothesis test conducted. The null hypothesis states that the means do not differ and consequently there is no statistically significant difference between the number of observations used, formally stated as (4.3). Alternatively, (4.4) the means differ significantly and therefore the number of observations used matter.

$$H_0 : \mu_1 = \mu_2 = \mu_3 \tag{4.3}$$

$$H_1 : \mu_i \neq \mu_j, \quad i \neq j, \quad i, \ j \in \{1, \dots, 3\}$$
(4.4)

The 10 experiments that are used to determine whether 100 observations per solution are sufficient and are given in Table D.14, a summary of which is presented in Table D.15. The ANOVA test results are summarised in Table 4.4. Note that the simulation time for the OMP is *one simulation day*, denoted by 01:00:00:00.

 Table 4.4:
 The results of the ANOVA test for the OMP.

		Served trains										
Source of Variation	SS	df	MS	F	Fcrit	<i>p</i> -value	\mathbf{SS}	df	MS	F	Fcrit	p-value
Between Groups	48 677 990.34	9	5408665.59	613461121.17	2.39	0	16.13	9	1.79	190.39	2.39	0
Within Groups	0.18	20	0.01				0.19	20	0.01			
Total	48677990.51	29					16.32	29				

Table 4.5: The experiments that are statistically significantly different for the OMP.

	<i>p</i> <	< 0.05
	Total cost	Served trains
Exp 1	\checkmark	\checkmark
Exp 3		\checkmark
Exp 4	\checkmark	\checkmark
Exp 5		\checkmark
Exp 6	\checkmark	\checkmark
Exp 7		\checkmark
Exp 8	\checkmark	\checkmark
Exp 9		\checkmark
$Exp \ 10$	\checkmark	\checkmark

Since F and p < 0.05, the outcome is to reject H_0 which indicates that the mean of at least one pair is significantly different and therefore a *post hoc* test is required to determine which. As mentioned, the multiple comparisons are used to determine the significant pair(s). The *t*-test concluded that there is a significant difference for $Exp \ 1$ and Exp3-10, as seen in Table 4.5. $Exp \ 1$ is used to visually present the differences in terms of CIs for the total cost (Figure 4.4a) and served trains (Figure 4.4b) objectives, for 10, 100 and 1000 observations as well as the corresponding *p*-values. From Figures 4.4a and 4.4b it can be seen that for *Exp 1*, 10 observations per solution (A) are *statistically significantly different* to 100 and 1000 observations per solution (B and C, respectively), *i.e.* p < 0.05.



Figure 4.4: The respective CI plots and p-value tables obtained for the work-in-progress and throughput objectives, respectively, for the OMP.

Therefore, 10 observations per solution would not be enough to say with 95% confidence that the mean generated by 10 observations is the same as the mean generated by 100 or 1000 observations. However, B and C are not statistically significantly different and therefore, 100 observations per solution is sufficient for OMP.

Problem complexity

To summarise, the OMP has two objectives and *three* decision variables, with a search space cardinality of $3\,375$ solutions. The problem has *five* sources of randomness, namely shovel processing time, crusher processing time, the processing time for loader 1 and loader 2, and an inspection processing time. It is also important to note that each search technique only evaluates $1\,000$ solution evaluations, *i.e.* only 29.63% of the total solution space is searched.

Exhaustive enumeration

To obtain the true Pareto set, exhaustive enumeration was performed with 100 simulation observations per {Shovels, Crushers, Trucks} combination. All decision variables were adjusted in steps of 1 unit from 1 to 15, thus exploring a solution space of 3 375 possibilities. The resulting front (containing 36 points) is considered the true Pareto front and is shown in Figure 4.5. The true hyperarea for the OMP is 14 485.04, determined with reference point {2800, 3.8}.



Figure 4.5: The true Pareto front of the OMP obtained by method of exhaustive enumeration.

Chapter 4. Simulation Models and Statistical Prerequisites

4.2.2 The (s, S) inventory problem

There are two conflicting objectives that need to be optimised, namely minimise total inventory cost while maximising service level. The decision variables are the reorder point (s) and reorder quantity (S). The sensible upper bound was found as $s, S \in \{1, \ldots, 500\}$, resulting in a discrete search space cardinality of 250 000 solutions.

The MOO question can be formulated as: what values of s and S result in a minimum total inventory cost and maximum service level? Again, in the context of simulation optimisation, a search algorithm samples values for s and S within the specified range, the values are then evaluated by the simulation model and point estimators for the objectives are returned. The evaluation of a decision variable combination s, S results in a corresponding value for total inventory cost and service level.

Observations per solution

A similar process is followed as documented for the OMP, the null (4.3) and alternative (4.4) hypothesis stated remain the same, but for the IP. The 10 experiments used to determine if 100 observations per solution are sufficient are given in Table D.1, where group A represents 10 observations, B-100 observations and C-1000 observations, a summary of which is presented in Table D.2. The ANOVA test results are summarised in Table 4.6.

Table 4.6: The results of the ANOVA test for the IP.

			Total inve	entory cost	Service level							
Source of variation	SS	df	MS	F	Fcrit	<i>p</i> -value	SS	df	MS	F	Fcrit	p-value
Between groups	8998297	9	999810.82	18 942.63	2.39	0	15051.09	9	1672.34	9350.76	2.39	0
Within groups	1055.62	20	52.78				3.58	20	0.18			
Total	8999353	29					15054.66	29				

Since $F \operatorname{crit} \langle F \text{ and } p \langle 0.05 \rangle$, the outcome is to reject H_0 which indicates that the mean of at least one pair is significantly different and therefore a *post hoc* test is required to determine which pair. Again, multiple comparisons *t*-test is used to determine the significant pair(s). The *t*-test concluded that there is a significant difference for $Exp \ 6$ for the total inventory cost objective and $Exp \ 10$ for the service level objective. $Exp \ 6$ and $Exp \ 10$ are used to visually present the differences in terms of CIs for the total inventory cost (Figure 4.6a) and service level (Figure 4.6b) objectives, for 10, 100 and 1 000 observations as well as the corresponding *p*-values. From Figures 4.6a and 4.6b it can be seen that for $Exp \ 6$ and $Exp \ 10$, 10 observations per solution (A) are statistically significantly different to 100 observations per solution (B), *i.e.* p < 0.05.



Figure 4.6: The respective CI plots and *p*-value tables obtained for the total inventory cost and service level objectives, respectively, for the IP.

Therefore, 10 observations per solution would not be enough to say with 95% confidence that the mean generated by 10 observations is the same as the mean generated by 100 observations. However, B and C are not statistically significantly different and therefore 100 observations per solution are sufficient for the IP.

Problem complexity

To summarise, the IP has two objectives and two decision variables, with a search space cardinality of 250 000 solutions. The problem has *three* sources of randomness (or stochastic elements), namely the arrival rate, customer demand and the inventory replenishment lead times. It is also important to note that the search techniques only evaluate a small percentage of the solution space. Specifically, only 1 000 evaluations were allowed, *i.e.* only 0.4% of the total solution space is searched.

Exhaustive enumeration

To obtain the true Pareto set, exhaustive enumeration was performed with 100 simulation observations per (s, S) combination. Both s and S were adjusted in steps of 1 unit from 1 to 500, thus exploring a solution space of 250 000 possibilities. The resulting front (containing 1 393 points) is considered the true Pareto front \mathcal{P}_T for this problem and is shown in Figure 4.7. The true hyperarea for the IP is 114 443.61, determined with reference point {2500, 20}.



Figure 4.7: The true Pareto front of the IP obtained by method of exhaustive enumeration.

4.2.3 The buffer-allocation problem: five machines

For each BAP instance, there are two conflicting objectives that need to be optimised, namely minimise *work-in-progress* while maximising *throughput*. The decision variables are the buffer spaces (or niches) buffer 1–4 with a sensible upper bound that was found at 17, as described previously, resulting in a discrete search space cardinality of 83 521 solutions.

The MOO question for the BAP can be formulated as: what values of the buffer spaces result in the minimum work-in-progress and maximum throughput? This is the same for all the BAP instances, however the buffer spaces differ for each instance. Again, in the context of simulation optimisation, a search algorithm samples values for buffer 1–4 within the specified range. The values are then evaluated by the simulation model and point estimators for the objectives are returned. The evaluation of a decision variable combination {buffer 1, buffer 2, buffer 3, buffer 4}, for the sake of brevity {B1, B2, B3, B4}, results in a corresponding value for the work-in-progress and throughput.

Observations per solution

Again, a similar process is followed as documented for the OMP, the null (4.3) and alternative (4.4) hypotheses stated remains the same, but for BAP5. The 10 experiments that are used to determine whether 100 observations per solution are sufficient are given in Table D.5, a summary of which is presented in Table D.6. The ANOVA test results are summarised in Table 4.7.

	Work-in-progress							Throughput					
Source of Variation	\mathbf{SS}	df	$_{\mathrm{MS}}$	F	Fcrit	p-value	\mathbf{SS}	df	MS	F	Fcrit	p-value	
Between Groups	9.46	9	1.05	468.35	2.39	0	1252.24	9	139.14	145.24	2.39	0	
Within Groups	0.04	20	0				19.16	20	0.96				
Total	9.51	29					1271.39	29					

 Table 4.7:
 The results of the ANOVA test for BAP5.

Since F crit $\langle F$ and $p \langle 0.05$, the outcome is to reject H_0 which indicates that the mean of at least one pair is significantly different and therefore a *post hoc* test is required to determine which. Again, multiple comparisons are used to determine the significant pair(s). The *t*-test concluded that there is a significant difference for $Exp \ 1$, $Exp \ 3-4$ and $Exp \ 7-10$, as seen in Table 4.8. Exp1 is used to visually present the differences in terms of CIs for the work-in-progress (Figure 4.8a) and throughput (Figure 4.8b) objectives, for 10, 100 and 1 000 observations as well as the corresponding *p*-values.

Table 4.8: The experiments that are statistically significantly different for BAP5.

	p < 0.05								
	Work-in-progress	Throughput							
Exp 1	\checkmark	✓							
Exp 3		\checkmark							
Exp 4		\checkmark							
Exp 7	\checkmark								
Exp 8	\checkmark								
Exp 9	\checkmark								
Exp 10		\checkmark							

From Figures 4.8a and 4.8b it can be seen that for $Exp \ 1$, 10 observations per solution (A) are statistically significantly different to 1 000 observations per solution (C), *i.e.* p < 0.05. Therefore, 10 observations per solution would not be enough to say with 95% confidence that the mean generated by 10 observations is the same as the mean generated by 1 000 observations. However, B and C are not statistically significantly different and therefore, 100 observations per solution is sufficient for BAP5.





Figure 4.8: The respective CI plots and *p*-value tables obtained for the work-in-progress and throughput objectives, respectively, for the BAP5.

Problem complexity

To summarise, BAP5 has two objectives and four decision variables, with a search space cardinality of 83 521 solutions. The problem has *fifteen* sources of randomness, namely five machine processing times, five failure distributions and five mean time to repair (MTTR) distributions. Only a small percentage of the total search space is explored, specifically only 1000 evaluations were allowed, *i.e.* only 1.2% of the total solution space is searched.

Exhaustive enumeration

To obtain the true Pareto set, exhaustive enumeration was performed with 100 simulation observations per {B1, B2, B3, B4} combination. All buffers were adjusted in steps of 1 unit from 1 to 17, thus exploring a solution space of 83521 possibilities. The resulting front (containing 306 points) is considered the true Pareto front \mathcal{P}_T for this problem and is shown in Figure 4.9. The true hyperarea for the BAP is 33.84, determined with reference point {2.5, 70}.



Figure 4.9: The true Pareto front of BAP5 obtained by method of exhaustive enumeration.

4.2.4 The buffer-allocation problem: 10 machines

The decision variables are the buffer spaces buffer 1–9 (B1–B9). The sensible upper bound was found at 10, resulting in a discrete search space cardinality of 1 000 000 000 solutions. Again, in the context of simulation optimisation, a search algorithm samples values for B1–9 within the specified range. The values are then evaluated by the simulation model and point estimators for the objectives are returned. The evaluation of a decision variable combination {B1, B2, B3, B4, B5, B6, B7, B8, B9}, results in a corresponding value for the work-in-progress and throughput.

Observations per solution

Again, a similar process is followed as documented for the OMP, the null (4.3) and alternative (4.4) hypothesis stated remain the same, but for BAP10. The 10 experiments that are used to determine whether 100 observations per solution are sufficient are given in Table D.8, a summary of which is presented in Table D.9. The ANOVA test results are summarised in Table 4.9.

Table 4.9: The results of the ANOVA test for BAP10.

	Work-in-progress							Throughput					
Source of Variation	\mathbf{SS}	df	$_{\mathrm{MS}}$	F	Fcrit	p-value	\mathbf{SS}	df	MS	F	Fcrit	p-value	
Between groups	4.47	9	0.50	703.67	2.39	0	730.72	9	81.19	17.32	2.39	0	
Within groups	0.01	20	0				93.77	20	4.69				
Total	4.48	29					824.49	29					

Since F crit < F and p < 0.05, the outcome is to reject H_0 which indicates that the mean of at least one pair is significantly different and therefore a *post hoc* test is required to determine which. Again, multiple comparisons are used to determine the significant pair(s). The *t*-test concluded that there is a significant difference for $Exp \ 1-8$ and 10, as seen in Table 4.10.

Table 4.10: The experiments that are statistically significantly different for BAP10.

	p < 0.0)5
	Work-in-progress	Throughput
Exp 1	\checkmark	
Exp 2	\checkmark	\checkmark
Exp 3		\checkmark
Exp 4		\checkmark
Exp 5		\checkmark
Exp 6		\checkmark
Exp 7		\checkmark
Exp 8		\checkmark
Exp 10		\checkmark

 $Exp\ 2$ is used to visually present the differences in terms of CIs for the work-in-progress (Figure 4.10a) and throughput (Figure 4.10b) objectives, for 10, 100 and 1000 observations as well as the corresponding *p*-values.



Figure 4.10: The respective CI plots and *p*-value tables obtained for the work-in-progress and throughput objectives, respectively, for BAP10.

From Figures 4.10a and 4.10b it can be seen that for $Exp \ 2$, 10 observations per solution (A) are statistically significantly different to 100 and 1000 observations per solution (B and C, respectively), *i.e.* p < 0.05.

Therefore, 10 observations per solution would not be enough to say with 95% confidence that the mean generated by 10 observations is the same as the mean generated by 100 or 1000 observations. However, B and C are not statistically significantly different and therefore, 100 observations per solution is sufficient for BAP10.

Problem complexity

To summarise, BAP10 has two objectives and nine decision variables, with a search space cardinality of 1×10^9 solutions. The problem has *thirty* sources of randomness, namely ten machine processing times, ten failure distributions and ten mean time to repair (MTTR) distributions. It is also important to note that the search techniques only evaluate a small percentage of the solution space. Specifically, only 1 000 evaluations were allowed, *i.e.* only 0.0001% of the total solution space is searched.

Exhaustive enumeration

An exhaustive search approach involves searching through the entire solution space (*i.e.* 100 simulation observations per {B1, B2, B3, B4, B5, B6, B7, B8, B9} combination and all buffers adjusted in steps of 1 unit from 1 to 10) to find the true Pareto set. This approach is not feasible for large decision spaces, such as 1×10^9 possibilities, as it becomes computationally time consuming and practically impossible. Therefore, design of experiments was used to obtain a good approximate of the true Pareto set which for the purposes of this study will represent the true Pareto set. The resulting front (containing 120 points) is considered the true Pareto front and is presented in Figure 4.11. The true hyperarea for BAP10 is 22.66, determined with reference point {1.4, 60}.



Figure 4.11: The true Pareto front of BAP10 obtained by design of experiments.

4.2.5 The non-linear buffer-allocation problem: 16 machines

The decision variables are the buffer spaces buffer 1–15 (B1–B15). The sensible upper bound was found at 10, resulting in a discrete search space cardinality of 1×10^{15} solutions. Again, in the context of simulation optimisation, a search algorithm samples values for B1–15 within the specified range. The values are then evaluated by the simulation model and point estimators for the objectives are returned. The evaluation of a decision variable combination {B1, B2, B3,

B4, B5, B6, B7, B8, B9, B10, B11, B12, B13, B14, B15}, results in a corresponding value for the work-in-progress and throughput.

Obervations per solution

Again, a similar process is followed as documented for the OMP, the null (4.3) and alternative (4.4) hypothesis stated remain the same, but for BAP16. The 10 experiments that are used to determine whether 100 observations per solution are sufficient are given in Table D.11, a summary of which is presented in Table D.12. The ANOVA test results are summarised in Table 4.11.

Table 4.11: The results of the ANOVA test for BAP16.

Work-in-progress							Throughput					
Source of Variation	\mathbf{SS}	df	$_{\mathrm{MS}}$	F	Fcrit	p-value	\mathbf{SS}	df	MS	F	Fcrit	p-value
Between Groups	2.67	9	0.30	193.14	2.39	0	34889.22	9	3876.58	101.25	2.39	0
Within Groups	0.03	20	0				765.77	20	38.29			
Total	2.71	29					35654.99	29				

Since F crit $\langle F$ and $p \langle 0.05$, the outcome is to reject H_0 which indicates that the mean of at least one pair is significantly different and therefore a *post hoc* test is required to determine which. Again, multiple comparisons are used to determine the significant pair(s). The *t*-test concluded that there is a significant difference for $Exp \ 2-4$ and 8-10, as seen in Table 4.12.

Table 4.12: The experiments that are statistically significantly different for BAP16.

	p < 0.05								
	Work-in-progress	Throughput							
Exp 2		\checkmark							
Exp 3	\checkmark	\checkmark							
Exp 4	\checkmark								
Exp 8	\checkmark								
$\mathrm{Exp}\ 10$	\checkmark	\checkmark							

Exp 3 is used to visually present the differences in terms of CIs for the work-in-progress (Figure 4.12a) and throughput (Figure 4.12b) objectives, for 10, 100 and 1000 observations as well as the corresponding *p*-values. From Figures 4.12a and 4.12b it can be seen that for *Exp* 10, 10 observations per solution (A) are *statistically significantly different* to 100 and 1000 observations per solution (B and C, respectively), *i.e.* p < 0.05.



Figure 4.12: The respective CI plots and *p*-value tables obtained for the work-in-progress and throughput objectives, respectively, for the BAP16.

Therefore, 10 observations per solution would not be enough to say with 95% confidence that the mean generated by 10 observations is the same as the mean generated by 100 or 1000 observations. However, B and C are not statistically significantly different and therefore 100 observations per solution is sufficient for BAP16.

Problem complexity

To summarise, BAP16 has two objectives and *fifteen* decision variables, with a search space cardinality of 1×10^{15} solutions. The problem has *fifty-one* sources of randomness, namely sixteen machine processing times, sixteen failure distributions and sixteen mean time to repair (MTTR) distributions. Lastly, the routing from B1, B3 and B7 are determined stochastically, *i.e.* there is a 50:50 chance for the part to go from B1 to M1 or M2, B3 to M4 or M5 and B7 to M8 or M9, as seen in Figure D.8. Again, note that the search techniques only evaluate a small percentage of the solution space. Specifically, only 1000 evaluations were allowed, *i.e.* only $1 \times 10^{-10}\%$ of the total solution space is searched.

Exhaustive enumeration

An exhaustive search approach involves searching through the entire solution space (*i.e.* 100 simulation observations per {B1, B2, B3, B4, B5, B6, B7, B8, B9, B10, B11, B12, B13, B14, B15} combination and all buffers adjusted in steps of 1 unit from 1 to 10) to find the true Pareto set. However, as mentioned, this approach is not feasible for large decision spaces, such as 1×10^{15} possibilities. Therefore, design of experiments was used to obtain a good approximate of the true Pareto set which for the purposes of this study will represent the true Pareto set. The resulting front (containing 132 points) is considered the true Pareto front and is presented in Figure 4.13. The true hyperarea for BAP16 is 214.33, determined with reference point {1.4, 350}.



Figure 4.13: The true Pareto front of BAP16 obtained by design of experiments.

4.3 Conclusion: Chapter 4

This chapter discussed some of the statistical prerequisites required to understand the analysis of simulation outputs. The respective simulations models were also described in terms of their inputs, outputs, upper bounds, search space cardinality, complexity and the true Pareto front was given. This chapter also set out to determine whether 100 observation per solution are sufficient. Table 4.13 summarises the results of the statistical tests conducted in this chapter. *Overall*, 100 observations per solution are sufficient for all five problems, whereas 10 observations were not and 1 000 are not computationally feasible. Also, with 100 observations per solution the central limit theorem can be applied and the observed solutions are approximately normally distributed.

Table 4.13: Summary of the sufficient number of observations per solution for the respective problems.

	0	bserva	tions
	10	100	1000
IP	x	\checkmark	\checkmark
BAP5	x	\checkmark	\checkmark
BAP10	x	\checkmark	\checkmark
BAP16	x	\checkmark	\checkmark
OMP	×	\checkmark	\checkmark

To revise, the more observations per solution, the smaller the CI becomes, however the effort required is greater, *i.e.* the values for 100 and 1000 observations per solution may differ numerically, but statistically they are considered the same. However, there is a significant difference in the computational effort required to run 1000 observations per solution when compared with 100. Also, it is more important to obtain an approximate Pareto set showing a trend, rather than having more accurate estimations of the objectives, *i.e.* it is more important to find approximations that are considered accurate enough in the minimum amount of simulation runtime possible.

Next, Chapter 5 documents the metaheuristics chosen as LLHs as well as the proposed hyperheuristics.

CHAPTER 5

Metaheuristics and Hyperheuristics

5.1	The main concepts for metaheuristics	
	5.1.1	Initial solution
	5.1.2	Solution representation
	5.1.3	Stopping condition
5.2	2 Multi-objective cross-entropy method	
	5.2.1	Preliminaries for the MOOCEM
	5.2.2	The MOOCEM algorithm 76
5.3	Non-dominated sorting genetic algorithm II	
	5.3.1	The NSGA-II
	5.3.2	Selection operators
	5.3.3	$Crossover operators \dots \dots \dots \dots \dots \dots 91$
	5.3.4	$Mutation operators \dots \dots$
5.4	1 Dominance-based bi-objective simulated annealing $\ldots \ldots \ldots \ldots \ldots$	
	5.4.1	The DBMOSA algorithm
	5.4.2	The annealing schedule
	5.4.3	The initial temperature
	5.4.4	Neighbourhood move operators
5.5	The BOCEGAH algorithm	
5.6	The BOSAH algorithm	
5.7	Summary	

The previous chapter introduced the simulation optimisation problems studied as well as statistical prerequisites required to conduct a simulation study.

This chapter serves to fulfil Objective 4 and 5 as stated in Chapter 1. Metaheuristics are constantly being improved upon, either by designing new approaches or by extending current approaches, to better and more efficiently solve COPs. Consequently, deciding which metaheuristic (or heuristic operators) to use is not a trivial task. This chapter presents three metaheuristics from different classes in an attempt to represent the diversity of algorithms available in the literature, namely the multi-objective cross-entropy method (MOOCEM), the non-dominated sorting genetic algorithm II (NSGA-II) and the dominance-based multi-objective simulated annealing (DBMOSA) algorithm. First, some main concepts of S- and P-metaheuristics are discussed with specific focus on their implementation in this study. Next, each metaheuristic is discussed, in great detail, in the context of simulation optimisation and its integration in Tecnomatix.

Next, the two hyperheuristics proposed in this study are presented, first for population-based search and then for single-solution based search. The MOOCEM and the NSGA-II are employed for the population-based search hyperheuristic and DBMOSA and three move operators are employed for the single-solution based search hyperheuristic. Each hyperheuristic is also discussed in the context of simulation optimisation and its integration in Tecnomatix. The chapter closes with a summary of its contents.

5.1 The main concepts for metaheuristics

In this section a discussion follows on the basic concepts for metaheuristics, namely initial solution(s), solution representations and stopping conditions. These concepts are considered when designing *any* metaheuristic [93]. The concept of solution dominance and archiving, as discussed in §2.2.1 and §2.2.3, forms part of all the LLHs discussed in this chapter.

5.1.1 Initial solution

Generating initial solutions is considered a crucial step when designing a metaheuristic, as the initial solution(s) have a great impact on the exploration-exploitation capabilities of the metaheuristic and can therefore affect its performance [93]. In this regard, it is important to consider the strategy employed to ensure diverse initial solutions are generated to prevent premature convergence [134].

The two main strategies are *random* and *greedy*. The random strategy randomly generates initial solutions, whereas the greedy strategy finds determines an initial solution that is considered to be of *good quality* by method of applying a greedy algorithm to determine the initial solution. Again, a trade-off exists between the amount of exploration and exploitation. Hybrid strategies combine both random and greedy strategies, *i.e.* a pool of initial solutions are generated using both random and greedy algorithms [93, 134].

The strategy followed for generating initial populations for the P-metaheuristics is the random strategy. The initial populations are generated using the *normal* and *Poisson distributions*. It is important to note that the random seed values are different for every run to ensure that the P-metaheuristics start with a different initial population for each run. This is done for both MOOCEM and NSGA-II. The initial solution for the S-metaheuristic is randomly selected in the range of possible decision variable values. If a solution has been used as an initial solution, it is removed from the set of possible initial solutions, ensuring that DBMOSA starts at different initial solutions for every run.

5.1.2 Solution representation

Deciding on a solution representation (or encoding) is a fundamental design question to consider during the development of a metaheuristic. The representation influences the efficiency and effectiveness of a metaheuristic and constitutes careful consideration. The representation must be suitable and relevant to the optimisation problem, *i.e.* COPs. The efficiency is also related to the search operators applied to the representation, for example move operators or crossover operators. In conclusion, the choice of solution representation requires consideration. Figure 5.1 highlights the main solution representations in the literature, namely binary encodings, vector of discrete values, permutations and vector of real values [93, 250].

For the sake of completeness, Figure 5.1 includes the representation for a vector of real values, however, it is not applicable for COPs and is excluded from further discussion.

In Figures 5.2a–5.2c, an illustration is presented of the neighbourhoods that correspond to the solution representations applicable for discrete variable optimisation. A binary representation, represents a solution in a vector of bits, where a neighbour is created by flipping a single bit, as illustrated in Figure 5.2a. The neighbourhood is typically based on a Hamming distance of 1, where the size of the neighbourhood is the size of the vector.


Figure 5.1: Some classical solution representations for optimisation problems: vector of binary values, vector of discrete values, vector of real values and permutation, adapted from [250].

In Figure 5.2b, an illustration is presented of the neighbourhood for the solution $\{4, 3, 4\}$ with a lower bound of 0 and an upper bound of 4, for all decision variables. The size of the neighbourhood is n(k-1) = 12, where n = 3 represents the number of decision variables and the cardinality of the decision variables, is denoted by k = 5. The neighbourhood in Figure 5.2c is created by a pair-wise exchange, where the order is of significance. The neighbourhood size depends on the permutation length n and can be calculated as $\frac{n(n-1)}{2}$ [250].



Figure 5.2: Illustration of the neighbourhood for (a) binary encoding, (b) discrete vectors and (c) permutation representations, adapted from [250].

A majority of the COPs studied either have permutation representations (e.g. travelling salesman problem (TSP)) or discrete vector representations (e.g. quadratic assignment problem (QAP)),

which is attributable to the number of metaheuristics that are available and the simplicity of manipulating these types of representations [1, 148]. The simulation optimisation problems studied are considered COPs. Solutions are evaluated by the simulation model which takes as input a vector of discrete values and because the search operators can be modified to manipulate discrete vectors, discrete vector representation is chosen.

5.1.3 Stopping condition

The search terminates if the given stopping condition is met. A stopping condition may be a fixed number of iterations, or a convergence measure [93]. The stopping conditions of the LLHs have been modified to control explicitly the duration of each algorithm and to facilitate fair comparisons between the algorithms. A fixed number of iterations is adopted, namely 1000 solution evaluations.

The following sections document each LLH with specific focus on the integration of simulation and optimisation, starting with MOOCEM, then NSGA-II and finally DBMOSA.

5.2 Multi-objective cross-entropy method

The single-objective cross-entropy method (CEM) [217] has been adapted to solve MOOPs. The multi-objective extension, called MOOCEM, was purposefully designed by Bekker and Aldrich [22] to solve (computationally expensive time-dependent) stochastic multi-objective simulation optimisation problems — naturally, motivating its inclusion in this study. The CEM is based on *Importance Sampling* and the *Kullback–Leibler* (or cross-entropy) distance [58]. For a theoretical discussion, the reader is referred to [217]. The following section introduces some preliminaries specific to the MOOCEM algorithm.

5.2.1 Preliminaries for the MOOCEM

The literature does not prescribe a specific probability distribution for the CEM [21, 22]. In this study the normal and Poisson distributions are used for both population initialisation and to sample decision variable values from. To enable sampling of feasible decision variable values, it is necessary to truncate the probability distribution so as to sample in the range $[l_i, u_i]$ for decision variable *i*.

The truncated normal (or Poisson) distribution on the given range $[0, u_i]$ is obtained by normalising the fundamental normal (or Poisson) distribution. This is described next, followed by how to sample from these distributions. The probability density function (pdf) for the normal distribution is given as

$$f(x) = \frac{1}{\sigma\sqrt{2\pi}} \exp^{-\frac{1}{2}(\frac{x-\mu}{\sigma})^2}, \quad -\infty < x < \infty,$$
(5.1)

where σ denotes the standard deviation and μ the mean. The pdf for the truncated normal distribution with μ_i and σ_i , on the range [a, b], is given as

$$f_{t,i}(x, \ \mu_i, \ \sigma_i, \ a, \ b) = \frac{f(x)}{\int_a^b f(x) dx}, \ a \le x \le b,$$
 (5.2)

where $f(\cdot)$ is the pdf, as defined in (5.1).

74

The probability mass function (pmf) for the Poisson distribution can be expressed as

$$f(x) = \frac{\lambda^x \exp^{-\lambda}}{x!}, \quad x = 0, 1, 2, \dots$$
 (5.3)

where λ is the mean. Note that μ and σ (or λ) are arbitrarily initialised. The pmf for the truncated Poisson distribution with rate λ_i , on the range [a, b], is given as

$$f_{t,i}(x, \lambda_i, a, b) = \frac{e^{-\lambda_i} \lambda_i^x}{\sum_{x=a}^b \frac{e^{-\lambda_i} \lambda_i^x}{x!} x!}, \quad x = a, \dots, b.$$
(5.4)

The pseudo-code description of truncating a probability distribution is presented in Algorithm 5.1. The truncation process is similar for the normal and Poisson distributions, therefore one algorithm is sufficient to describe the process for both probability distributions. In Algorithm 5.1 the truncation process is first described for the normal distribution and then for the Poisson distribution in brackets.

First, the pdf (or pmf) values $f_i(x)$ for normal (or Poisson) distribution are calculated, using (5.1) (or (5.3)), for decision variable *i* in the range [0, u_i]. Next, each $f_i(x)$ value is normalised, by dividing it by the sum of the $f_i(x)$ values in the range $[l_i, u_i]$. This results in the truncated pdf (or pmf) values $f_{t,i}(x)$. Now, the truncated cdf values $F_{t,i}(x)$ can be calculated as the sum of $f_{t,i}(x)$, where x is in the range $[l_i, u_i]$ and $F_{t,i}(l_i) = f_{t,i}(l_i)$, $F_{t,i}(l_i + 1) = f_{t,i}(l_i) + f_{t,i}(l_i + 1)$ and so on. To validate that the distributions are truncated correctly, the cdf value for the upper bound should be equal to 1, *i.e.* $F_{t,i}(u_i) = 1$.

Algorithm 5.1: Truncate a probability distribution			
Input : For the normal distribution, specify the μ_i and σ_i and (for the Poisson distribution,			
specify λ_i) also specify the upper and lower bound $[l_i, u_i]$	for decision variable i .		
Output : The truncated normal (or Poisson) distribution.			
1 $sum \leftarrow 0;$	// Initialise sum		
2 for $x := 0$ to u_i do			
3 $\[$ Calculate the pdf (or pmf) value $f_i(x)$ for x , using (5.1) (or (5.3))); $// x \in [0, u_i]$		
4 for $x := l_i$ to u_i do			
5 $\lfloor sum \leftarrow sum + f_i(x);$ // Dete	ermine the sum, $x \in [l_i, \ u_i]$		
6 for $x := l_i$ to u_i do			
7 $\[\]$ Calculate the truncated pdf (or pmf) value, $f_{t,i}(x) \leftarrow f_i(x)/sum$;		
8 Calculate the first truncated cdf value, $F_{t,i}(l_i) \leftarrow f_t(l_i)$;			
9 for $x := l_i + 1$ to u_i do			
10 $\[\]$ Calculate the truncated cdf value, $F_{t,i}(x) \leftarrow f_{t,i}(x) + f_{t,i}(x-1);$	// $x \in [l_i + 1, u_i]$		

The next step is to sample from the truncated normal (or Poisson) distribution which is described in Algorithm 5.2. To sample from the truncated normal (or Poisson) distribution, a random number $r_u \sim \mathcal{U}(0, 1)$ is generated and $x \leftarrow l_i$. If $r_u < F_{t,i}(x)$, where x is in the range $[l_i, u_i]$ for decision variable i, then x is incremented. The algorithm terminates when $r_u \ge F_{t,i}(x)$ and x is returned as the random sample in the range $[l_i, u_i]$.

Sampling from these distributions is simple and the sample values, for decision variable i on the defined range, are easily obtained. Figures 5.3a and 5.3b illustrate the truncated normal distribution with μ_i and σ_i and the Poisson distribution with λ_i , for decision variable i, where x is defined in the range [-1, 2].



(a) Truncated normal distribution (b) Truncated Poisson distribution

Figure 5.3: An example illustrating the (a) truncated normal distribution with μ_i and σ_i and (b) truncated Poisson distribution with λ_i , for decision variable *i*, where *x* is defined in the range [-1, 2], adapted from [22].

The next section provides a detailed description of the MOOCEM algorithm, supported by the user-interface of the MOOCEM tool used in Tecnomatix, to illustrate the integration of simulation and optimisation.

5.2.2 The MOOCEM algorithm

Figures 5.4a and 5.4b aids with elucidating the simulation optimisation process *via* the MOOCEM tool as it is used in Tecnomatix. The steps for using the MOOCEM tool are as follows: starting with the *Definition tab* as shown in Figure 5.4a.

- 1. Define the number of decision variables (n), from the drop-down list, then click the *Define Input Variables* button and define each decision variable in terms of its path and lower and upper bounds, as seen in Figure 5.5a.
- 2. Click the *Define Output Variables* button and define each output variable in terms of its path and select the optimisation direction from the drop-down list, *i.e. Min* or *Max* as seen in Figure 5.5b.

The next step is to navigate to the *Parameterisation tab*, as shown in Figure 5.4b:

3. Input the Population size (N), Number of loops, Smoothing parameter, Probability of inversion and Epsilon.

76



Figure 5.4: The MOOCEM tool user-interface.

The above-mentioned parameters are used as input to Algorithm 5.4. To execute the MOOCEM algorithm and optimise the simulation problem at hand click the *Start* button. The *Evaluation* tab is where the simulation time for the specific simulation problem is specified as discussed in Chapter 4.



(b) The output variable table

FIGURE 5.5: The user-interface for the input and output tables used in Tecnomatix.

As mentioned, the initial population is generated randomly. The Poisson distribution is a discrete probability distribution, denoted by $X \sim \text{Pois}(\lambda)$, where λ is the mean. The Poisson distribution is unable to generate sample values larger than 130, therefore if the upper bound of a decision variable exceeds 130, as in the case of the IP, the normal distribution is used. The normal distribution is a continuous probability distribution, denoted by $X \sim \mathcal{N}(\mu, \sigma^2)$, where μ is the mean and σ the standard deviation.

When sampling from the normal distribution, discrete intervals are used to enable the generation of discrete sample values and because decision variable i has a lower and upper bound, it is necessary to truncate the distributions as to contain the search to only generating sample values within the specified bounds, as discussed in §5.2.1.

The algorithm starts by initialising the iteration counter $t \leftarrow 1$, the current loop $k \leftarrow 1$, the current number of evaluations $NumEvaluations \leftarrow 0$ and a boolean variable $NotTerminate \leftarrow$ true. Next, the parameters specific to the probability distribution are initialised [21, 22]. If the normal distribution is used, then the mean μ_i and standard deviation σ_i is initialised, for decision variable *i*. If the Poisson distribution is used, then the mean λ_i is initialised for decision variable *i*. The following discussion assumes the normal distribution is used and therefore a mean and standard deviation are required.

1. Initialise μ_i and σ_i .

Initialise $\mu_i \leftarrow l_i + (u_i - l_i)r_u$, where l_i is the lower bound and u_i the upper bound for decision variable *i*, for $i \in \{1, ..., n\}$ and r_u is a random number, between 0 and 1, generated from the uniform distribution, denoted by $r_u \sim \mathcal{U}(0, 1)$. Next, initialise an arbitrarily large value for σ_i , for example $\sigma_i \leftarrow 10(u_i - l_i)$ [21, 22]. (In the case of the Poisson distribution, λ_i is arbitrarily initialised as $\lambda_i \leftarrow u_i r_u$).

- 2. Construct the truncated probability distribution for each decision variable *i* for μ_i and σ_i , using Algorithm 5.1. Refer to §5.2.1.
- 3. Generate an initial population, of size N, by sampling from the truncated probability distribution, using Algorithm 5.2 for each decision variable i. Again, refer to §5.2.1.
- 4. Evaluate the initial population, *i.e.* the simulation model evaluates N decision variable combinations.

Now, the loop initiates, while the current loop k is less than the user-defined Number of loops, denoted by k_{max} , do the following:

- 5. Rank the solutions obtained by the simulation model, using Algorithm 5.3 for specified ranking threshold ρ_E .
- 6. Append the solutions with ranking values not exceeding a specified threshold value ρ_E to a so-called elite vector, titled *Elite*.

Algorithm 2.1 is adapted to consider the ranking threshold ρ_E , as given in Algorithm 5.3.

Consequently, *Elite* represents the current weakly non-dominated set. Solution x_1 is said to weakly dominate x_2 (denoted by $x_1 \leq x_2$) if $f_k(x_1) \leq f_k(x_2)$ for $k \in \{1, \ldots, m\}$ [162]. To ensure exploration and exploitation during the search, the initial ranking threshold is selected as $\rho_E = 2$, *i.e. Elite* contains solutions with rank $\rho \in \{0, 1, 2\}$. The ranking value of the solution indicates the number of solutions that dominate it [21, 22]. For example, a solution with $\rho = 2$ is dominated by two solutions in the population.

- 7. If the elite vector is not an empty set, denoted as $Elite \neq \emptyset$, then update μ_i and σ_i according to (5.5).
- 8. Construct the truncated probability distribution for each decision variable *i* for μ_i and σ_i , using Algorithm 5.1.
- 9. Generate the new population of N possible solutions by sampling from the truncated probability distribution, using Algorithm 5.2 for each decision variable i and continue from step 19 onwards.

Algorithm 5.3: Ranking threshold adaptation of the Pareto ranking algorithm, for a *min-max* bi-objective optimisation problem

: A population of solutions \mathcal{P} and ranking threshold ρ_E . Input **Output** : The elite vector containing solutions with rank $0 - \rho_E$. 1 $\mathcal{P} \leftarrow \mathsf{sort}(\mathcal{P}, 1);$ // Sort the first objective in descending order for $i \leftarrow 1$ to $|\mathcal{P}|$ do 2 // Initialise Pareto rank to zero penalty $\leftarrow 0$; 3 for $j \leftarrow (i+1)$ to $|\mathcal{P}|$ do if $f_j^2 \prec f_i^2$ then // *i.e.* if $f_j^2 \ge f_i^2$ $penalty \leftarrow penalty + 1;$ $\mathbf{4}$ $\mathbf{5}$ // Increment the penalty 6 // Rank of solution i7 $\rho_i \leftarrow penalty;$ 8 for $i \leftarrow 1$ to $|\mathcal{P}|$ do if $\rho_i = \rho_E$ then 9 Elite \leftarrow Elite \cup {i}; // Solutions with rank $0-\rho_E$ 10

10. Check for duplicate solutions and replace them with non-duplicate solutions.

The parameter vectors (μ_i, σ_i) are smoothed by means of the smoothing function and the decision variable values in the elite vector, given as

$$\hat{\boldsymbol{v}}_t = \alpha \hat{\boldsymbol{v}}_t + (1 - \alpha) \hat{\boldsymbol{v}}_{t-1},\tag{5.5}$$

where t is the current iteration and α is the smoothing constant, typically in the range [0.6, 0.9]. For example, the value of $\sigma_{i,t}$, for the first decision variable, is updated as follows

$$\hat{\sigma}_{1,t} = \alpha \tilde{\sigma}_{1,t} + (1-\alpha)\hat{\sigma}_{1,t-1},\tag{5.6}$$

after iteration t [21, 22]. The idea behind smoothing μ_i and σ_i is to account for some information from the previous iteration.

11. If $Elite \neq \emptyset$ and k > 1, then construct a histogram for each decision variable i using the values in *Elite*.

The histograms are used to guide the search and inform the MOOCEM on what regions it needs to explore next. The histogram concept is implemented as follows and explained using Figure 5.6 for a decision variable i defined in the range $[l_i, u_i]$. In this example, there are r = 5 equally sized classes that are formed between the two boundary classes (*i.e.* the first and last class), resulting in a total of r + 2 classes [21, 22].

The first step is to determine the boundaries of the classes. For the first class, the lower bound is set equal to l_i (the lower bound of decision variable i) and the upper bound is set equal to the minimum value found in *Elite*, *i.e.* $\min(Elite(\cdot, i))$. Similarly, for the last class, the upper bound is set equal to u_i (the upper bound of decision variable i) and the lower bound is set equal to the maximum value found in *Elite*, *i.e.* $\max(Elite(\cdot, i))$ [21, 22].

Next, the equally sized class intervals can be calculated as

 $\frac{\max(\textit{Elite}(\cdot,i)) - \min(\textit{Elite}(\cdot,i))}{r}$



Figure 5.6: An example illustrating the histogram concept for decision variable i and r = 5 equally sized classes, with a total of r + 2 classes, adapted from [22].

In summary, consider again the example in Figure 5.6, suppose the class boundaries are recorded in a vector $C_i = \{c_{i1}, c_{i2}, \ldots, c_{i(r+2)}, c_{i((r+2)+1)}\}$, where c_{i1} corresponds to the lower boundary of the first class $(i.e. c_{i1} = l_i)$ and $c_{i((r+2)+1)}$ corresponds to the upper boundary of the last class $(i.e. c_{i((r+2)+1)} = u_i)$. Note that C_i contains r + 3 elements because there are r + 2 classes in the histogram, also, the width of the first class $([c_{i1}, c_{i2}])$ and last class $([c_{i(r+2)}, c_{i((r+2)+1)}])$ may differ from each other, whereas the r classes inbetween have equal widths [21, 22].

12. Determine the frequency value $\tau_{i\kappa}$ for each class, for decision variable *i*, that is present in *Elite*.

Informally, the frequency of class $c_{i1} - c_{i2}$ is determined by counting how many solutions in *Elite* are smaller than or equal to c_{i2} . Next, the frequency of class $c_{i2} - c_{i3}$ is determined by counting how many solutions in *Elite* are greater than c_{i2} and smaller than or equal to c_{i3} and so on, until the frequency of class $c_{i7} - c_{i8}$ is determined.

Formally: X_{ij} belongs to the class $[c_{i\kappa}, c_{i(\kappa+1)})$ if $c_{i\kappa} < X_{ij} \le c_{i(\kappa+1)}$, where $1 \le \kappa \le r+2$ and j is the number of solutions in *Elite*. To conclude, the histogram frequency value τ_{i1} represents the frequency count of decision variable i in the range $[c_{i1}, c_{i2}), \tau_{i2}$ represents the count in the range $[c_{i2}, c_{i3})$, and so on. The new population is formed proportionally, according to the class frequencies for each decision variable [21, 22].

13. Calculate the proportions that determine the number of solutions to generate in each class, respectively, for decision variable i.

Suppose the elite vector contains E_r solutions and $\tau_{i\kappa}$ occurrences in class $[c_{i\kappa}, c_{i(\kappa+1)})$ for a given decision variable. If E_r is not equal to N, then $\lfloor N_{\tau_{ik}}/E_r \rfloor$ values are created from this range. Note that the proportional numbers may not add up to N because of the rounding down calculation, however the difference is simply added to the last class [21, 22].

5.2. Multi-objective cross-entropy method

14. Calculate the temporary $\mu'_{i\kappa}$ and $\sigma'_{i\kappa}$ values associated with the specific histogram class ranges, *i.e.* for the class $[c_{i\kappa}, c_{i(\kappa+1)})$ corresponding to the decision variable *i*. The parameter estimators are given as

$$\mu_{i\kappa}' = c_{i\kappa} + r_u (c_{i(\kappa+1)} - c_{i\kappa}), \text{ and}$$
(5.7)

$$\sigma_{i\kappa}' = c_{i(\kappa+1)} - c_{i\kappa}, \tag{5.8}$$

where $1 \le \kappa \le r + 2$ and $r_u \sim \mathcal{U}(0, 1)$ [21, 22].

Since *Elite* is used to determine the frequencies, according to the rankings returned by the candidates, classes that do not contribute to the elite vector are effectively eliminated from the search (as the search progresses). For this reason, the histogram frequencies are purposefully inverted to prevent premature convergence albeit with low probability. This is done for each iteration t. Note that the probability of inverting the histogram frequencies is denoted by p_h , where the range is typically [0.1, 0.3] [21, 22].

15. Generate a random number $r_u \sim \mathcal{U}(0, 1)$. If $r_u < p_h$ the histogram frequencies are inverted, otherwise the frequencies remain the same and the new population may be generated accordingly.

For decision variable *i*, the maximum frequency over all class frequencies is determined, denoted as τ_{max} . This corresponds to the frequency of class $c_{i4}-c_{i5}$, as seen in Figure 5.7a. The frequency, for each class ($\tau_{i\kappa}$), is then subtracted from τ_{max} , resulting in an inverted histogram, as shown in Figure 5.7b. After the frequency inversion, search ranges with small proportions of population candidate allocations receive higher proportions of allocations, while search ranges with high proportions of population allocations receive fewer allocations, *i.e.* forcing the algorithm to explore the regions possibly eliminated during the search [21, 22].



Figure 5.7: An example illustrating the inversion of the histogram frequencies at iteration t for decision variable i, adapted from [22].

The number of classes increases as the search progresses which renders it possible to maintain good combinations of decision variable values because the class ranges become smaller and consequently more fine-tuned. The number of classes should not become too large because the algorithm will become inefficient and may not be larger than the decision variable bound range. For example, the number of classes for the BAP5 with l = 1 and u = 17, may not exceed u - l = 16 because then the boundaries of the classes become real values. Finally, the new population can be generated by sampling from the relevant truncated probability distribution and class frequencies [21, 22].

- 16. Construct the truncated probability distribution for each decision variable *i* for each class, using $\mu'_{i\kappa}$ and $\sigma'_{i\kappa}$ and Algorithm 5.1.
- 17. Generate the new population of possible solutions by sampling from the truncated probability distribution for each class, using Algorithm 5.2.
- 18. Check for duplicate solutions and then replace them with non-duplicate solutions.
- 19. Evaluate the new population, *i.e.* the simulation model evaluates N new decision variable combinations.

This concludes the loop which is repeated several times. If all $\sigma_{i,t} < \epsilon_c$, then the ranking threshold is set to $\rho_E \leftarrow 1$, *i.e. Elite* contains solutions with $\rho \in \{0, 1\}$, and the number of loops k is incremented, consequently the number of classes of the histograms is incremented, $r \leftarrow k + 5$. MOOCEM usually terminates when the value for σ_i of each decision variable has decreased below a user-specified threshold ϵ_c .

As mentioned, the stopping condition employed in this study is fixed at 1 000 solution evaluations (or when $k = k_{\text{max}}$). At termination, the ranking threshold is set to $\rho_E \leftarrow 0$ and *Elite* is ranked one final time, using Algorithm 5.3, and returned as an approximate of the true Pareto set of solutions to the bi-objective simulation optimisation problem under consideration [21, 22]. The pseudo-code description of the entire process described above is provided in Algorithm 5.4, with specific reference to the above-mentioned steps.

Although it is not explicitly reported, when a new generation is generated, the algorithm has to check if duplicate solutions are present, and if so, replace them with non-duplicate solutions. This is a critical step to ensure that the computationally expensive function evaluator (*i.e.* the simulation model) does not evaluate the same solution. Even though checking for duplicate solutions may add to the computational time, it is less computationally expensive than unnecessarily evaluating a solution, this is done for each algorithm.

In conclusion, the properties of the MOOCEM algorithm are stated. The MOOCEM algorithm

- 1. is a population-based search technique,
- 2. has a statistical basis,
- 3. uses an archive (*Elite*) to preserve weakly-dominated solutions during the search and finally non-dominated solutions after the search terminates,
- 4. employs a Pareto ranking scheme to find non-dominated solutions,
- 5. achieves proximity to the true Pareto front via the convergence mechanism of the CEM,
- 6. and ensures diversity of the search via the inversion of the histogram (with probability p_h) which determines the search regions.

Next, the NSGA-II algorithm is discussed.

Algorithm 5.4: The MOOCEM algorithm, adapted from [21, 22], for the simulation optimisation process

	Input : Previously described using Figures 5.4a and 5.4b.		
	Output : The approximate Pareto set (\mathcal{P}_S) for the given	simulation problem.	
1	Initialise the iteration counter $t \leftarrow 1$;		
2	Initialise the current loop $k \leftarrow 1$;		
3	Initialise $NumEvaluations \leftarrow 0;$ //	The current number of evaluations	
4	Initialise $NotTerminate \leftarrow true;$	// Boolean variable	
5	Initialise μ_i and σ_i ;	// Refer to step 1	
6	$Elite \leftarrow \emptyset;$	<pre>// Initialise Elite vector</pre>	
7	Construct the truncated distribution, using Algorithm 5.1	; // Refer to step 2	
8	Generate an initial population, using Algorithm 5.2;	// Refer to step 3	
9	Evaluate the initial population;	// Refer to step 4	
10	while $k \leq k_{max}$ and $NumEvaluations < MaxEvaluations$	ions do	
11	$t \leftarrow t + 1;$	// Increment t	
12	Rank the solutions, for $\rho_E = 2$, using Algorithm 5.3;	// Refer to step 5-6	
13	if $Elite \neq \emptyset$ then	-	
14	Update μ_i and σ_i , using (5.5);	// Refer to step 7	
15	if All $\sigma_{i,t} < \epsilon_c$ then		
16	$ NotTerminate \leftarrow false;$		
17	Not Terminate \leftarrow (Not Terminate and (Nt \leq MaxE	valuations/2));	
18	$NumEvaluations \leftarrow NumEvaluations + N;$		
19	if $NotTerminate = false$ then		
20	Rank the solutions, for $\rho_E = 1$, using Algorithm 5.	3;	
21	$k \leftarrow k+1;$		
22	Update μ_i and σ_i , as explained in Step 1;		
23	$t \leftarrow 0;$		
24	$\ \ NotTerminate \leftarrow true;$		
25	else if $NotTerminate = true$ then		
26	if $Elite \neq \emptyset$ and $k > 1$ then		
27	Follow the steps described in 11–19;	<pre>// Refer to step 11-19</pre>	
28			
29	Construct the truncated distribution, using Al	gorithm 5.1: // Refer to step 8	
30	Generate a new population, using Algorithm	5.2: // Refer to step 9	
31	Evaluate the new population;	// Similar to step 19	
32	Rank the solutions in <i>Elite</i> , for $\rho_E \leftarrow 0$, using Algorithm	5.3;	
33	$\mathcal{P}_S \leftarrow Elite;$	// Approximate Pareto set	
	₩ /	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	

5.3 Non-dominated sorting genetic algorithm II

Many multi-objective extensions of the single-objective genetic algorithm have been proposed to solve MOOPs and some have even been improved upon. Deb *et al.* [67] developed the NSGA-II to improve upon NSGA (its predecessor) [238] and is classified under evolutionary algorithms (EAs). NSGA-II was one of the first algorithms to purposefully maintain the diversity of the solution set with the focus of converging towards the true Pareto front [238]. EAs are inspired by the principles of natural selection and evolution [10]. For an introductory survey on EAs the reader is referred to [11], whereas for an introduction to multi-objective EAs (MOEAs), the reader is referred to [47].

The following section starts by presenting a detailed description of the NSGA-II algorithm, supported by the user-interface of the NSGA-II tool used in Tecnomatix, to illustrate the in-

83

tegration of simulation and optimisation. Thereafter, the selection, crossover and mutation operators are discussed. The crossover and mutation operators considered in this study, are real-coded operators applicable to the discrete search space.

5.3.1 The NSGA-II

Figures 5.8a and 5.8b aids with elucidating the simulation optimisation process *via* the NSGA-II tool as it is used in Tecnomatix. The first two steps for using the NSGA-II tool are the same as for the MOOCEM tool as described in §5.2.2, refer to steps 1–2. Accordingly, starting with the *Definition tab* as shown in Figure 5.8a, follow the first two steps of the MOOCEM tool, *i.e.* steps 1 and 2. After steps 1 and 2 have been completed continue with step 3 as described here:

3. Select the selection operator to apply, *i.e.* choose between *Binary tournament selection* and *Rank selection*.

🖬 NSGA_II 🛛 🕹 🗙	🖺 NSGA_II 🛛 🗙
Show	Show
Start Stop Reset	Start Stop Reset
Experiment: Observation:	Experiment: Observation:
Definition Parameterisation Evaluation	Definition Parameterisation Evaluation
Decision Variables: Number of decision variables: 2 • Define Input Variables	Population: Population size: 100 Number of generations: 10
Objective Functions:	Maximum evaluations: 1000
Define Output Variables	Operators:
Binary Tournament Selection	Crossover probability: 1.0 -
Rank Selection	Polynomial Mutation *
Observations per solutions: 100 *	Mutation probability: 0.1 *
OK Cancel Apply	OK Cancel Apply
(a) Definition tab	(b) Parameterisation tab

Figure 5.8: The NSGA-II tool user-interface.

The next step is to navigate to the *Parameterisation tab*, as shown in Figure 5.8b:

4. Input the Population size (N), Number of generations, Crossover probability, Mutation operator and Mutation probability.

The mutation operator is selected from a drop-down list as shown in Figure 5.9, where the choice is *Polynomial mutation* or *Dynamic mutation*. The above-mentioned parameters are used as input to Algorithm 5.7. To execute the NSGA-II algorithm and optimise the simulation problem at hand, click the *Start* button.

The algorithm starts by initialising the generation counter, $t \leftarrow 0$. Recall that the initial population is generated randomly, using the truncated normal or Poisson distribution as described in §5.2.1, where μ_i and σ_i or λ_i are arbitrarily initialised, as suggested in step 1.

Ope	erators:		
	Crossover probability:	1.0	*
	Polynomial Mutation		*
	Polynomial Mutation		
	Dynamic Mutation		

Figure 5.9: The mutation operators.

- 5. Generate the initial parent population \mathcal{P}_0 , of size N.
- 6. Evaluate the initial population, *i.e.* the simulation model evaluates N decision variable combinations.

If t = 0, then \mathcal{P}_0 is the mating pool, *i.e.* no selection operator is applied and the offspring population \mathcal{Q}_0 is simply generated from \mathcal{P}_0 .

- 8. Generate the offspring population \mathcal{Q}_0 , of size N, by applying the reproduction operators, namely the crossover operator with probability p_c and mutation operator with probability p_m to the mating pool (\mathcal{P}_0) .
- 9. Check for duplicate solutions and then replace them with non-duplicate solutions.

A random number $r_u \sim \mathcal{U}(0, 1)$ is generated. If $r_u < p_c$, crossover is performed, otherwise the solution remains the same, *i.e.* no modification is made. Similarly, if $r_u < p_m$, mutation is performed. Note that different random numbers r_u are generated for each decision step.

9. Evaluate the offspring population (Q_0) , *i.e.* the simulation model evaluates N decision variable combinations.

Now, the loop initiates, while the current generation $t < t_{\text{max}}$, where t_{max} is the fixed number of iterations permitted, divided by the population size N, *i.e.* $t_{\text{max}} = 10$. Do the following:

- 10. Combine the parent and offspring populations, *i.e.* $\mathcal{R}_t \leftarrow \mathcal{P}_t \cup \mathcal{Q}_t$, of size 2N.
- 11. Assign fitness values to the combined population \mathcal{R}_t .

The fitness values are used to determine which solutions should constitute the mating pool. NSGA-II generally follows the same structure as MOEAs, however, its *fitness assignment* and *selection procedures* distinguish it from other MOEAs. Each solution in the population is assigned two attributes that constitute its fitness, namely a Pareto rank and crowding distance. The procedure is as follows: The first step is to apply the fast non-dominated sorting algorithm (FNSA), described in Algorithm 5.5, in order to rank the solutions in \mathcal{R}_t according to their Pareto rank and partition them into their respective fronts. Thereafter, the solutions within each rank (or front) are ranked again according to their crowding distance by applying the crowding distance sorting and assignment algorithm, as described in Algorithm 5.6 [67]. Algorithm 2.1 is adapted to return the non-dominated fronts $\mathcal{F}_1, \ldots, \mathcal{F}_k$, as seen in Algorithm 5.5. The nature of Algorithm 5.5 is described using Figure 5.10.

The population of solutions in Figure 5.10a is ranked and partitioned into their respective sets and fronts (as depicted in Figure 5.10b). Front 1 (denoted by \mathcal{F}_1), contains all the solutions with a rank value of zero and Pareto rank, *i.e.* $\rho_1 = 0$ (denoted by \bullet), forming the Pareto front.

Algorithm 5.5: Fast non-dominated sorting algorithm, for a *min-max* bi-objective optimisation problem [67]



Output : The population of \mathcal{P} partitioned into k successive non-dominated fronts $\mathcal{F}_1, \ldots, \mathcal{F}_k$ with a corresponding Pareto rank ρ_1, \ldots, ρ_k and each solution x is also assigned a Pareto rank ρ_x , where $\mathcal{P}_S \leftarrow \mathcal{F}_1$, containing all the solutions with $\rho = 1$.





Figure 5.10: An illustration of candidate solutions partitioned into their respective sets and fronts and ranked accordingly, using Algorithm 5.5.

Next, \mathcal{F}_2 comprises all \bullet solutions with $\rho_2 = 1$, \mathcal{F}_3 comprises all \bullet solutions with $\rho_3 = 2$, and so forth until Front 5 (denoted by \bullet) is formed containing all the solutions with $\rho_5 = 4$.

As mentioned, Algorithm 5.6 is applied to each non-dominated front obtained by the FNSA. The algorithm takes as input the non-dominated front \mathcal{F} and returns the crowding distances d_1, \ldots, d_ℓ as output, where ℓ denotes the number of solutions in \mathcal{F} . The first step is to initialise all the crowding distances to zero. Thereafter, the solutions in each objective are sorted in ascending order irrespective of the optimisation context. Next, the distances of the boundary solutions are set to infinity. For the intermediary solutions (*i.e.* 2 to $\ell - 1$), the crowding distance is assigned by calculating the normalised distance between the solutions above and below its rank, using (5.9). The boundary solutions are always added first, with equal probability.

Algorithm 5.6: Crowding distance sorting and assignment algorithm [67] **Input** : A non-dominated front \mathcal{F} of cardinality ℓ . **Output** : The crowding distances d_1, \ldots, d_ℓ for each solution in front \mathcal{F} . // Number of solutions in front ${\cal F}$ 1 $\ell \leftarrow |\mathcal{F}|;$ for $i \leftarrow 1$ to ℓ do $\mathbf{2}$ $d_i \leftarrow 0;$ // Initialise crowding distances to zero 3 4 for f_k to $|f_m|$ do $I^k \leftarrow \mathsf{sort}(\mathcal{F}, k);$ 5 6 $d_1 = d_\ell = \infty;$ for $i \leftarrow 2$ to $(\ell - 1)$ do // Apply to all solutions between the endpoints 7 8 $d_i \leftarrow d_i + \frac{(f_k^{I[i+1]} - f_k^{I[i-1]})}{(f_k^{\max} - f_k^{\min})}$ (5.9)

The fitness values of two solutions may be compared according to the crowded comparison operator, given that each solution i is assigned a Pareto rank ρ_i and crowding distance d_i . Accordingly, solution a is said to dominate solution b, denoted as $a \prec_{cc} b$ if one of two scenarios, as stated in (5.10), are true. The first: if the solutions have different Pareto ranks (*i.e.* $\rho_a < \rho_b$) then $a \prec_{cc} b$, the solution with the lowest Pareto rank is preferred, *i.e.* solution a. The second: if two solutions a and b have the same Pareto ranks (*i.e.* $\rho_a = \rho_b$), then the solution with the greater crowding distance is preferred as they promote more diversity, *i.e.* if $d_a > d_b$ then $a \prec_{cc} b$ and solution a is preferred [67]. The crowded comparison operator is formally given as

$$a \prec_{cc} b$$
 if $\begin{cases} \rho_a < \rho_b, \text{ or} \\ \rho_a = \rho_b \text{ and } d_a > d_b. \end{cases}$ (5.10)

During the entire search process, an archive is maintained and captured in the elite vector *Elitist* and includes all the non-dominated solutions found at each generation t, *i.e.* the solutions in \mathcal{F}_1 with $\rho = 1$.

12. Maintain the elite vector by appending \mathcal{F}_1 to *Elitist*, denoted as *Elitist* \leftarrow *Elitist* $\cup \mathcal{F}_1$.

Next, the mating pool \mathcal{P}_{t+1} can be formed based on the crowded comparison operator \prec_{cc} fitness measure, as previously described.

13. Create the new mating pool \mathcal{P}_{t+1} by applying the selection operator.

The selection process is depicted in Figure 5.11. A selection operator is applied to the combined population \mathcal{R}_t , consisting of 2N solutions. The selection operator selects N solutions to form the mating pool, thereafter two offspring solutions are generated from two parent solutions, *i.e.* N parents are used to generate N offsprings. The selection operator selects solutions based on the crowded comparison operator.

It can be seen that the new population consists of the solutions in \mathcal{F}_1 and \mathcal{F}_2 because the solutions with the lower Pareto ranks are included first. A subset of \mathcal{F}_3 also forms part of the new population, *i.e.* $\mathcal{P}_{t+1} = \{\mathcal{F}_1, \mathcal{F}_2, D\}$, where $D \subset \mathcal{F}_3$.

Next, the offspring population can be generated from \mathcal{P}_{t+1} .

Chapter 5. Metaheuristics and Hyperheuristics



Figure 5.11: The fitness assignment procedure followed by NSGA-II to generate population \mathcal{P}_{t+1} , indicating the partitioning of the parent and offspring populations \mathcal{P}_t and \mathcal{Q}_t , adapted from [67].

- 14. Generate the offspring population Q_t by applying the crossover operator with probability p_c and mutation operator with probability p_m to the solutions in the mating pool.
- 15. Check for duplicate solutions and then replace them with non-duplicate solutions.

Again, a random number $r_u \sim \mathcal{U}(0, 1)$ is generated. If $r_u < p_c$ crossover is performed on decision variable *i*, otherwise, no modification is made. Similarly, a random number $r_u \sim \mathcal{U}(0, 1)$ is generated, *i.e.* different random numbers r_u are generated for each decision step. If $r_u < p_m$ then mutation is performed on decision variable *i*, otherwise, no modification is made.

- 15. Evaluate the offspring population Q_t , *i.e.* the simulation model evaluates N new decision variable combinations.
- 16. Increment the generation counter, $t \leftarrow t + 1$.

This concludes the loop which is repeated until the stopping condition $t \ge t_{\text{max}}$ is met. At termination, the archive *Elitist* is ranked, according to Algorithm 2.1 and returned as an approximate of the true Pareto set of solutions to the bi-objective simulation optimisation problem under consideration, *i.e.* $\mathcal{P}_S \leftarrow Elitist$. Recall that (although it is not explicitly reported) when a new generation is generated, the algorithm has to check if duplicate solutions are present, and if so, replace them with non-duplicate solutions to ensure that the expensive function evaluator does not unnecessarily evaluate a solution and is done for each algorithm. A pseudo-code description of the entire process described above is provided in Algorithm 5.7, with specific reference to the above-mentioned steps.

During each generation, solutions are selected based on fitness values (and not objective function values) to form the parent population [149]. The crowded comparison operator fitness measure (\prec_{cc}) enables the selection process used in NSGA-II to favour the exploration of diverse solutions [67], refer to the discussion around step 11. Next, an offspring population is created by applying reproduction operators (crossover and mutation) to the solutions in the parent population.

Α	lgorithm 5.7 : NSGA-II, adapted from [67, 222], for the s	imulation optimisation process			
	Input : Previously described using Figures 5.8a and 5.8b.				
	Output : The approximate Pareto set (\mathcal{P}_S) for the given simulation	lation problem.			
1	$t \leftarrow 0;$ // Init	tialise the generation counter			
2	$Elitist \leftarrow \emptyset;$	// Initialise Elitist			
3	Generate an initial population \mathcal{P}_0 ;	// Refer to step 5			
4	Evaluate \mathcal{P}_0 ;	// Refer to step 6			
5	Generate the offspring population \mathcal{Q}_0 ;	// Refer to step 8			
6	Evaluate \mathcal{Q}_0 ;	// Refer to step 9			
7	while $t < t_{max} \operatorname{do}$				
8	$\mathcal{R}_t \leftarrow \mathcal{P}_t \cup \mathcal{Q}_t;$	// Refer to step 10			
9	Assign fitness values, using Algorithm 5.5 and 5.6 respecti	vely; // Refer to step 11			
10	$Elitist \leftarrow Elitist \cup \mathcal{F}_1;$	// Refer to step 12			
11	Create the new mating pool of solutions (\mathcal{P}_{t+1}) ;	// Refer to step 13			
12	Generate offspring population \mathcal{Q}_{t+1} ;	// Refer to step 14			
13	Evaluate Q_{t+1} ;	// Refer to step 15			
14	$t \leftarrow t+1;$	// Refer to step 16			
15	Rank the solutions in <i>Elitist</i> , using Algorithm 2.1;	// Get non-dominated solutions			
16	$\mathcal{P}_S \leftarrow Elitist;$	// Approximate Pareto set			

Finally, a replacement scheme determines which solutions of the population will survive from the offspring and parent populations. This process, as depicted in Figure 5.12, is iterated until a stopping condition is met [93].



Figure 5.12: An illustration of a generation in EAs, adapted from [93].

Selection pressure is the paradigm in which solutions that have superior fitness values are selected with a higher probability, without it — the search process is arbitrary. Selection pressure and diversity preservation are complimentary [255]. The selection pressure guides the population to better solutions resulting in a loss of population diversity. Conversely, preserving population diversity means selecting worse solutions thereby negating the effect of selection pressure. Consequently, a trade-off exits and should be balanced to obtain a diverse population in close proximity to the true Pareto front, *i.e.* aiding the exploration and exploitation of the search and not premature convergence [171, 175]. The selection operators considered for its inclusion in this study are discussed next.

5.3.2 Selection operators

As mentioned, the selection operator determines which solutions in the parent population are used to reproduce (by method of reproduction operators) to create the offspring population. This strategy is based on the principle of *survival of the fittest*, where superior parents typically result in better and fitter offsprings. This selection process is critical as it guides the search toward the true Pareto front [93]. There are many selection strategies, namely, tournament selection, rank selection and roulettewheel selection, refer to [11, 14, 32, 103]. Roulette-wheel selection is dependent on the optimisation direction, whereas tournament and rank selection are not. For more information regarding roulette-wheel selection, the reader is referred to [209]. During tournament selection, k solutions are selected at random from the combined population R_t , which compete against each other. The solution with the best fitness value, as discussed in step 11, wins and is included in the mating pool. In this study, the tournament size (or number of solutions competing) is k = 2, referred to as binary tournament selection (BTS) [93, 209].

Rank selection (RS) simply selects the solutions with the best fitness values, as discussed in step 11 in §5.3.1, *i.e.* the mating pool consists of $\mathcal{F}_1, \mathcal{F}_2, \ldots$ until the mating pool consists of N solutions. Note that binary tournament selection enables solutions with lower fitness values to enter the mating pool. Rank selection also enables solutions with lower fitness values to enter the mating pool, however, solutions with the greatest fitness values are selected first and consequently, if the number of solutions in Front 1 is equal to the population size $(i.e. |\mathcal{F}_1| = N)$ then only non-dominated solutions are selected to enter the mating pool.

The decision to incorporate discrete vector representations is further motivated by the disadvantages of binary representations. The process in Figure 5.13 is as follows: crossover and mutation are performed in respect of the binary string (the genotype or encoded solution), which is then decoded (as the phenotype) and evaluated by the simulation model evaluator, *i.e.* the reproduction operators act on the genotype level while the simulation model evaluator uses the phenotype of the associated genotype. At this point the selection operator is applied and a mating pool created.

The selected parents are then encoded and crossover as well as mutation are performed on the encoded solutions, and so on. Real-coded reproduction operators have been suggested in the literature [64, 66, 69, 70] and adapted for discrete vector representations due to the unnecessary computational inefficiency of encoding and decoding solutions.



Figure 5.13: Illustration of the effect that changing a single bit has on the value the string represents, adapted from [93].

Furthermore, in order to move to a neighbouring solution, more than one bit might need to be changed which introduces incumbrance, as changing one bit does not translate into a neighbour. An example is presented in Figure 5.13, with specific focus on the binary string $[1\ 1\ 1\ 1] = 15$ — by converting a single bit the string can take the following values $\{7, 11, 13, 14\}$, two of which are neighbours and two that are not. There is, however, the option of converting the binary code to so-called Gray code, where changing one bit would result in a neighbouring solution.

However, the challenge remains of keeping the genotypes within the bounds of the decision variables and requires considerable effort. Consequently, a so-called direct encoding is used, where the genotype is similar to the phenotype. The type of operations that can be applied to a solution depends on its representation, *i.e.* if the solution representation is discrete then the application of the reproduction operators should modify the discrete value and return a discrete value for the solution and is dependent on the problem under consideration. Many optimisation problems have discrete search spaces [221, 252], as do the simulation problems in this study.

For this reason, techniques exist to adapt continuous heuristics, for example the recombination operators, for the discrete search space [52]. In this study, the technique of rounding off is used, *i.e.* the value generated by the continuous heuristic is simply rounded off to the nearest integer value. This approach was first used by Yoshida *et al.* [261] and is used in this study due to its simplicity and low computational cost.

The real-coded crossover operators that have been adapted for discrete search spaces and that are considered for its inclusion in this study, are discussed next.

5.3.3 Crossover operators

The next step in creating the offspring population is *crossover*. A crossover probability p_c is specified by the user and represents the proportion of crossover operations, *i.e.* p_c determines whether or not the crossover operator is applied. Crossover operators are *n*-ary operators [93]. In this study, the crossover operators are 2-ary (or binary), *i.e.* the crossover operator modifies two solutions in the parent population, simultaneously.

The general procedure is as follows: generate a random number $r_u \sim \mathcal{U}(0, 1)$, if $r_u < p_c$ the crossover is applied to two parent solutions. Crossover creates two new offspring solutions by using parts of both parent solutions with the purpose of inheriting some characteristics of both parents. The offsprings may potentially be generated far from the region being sampled, depending on how far the parent solutions are from each other and the crossover operator used, consequently introducing diversity into the population [104]. The primary task of the crossover operator is therefore to search the neighbourhood of the parent solutions without any bias to specific neighbourhoods. It is the task of the selection operator to guide the search towards better, fitter neighbourhoods [64]. As mentioned, if $r_u \ge p_c$, then the solution remains unchanged, *i.e.* the parents become the offsprings.

There are several real-coded crossover operators, namely simplex crossover [247], simulated binary crossover (SBX) [62, 66], Laplace crossover (LX) [69] and blend crossover, to name a few. SBX and LX are considered parent-centric, *i.e.* each parent has an equal probability of creating an offspring in its neighbourhood [93].

Simulated binary crossover

SBX uses a probability distribution function $P(\beta)$ to simulate the single-point crossover operator, as used in binary-coded operators [62, 66], to generate two offspring solutions.

The probability distribution function is given as

$$P(\beta) = \begin{cases} 0.5(\eta+1)\beta^{\eta}, & \text{if } \beta \le 1\\ 0.5(\eta+1)\beta^{\frac{1}{\eta+2}}, & \text{otherwise.} \end{cases}$$
(5.11)

The distribution index η controls the distance between the offsprings and the parents, where a large η results in a higher probability of generating offsprings that are near their parents. For MOO the interval value for η is suggested as [5, 10] [93]. The procedure of computing the offspring solutions x_1^{t+1} and x_2^{t+1} from parent solutions x_1^t and x_2^t is described in Algorithm 5.8, assuming $r_u < p_c$. First, a random number $r_u \sim \mathcal{U}(0, 1)$ is generated, thereafter the ordinate β can be determined by solving for (5.12) using r_u and specified η . Finally, the offsprings can be computed. For more detail, the reader is referred to [62, 66, 93].

Algorithm 5.8: Simulated binary crossover, adapted for the discrete search space [62, 66]

Input : Two parent solutions x_1^t and x_2^t and a specified value for the distribution index η . **Output** : Two discrete valued offspring solutions x_1^{t+1} and x_2^{t+1} .

- **1** Generate a random number, $r_u \sim \mathcal{U}(0, 1)$;
- ${\bf 2}\;$ Calculate β as

$$\beta = \begin{cases} 2r_u^{\frac{1}{\eta+1}}, & \text{if } r_u \le 0.5\\ \left[\frac{1}{2(1-r_u)}\right]^{\frac{1}{\eta+1}}, & \text{otherwise.} \end{cases}$$
(5.12)

3 Compute offsprings as

$$\begin{aligned} x_1^{t+1} &= 0.5 \lfloor (1+\beta) x_1^t + (1-\beta) x_2^t \rfloor \\ x_2^{t+1} &= 0.5 \lfloor (1-\beta) x_1^t + (1+\beta) x_2^t \rfloor \end{aligned}$$

Laplace crossover

The LX operator uses the Laplace distribution to create offsprings. The procedure of computing the offspring solutions x_1^{t+1} and x_2^{t+1} from parent solutions x_1^t and x_2^t is described in Algorithm 5.9, assuming $r_u < p_c$.

Algorithm 5.9:	Laplace crossover	, adapted for the	e discrete search space	[69]
----------------	-------------------	-------------------	-------------------------	------

Input : Two parent solutions x_1^t and x_2^t and specified values for the location parameter a and the scale parameter b.

- **Output** : Two discrete valued offspring solutions x_1^{t+1} and x_2^{t+1} .
- 1 Generate a random number, $r_u \sim \mathcal{U}(0,1)$;
- **2** Calculate β as

$$\beta = \begin{cases} a - b \log_e(r_u), \text{ if } r_u \le 0.5\\ a + b \log_e(r_u), \text{ otherwise.} \end{cases}$$
(5.13)

3 Compute offsprings as

$$\begin{split} x_1^{t+1} &= x_1^t + \lfloor \beta \ \mathsf{abs}(x_1^t - x_2^t) \rfloor \\ x_2^{t+1} &= x_2^t + \lfloor \beta \ \mathsf{abs}(x_1^t - x_2^t) \rfloor \end{split}$$

Similar to SBX, the value of b (the scale parameter) influences the distance between the offsprings and the parents. However, for smaller values of b, offsprings are likely to be near the parents. For fixed values of a and b, the offspring are created proportional to the parents, *i.e.* the offspring are expected to be near each other if the parents are near each other, and *vice versa* [69]. First, a random number $r_u \sim \mathcal{U}(0, 1)$ is generated, thereafter β can be determined by solving for (5.13) using r_u and specified values for a and b, then the offsprings can be computed.

Blend crossover

The BX procedure of computing offsprings x_1^{t+1} and x_2^{t+1} from x_1^t and x_2^t is described in Algorithm 5.10, assuming $r_u < p_c$. First, a random number $r_u \sim \mathcal{U}(0, 1)$ is generated, thereafter the offsprings are created randomly in the range $[x_1^t - r_u(x_2^t - x_1^t), x_2^t + r_u(x_2^t - x_1^t)]$ [203].

Algorithm 5.10: Blend crossover, adapted for the discrete search space [203]

Input : Two parent solutions x_1^t and x_2^t , where $x_1^t < x_2^t$.

Output : Two discrete valued offspring solutions x_1^{t+1} and x_2^{t+1} .

1 Generate a random number, $r_u \sim \mathcal{U}(0,1)$;

2 Compute offsprings as

$$x_1^{t+1} = \lfloor r_u x_1^t + (1 - r_u) x_2^t \rfloor$$
$$x_2^{t+1} = \lfloor (1 - r_u) x_1^t + r_u x_2^t \rfloor$$

Upon initial experimentation, the LX and BX operators proved to be troublesome, generating infeasible offsprings and proving difficult to keep values within bound irrespective of the user-specified parameter values (*e.g.* the location or scale parameters). Consequently, LX and BX are excluded from further consideration.

Next, the real-coded mutation operators adapted for discrete search spaces and considered for its inclusion in this study, are discussed.

5.3.4 Mutation operators

After crossover, the next step in creating the offspring population, is *mutation*. A mutation probability p_m is specified by the user and is typically markedly lower than the crossover probability. In contrast to crossover operators, mutation operators are *unary operators*, modifying a single solution of the parent population independent of the remainder of the population [65]. The general procedure is as follows: a uniform random number $r_u \sim \mathcal{U}(0,1)$ is generated, if $r_u < p_m$ the solution is mutated, *i.e.* the current variable value (which is the parent if crossover was not performed or offspring if crossover was performed) is perturbed to a neighbouring value (the offspring). This is performed for each respective decision variable and each solution.

Mutation introduces and maintains diversity in the population [104, 119]. At the start of the search, it may be desirable that the mutation operator favours values markedly different from the current solution, but as the search progresses it should favour values more similar to the parent solutions [203]. As mentioned, if $r_u \ge p_m$, then the solution remains unchanged. There are several real-coded mutation operators, namely random mutation [57], Gaussian mutation [225], polynomial mutation (PM) [59, 63], power mutation [70], dynamic mutation (DM) [203], non-uniform mutation, to name a few. PM and DM operators create solutions within the user-specified range [l, u], naturally motivating their inclusion in this study.

Polynomial mutation operator

PM uses the polynomial probability distribution, with a user-defined distribution index ($\eta_m \approx 20$), to perturb a solution near the parent (or crossover solution) [93]. The mutation operator creates a solution x' in the range [l, u], by mutating x as described in Algorithm 5.11, assuming

 $r_u < p_m$. First, a random number $r_u \sim \mathcal{U}(0, 1)$ is generated, thereafter δ can be determined by solving for (5.14) using r_u and specified η_m . Finally, the mutated offspring can be computed.

Algorithm 5.11: Polynomial mutation, adapted for the discrete search space [65, 93]

- **Input** : Solution x and distribution index η_m . If crossover was not applied, then x is the parent solution, otherwise x is the offspring solution created by from the crossover operation. **Output** : Mutated offspring x'.
- **1** Generate a random number, $r_u \sim \mathcal{U}(0, 1)$;
- **2** Calculate δ as

$$\delta = \begin{cases} 2r_u^{\frac{1}{\eta_m+1}} - 1, & \text{if } r_u < 0.5\\ 1 - 2(1 - r_u)^{\frac{1}{\eta_m+1}}, & \text{otherwise.} \end{cases}$$
(5.14)

 ${\bf 3} \ {\rm Apply \ mutation \ operator}$

$$x' = \begin{cases} x + \lfloor (x-l)\delta \rfloor, & \text{if } r_u \le 0.5, \\ x + \lfloor (u-x)\delta \rfloor, & \text{otherwise.} \end{cases}$$

Dynamic mutation

As mentioned, it may be desirable to start out with a mutated solution where all possible values are equally probable (called uniform mutation) and as the search progresses to focus the search near the parent solutions [203]. The mutated solution x' is created from solution x, for $r_u \sim \mathcal{U}(0,1)$, as described in Algorithm 5.12, assuming $r_u < p_m$. First, a random number $r_u \sim \mathcal{U}(0,1)$ is generated, thereafter α can be determined by solving for (5.15) using r_u and specified γ . Finally, the mutated offspring can be computed.

Algorithm 5.12: Dynamic mutation, adapted for the discrete search space [203]

Input : Solution x and mutation parameter γ . If crossover was not applied, then x is the parent solution, otherwise x is the offspring solution created by from the crossover operation. **Output** : Mutated offspring x'.

- **1** Generate a random number, $r_u \sim \mathcal{U}(0, 1)$;
- **2** Calculate ξ as

$$\xi = \left(1 - \frac{t - 1}{t_{\max}}\right)^{\gamma},\tag{5.15}$$

where t is the current generation count, t_{max} the total number of generations and β the user specified mutation parameter. For the purposes of this study $\beta = 5$. Apply mutation operator

$$x' = \begin{cases} l + \lfloor (r_u - l)^{\xi} (x - l)^{1 - \xi} \rfloor, & \text{if } r_u \le x, \\ u - \lfloor (u - r_u)^{\xi} (u - x)^{1 - \xi} \rfloor, & \text{otherwise.} \end{cases}$$

In conclusion, the properties of the NSGA-II algorithm are stated. The NSGA-II algorithm

- 1. is a population-based search technique,
- 2. has a biological basis,

- 3. uses an archive to preserve non-dominated solutions (*Elitist*) during the search,
- 4. employs a Pareto ranking scheme to find non-dominated solutions,
- 5. achieves proximity to the true Pareto front via the selection operator,
- 6. and ensures diversity of the search *via* the crowded comparison operator, crossover and mutation operators.

Finally the DBMOSA algorithm is discussed.

5.4 Dominance-based bi-objective simulated annealing

The single-objective SA algorithm has been adapted to solve MOOPs, DBMOSA being one such manifestation. SA was first proposed by Kirkpatrick *et al.* [143] and later Smith *et al.* developed the multi-objective adaptation DBMOSA [233]. For a summative description of the different multi-objective extensions of SA, the reader is referred to [236]. The following section provides a detailed description of the DBMOSA algorithm, supported by the user-interface of the DBMOSA tool used in Tecnomatix, to illustrate the integration of simulation and optimisation.

5.4.1 The DBMOSA algorithm

Figures 5.14a and 5.14b are purposefully included to elucidate the simulation optimisation process *via* the DBMOSA tool as it is used in Tecnomatix. The steps for using the DBMOSA tool are as follows: starting with the *Definition tab* as shown in Figure 5.14a.

🚍 DBMOSA	X BBMOSA X
Show	Show
Start Stop Reset	Start Stop Reset
Experiment: Observation:	Experiment: Observation:
Definition Parameterisation Evaluation	Definition Parameterisation Evaluation
Decision Variables: Number of decision variables: 2 * Define Input Variables	Initialise: Probability of accepting non-improving solutions: 0.80
Objective Functions: Define Output Variables	Calculate the Initial Temperature
Neighbouring Moves:	Cooling: 0.75 , [0.75,1) Accepts: 7 •
Move 1 Observations per solutions: 100	Heating: 1.2 , (1,1.25) Attempts: 10 • Stopping Condition: 15 (poor epochs)
OK Cancel Apply	OK Cancel Apply
(a) Definition tab	(b) Parameterisation tab

Figure 5.14: The DBMOSA tool user-interface.

- 1. Define the number of decision variables, then click the *Define Input Variables* button and define the variables in terms of their paths, lower and upper bounds and initial solutions, as seen in Figure 5.15.
- 2. Define the output variables, click the *Define Output Variables* button, in terms of their paths and select the optimisation direction, *i.e. Min* or *Max* as seen in Figure 5.5b.
- 3. Choose the neighbouring move operator that should be applied. Refer to §5.4.4 for the detailed documentation regarding the respective move operators.

Decision variable (DV) path:	Lower bound	Upper bound	Initial solution

Figure 5.15: The user-interface for the decision variable table used for DBMOSA in Tecnomatix.

The next step is to navigate to the *Parameterisation tab*, as shown in Figure 5.14b:

- 4. Input the *Cooling* and *Heating* schedule parameters, select the number of *Accepts* before the temperature should decrease and the number of *Attempts* before it should increase.
- 5. Input the Stopping Condition, i.e. the number of poor epochs, denoted by maxpoor Epochs.
- 6. Select the probability of accepting non-improving moves and then click the *Calculate the Initial Temperature* button, refer to §5.4.3.

After the initial temperature has been calculated the *Start* button appears. The above-mentioned parameters are used as input to Algorithm 5.13, click the *Start* button to execute the DBMOSA algorithm and optimise the simulation problem at hand. As mentioned, the initial solution is generated randomly, represented by \boldsymbol{x} . The algorithm starts by initialising the iteration counter, $t \leftarrow 1$, the counters $accepts \leftarrow 1$ (since the initial solution is accepted), $attempts \leftarrow 0$, poor epochs, $poorEpochs \leftarrow 0$ and epochs, $epochs \leftarrow 0$.

1. Evaluate the initial solution (\mathbf{x}) , *i.e.* the simulation model evaluates \mathbf{x} .

An archive \mathcal{A} is maintained throughout the search process and contains all the non-dominated solutions found. Because \boldsymbol{x} is the first solution evaluated it is considered non-dominated and enters the archive. However, as the search progresses, \boldsymbol{x} might become dominated after which it is removed from the archive.

2. Initialise the archive to contain the initial solution, *i.e.* $\mathcal{A} \leftarrow \boldsymbol{x}$.

Now the loop initiates, while the current iteration is less than or equal to the fixed number of iterations permitted (*i.e.* $t \leq 1000$), do the following:

- 3. Increment the iteration counter, $t \leftarrow t + 1$.
- 4. Generate a neighbouring solution x' from the previously accepted solution x according to the selected move operator (Move operator 1–4), denoted by $x' \leftarrow \text{move operator}(x)$. Refer to §5.4.4.

6. Evaluate x', *i.e.* the simulation model evaluates x'.

Smith *et al.* [233] proposed the use of an energy function which is defined in terms of the proposed archive $(\tilde{\mathcal{A}})$ containing all the non-dominated solutions found as well as the current and neighbouring solutions, *i.e.* $\tilde{\mathcal{A}} = \mathcal{A} \cup \{x\} \cup \{x'\}$. Consequently, the change in energy for the current solution x and neighbouring solution x' may be expressed as

$$\Delta_E(\boldsymbol{x'}, \boldsymbol{x}) = \frac{|\tilde{\mathcal{A}}_{\boldsymbol{x'}}| - |\tilde{\mathcal{A}}_{\boldsymbol{x}}|}{|\tilde{\mathcal{A}}|}, \qquad (5.16)$$

where the difference is normalised by dividing by $|\tilde{\mathcal{A}}|$ (the number of solutions in the archive). It has been reported that normalising the energy promotes both coverage of and convergence towards the true Pareto set. The archive $\tilde{\mathcal{A}}$ includes the current solution to ensure that $\Delta_E(\mathbf{x'}, \mathbf{x}) < 0$ when the neighbouring solution dominates the current solution (*i.e.* $\mathbf{x'} \prec \mathbf{x}$) [233].

In Figure 5.16 an illustration is presented of the nature of the energy measure for the current solution \boldsymbol{x} and neighbouring solution $\boldsymbol{x'}$, where the number of solutions in $\tilde{\mathcal{A}}$ is 11 (denoted as $|\tilde{\mathcal{A}}| = 11$) and $|\mathcal{A}| = 9$. It can be seen that \boldsymbol{x} is dominated by 3 solutions in the archive (denoted as $|\tilde{\mathcal{A}}_{\boldsymbol{x}}| = 3$), and $\boldsymbol{x'}$ is dominated by 1 solution (denoted as $|\tilde{\mathcal{A}}_{\boldsymbol{x'}}| = 1$). The resulting change in energy of $\Delta_E(\boldsymbol{x'}, \boldsymbol{x}) = \frac{1-3}{11} = -\frac{2}{11}$. Therefore, accepting $\boldsymbol{x'}$ as the new current solution for the next iteration would encourage exploration by guiding the search towards less explored neighbourhoods of the non-dominated front (this is because $\boldsymbol{x'}$ is dominated by fewer solutions than \boldsymbol{x}) [128].



Figure 5.16: An example illustrating the energy measure of current solution x and neighbouring solution x' for a min-min bi-objective problem with objective functions f_1 and f_2 , adapted from [128].

- 5. Define the proposed archive $\tilde{\mathcal{A}}$ that includes the current and the neighbouring solution, *i.e.* $\tilde{\mathcal{A}} \leftarrow \{\boldsymbol{x}, \boldsymbol{x'}\}$. The number of solutions in the archive is (currently) two.
- 6. Define the subset of solutions in \mathcal{A} that dominates \boldsymbol{x} (in terms of objective function values), *i.e.* $\tilde{\mathcal{A}}_{\boldsymbol{x}} = \{ \mathbf{y} \in a\mathcal{P}_S \mid \mathbf{y} \prec \boldsymbol{x} \}.$
- 7. Define the subset of solutions in $\tilde{\mathcal{A}}$ that dominates $\mathbf{x'}$ (in terms of objective function values), *i.e.* $\tilde{\mathcal{A}}_{\mathbf{x'}} = \{\mathbf{y} \in a\mathcal{P}_S \mid \mathbf{y} \prec \mathbf{x'}\}.$
- 8. Calculate the change in energy (Δ_E) for \boldsymbol{x} and $\boldsymbol{x'}$ using (5.16).

Every solution that is *accepted* during the search, is a candidate for inclusion in the archive (\mathcal{A}) . If the accepted solution is not dominated by any solution in the archive, it is included. If the accepted solution dominates solutions currently in the archive, it is included, and the dominated solutions are removed [233]. If, however the accepted solution is dominated by any other solution in the archive, it is not included, as discussed in §2.2.3.

10. The objective values of \boldsymbol{x} and $\boldsymbol{x'}$ are compared and if $\boldsymbol{x'}$ is not dominated by any other solution in the archive, *i.e.* $|\mathcal{A}_{\boldsymbol{x'}}| = 0$, it is accepted as the new current solution for the next iteration (with a probability of 1). Otherwise, $\boldsymbol{x'}$ is accepted stochastically, according to the *Metropolis acceptance rule* [180], where the probability of accepting a non-improving neighbouring solution is given in (5.17).

To elucidate this further, take for example a bi-objective max-max problem with two decision variables $\{x_1, x_2\}$, if $f(\mathbf{x'_1}, \mathbf{x'_2}) \ge f(\mathbf{x_1}, \mathbf{x_2})$, then $\mathbf{x'}$ is accepted with a probability 1, otherwise $\mathbf{x'}$ is accepted according to the Metropolis acceptance rule, given as

$$P_r(\boldsymbol{x'}) = \min\left\{1, \, \exp^{\left(-\frac{\Delta_E(\boldsymbol{x'},\boldsymbol{x})}{q_t}\right)}\right\},\tag{5.17}$$

where q_t is the temperature at iteration t [233].

11. Generate a random number $r_u \sim \mathcal{U}(0, 1)$. If $r_u < P_r(\mathbf{x'})$ then the neighbouring solution enters the archive $(\tilde{\mathcal{A}})$ and becomes the new current solution, *i.e.* $\mathbf{x} \leftarrow \mathbf{x'}$.

DBMOSA is executed iteratively, each iteration is known as an epoch, where the temperature q_t remains constant during an epoch and only increases or decreases when an epoch is incremented. An epoch is incremented if either *maxAccepts* or *maxAttempts* are reached. Suppose the maximum number of attempts (or accepts) are defined as 5, denoted as *maxAttempts* \leftarrow 5 (or *maxAccepts* \leftarrow 5), then *epochs* is incremented if *attempts* = 5 (or if *accepts* = 5), irrespective if the attempts or accepts occured successively.

- 12. If the solution is accepted then *accepts* is incremented, *i.e.* $accepts \leftarrow accepts + 1$.
- 13. If accepts $\geq maxAccepts$ then $epochs \leftarrow epochs + 1$ and the temperature q_t is cooled. Refer to §5.4.2.
- 14. If the solution is attempted then attempts is incremented, *i.e.* attempts \leftarrow attempts + 1.
- 15. If attempts $\geq maxAttempts$ then $epochs \leftarrow epochs + 1$ and the temperature is heated. Refer to §5.4.2.

Again, suppose $maxAttempts \leftarrow 5$, then *poorEpochs* is incremented when maxAttempts number of *successive epochs* have elapsed without accepting a solution. This is illustrated in the example given in Table 5.1. It can be seen that after 5 solution evaluations are attempted, *epochs* is incremented and if 5 epochs are incremented consecutively, without accepting a solution then *poorEpochs* is incremented, *i.e.* if 25 attempts are made then *poorEpochs* \leftarrow *poorEpochs*.

16. If a solution is attempted (*i.e.* not accepted) maxAttempts consecutive number of times then poor Epochs is incemented, *i.e.* poor Epochs \leftarrow poor Epochs + 1.

	attemps		epochs		poorEpochs
1	\checkmark		0		0
2	\checkmark		0		0
3	\checkmark		0		0
4	\checkmark		0		0
5	\checkmark	+1	1		0
:	÷		÷		:
21	\checkmark		4		0
22	\checkmark		4		0
23	\checkmark		4		0
24	\checkmark		4		0
25	\checkmark	+1	5	+1	1

Table 5.1: An example illustrating the incrementing process followed for an epoch and a poorEpoch.

This concludes the loop. The algorithm terminates when the current iteration is greater than the maximum number of iterations permitted, *i.e.* $t > t_{\text{max}}$, where $t_{\text{max}} = 1\,000$ solution evaluations. At termination, the archive \mathcal{A} is returned as an approximate of the true Pareto set of solutions to the bi-objective simulation optimisation problem under consideration, *i.e.* $\mathcal{P}_S \leftarrow \mathcal{A}$.

It is worth mentioning that there are other stopping conditions that may be implemented, namely terminating when the number of *poorEpochs* $\geq maxPoorEpochs$ (as depicted in the user-interface) or when $q_t \geq q_f$, where q_f is a user-specified final temperature. Specific reference is made to this in Algorithm 5.13.

Although it is not explicitly reported, when a neighbouring solution is generated, the algorithm has to check if it is a duplicate solution, and if so, replace it with a non-duplicate solution. Again, this is done to ensure that the expensive function evaluator does not unnecessarily evaluate a solution and is done for each algorithm.

A pseudo-code description of the entire process described above is provided in Algorithm 5.13, with specific reference to the above-mentioned steps.

The temperature q_t controls the probability of accepting non-improving solutions. If the temperature is high, the probability of accepting a non-improving solution is greater and *vice versa*. The temperature is controlled by the chosen *annealing schedule*, this process and the annealing schedule adopted in this study is discussed next.

5.4.2 The annealing schedule

The annealing schedule determines by how much the temperature is decreased (using the cooling schedule) or increased (using the reheating schedule). Recall that the temperature is only decreased or increased if an epoch is incremented, which is based on what happens during the search, *i.e.* how many solutions are accepted or not. If an excessive number of solutions are accepted, the cooling schedule decreases the temperature at some rate and, conversely, if too few solutions are accepted, the reheating schedule increases the temperature at some rate (depending on the annealing schedule).

Intuitively, the cooling schedule encourages exploitation, since the probability of accepting a non-improving solution becomes less significant. On the other hand, the reheating schedule encourages exploration, since the probability of accepting a non-improving solution becomes more significant [169, 233], as depicted in Figure 5.17.

Algorithm 5.13: The DBMOSA algorithm adapted from [233], for the simulation optimisation process

: Previously described using Figures 5.14a and 5.14b. Input **Output** : The approximate Pareto set (\mathcal{P}_S) for the given simulation problem. **1** Initialise the iteration counter, $t \leftarrow 1$; **2** Initialise counters, $accepts \leftarrow 1$, $attempts \leftarrow 0$ and $poorEpochs \leftarrow 0$; **3** Initialise the number of epochs, $epochs \leftarrow 0$; 4 Evaluate the initial solution x; // Refer to step 1 5 Initialise the archive, $\mathcal{A} \leftarrow \{x\}$; // Refer to step 2 while $t \le t_{max}$ do // Other stopping condition(s) (e.g. poorEpochs < maxPoorEpochs) 6 $t \leftarrow t + 1;$ // Refer to step 3 7 Generate a neighbouring solution x'; 8 // Refer to step 4 9 Evaluate x': // Refer to step 6 Calculate the change in energy $\Delta_E(\mathbf{x'}, \mathbf{x})$, according to (5.16); // Refer to step 5-8 $\mathbf{10}$ if $|\mathcal{A}_{\boldsymbol{x}'}| = 0$ then 11 Add x' to the archive, $\mathcal{A} \leftarrow \mathcal{A} \cup \{x'\}$; // Refer to step 10 12 $\mathbf{13}$ $x \leftarrow x';$ for $y \in \mathcal{A}$ do 14 $\mathbf{15}$ if $x \prec y$ then $\mathcal{A} \leftarrow \mathcal{A} \setminus \{y\};$ // Remove y from archive 16 $accepts \leftarrow accepts + 1;$ // Refer to step 12 17 else 18 $| A \leftarrow |\tilde{\mathcal{A}}| + 1;$ // Increment the number of solutions in the archive 19 $\mathbf{20}$ else Generate a random number, $r_u \sim \mathcal{U}(0, 1)$; // Refer to step 11 21 $\mathbf{22}$ if $r_u < P_r(\mathbf{x'})$ then Add x' to the archive, $\tilde{\mathcal{A}} \leftarrow \tilde{\mathcal{A}} \cup \{x'\}$ 23 $x \leftarrow x';$ $\mathbf{24}$ $A \leftarrow |\tilde{\mathcal{A}}| + 1;$ // Increment the number of solutions in the archive 25 $\mathbf{26}$ $accepts \leftarrow accepts + 1;$ // Refer to step 12 27 else $attempts \leftarrow attempts + 1;$ // Refer to step 14 28 **29** if $accepts \geq maxAccepts$ then $epochs \leftarrow epochs + 1;$ // Refer to step 13 30 // According to cooling schedule, refer to §5.4.2 31 Decrease the temperature; 32 accepts $\leftarrow 0$; // Reset counter **33** else if $attempts \geq maxAttempts$ then $epochs \leftarrow epochs + 1;$ // Refer to step 15 34 // According to reheating schedule, refer to §5.4.2 Increase the temperature; $\mathbf{35}$ attempts $\leftarrow 0$; // Reset counter 36 **37** if successiveAttempts = maxAttempts then $poorEpochs \leftarrow poorEpochs + 1;$ // Refer to step 16 38 **39** $\mathcal{P}_S \leftarrow \mathcal{A};$ // Approximate Pareto set

The choice between different annealing schedules involves a compromise between the quality of the solutions found and the speed of cooling [93]. SA is computationally analogue to the process of physically annealing a metal, *i.e.* to slowly cool a metal to induce an equilibrium state of low-energy. Similarly, if the temperature is decreased sufficiently slow, then betterquality solutions are obtained, however requires significantly more computational time. Some annealing schedules include *linear* schedules, *exponential* schedules [143], *geometric*, *Geman*

100

and Geman or logarithmic annealing schedules [92], nonmonotonic schedules [110] and adaptive schedules [93]. The geometric schedule is considered the most popular of the annealing schedules and is used for both cooling and reheating.



Figure 5.17: An illustration of how the probability of accepting non-improving moves changes as the temperature changes, resulting in either exploration or exploitation of the search.

The geometric cooling schedule is given by,

$$q_{t+1} \leftarrow \alpha q_t, \ \alpha \in \left[\begin{array}{c} 3\\ \overline{4}, 1 \end{array} \right),$$
(5.18)

where α is the cooling parameter. The geometric reheating schedule is given by,

$$q_{t+1} \leftarrow \beta q_t, \ \beta \in \left(1, \frac{5}{4}\right],$$

$$(5.19)$$

where β is the reheating parameter [233].

It has been reported that the overall *cooling rate* is more important than the specific cooling function. The initial temperature q_0 should also be considered. If the starting temperature is notably high, the search resembles random search, otherwise, the search resembles local search. Consequently, the initial temperature should be high enough to allow for sufficient exploration without being completely random [93].

5.4.3 The initial temperature

As mentioned previously, the DBMOSA tool determines an initial temperature by pressing the *Calculate Initial Temperature* button. The methodology followed is briefly described in this section. Many have proposed methods for determining a suitable initial temperature that is specific to the problem being optimised, refer to [127, 143, 249].

Typically, an initial temperature is found where the probability of accepting non-improving solutions is close to 1. The algorithm randomly generates trial solutions until the neighbouring solution is dominated by the solutions currently in the archive, thereafter the initial temperature q_0 can be determined, based on the probability chosen, by solving for

$$q_0 = -\frac{\Delta_E}{\ln P_r(\boldsymbol{x'})}.$$
(5.20)

In this study the initial temperature is chosen arbitrarily as $q_0 = 10$, which is relatively high so as to encourage exploration at the start of the search.

The motivation is to facilitate the hyperparameter tuning with specific focus on the *move op*erator (or neighbouring function) used and no other parameters that may influence the performance of the algorithm, *i.e.* observe the effect on the performance of DBMOSA based on the move operator used. It may be that the chosen initial temperature is better for a specific problem, however, this does not influence the analysis of which move operators are best, since single-problem comparisons are done.

5.4.4 Neighbourhood move operators

In order to apply the DBMOSA algorithm, a perturbation method (or neighbouring move operator) has to be defined [222]. The move operators reported for SA typically perform a *random perturbation* of the current solution \boldsymbol{x} , according to a set of feasible moves while considering the specific combinatorial context of the problem. The move operators should keep some part of the solution and strongly perturb another part to escape local optima and possibly move to more attractive neighbourhoods, as depicted in Figure 5.18.



Figure 5.18: The perturbation of solution x for a minimisation problem, adapted from [93].

Consequently, a neighbour is generated by the application of a *move operator* that performs a small perturbation of the solution, where perturbation is a random move considering the set of feasible moves [93]. The motivation behind the proposed move operators is the lack of move operators applicable for discrete search spaces.

Move operator 2 simply uses a linear regression model to guide the search. Move operators 3 and 4 borrow mutation operators from the evolutionary computation literature and adapt them for discrete search spaces. These constitute the set of neighbourhood move operators proposed for the DBMOSA algorithm and are discussed in the following sections, respectively. Note that all move operators consider the solution representation and generate feasible solutions.

Move operator 1: Temperature dependent move

Move operator 1 was designed to change as the temperature parameter q_t changes and was designed during a quantitative pilot study. Consequently, when the temperature is high, the corresponding move should be large as to encourage exploration, and conversely, when the temperature is small the corresponding move should be small as to encourage exploitation. The proposed move operator is described in Algorithm 5.14 and is applied to the solution previously accepted into the archive.

5.4. Dominance-based bi-objective simulated annealing

Algorithm 5.14: Neighbour	ing move 1		
Input : Previously accept	ed solution x .		
Output : Neighbouring solu	tion x' .		
1 Generate a random number	$r, r_u \sim \mathcal{U}(0,1);$	// Random number b	between 0 and 1
2 if $r_u < 0.5$ then			
3			
	$x \leftarrow x + \min(u - x)$	x - l, stepsize)	(5.21)
4 else			
5			
	$x \leftarrow x - \min(u - x)$	x - l, stepsize)	(5.22)
L			

Table 5.2: The temperature ranges used to specify the stepsize of a move.

q_t	stepsize
$q_t \ge 0.75q_0$	(0.3 - 0.39)u
$0.5q_0 \le q_t < 0.75q_0$	(0.20 - 0.29)u
$0.25q_0 \le q_t < 0.5q_0$	(0.10 - 0.19)u
$q_t < 0.25q_0$	(0.01 - 0.09)u

Suppose that the initial temperature for a given problem is $q_0 = 2$ and the current temperature for iteration t is $q_t = 1.6$. Then the relationship $\frac{q_t}{q_0} = 0.8$, *i.e.* the temperature range is $q_t \ge 0.75q_0$ as seen in Table 5.2, and the corresponding step size is between 30%–39% of the upper bound for decision variable *i*. Also, suppose that the lower and upper bounds for decision variable *i* are l = 0 and u = 100 and currently x = 50. The step size can be is determined by first generating a random number between 0.3 and 0.39. Suppose 0.32 was generated and that $r_u > 0.5$ then the neighbouring solution x' is created by adding the minimum of expression in (5.21) as

$$x = 50 + \min(100 - 50, 50 - 0, (0.32)100)$$

= 82,

otherwise the neighbouring solution x' is created by subtracting the minimum of expression (5.22) as

$$x = 50 - \min(100 - 50, 50 - 0, (0.32)100)$$

= 5.

Move operator 2: Linear regression

Move operator 2 applies a linear regression model to the solutions that are evaluated and then uses the coefficients for the respective decision variables to guide the search. The regression model is updated after every 50 solution evaluations. The motivation behind move operator 2 is its ease of implementation when compared with other regression models. Linear regression is discussed in Chapter 3. The idea is to use the coefficients in respect of the decision variable's effect on the two objectives to determine how far the decision variables should move from its current position.

Move operator 3: Dynamic and polynomial mutation

As mentioned, move operator 3 and 4 are borrowed from evolutionary computation literature and adapted to form move operators that mutate discrete vector representations. Polynomial and dynamic mutation are combined to form move operator 3 discussed in §5.3.4 and §5.3.4. The mutation operators are applied as described in Algorithm 5.15, to the solution previously accepted into the archive. For every decision variable, a random number $r_u \sim \mathcal{U}(0,1)$ is generated and if $r_u < 0.5$ polynomial mutation is applied, otherwise, dynamic mutation is applied, however (5.15) is adapted as $\xi = 1 - |(\text{counter mod } 100)/(10)|^5$, where counter is the current solution. By combining polynomial mutation and dynamic mutation, the hope is that a balance between exploitation and exploration is established.

Algorithm 5.15: Neighbouring move 3	
Input : Previously accepted solution x .	
Output : Neighbouring solution x' .	
1 Generate a random number, $r_u \sim \mathcal{U}(0, 1)$;	// Random number between 0 and 1
2 if $r_u < 0.5$ then	
3 Polynomial mutation;	// See Section 5.3.4
4 else	
5 Dynamic mutation;	// See Section 5.3.4

Move operator 4: Temperature dependent dynamic mutation

Move operator 4 adapts the dynamic mutation operator (discussed in $\S5.3.4$) to determine a move based on the relationship between the initial temperature and the current temperature. Consequently, (5.15) is adapted to include the temperature parameters and is given as

$$\xi = 1 - \left(\frac{q_0 - q_t}{q_0}\right), \tag{5.23}$$

where q_0 is the initial temperature and q_t is the temperature at iteration t. The smaller the value of ξ the closer the neighbouring solution x' is created to the previously accepted solution x.

In conclusion, the properties of the DBMOSA algorithm are stated. The DBMOSA algorithm

- 1. is a single-solution based search technique,
- 2. has a physics basis,
- 3. uses an archive to preserve non-dominated solutions (\mathcal{A}) during the search,
- 4. uses an archive $(\hat{\mathcal{A}})$ to control the probability of acceptancing non-improving solutions,
- 5. employs a Pareto ranking scheme to find non-dominated solutions,
- 6. achieves proximity to the true Pareto front via the cooling schedule,
- 7. and ensures diversity of the search *via* the energy measure and the Metropolis acceptance rule which determine whether or not a non-improving solution is accepted into the archive.

Recall that, based on the NFL theorem, no single algorithm outperforms all other algorithms when considering a diverse set of optimisation problems, irrespective of the performance metric used [257]. Consequently, the implementation of hyperheuristics proposed a way to try to mitigate, to some extent, the effect of the NFL theorem by combining more than one algorithm. AMALGAM was used as inspiration for both the population-based and single-solution based search hyperheuristics, as proposed in this study. As mentioned, AMALGAM is a population-based search hyperheuristic proposed by Vrugt and Robinson [254], which incorporates multiple metaheuristics (or LLHs) simultaneously and has also been adapted and applied to single-solution based searches, as proposed in [128].

The choice of hyperheuristic framework (or method for managing the LLHs) is crucial as it is often difficult to determine which LLH to apply to the current search space and serves as the motivation for using AMALGAM as inspiration. AMALGAM applies the LLHs in parallel (or proportionally) rather than successively. The heuristic selection procedure proposed in AMAL-GAM determines how many solutions each LLH may generate during each generation (or loop), *i.e.* given a population size N, then each LLH is allocated a proportion of N (based on its individual past performance) to contribute to and form the offspring population [38, 76]. The move acceptance procedure implemented in AMALGAM is that of NSGA-II, namely Pareto rank and crowding distance (as discussed in §5.3.1).

The next sections document each hyperheuristic, first, a detailed description of the populationbased search hyperheuristic inspired by AMALGAM is given, called the **bi-o**bjective **c**ross**e**ntropy and **g**enetic **a**lgorithm **h**yperheuristic (BOCEGAH). Next, a detailed description of the single-solution based search hyperheuristic, also inspired by AMALGAM is given, called the **bi-o**bjective **s**imulated **a**nnealing **h**yperheuristic (BOSAH). Each description is supported by the user-interface of the *Hyperheuristic tool* used in Tecnomatix, to illustrate the integration of simulation and optimisation.

5.5 The BOCEGAH algorithm

The typical procedure for solving MOOPs (approximately) by means of a population-based search approach is to implement a single algorithm that evolves a population of solutions (iteratively). AMALGAM follows a similar approach, however, multiple algorithms are implemented in parallel and each evolve a proportion of the population [254]. The MOOCEM and the NSGA-II are also applied in parallel (or simultaneously), evolving their own (partial) populations, enabling them to (ultimately) exploit their shared advantages while compensating for their respective weaknesses [192]. To conclude, by applying the LLHs in parallel, the hyperheuristic is able to determine the extent to which each LLH may contribute to the search, based on its past performance and effectively allocate solutions to each LLH for the current search space.

The BOCEGAH implements AMALGAMs heuristic selection procedure, however, each LLH (*i.e.* the MOOCEM and the NSGA-II) maintain their own move acceptance strategies as designed (specifically) for them. The reasoning pertains to the trade-off between exploration and exploitation of the search space and algorithms have to be designed to balance this trade-off. The simulation optimisation process (for population-based search) is illustrated using the Hyperheuristic tool in Tecnomatix, as shown in Figures 5.19a and 5.19b. Note that the same tool user-interface is used for both hyperheuristics. The steps for using the Hyperheuristic tool are as follows: starting with the *Definition tab* as shown in Figure 5.19a.

1. Define the number of decision variables (n), from the drop-down list, then click the Define

Input Variables button and define each decision variable in terms of its path and lower and upper bounds (as previously shown in Figure 5.5a).

2. Click the *Define Output Variables* button and define each output variable in terms of its path and select the optimisation direction from the drop-down list, *i.e. Min* or *Max* (as previously shown in Figure 5.5b).

🚍 HyperHeuristic	×	🗮 HyperHeuristic	×
Show		Show	
Start Stop Reset		Start Stop Reset]
Experiment: Observation:		Experiment: Observation:	
Definition Parameterization Evaluation		Definition Parameterization Evaluation	
Decision Variables: Number of decision variables: 2 • Define Input Variables		 Single solution-based search Population-based search 	
Objective Functions:		Population-based:	
Define Output Variables		Population size: 100	
Observations per solutions: 100 *	_	Maximum evaluations: 1000	
OK Cancel Apply		OK Cancel Apply	
(a) Definition tab		(b) Parameterisation tab	

Figure 5.19: The Hyperheuristic tool user-interface for population-based search.

The next step is to navigate to the *Parameterisation tab*, as shown in Figure 5.19b:

3. Select Population-based search, enter the Population size (N) and Maximum evaluations permitted.

Recall that the population size N = 100 and a fixed number of 1000 solution evaluations are permitted in this study. The above-mentioned parameters are used as input to Algorithm 5.16 as well as the best hyperparameter combination specific to MOOCEM and NSGA-II, as reported in Chapter 6. To execute the Hyperheuristic tool for population-based search and solve for the simulation problem at hand, click the *Start* button. A similar process is followed to execute and evaluate the Hyperheuristic tool as for the individual LLHs, where the initial population is also generated randomly.

Start with the initialisation of the parameters for the MOOCEM algorithm (as described in §5.2.2):

- 1. Initialise the elite vector, $Elite \leftarrow \emptyset$.
- 2. Initialise μ_i and σ_i (refer to §5.2.2 step 1).

Next, initialise the parameters for the NSGA-II algorithm (as described in $\S5.3.1$):

3. Initialise the elite vector $(Elitist \leftarrow \emptyset)$.

Next, initialise the parameters specific to the BOCEGAH algorithm. First, initialise the loop counter, $l \leftarrow 0$ and the number of LLHs (or heuristic components) considered, $nLLH \leftarrow 2$. Also, initialise the elite vector $Elites \leftarrow \emptyset$, used to combine the elite vectors of MOOCEM and NSGA-II at termination. Now, the general procedure starts.

- 4. Generate an initial population (of size N) by sampling from a truncated probability distribution, using Algorithm 5.1 and 5.2 as discussed in §5.2.1.
- 5. Evaluate the initial population, *i.e.* the simulation model evaluates N decision variable combinations.
- 6. Increment the loop counter, $l \leftarrow l + 1$.

At the start of the search procedure each LLH (*i.e.* MOOCEM and NSGA-II) is allocated the same number of solutions, *i.e.* for l = 1 and N = 100, each LLH is allocated N/2 = 50(offspring) solutions, denoted by $N_{h,l}$ for $h \in \{1, 2\}$, where h = 1 is the MOOCEM and h = 2is the NSGA-II. The allocations are updated for every loop l.

To summarise, for loop l = 1, MOOCEM generates 50 offspring solutions (referred to as offspring population $Q_{1,1}$) from the 50 solutions in the random initial population, and similarly, NSGA-II generates 50 offspring solutions ($Q_{2,1}$) from the remaining 50 solutions in the random initial population.

7. Initialise the allocations $N_{h,l}$ for $h \in \{1, 2\}$ and loop l = 1, *i.e.* $N_{h,1} \leftarrow N/2$.

Now, the loop initiates, while the current loop l < 10 (resulting in 1 000 solution evaluations), do the following. Generate the offspring populations $\mathcal{Q}_{h,l}$ for $h \in \{1, 2\}$. The process of generating the offspring populations is first described for the MOOCEM and then for the NSGA-II. The MOOCEM generates (partial) offspring population $\mathcal{Q}_{1,l}$ of size $N_{1,l}$ by following steps 11–30 in Algorithm 5.4. Also, refer to the supporting documentation given in §5.2.2.

8. Generate offspring population $\mathcal{Q}_{1,l}$ of size $N_{1,l}$ for loop l by following steps 11–30 in Algorithm 5.4.

Next, the NSGA-II generates the other partial offspring population $Q_{2,l}$ of size $N_{2,l}$ by following steps 8–12 and 14 in Algorithm 5.7 for loop l. Also, refer to the supporting documentation given in §5.3.1.

9. Generate offspring population $\mathcal{Q}_{2,l}$ of size $N_{2,l}$ for loop l by following steps 8–12 and 14 in Algorithm 5.7.

After the offspring populations $Q_{1,l}$ and $Q_{2,l}$ are generated, it is necessary to check if duplicate solutions are present and then replace them with non-duplicate solutions by generating new (offspring) solutions, as described in steps 8 and 9. This is a critical step that ensures that the expensive function evaluator does not unnecessarily evaluate solutions.

10. Check for duplicate solutions and then replace them with non-duplicate solutions.

Next, combine the individual offspring populations $\mathcal{Q}_{h,l}$ for $h \in \{1, 2\}$ and then evaluate the combined population, denoted by $\mathcal{P}_{C,l}$.

- 11. Combine the offspring populations generated in step 8 and 9 for loop $l, i.e. \mathcal{P}_{C,l} \leftarrow \mathcal{Q}_{1,l} \cup \mathcal{Q}_{2,l}$.
- 12. Evaluate the combined population $\mathcal{P}_{C,l}$ of size N, for loop l, *i.e.* the simulation model evaluates N decision variable combinations.

Next, determine the new allocations for loop l + 1, following steps 13–15.

13. Rank the combined population $\mathcal{P}_{C,l}$ for loop l, using Algorithm 5.5.

Next, determine how many solutions from $\mathcal{Q}_{1,l}$ and $\mathcal{Q}_{2,l}$ contributed to the new population, based on the non-dominated solutions found in step 13. The contributions are referred to as $C_{h,l}$ for $h \in \{1, 2\}$.

14. Determine the contributions $C_{h,l}$ for $h \in \{1, 2\}$, for loop l.

Now, the new allocations can be determined. This is where the heuristic selection procedure proposed in AMALGAM [254] is implemented. Each LLH is allocated a number of solutions based on their contributions $(C_{h,l})$. Suppose LLH $h \in \{1, \ldots, nLLH\}$ contributes $C_{h,l}$ solutions during loop l, then let the number of solutions that LLH $h \in \{1, 2\}$ may generate during loop l+1 be given by

$$N_{h,l+1} = \frac{\frac{C_{h,l}}{N_{h,l}}}{\sum_{h=1}^{nLLH} \frac{C_{h,l}}{N_{h,l}}} N.$$
(5.24)

When l = 1, then $N_{1,1}, N_{2,1} = 50$ because (as mentioned) each LLH $h \in \{1, 2\}$ is given equal opportunity at the start. Now, suppose h = 1 (MOOCEM) contributed 20 solutions (*i.e.* $C_{1,1} = 20$) and h = 2 (NSGA-II) contributed 15 solutions (*i.e.* $C_{2,1} = 15$). Then, the new allocation for l = 2 and h = 1 (*i.e.* $N_{1,2}$) can be calculated using (5.24), as depicted in Figure 5.20.



Figure 5.20: An example illustrating the heuristic selection procedure implemented for the BOCEGAH.

Accordingly, h = 2 (NSGA-II) is allocated $N_{2,2} = 43$ solutions. Note that provision is made to ensure that the number of solutions allocated to each LLH h for loop l adds up to the population size, *i.e.* $N_{1,l} + N_{2,l} = N$. Also, to ensure that a LLH is not eliminated from the search, a minimum allocation is set at 5, *i.e.* the minimum number of offspring solutions that LLH h may generate is $N_{h,l} = 5$. If the allocations for either MOOCEM or NSGA-II is less than 5, the allocation is updated to equal 5. For example, if $\{N_{1,2}, N_{2,2}\} = \{2, 98\}$ then it is updated so that $\{N_{1,2}, N_{2,2}\} = \{5, 95\}$.
109

- 15. Update the allocations $N_{h,l}$ for $h \in \{1, 2\}$ for loop l, using (5.24).
- 16. Increment the loop counter, $l \leftarrow l + 1$.

This concludes the loop. At termination, the elite vectors for MOOCEM (*Elite*) and NSGA-II (*Elitist*) are joined in elite vector *Elites*, *i.e. Elites* \leftarrow *Elite* \cup *Elitist*. Finally, the solutions in *Elites* are ranked using Algorithm 2.1 and returned as an approximate of the true Pareto set of solutions to the bi-objective simulation optimisation problem under consideration, *i.e.* $\mathcal{P}_S \leftarrow Elites$.

A pseudo-code description of the entire process described above is provided in Algorithm 5.16, with specific reference to the above-mentioned steps.

Α	lgorithm 5.16: The BOCEGAH algorithm for population-	based search
	Input : Previously described using Figures 5.19a and 5.19b.	
	Output : The approximate Pareto set (\mathcal{P}_S) for the given simulat	ion problem.
	/* MODCEM initialisation	*/
1	Refer to step 1-2;	
	/* NSGA-II initialisation	*/
2	Refer to step 3;	
	<pre>/* Hyperheuristic initialisation</pre>	*/
3	$l \leftarrow 0;$ //	Initialise the loop counter
4	$nLLH \leftarrow 2;$ // Ir	itialise the number of LLHs
5	$Elites \leftarrow \emptyset; \qquad \qquad //$	Initialise the elite vector
	/* General procedure starts	*/
6	Generate the initial population (of size N) using Algorithm 5.1;	// Refer to step 4
7	Evaluate initial population;	// Refer to step 5
8	$l \leftarrow l + 1;$ // Increment the lo	op counter, refer to step 6
9	Initialise the allocations, $N_{h,l}$ for $h \in \{1, 2\}$;	// Refer to step 7
10	while $l < 10$ do	
11	Follow steps 11–30 as described in Algorithm 5.4;	// Refer to step 8
12	Follow steps 8–12 and 14 as described in Algorithm 5.7;	// Refer to step 9
13	Combine onspring populations, $\mathcal{P}_{C,l} \leftarrow \mathcal{Q}_{1,l} \cup \mathcal{Q}_{2,l}$; Evaluate the combined offenning population $\mathcal{P}_{T,l}$ of size N:	// Refer to step 11
14	Bank the solutions in $\mathcal{D}_{\alpha\alpha}$ using Algorithm 5.5:	// Refer to step 12
16	Determine the contributions C_{1} , for $h \in \{1, 2\}$.	// Refer to step 14
17	Undate allocations $N_{i,j}$ for $h \in \{1, 2\}$ using (5.24):	// Refer to step 15
18	$l \leftarrow l + 1$	// Refer to step 16
10		
19	$Elites \leftarrow Elite \cup Elitist;$	// Combine elite vectors
20	Rank the solutions in <i>Elites</i> ;	// Using Algorithm 2.1
21	$\mathcal{P}_S \leftarrow Elites;$	// Approximate Pareto set

The next section documents BOSAH with specific focus on the integration of simulation and optimisation.

5.6 The BOSAH algorithm

The typical procedure for solving MOOPs (approximately) by means of a single-solution based search approach is to implement a single algorithm that perturbs a single solution (iteratively). For that reason, the heuristic selection procedure proposed for AMALGAM is adapted for successive heuristic assignment.

Each LLH (or move operator) is assigned successively instead of simultaneously to facilitate single-solution based search. The heuristic selection implemented in AMALGAM is employed to determine how many consecutive solutions a LLH may be assigned for. Three LLHs are assigned sequentially, *i.e.* move operator $1-2-3-1\cdots$, each generating their assigned number of solutions. For this reason, another step is proposed as part of the heuristic selection procedure to ensure that if a LLH is performing poorly then instead of completing its assigned number of solutions, the next move operator in the sequence is assigned the rest of its allocations. The purpose of this step is to ensure that the search does not stagnate (due to the NFL theorem), because the LLHs are assigned sequentially, it is possible that the best move operator for the current point in the search is not assigned and should be addressed accordingly.

The simulation optimisation process (for single-solution based search) is illustrated using the Hyperheuristic tool in Tecnomatix as shown in Figures 5.21a and 5.21b. The steps for using the Hyperheuristic tool are as follows: starting with the *Definition tab*, as shown in Figure 5.21a.

HyperHeuristic X	HyperHeuristic ×
how	Show
Start Stop Reset	Start Stop Reset
Experiment: Observation:	Experiment: Observation:
Definition Parameterization Evaluation	Definition Parameterization Evaluation
Decision Variables: Number of decision variables: 2 Define Input Variables	 Single solution-based search Population-based search
Objective Functions:	Population-based:
Define Output Variables	Population size: 100
Observations per solutions: 100 -	Maximum evaluations: 1000
OK Cancel Apply	OK Cancel Apply
(a) Definition tab	(b) Parameterisation tab

Figure 5.21: The Hyperheuristic tool user-interface for single-solution based search.

- 1. Define the number of decision variables (n), from the drop-down list, then click the *Define Input Variables* button and define each decision variable in terms of its path, lower and upper bounds and initial solutions (as previously shown in Figure 5.15).
- 2. Click the *Define Output Variables* button and define each output variable in terms of its path and select the optimisation direction from the drop-down list, *i.e. Min* or *Max* (as previously shown in Figure 5.5b).

The next step is to navigate to the *Parameterisation tab*, as shown in Figure 5.19b:

3. Select Single-solution based search and enter the Maximum evaluations permitted.

Recall that a fixed number of 1 000 evaluations are permitted in this study. The above-mentioned parameters are used as input to Algorithm 5.17 as well as the three best move operators for DBMOSA as found in Chapter 6. To execute the Hyperheuristic tool for single-solution based

search and optimise the simulation problem at hand click the *Start* button. A similar process is followed to execute and evaluate the Hyperheuristic tool as for the DBMOSA and its move operators (LLHs), where the initial solution is also generated randomly, represented by \boldsymbol{x} .

Start with the initialisation of the parameters for the DBMOSA algorithm (as described in §5.4):

1. Initialise the counters, accepts (*accepts* \leftarrow 1), attempts (*attempts* \leftarrow 0), poor epochs (*poorEpochs* \leftarrow 0) and epochs (*epochs* \leftarrow 0).

Next, initialise the parameters specific to the BOSAH algorithm. First initialise the iteration counter, $t \leftarrow 1$, and the number of LLHs (or heuristic components) considered, $nLLH \leftarrow 3$. Next, initialise the move sequence counter (denoted by i) and the move operator (denoted by m) assigned at the start of the search. Note that move operator $m \in \{1, 2, 3\}$, where m = i implies move operator i.

- 2. Initialise the move sequence counter, $i \leftarrow 1$.
- 3. Initialise the move operator assigned, $m \leftarrow 1$.

The search starts with move operator 1, denoted as m = 1. Initially, each LLH (*i.e.* move operator 1–3 or $m \in \{1, 2, 3\}$) is allocated the same number of solutions that it is permitted to generate. The allocations are denoted by $N_{m,i}$ and are initialised to 20, as illustrated in Table 5.3, *i.e.* $N_{m,1} = 20$ for $m \in \{1, 2, 3\}$. Note that the initial allocations were chosen arbitrarily.

 Table 5.3: An example illustrating the initial allocations for the BOSAH.

i	m	$N_{m,i}$	$T_{m,i}$
	1	20	20
1	2	20	40
	3	20	60

4. Initialise the allocations $N_{m,i}$ for $m \in \{1, 2, 3\}$, for move sequence i = 1.

The allocations are updated after each move sequence *i*, where a move sequence consists of move operator m = 1 generating $N_{1,i}$ solutions, then m = 2 generating $N_{2,i}$ solutions and lastly m = 3 generating $N_{3,i}$ solutions. This process continues until 1000 solutions have been generated and evaluated. Now, the general procedure starts. Recall, that an archive \mathcal{A} is maintained throughout the search and contains all the non-dominated solutions generated by the individual move operators (or LLHs). The random initial solution \boldsymbol{x} is evaluated and inserted into the archive, $\mathcal{A} \leftarrow \{\boldsymbol{x}\}$.

- 5. Evaluate the initial solution x, *i.e.* the simulation model evaluates x.
- 6. Initialise the archive to contain the initial solution, $\mathcal{A} \leftarrow \{x\}$.

As part of the heuristic selection strategy, it is necessary to count the number of non-dominated solutions found per move operator assignment, *i.e.* the number of non-dominated solutions found by move operator m for move sequence i and stored in variable NDS. Initially $NDS \leftarrow 1$ since one solutions has been evaluated. Note that NDS is initialised to zero every time a new move operator m is assigned.

7. Initialise the non-dominated solutions counter, $NDS \leftarrow 1$.

As part of the proposed heuristic selection procedure, a rate (denoted by $rate_t$) of non-dominated solutions is updated at each iteration t, *i.e.* $rate_t$ is the number of non-dominated solutions found divided by t (the number of solutions evaluated so far). Initially, $rate_t \leftarrow 1$ as the initial solution is considered non-dominated.

8. Initialise the rate of non-dominated solutions found, $rate_1 \leftarrow 1$.

Now, the loop initiates, while $t \leq 1000$ do the following. Increment the iteration counter, $t \leftarrow t + 1$. The next step, as described in §5.4, is to generate the neighbouring solution \mathbf{x}' using a move operator. At each iteration t, the algorithm checks if move operator m should be employed to generate neighbouring solution \mathbf{x}' .

To facilitate the heuristic selection procedure (of the move operators $m \in \{1, 2, 3\}$) three steps are required. The first step ensures that each move operator (m) generates its allocated number of solutions $(N_{m,i})$. The second step is proposed in this study to ensure that the search does not stagnate, making it a novel contribution. The third step determines the new allocations for the next move sequence i and is similar to the strategy implemented in BOCEGAH and proposed for AMALGAM [254]. The heuristic selection procedure is described in steps 9–20.

The *first* step

Suppose t = 15 (*i.e.* 15 solutions have been evaluated) then from Table 5.3 it can be seen that it is currently the first move sequence (*i.e.* i = 1) and move operator m = 1 is assigned. Also, from Table 5.3, it can be seen that move operator m = 1 is allocated 20 solutions (*i.e.* $N_{1,1} = 20$). Therefore, once the current iteration t reaches the cumulative total number of solutions that move operator m for move sequence i may generate (*i.e.* $T_{m,i}$), then the next move operator in the move sequence is assigned (*i.e.* $m \leftarrow m + 1$).

If $t = N_{m,i} + 1$, then execute steps 9–11:

- 9. Assign the number of non-dominated solutions (NDS) contributed by move operator m for move sequence i, denoted by $C_{m,i} \leftarrow NDS$.
- 10. Re-initialise the NDS counter to zero, $NDS \leftarrow 0$.
- 11. Increment the move operator assigned, $m \leftarrow m + 1$.

The *second* step

The motivation behind the second heuristic selection step is to negate the effect of the NFL theorem by playing to the strengths of the individual move operators, *i.e.* if move operator m = 1 is not generating (enough) solutions that are evaluated as non-dominated then the rest of its allocation should be transferred to the next move operator in the move sequence (*i.e.* m = 2) and so forth. Ultimately, compensating for their respective weaknesses and exploiting their individual strengths.

Specifically, when the current move operator m has generated half of its allocated solutions $(i.e. \ t = N_{m,i}/2 + T_{m-1,i})$, then test if the move operator's rate of generating non-dominated solutions $(i.e. \ \frac{NDS}{N_{m,i}/2})$ is smaller than the overall rate $(rate_t)$. If this is true, then move operator

m is performing below the search standard and should not continue to generate the rest of its allocated solutions. Rather, the rest of the solutions are transferred to the next move operator (i.e. m+1) in the same move sequence i. However, if m = nLLH, then the rest is not transferred over to the next move sequence.

Formally, the overall rate of non-dominated solutions found, $rate_t$ can be determined as

$$rate_t = \frac{\sum C_{m-1,i} + NDS}{t},\tag{5.25}$$

where $m \in \{1, 2, 3\}$, $\sum C_{m-1,i}$ is the sum of the individual contributions up until m-1 for move sequence i = 1 to the current move sequence i, $NDS \leftarrow 3$ (currently) and t is the current iteration. If m = 1 and i = 2, then the contribution $C_{m-1,i}$ is represented by $C_{3,1}$, *i.e.* the previous move sequence.

Suppose that the number of non-dominated solutions found by move operator m = 2 is 3 (denoted by NDS = 3) out of the $N_{2,2}/2 = 10$ solutions generated (for move sequence i = 2), then the rate at t = 95 can be calculated using (5.25) as depicted in Figure 5.22. Consequently, $\frac{NDS}{N_{2,2}/2} = 0.3$ and 0.3 < 0.693.

$$rate_{95} = \frac{\left(\sum C_{m-1,i}\right) + NDS}{t} = \frac{\left(15 + 12 + 9 + 22\right) + 3}{95} = 0.693 \qquad \left(5.25\right) \leftarrow \cdots - \left(\frac{2}{2} \quad 20 \quad 17 \quad 105\right) \\ = 0.693 \qquad \left(\frac{i \quad m \quad N_{m,i} \quad C_{m,i} \quad T_{m,i}}{1 \quad 20 \quad 15 \quad 20} \right) \\ \hline i \quad 20 \quad 15 \quad 20 \\ 1 \quad 2 \quad 20 \quad 12 \quad 40 \\ 3 \quad 20 \quad 9 \quad 60 \\ \sum \quad 36 \quad 60 \\ \sum \quad 36 \quad 60 \\ \sum \quad 36 \quad 60 \\ \sum \quad 22 \quad 85 \\ = \frac{\frac{C_{1,1}}{N_{1,1}}}{\sum_{m=1}^{3} \frac{C_{m,1}}{N_{m,1}}} T_{3,1} \\ (5.26) = \frac{\frac{15}{20}}{\frac{15}{20} + \frac{12}{20} + \frac{9}{20}} 60 \\ = \frac{15}{20} + \frac{12}{20} + \frac{9}{20} + \frac{9}{20} + \frac{12}{20} + \frac{12}{20} + \frac{9}{20} + \frac{12}{20} + \frac{9}{20} + \frac{12}{20} + \frac{12}{20} + \frac{9}{20} + \frac{12}{20} + \frac{12}{20} + \frac{12}{20} + \frac{12}{20} + \frac{9}{20} + \frac{12}{20} + \frac{12}{20} + \frac{12}{20} + \frac{9}{20} + \frac{12}{20} + \frac{12}{2$$

Figure 5.22: An example illustrating the heuristic selection procedure proposed for the BOSAH.

If i > 1 and $t = N_{m,i}/2 + T_{m-1,i}$ and $\frac{NDS}{N_{2,2}/2} < rate_t$ and $N_{m,i} > 5$, then execute steps 12–16:

- 12. Update the number of allocations assigned to move operator m, *i.e.* $N_{m,i} \leftarrow N_{m,i}/2$.
- 13. Assign the non-dominated solutions (NDS) contributed by move operator m for move sequence $i, i.e. C_{m,i} \leftarrow NDS$.
- 14. Re-initialise the NDS counter to zero, $NDS \leftarrow 0$.

Consequently, more operator m = 2 is assigned fewer solutions, *i.e.* $N_{2,2} \leftarrow 10$ instead of 20. Also, the contribution of more operator m = 2 becomes $C_{2,2} \leftarrow 3$. Next, the next more operator m = 3 in the more sequence is assigned and its allocation updated as $N_{3,2} \leftarrow 15 + 20/2$.

- 15. Increment the move operator, $m \leftarrow m + 1$.
- 16. Update the number of allocations assigned to move operator m for move sequence i, i.e. $N_{m,i} \leftarrow N_{m,i} + N_{m-1,i}/2.$

<u>The</u> third step

Recall that there is a sequence of move operators being assigned, called a move sequence. The sequence is as follows, move operator m = 1, then m = 2 and lastly m = 3, until termination is reached. Therefore, if the increment of move operator m results in m = nLLH + 1, then a move sequence is complete and move operator m = 1 is assigned again and the move sequence counter is incremented, $i \leftarrow i + 1$. Next, determine the new allocations $N_{m,i}$ for move sequence i and move operators $m \in \{1, 2, 3\}$, using (5.26).

$$N_{m,i} = \frac{\frac{C_{m,i-1}}{N_{m,i-1}}}{\sum_{m=1}^{3} \frac{C_{m,i-1}}{N_{m,i-1}}} T_{3,1}.$$
(5.26)

Figure 5.22 aids with elucidating the process followed to determine the new allocation for move sequence i = 2, move operator m = 1, *i.e.* $N_{1,2}$. In the example, the contributions are $C_{m,1} = \{15, 12, 9\}$, *i.e.* move operator m = 1 contributed $C_{1,1} = 15$ (non-dominated) solutions from $N_{1,1} = 20$. From this $N_{1,2} = 25$ is calculated using (5.26) as shown in Figure 5.22.

Following a similar process for $N_{2,2}$ and $N_{3,2}$ results in allocations $N_{2,2} = 20$ and $N_{3,2} = 15$. Note that the total number of allocations should be equal to the sum of the individual allocations for the first move sequence, namely 60 solutions, which is arbitrarily chosen.

If m = nLLH + 1, then execute steps 17–19:

- 17. Restart the move operator sequence, $m \leftarrow 1$.
- 18. Increment the move sequence, $i \leftarrow i + 1$.
- 19. Update the allocations $N_{m,i}$ for move sequence i and $m \in \{1, 2, 3\}$, using (5.26).

A minimum allocation threshold n_a is set at $n_a = 5$ and is required so that none of the move operators recieve an allocation of 0, and thereby be eliminated from the search. Consequently, if step 19 results in an allocation of less than five solutions, then it is simply set to 5 and the other allocations are updated accordingly.

20. If $N_{m,i} < n_a$, then $N_{m,i} \leftarrow n_a$ for $m \in \{1, 2, 3\}$.

Finally, the neighbouring solution x' can be generated. Recall that the neighbouring solution is generated from the previously accepted solution (as discussed in §5.4).

- 21. Generate the neighbouring solution $\mathbf{x'}$ according to move operator m, denoted by move operator_m(\mathbf{x}).
- 22. Check if a duplicate solution are present and then replace it with a non-duplicate solution.

After the neighbouring solution x' is generated using move operator m, it is necessary to check if a duplicate solution is present. If so, then replace it with a non-duplicate solution by generating a new neighbouring solution, (again) using move operator m. This step ensures that the expensive function evaluator does not unnecessarily evaluate a solution. Now, evaluate the non-duplicate neighbouring solution x'.

Α	Algorithm 5.17: The BOSAH method for single-solution based search						
	Input : Previously described using Figures 5.19a and 5.21b.						
	Output : The approximate Pareto set (\mathcal{P}_S) for the given simulation	ı problem.					
	/* DBMOSA initialisation	*/					
1	Refer to step 1;						
	/* Hyperheuristic initialisation	*/					
2	$t \leftarrow 1;$ // Ini	tialise iteration counter					
3	$nLLH \leftarrow 3;$ // Init	ialise the number of LLHs					
4	Initialise the move sequence counter, $i \leftarrow 1$;	// Refer to step 2					
5	Initialise move operator assigned, $m \leftarrow 1$;	// Refer to step 3					
6	Initialise allocations;	// Refer to step 4					
	/* General procedure starts	*/					
7	Evaluate the initial solution x ;	// Refer to step 5					
8	Initialise the archive, $\mathcal{A} \leftarrow \{x\}$;	// Refer to step 6					
9	$NDS \leftarrow 1;$	// Refer to step 7					
10	Initialise the $rate_1 \leftarrow 1$;	// Refer to step 8					
11	while $t \le 1000$ do						
12	$t \leftarrow t+1;$ // Increm	ent the iteration counter					
13	If $t = T_{m,i} + 1$ then						
14	Assign the contribution of $m, C_{m,i} \leftarrow NDS;$	// Refer to step 9					
15	$NDS \leftarrow 0;$	// Refer to step 10					
16		// Refer to step 11					
17	if $i > 1$ and $t = N_{m,i}/2 + T_{m-1,i}$ and $\frac{NDS}{N_{m,i}/2} < rate_t$ and $N_{m,i}$	$_i > 5$ then					
18	Update allocations;	<pre>// Refer to step 12</pre>					
19	Assign the contribution of $m, C_{m,i} \leftarrow NDS;$	<pre>// Refer to step 13</pre>					
20	$NDS \leftarrow 0;$	// Refer to step 14					
21	$m \leftarrow m+1;$	// Refer to step 15					
22	Update allocations;	// Refer to step 16					
23	$\mathbf{if} m = nLLH + 1$ then						
24	$m \leftarrow 1$:	// Refer to step 17					
25	$i \leftarrow i + 1;$	// Refer to step 18					
26	Update allocations, $N_{m,i}$ for $m \in \{1, 2, 3\}$;	// Refer to step 19					
27	if $N_{m,i} < 5$ then	· · · · ·					
28	$N_{m,i} \leftarrow 5$ and update allocations accordingly;	// Refer to step 20					
29	Generate neighbouring solution x' according to move operator n	n: // Refer to step 21					
30	Evaluate neighbouring solution x' :	// Refer to step 23					
31	Follow steps 8–36 as described in Algorithm 5.13:	// Refer to step 24					
32	Update $rate_t$, using (5.25);	// Refer to step 25					
33	$\mathcal{P}_s \leftarrow \mathcal{A};$	// Approximate Pareto set					

23. Evaluate the neighbouring solution (x'), *i.e.* the simulation model evaluates x.

Next, execute the move acceptance mechanism of the DBMOSA algorithm, by following steps 8–36 in Algorithm 5.13, as discussed in §5.4. Naturally, the move acceptance strategy was chosen as simulated annealing.

- 24. Apply the move acceptance mechanism, by following steps 8–36 in Algorithm 5.13, refer to §5.4.
- 25. Update $rate_t$, using (5.25).

This concludes the loop. At termination, the archive \mathcal{A} is returned as an approximate of the true

Pareto set of solutions to the bi-objective simulation optimisation problem under consideration, *i.e.* $\mathcal{P}_S \leftarrow \mathcal{A}$. A pseudo-code description of the entire process described above is provided in Algorithm 5.17, with specific reference to the above-mentioned steps and concludes the documentation of both hyperheuristics proposed in this study.

5.7 Summary

This chapter discussed some of the main concepts of S- and P-metaheuristics, with specific focus on their implementations for each respective metaheuristic considered in this study. The MOOCEM, the NSGA-II and the DBMOSA algorithms were discussed in the context of simulation optimisation and their integration in Tecnomatix. Next, the two hyperheuristics were presented, first for population-based search employing MOOCEM and NSGA-II as LLHs and then for single-solution based search employing three move operators.

The next chapter determines which hyperparameter combination, for each metaheuristic respectively, should be employed for the respective hyperheuristics.

CHAPTER 6

Algorithmic Parameter Evaluation

6.1	Introd	luction to multi-objective statistical analysis					
6.2	Determining the MOOCEM algorithm hyperparameters						
	6.2.1	Open mine problem					
	6.2.2	(s,S) Inventory problem					
	6.2.3	Buffer allocation problem: five machines					
	6.2.4	Buffer allocation problem: 10 machines					
	6.2.5	Non-linear buffer allocation problem: 16 machines					
6.3	Deteri	mining the NSGA-II hyperparameters					
	6.3.1	<i>Open mine problem</i>					
	6.3.2	(s,S) Inventory problem					
	6.3.3	Buffer allocation problem: five machines					
	6.3.4	Buffer allocation problem: 10 machines					
	6.3.5	Non-linear buffer allocation problem: 16 machines					
6.4	Deteri	mining the DBMOSA algorithm hyperparameters					
6.5	Conclusion: Chapter 6						

The previous chapter discussed the hyperheuristics BOCEGAH and BOSAH and their respective LLHs with specific reference to the simulation optimisation process as it occurs in Tecnomatix.

This chapter documents the empirical (and statistically sound) hyperparameter study conducted which is deemed necessary due to the high number of hyperparameter combinations (or configurations) that algorithms have and a lack of insight on how to choose them, attributable to their complex interactions. Moreover, when considering more than one problem it becomes markedly challenging to suggest a common hyperparameter combination suitable for the subset of problems considered. Consequently, deciding which metaheuristic (or heuristic operators) to use is not a trivial task.

The first section introduces the non-parametric hypothesis test and corresponding *post hoc* test followed to determine the best hyperparameter combinations. Next, the hypothesis tests and *post hoc* tests (where necessary) are documented for the MOOCEM, then the NSGA-II and lastly for the DBMOSA, for all five discrete-event dynamic stochastic simulation optimisation problems, as discussed in Chapter 4. After the individual tests have been conducted (for each algorithm), the results are consolidated in order to make an informed decision regarding the hyperparameter combination when considering all five simulation problems. The chapter closes with a conclusion of its contents.

6.1 Introduction to multi-objective statistical analysis

As mentioned in §2 the term *quality performance* is used, instead of the term *performance* which pertains to both time and quality. In this study *only* quality is assessed based on an algorithm's performance in the objective (or solution) space for a fixed number of evaluations (chosen as

1000). As mentioned (in Chapter 5) the solutions obtained by the MOOCEM, the NSGA-II and the DBMOSA are referred to as approximation sets which aim to approximate the true Pareto set. In this chapter, the approximation sets are compared with each other based on the obtained hyperareas and number of non-dominated solutions in the approximation sets.

Recall that each algorithm contains stochastic elements and operate on stochastic problems (in this study), therefore, the resulting approximation sets vary from experimental run to experimental run. In order to derive significant statistical results, many experimental runs must be carried out, at least 10 or more than 100 if possible [93]. However, due to the limited timeframe of this study, 100 runs are carried out (unless stated otherwise). Consequently, any statements about the quality of an experimental run is probabilistic in nature, *i.e.* the estimated parameter values may be numerically different, however, statistical tests are required to determine whether they are statistically different. For the sake of brevity, experimental runs are hereafter simply referred to as runs.

There are two main types of statistical tests in the literature: *parametric* and *nonparametric* tests. The decision between the two tests depends on the assumptions made about the properties of the underlying population (or distribution) [121]. To elaborate, parametric tests assume that the data fulfills three conditions, namely *normality*, *independency* and *homoscedasticity* [229]. However, these conditions cannot be assumed for output data generated by stochastic optimisation algorithms and therefore nonparametric tests are considered [48].

Recall (from Chapter 4) that the possible outcomes of a hypothesis test is that either the data supports the research question stated as the *null hypothesis* H_0 (which is assumed to be true) or disproves the research prediction stated as the *alternate hypothesis* H_1 [71]. The Friedman test is regarded an omnibus test, *i.e.* more than three groups of data is to be analysed and is considered the nonparametric counterpart of the parametric two-way (repeated measures) analysis of variance (ANOVA), performed on ordinal (or ranked) data [89].

The test statistic suggested by Friedman is given as [231]

$$\chi_F^2 = \frac{12}{nk(k+1)} \sum_{k=1}^k R_k^2 - 3n(k+1), \tag{6.1}$$

where $R_k = \sum_{i=1}^n R_{ik}$ is the sum of the ranks for sample k over the n observations and χ_F^2 is approximately χ^2 -distributed. At a significance level of $\alpha = 0.05$, the null hypothesis is rejected if $\chi_F^2 \ge \chi_{k-1,1-\alpha}^2$, where $\chi_{k-1,1-\alpha}^2$ is the $(1 - \alpha)$ quantile of the Chi-square distribution with k - 1 degrees of freedom [206].

Iman and Davenport [129] noted that the χ^2 approximation was occasionally poor or too conservative and is therefore also included. The test statistic suggested by Iman and Davenport is referred to as *Iman-Davenport* is given by [129]

$$F_{ID} = \frac{(n-1)\chi_F^2}{n(k-1) - \chi_F^2}.$$
(6.2)

Similarly, at $\alpha = 0.05$, the null hypothesis is rejected if $F_{ID} \geq F_{k-1,(k-1)(n-1),1-\alpha}$, where $F_{k-1,(k-1)(n-1),1-\alpha}$ is the $(1-\alpha)$ quantile of the *F*-distribution with (k-1) and (k-1)(n-1) degrees of freedom.

Both statistical tests yield a *p*-value, and if $p < \alpha$, then H_0 is rejected. Indicating that at least one of the summed ranks are statistically significantly different, however, it does not indicate which groups. Therefore, a *post hoc* test is required to determine where the differences are.



Figure 6.1: Non-parametric tests and post hoc procedures for $N \times N$ comparisons.

The Nemenyi, formally called the Wilcoxon-Nemenyi-McDonald-Thompson *post hoc* test is categorised as more powerful than that of Holm and Hommel *post hoc* tests [121]. The procedure of determining the individual differences involves performing two-tailed pairwise significance tests between all pairs of samples, using the Friedman ranks. The test is performed using rank sums, and H_0 is rejected if

$$|R_{.k} - R_{.j}| \ge q_{k, \ n-k, \ 1-\alpha} \sqrt{\frac{nk(k+1)}{12}},\tag{6.3}$$

where $q_{k, n-k, 1-\alpha}$ is the $(1-\alpha)$ of the studentised range distribution with k and (n-k) degrees of freedom and requires equal sample sizes, *i.e.* $n_1 = n_2 = \ldots = n_k$ for each group k.

The Nemenyi *post hoc* test controls the familywise Type I error rate, correcting for the multiple inferences made by dividing the calculated *p*-values by the number of combinations determined as $\frac{k(k-1)}{2}$, thereby ensuring that the experiment-wide significance level $\alpha = 0.05$ is not exceeded.

The hypothesis testing procedure followed in this study is listed below:

- 1. State the null H_0 and alternate hypotheses H_1 . Formally stated as,
 - H_0 : There is **no** significant difference between the ranked sums of the groups being compared.
 - H_1 : There is a significant difference between the ranked sums of the groups being compared and a post-hoc test is required to determine which groups.
- 2. State the level of significance, alpha: $\alpha = 0.05$.
- 3. Calculate the degrees of freedom:

Friedman df = k - 1Iman-Davenport $df = \{k - 1, (k - 1)(n - 1)\}$

where k is the number of groups being compared and n is the number of observations. For the purpose of this paper n = 100 (unless stated otherwise), and k varies from algorithm to algorithm.

- 4. Get the corresponding critical values for both distributions, *i.e.* the χ^2 -distribution and *F*-distribution, as χ^2_{k-1} and $F_{k-1,(k-1)(n-1)}$, respectively.
- 5. Calculate the test statistics for Friedman using (6.1) and Iman-Davenport using (6.2).

- 6. Determine the *p*-values.
- 7. Reject H_0 or fail to reject H_0 .

Friedman The null hypothesis is rejected if $\chi_F^2 \ge \chi_{k-1,\alpha}^2$ or if $p < \alpha$. **Iman-Davenport** The null hypothesis is rejected if $F_{ID} \ge F_{k-1,(k-1)(n-1),\alpha}$ or if $p < \alpha$.

6.2 Determining the MOOCEM algorithm hyperparameters

The MOOCEM contain many interacting hyperparameters to be adjusted and consequently, the purpose of this section is to determine the single best hyperparameter combination (for the MOOCEM) that results in the best algorithmic performance, across all five simulation problems. Algorithmic performance is compared based on the hyperareas obtained by the approximation fronts, as well as the number of non-dominated solutions in the fronts. The Friedman test and Iman-Davenport extension is used to compare these performances with regards to both measures. If the Friedman test or Iman-Davenport extension indicates significance, then the Nemenyi post-hoc test is conducted. Ultimately, the results are combined and used to decide which hyperparameter combination to implement in the BOCEGAH.

The population size N is typically chosen in the range [30, 100], in this study N = 100. Bekker [21] found that $\epsilon_c \in [0.1, 1]$ was sufficiently small, accordingly $\epsilon_c = 0.1$. The smoothing parameter α is typically in the range [0.6, 0.9] and the probability of inverting the decision variable histograms p_h is typically in the range [0.1, 0.3] [21].

It is important to note that a different random number stream (RNS) is used for each run, and the same RNS is used for the respective hyperparameter combinations, *i.e.* RNS = 1 is used for run 1, RNS = 2 for run 2 and so on, until RNS = 100 for run 100. The same setting applies for all the hyperparameter combinations as well as all the problems. The RNS is responsible for generating the initial population and therefore needs to be consistent for the same runs across the hyperparameter combinations. This applies for both the MOOCEM and the NSGA-II.

The MOOCEM parameters that are considered for the hyperparameter study are summarised in Table 6.1, where the naming convention is adopted to help navigate this section, *i.e.* A1.1.1 refers to the hyperparameter combination $\alpha = 0.65$ and $p_h = 0.2$ for the OMP specifically. Nine hyperparameter combinations are explored per simulation problem, *i.e.* the number of groups being compared is k = 9. Moreover, each hyperparameter combination (or group) is run for 100 individual (simulation) runs (unless stated otherwise), *i.e.* n = 100 observations per group resulting in 100 individual hyperareas and number of non-dominated solutions. Also, each run consists of 1 000 solution evaluations (as mentioned previously) to facilitate fair comparison.

The hypothesis tests conducted are first discussed for the OMP, then the IP, BAP5, BAP10 and finally for BAP16. Thereafter, the results of the respective tests are combined and a single best hyperparameter combination is chosen.

6.2.1 Open mine problem

This study includes a wide range of problem sizes, from the smallest problem the OMP with a decision space of 3 375 solutions to the largest problem, BAP16 with a decision space of 1×10^{15} solutions. The OMP, which has a relatively small decision space, is included as a method to validate the algorithms, because if an algorithm is able to approximately solve the OMP, then it can be assumed that the algorithm will be able to solve the larger problems too.

Smoothing parameter	Probability of inversion	OMP	IP	BAP5	BAP10	BAP16
	$p_{h} = 0.2$	A1.1.1	A1.2.1	A1.3.1	A1.4.1	A1.5.1
$\alpha=0.65$	$p_{h} = 0.3$	A1.1.2	A1.2.2	A1.3.2	A1.4.2	A1.5.2
	$p_h = 0.4$	A1.1.3	A1.2.3	A1.3.3	A1.4.3	A1.5.3
	$p_h = 0.2$	A1.1.4	A1.2.4	A1.3.4	A1.4.4	A1.5.4
$\alpha = 0.70$	$p_{h} = 0.3$	A1.1.5	A1.2.5	A1.3.5	A1.4.5	A1.5.5
	$p_h = 0.4$	A1.1.6	A1.2.6	A1.3.6	A1.4.6	A1.5.6
	$p_{h} = 0.2$	A1.1.7	A1.2.7	A1.3.7	A1.4.7	A1.5.7
$\alpha = 0.75$	$p_{h} = 0.3$	A1.1.8	A1.2.8	A1.3.8	A1.4.8	A1.5.8
	$p_{h} = 0.4$	A1.1.9	A1.2.9	A1.3.9	A1.4.9	A1.5.9

Table 6.1: The hyperparameter search space and corresponding naming convention adopted refer to the specific hyperparameter combinations for the specific simulation problem, resulting in the entire hyperparameter search space considered for the MOOCEM.

Note that for the OMP, only 40 simulation runs per hyperparameter combination were executed, *i.e.* n = 40 observations. The reason for not executing 100 simulation runs as with the other problems is due to the time constraint of the project and the purpose of its inclusion in this study. The critical values for k = 9 and n = 40 for the Chi-square and *F*-distributions are $\chi_8^2 = 15.51$ and $F_{8,312} = 2.68$, respectively.

When calculating the hyperareas for the various approximation fronts, larger hyperarea values (than that of the true hyperarea) were obtained, this is demonstrated in Figure 6.2. Upon investigation of the approximation fronts, the discrepancy seems reasonable, and is attributable to the discontinuity of the approximation fronts and the method used to calculate the hyperarea of the front. From literature, it is clear that the better approximation front should have the largest hyperarea and consequently the hyperarea of the true approximation front should be the largest. For this reason, a linear adjustment is proposed to account for the discontinuity present in the OMPs approximation fronts, given as

$$I_{Hadj} = I_H \left(\frac{NDS}{NDS_{true}}\right), \text{ for } NDS \le NDS_{true}, \tag{6.4}$$

where I_H is the hyperarea calculated as described in §2.2.4, NDS is the number of nondominated solutions in the approximation front and NDS_{true} is the number of non-dominated solutions in the true Pareto front, refer to Chapter 4.

Note that the approximation front with the largest hyperarea does not necessarily correspond to the approximation front with the most non-dominated solutions found, *i.e.* it is possible for an approximation front to have more non-dominated solutions than that of the true nondominated front, because the non-dominated solutions obtained by the approximation fronts are not necessarily the true non-dominated solutions of the true Pareto front.

For example, consider the two Pareto fronts in Figure 6.2, the true Pareto front contains 36 nondominated solutions with a true hyperarea value of 14 485.04 presented by the blue dot area, and the other (larger) approximation front has 34 non-dominated solutions with a hyperarea value of 14 617.45 with the adittional red area. Due to the discontinuity, the approximation front is larger denoted by the red area. To account for this, the adjusted hyperarea is calculated by applying (6.4), such that

$$I_{Hadj} = 14617.45 \left(\frac{34}{36}\right)$$

= 13805.37.

The correction holds true for the case where the true Pareto set is found, since the obtained hyperarea would simply be multiplied by 1, as both sets contain 36 non-dominated solutions.



Figure 6.2: An example illustrating that a larger hyperarea value than the true hyperarea value can exist (for a worst approximation front), when the front is discontinuous.

Tables B.1 and B.2 present the adjusted hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 39–40 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. For example, the values in the columns of Table B.1 represent a snapshot of the adjusted hyperareas and their ranks obtained for $\alpha = 0.65$ and $p_h = 0.2$ (or A1.1.1) for *Run 1–10 and 39–40*. The boxplots in Figure 6.3 provide a graphical summary of the results for hyperparameter combinations A1.1.1–A1.1.9, as presented in Tables B.1 and B.2.



Figure 6.3: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A1.1.1–A1.1.9.

Table 6.2 shows the number of runs (out of 40) that obtained the true Pareto set, for each hyperparameter combination. For example, for $\alpha = 0.65$ and $p_h = 0.2$, the MOOCEM obtained the true Pareto set 24 out of the 40 times (*i.e.* 60% of the time). Consequently, validating that

the MOOCEM algorithm works correctly and will be able to solve the larger problems that follow this discussion. The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.3 and depicted in Figures B.1 and B.2 for A1.1.1–A1.1.9.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.3. The Friedman and Iman-Davenport test statistics for both the adjusted hyperareas (and the number of non-dominated solutions) indicate significance $(\chi_F^2 = \{22.11, 22.15\} > \chi_8^2, p < \alpha$ and $F_{ID} = \{2.89, 2.9\} > F_{8,312}, p < \alpha$). The outcome is to reject H_0 , which means that there exists a statistically significant difference between at least one of the groups. Consequently, the Nemenyi *post hoc* test is conducted to determine between which groups a statistically significant difference exists.

Table 6.2: The number of runs that the MOOCEM obtained the true Pareto set, for the corresponding hyperparameter combinations A1.1.1–A1.1.9.

A1.1.1	A1.1.2	A1.1.3	A1.1.4	A1.1.5	A1.1.6	A1.1.7	A1.1.8	A1.1.9
24	23	14	27	27	20	26	21	20

Table 6.3: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A1.1.1–A1.1.9.

	Friedman test	<i>p</i> -value	Iman-Davenport test	<i>p</i> -value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	22.11	0.01	2.89	0.004	Reject H_0
Non-dominated solutions	22.15	0.01	2.9	0.004	neject II0

The Nemenyi post hoc test is presented in Tables B.4 and B.5 and represent the upper triangle of the matrix that contains the adjusted *p*-values of the multiple pairwise comparisons for both the adjusted hyperareas and number of non-dominated solutions, respectively. The Friedman and Iman-Davenport extension both detected a significant difference (for $\alpha = 0.05$) between the groups, however upon further analysis, the Nemenyi post hoc test found that there was no statistically significant difference between the groups. Consequently, the hyperparameter combination employed do not influence the performance of the MOOCEM (on the OMP) and consequently the hyperparameter combinations A1.1.1–A1.1.9 are equally desirable.

6.2.2 (s, S) Inventory problem

Note that for the IP, BAP5, BAP10 and BAP16, 100 simulation runs per hyperparameter combination were executed, *i.e.* n = 100 observations per group. The critical values for k = 9 and n = 100 for the Chi-square and *F*-distributions are $\chi_8^2 = 15.51$ and $F_{8,792} = 1.95$, respectively. Note that the hyperareas are not adjusted for the IP, BAP5, BAP10 or BAP16, since the approximation fronts are not discontinuous. Tables B.6 and B.7 present the hyperareas and number of non-dominated solutions and their ranks for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests.

For example, the values in the columns of Table B.6 represent a snapshot of the hyperareas and their ranks obtained for $\alpha = 0.65$ and $p_h = 0.2$ (or A1.2.1) for *Run 1–10 and 99–100*. The boxplots in Figure 6.4 provide a graphical summary of the results for hyperparameter combinations A1.2.1–A1.2.9, as presented in Tables B.6 and B.7.



Figure 6.4: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A1.2.1–A1.2.9.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.8 and depicted in Figures B.3 and B.4 for A1.2.1–A1.2.9.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.4. The Friedman and Iman-Davenport test statistics do not indicate significance for the hyperareas ($\chi_F^2 = 6.76 < \chi_8^2$, $F_{ID} = 0.846 < F_{8,792}$, $p > \alpha$). Therefore, do not reject H_0 . The tests, however, do indicate significance for the number of non-dominated solutions found ($\chi_F^2 = 63.86 > \chi_8^2$, $F_{ID} = 8.59 > F_{8,792}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

Table 6.4: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A1.2.1–A1.2.9.

	Friedman test	p-value	Iman-Davenport test	<i>p</i> -value	Outcome	
	χ^2_F	p	F_{ID}	p		
Hyperarea	6.76	0.56	0.84	0.56	Fail to reject H_0	
Non-dominated solutions	63.86	0	8.59	0	Reject H_0	

The Nemenyi *post hoc* test results are presented in Table 6.5 and contains the adjusted *p*-values of the multiple pairwise comparisons for number of non-dominated solutions. The red *p*-values indicate statistical significance, for example the Nemenyi *post hoc* test found that there are statistically significant differences between A1.2.1 and A1.2.3, A1.2.5–A1.2.6 and A1.2.8–A1.2.9.

Table 6.5: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A1.2.1-A1.2.9.

	A1.2.2	A1.2.3	A1.2.4	A1.2.5	A1.2.6	A1.2.7	A1.2.8	A1.2.9
A1.2.1	1	0.02	0.72	0	0	1	0.04	0
A1.2.2		0.92	1	0.08	0	1	1	0.01
A1.2.3			1	1	0.17	1	1	1
A1.2.4				1	0	1	1	0.22
A1.2.5					1	0.19	1	1
A1.2.6						0	0.09	1
A1.2.7							1	0.01
A1.2.8								1

6.2.3 Buffer allocation problem: five machines

Tables B.9 and B.10 present the hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.5 provide a graphical summary of the results for hyperparameter combinations A1.3.1–A1.3.9, as presented in Tables B.9 and B.10.



Figure 6.5: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A1.3.1–A1.3.9.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.11 and depicted in Figures B.5 and B.6 for A1.3.1–A1.3.9.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.6. The Friedman and Iman-Davenport test statistics do not indicate significance for the hyperareas ($\chi_F^2 = 13.8 < \chi_8^2$, $F_{ID} = 1.74 < F_{8,792}$, $p > \alpha$) and therefore, fail to reject H_0 . The tests, however, do indicate significance for the number of non-dominated solutions found ($\chi_F^2 = 33.78 > \chi_8^2$, $F_{ID} = 4.37 > F_{8,792}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

Table 6.6: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A1.3.1–A1.3.9.

	Friedman test	<i>p</i> -value	Iman-Davenport test	<i>p</i> -value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	13.8	0.09	1.74	0.09	Fail to reject H_0
Non-dominated solutions	33.78	0	4.37	0	Reject H_0

The Nemenyi *post hoc* test is presented in Table 6.7 and contains the adjusted *p*-values of the multiple pairwise comparisons for number of non-dominated solutions. The red *p*-values indicate statistical significance, for example the Nemenyi *post hoc* test found that there was only statistically significant differences between A1.3.7 and A1.3.2–A1.3.3 and A1.3.6.

6.2.4 Buffer allocation problem: 10 machines

Tables B.12 and B.13 present the hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 99–100 as well as the sum of the ranks which is used in the

Table 6.7: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A1.3.1-A1.3.9.

	A1.3.2	A1.3.3	A1.3.4	A1.3.5	A1.3.6	A1.3.7	A1.3.8	A1.3.9
A1.3.1	1	1	1	1	1	0.27	1	1
A1.3.2		1	0.21	1	1	0	0.1	0.98
A1.3.3			0.17	1	1	0	0.08	0.83
A1.3.4				1	0.78	1	1	1
A1.3.5					1	0.15	1	1
A1.3.6						0.01	0.41	1
A1.3.7							1	1
A1.3.8								1

Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.6 provide a graphical summary of the results for hyperparameter combinations A1.4.1–A1.4.9, as presented in Tables B.12 and B.13.



Figure 6.6: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A1.4.1–A1.4.9.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are given in Table B.14 and presented in Figures B.7 and B.8 for A1.4.1–A1.4.9.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.8. The Friedman and Iman-Davenport test statistics for both the adjusted hyperareas (and the number of non-dominated solutions) indicate strong significance ($\chi_F^2 = \{529.1, 368.75\} > \chi_8^2$, $p < \alpha$ and $F_{ID} = \{193.36, 84.65\} > F_{8,792}$, $p < \alpha$). Therefore, the outcome is to reject H_0 , which means that there exists a significance difference between at least one of the groups in terms of both hyperarea and number of non-dominated solutions. Consequently, the Nemenyi post hoc test is conducted to determine between which groups a statistically significant difference exists.

The Nemenyi *post hoc* test is presented in Table 6.9 and contains the adjusted *p*-values of the multiple pairwise comparisons for both hyperareas and number of non-dominated solutions, since the test resulted in the same *p*-values. For example, the Nemenyi *post hoc* test found that there was a statistically significant difference between A1.4.1 and A1.4.4–A1.4.6 (for both the hyperareas and number of non-dominated solutions).

Table 6.8:	The Friedm	an and	Iman-Dave	enport test	results	based	on the	ranked	hyperareas	and	non-
dominated s	olutions obt	ained for	· each hype	erparamete	er combi	nation	A1.4.1	-A1.4.9.			

	Friedman test	p-value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	529.1	0	193.36	0	Reject Ho
Non-dominated solutions	368.75	0	84.65	0	neject m

Table 6.9: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the hyperareas and number of non-dominated solution found of the approximation fronts for the hyperparameter combinations A1.4.1-A1.4.9.

	A1.4.2	A1.4.3	A1.4.4	A1.4.5	A1.4.6	A1.4.7	A1.4.8	A1.4.9
A1.4.1	1	1	0	0	0	1	1	1
A1.4.2		1	0	0	0	1	1	1
A1.4.3			0	0	0	1	1	1
A1.4.4				1	1	0	0	0
A1.4.5					1	0	0	0
A1.4.6						0	0	0
A1.4.7							1	1
A1.4.8								1

6.2.5 Non-linear buffer allocation problem: 16 machines

Tables B.15 and B.16 present the hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.7 provide a graphical summary of the results for hyperparameter combinations A1.5.1–A1.5.9, as presented in Tables B.15 and B.16. The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.17 and depicted in Figures B.9 and B.10 for A1.5.1–A1.5.9.



Figure 6.7: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A1.5.1–A1.5.9.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.10. The Friedman and Iman-Davenport test statistics indicate significance for the hyperareas $(\chi_F^2 = 41.66 > \chi_8^2, F_{ID} = 5.44 > F_{8,792}, p < \alpha)$. Therefore, the outcome is to reject H_0 for the hyperareas, which means that there exists a significant difference between at least one of the groups in terms of the hyperarea value. Consequently, the Nemenyi *post hoc* test is

required to determine between which groups a statistically significant difference exists. The tests, however, do not indicate significance for the number of non-dominated solutions found $(\chi_F^2 = 10.28 < \chi_8^2, F_{ID} = 1.29 < F_{8,792}, p > \alpha)$ and therefore, fail to reject H_0 .

Table 6.10: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A1.5.1–A1.5.9.

	Friedman test	p-value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	41.66	0	5.44	0	Reject H_0
Non-dominated solutions	10.28	0.25	1.29	0.25	Fail to reject H_0

The Nemenyi *post hoc* test is presented in Table 6.11 and contains the adjusted *p*-values of the multiple pairwise comparisons for hyperareas. For example, the Nemenyi *post hoc* test found that there is a statistically significant difference between A1.5.1 and A1.5.3, A1.5.6 and A1.5.9.

Table 6.11: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the hyperareas of the approximation fronts for the hyperparameter combinations A1.5.1-A1.5.9.

	A1.5.2	A1.5.3	A1.5.4	A1.5.5	A1.5.6	A1.5.7	A1.5.8	A1.5.9
A1.5.1	1	0	1	1	0	1	1	0
A1.5.2		1	1	1	0.73	1	1	1
A1.5.3			0.02	0.33	1	0.21	0.63	1
A1.5.4				1	0.01	1	1	0.02
A1.5.5					0.14	1	1	0.38
A1.5.6						0.08	0.28	1
A1.5.7							1	0.24
A1.5.8								0.73

This concludes the individual (or problem-specific) analysis of the hyperparameter combinations for the MOOCEM. Finally, the single best hyperparameter combination can be determined for the MOOCEM, across all five simulation problems. Note that only the hyperareas are considered, since the hyperarea is the better indicator of the quality of an approximation front. Figure 6.8 represents the average hyperareas obtained for each respective hyperparameter combination (collectively referred to as A1.1–A1.9) for each simulation problem. The average number of non-dominated solutions obtained for hyperparameter combinations A1.1–A1.9 is presented in Figure B.11 for each simulation problem.

Figure 6.9 is used to explain the process followed to determine the single best hyperparameter combination (specifically for BAP10) for the MOOCEM. From the *p*-values given in Table 6.9 and included in the figure, it can be seen that hyperparameter combinations A1.4.4–A1.4.6 are statistically significantly different to A1.4.1–A1.4.3 and A1.4.7–A1.4.9 which is illustrated by the red rectangles in Figure 6.9. Because A1.4.4–A1.4.6 have the smaller average hyperareas they are eliminated from consideration. Accordingly, A1.4.1–A1.4.3 and A1.4.7–A1.4.9 are not statistically significantly different and are therefore considered. A similar process was followed for each problem and the outcome is presented in Table 6.12, where it is clear that hyperparameter combination A1.2–A1.3 and A1.7–A1.9 are the common denominators across all five simulation problems.

Next, the hyperparameter combination (*i.e.* A1.2–A1.3 and A1.7–A1.9) that obtains the largest average hyperarea (for most of the time) is chosen. The green checkmarks in Table 6.12 correspond to the hyperparameter combinations with the largest average hyperarea. Consequently, because hyperparameter combination A1.9 (or $\alpha = 0.75$ and $p_h = 0.4$) resulted in the largest

6.2. Determining the MOOCEM algorithm hyperparameters



Figure 6.8: The average hyperareas obtained for hyperparameter combinations A1.1–A1.9 for the respective simulation problems.

average hyperarea for four out of the five problems, it is employed in the BOCEGAH. Note that hyperparameter combinations A1.2–A1.3 and A1.7–A1.9 are not statistically significantly different, *i.e.* any one of these combinations could be used, however, one had to be chosen.

Interestingly, Bekker [21] found that the probability of inverting the decision variable histograms p_h is typically in the range [0.1, 0.3], however the hyperparameter study conducted for the MOOCEM on the five simulation problems concluded that $p_h = 0.4$ is the common best hyperparameter combination.

This concludes the empirical study determining the hyperparameter combination to be employed for the MOOCEM in the BOCEGAH. The results can be used as a guideline for choosing hyperparameters for the MOOCEM, when deciding on suitable hyperparameter combinations to employ, or is at least a good place to start.

The next section documents the extensive parametric study conducted for the NSGA-II.



Figure 6.9: An example illustrating the hyperparameter selection process followed for BAP10 based on the Friedman and Iman-Davenport extension hypothesis test and Nemenyi post hoc test conducted for the hyperareas obtained for hyperparameter combinations A1.4.1–A1.4.9.

Table 6.12: The hyperparameters for the MOOCEM that are considered for implementation in the BOCEGAH.

	A1.1	A1.2	A1.3	A1.4	A1.5	A1.6	A1.7	A1.8	A1.9
OMP	\checkmark	✓	✓	\checkmark	\checkmark	\checkmark	✓	\checkmark	\checkmark
IP	\checkmark								
BAP5	\checkmark								
BAP10	\checkmark	\checkmark	\checkmark				\checkmark	\checkmark	\checkmark
BAP16		\checkmark	\checkmark		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark

6.3 Determining the NSGA-II hyperparameters

The NSGA-II contains many interacting hyperparameters to be adjusted and consequently, the purpose of this section is to determine the single best hyperparameter combination for the NSGA-II that results in the best algorithmic performance, across all five simulation problems. Ultimately, the results of the Friedman and Iman-Davenport extension tests are combined and used to decide which hyperparameter combination to implement for NSGA-II and employ in the BOCEGAH.

- **Population size** N is typically chosen in the range [20, 100] because large populations result in better convergence toward the true Pareto front, however, the time complexity increases linearly with the size of the population [93]. In this study N = 100.
- **Crossover probability** p_c typically ranges between [0.3, 0.9]. In this study the $p_c \in \{0.9, 1\}$ is considered, *i.e.* crossover operations are applied to 90% and 100% of the parents.
- Mutation probability p_m typically ranges between [0.001, 0.01], where large probabilities could disrupt the search. Usually, the $p_m = 1/n_d$, where n_d is the number of decision variables, *i.e.* on average, one variable is mutated [93]. However, because the number of decision variables differs from simulation problem to simulation problem, ranging from 2 to 15, $p_m \in \{0.1, 0.2, 0.3\}$ is considered.

The NSGA-II parameters that are considered for the hyperparameter study are summarised in Table 6.13. A similar naming convention is adopted as before, where A2.1.1 refers to the hyperparameter combination rank selection, polynomial mutation, $p_c = 0.9$ and $p_m = 0.1$ for the OMP specifically. A search space of 24 combinations per simulation problem is explored, *i.e.* the number of groups being compared is k = 24.

The search space explores selection operators, binary tournament selection and rank selection, crossover operator simulated binary crossover, mutation operators polynomial mutation and dynamic mutation and (as mentioned) crossover probabilities $p_c \in \{0.9, 1\}$ and mutation probabilities $p_m \in \{0.1, 0.2, 0.3\}$ with a maximum of 1 000 solutions evaluations and a population size N = 100. Again, each hyperparameter combination (or group) is run for 100 individual (simulation) runs (unless stated otherwise), *i.e.* n = 100 observations per group resulting in 100 individual hyperareas and number of non-dominated solutions.

Table 6.13: The hyperparameter search space and corresponding naming convention adopted refer to the specific hyperparameter combinations for the specific simulation problem, resulting in the entire hyperparameter search space considered for the NSGA-II.

Selection	Mutation	Crossover	Mutation	OMP	IP	BAP5	BAP10	BAP16
operator	operator	probability	probability					
			$p_m = 0.1$	A2.1.1	A2.2.1	A2.3.1	A2.4.1	A2.5.1
lial I	$p_{c} = 0.9$	$p_m = 0.2$	A2.1.2	A2.2.2	A2.3.2	A2.4.2	A2.5.2	
	omo		$p_m = 0.3$	A2.1.3	A2.2.3	A2.3.3	A2.4.3	A2.5.3
c	lyn ıtat		$p_m = 0.1$	A2.1.4	A2.2.4	A2.3.4	A2.4.4	A2.5.4
tio	Po	$p_c = 1$	$p_m = 0.2$	A2.1.5	A2.2.5	A2.3.5	A2.4.5	A2.5.5
elec			$p_m = 0.3$	A2.1.6	A2.2.6	A2.3.6	A2.4.6	A2.5.6
k se			$p_m = 0.1$	A2.1.7	A2.5.7	A2.3.7	A2.4.7	A2.5.7
lan		$p_{c} = 0.9$	$p_m = 0.2$	A2.1.8	A2.2.8	A2.3.8	A2.4.8	A2.5.8
ц	nic		$p_m = 0.3$	A2.1.9	A2.2.9	A2.3.9	A2.4.9	A2.5.9
	/nau litat	$p_{c} = 1$	$p_m = 0.1$	A2.1.10	A2.2.10	A2.3.10	A2.4.10	A2.5.10
	DJ DJ		$p_m = 0.2$	A2.1.11	A2.2.11	A2.3.11	A2.4.11	A2.5.11
			$p_m = 0.3$	A2.1.12	A2.2.12	A2.3.12	A2.4.12	A2.5.12
			$p_m = 0.1$	A2.1.13	A2.2.13	A2.3.13	A2.4.13	A2.5.13
n	ial	$p_c = 0.9$	$p_m = 0.2$	A2.1.14	A2.2.14	A2.3.14	A2.4.14	A2.5.14
ctio	om		$p_m = 0.3$	A2.1.15	A2.2.15	A2.3.15	A2.4.15	A2.5.15
elee	lyn ıtat		$p_m = 0.1$	A2.1.16	A2.2.16	A2.3.16	A2.4.16	A2.5.16
nts	Po	$p_c = 1$	$p_m = 0.2$	A2.1.17	A2.2.17	A2.3.17	A2.4.17	A2.5.17
iem			$p_m = 0.3$	A2.1.18	A2.2.18	A2.3.18	A2.4.18	A2.5.18
rna			$p_m = 0.1$	A2.1.19	A2.2.19	A2.3.19	A2.4.19	A2.5.19
inary tour namic ttation		$p_{c} = 0.9$	$p_m = 0.2$	A2.1.20	A2.2.20	A2.3.20	A2.4.20	A2.5.20
		$p_m = 0.3$	A2.1.21	A2.2.21	A2.3.21	A2.4.21	A2.5.21	
	/na. .tat		$p_m = 0.1$	A2.1.22	A2.2.22	A2.3.22	A2.4.22	A2.5.22
Щ	D, D	$p_c = 1$	$p_m = 0.2$	A2.1.23	A2.2.23	A2.3.23	A2.4.23	A2.5.23
			$p_m = 0.3$	A2.1.24	A2.2.24	A2.3.24	A2.4.24	A2.5.24

The hypothesis tests conducted are first discussed for the OMP, then the IP, BAP5, BAP10 and finally for BAP16. Thereafter, the results of the respective tests are combined and a single best hyperparameter combination is chosen. Note that the previous sections (§6.2.1–§6.2.5) will be used as a template to succinctly present the remainder of the results.

6.3.1 Open mine problem

Recall that for the OMP, only 40 simulation runs per hyperparameter combination were executed, *i.e.* n = 40 observations. Accordingly, the critical values for k = 24 and n = 40 for the Chi-square and F-distributions are $\chi^2_{23} = 35.17$ and $F_{23,897} = 1.54$, respectively. Tables B.1 and B.2 present the adjusted hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 39–40 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.10 provide a graphical summary of the results for hyperparameter combinations A2.1.1–A2.1.24, as presented in Tables B.18 and B.19.



Figure 6.10: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A2.1.1–A2.1.24.

Table 6.14 shows the number of runs (out of 40) that obtained the true Pareto set, for each hyperparameter combination. For example, for hyperparameter combination A2.1.1, the NSGA-II obtained the true Pareto set 3 out of the 40 times (*i.e.* 7.5% of the time). Consequently, validating that the NSGA-II algorithm works correctly and will be able to solve the larger problems that follow this discussion.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.20 and depicted in Figures B.12–B.14 for A2.1.1–A2.1.24.

Table 6.14: The number of runs that the NSGA-II obtained the true Pareto set, for the respective hyperparameter combinations.

A2.1.1	A2.1.4	A2.1.7	A2.1.8	A2.1.9	A2.1.10	A2.1.11	A2.1.12	A2.1.20	A2.1.22	A2.1.23
3	2	2	4	1	4	1	1	3	1	2

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.15. The Friedman and Iman-Davenport test statistics for both the adjusted hyperareas (and the number of non-dominated solutions) indicate significance ($\chi_F^2 = \{304.76, 285.67\} > \chi_{23}^2$, $p < \alpha$ and $F_{ID} = \{19.32, 17.56\} > F_{23,897}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

Table 6.15: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A2.1.1–A2.1.24.

	Friedman test	<i>p</i> -value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	304.76	0	19.32	0	Reject Ho
Non-dominated solutions	285.67	0	17.56	0	neject m

The Nemenyi *post hoc* test is presented in Tables B.21 and B.22 and contains the adjusted p-values of the multiple pairwise comparisons for both the adjusted hyperareas and number of non-dominated solutions, respectively. The p-values in red indicate statistical significance, for example A1.1.1 and A1.1.13–A1.1.19, A1.1.21–A1.1.22 and A1.1.24.

6.3.2 (s, S) Inventory problem

Recall that for the IP, BAP5, BAP10 and BAP16, 100 simulation runs per hyperparameter combination were executed, *i.e.* n = 100 observations per group. The critical values for k = 24 and n = 100 for the Chi-square and F-distributions are $\chi^2_{23} = 35.17$ and $F_{23,2277} = 1.53$, respectively. Tables B.23 and B.24 present the hyperareas and number of non-dominated solutions and their ranks for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.11 provide a graphical summary of the results for hyperparameter combinations A2.2.1–A2.2.24, as presented in Tables B.23 and B.24.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are given in Table B.25 and presented in Figures B.15 and B.19 for A2.2.1–A2.2.24.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.16. The Friedman and Iman-Davenport test statistics both indicate significance for both the hyperarea and non-dominated solutions ($\chi_F^2 = \{171.73, 1030.87\} > \chi_{23}^2$, $F_{ID} = \{7.99, 80.41\} > F_{23,2277}, p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

The Nemenyi *post hoc* test is presented in Tables B.26 and B.27 and contains the adjusted *p*-values of the multiple pairwise comparisons for number of non-dominated solutions. The red *p*-values indicate statistical significance, for example there are statistically significant differences between A2.2.1 and A2.2.13, A2.2.15, A2.2.18 and A2.2.19.



Figure 6.11: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A2.2.1–A2.2.24.

Table 6.16: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A2.2.1–A2.2.24.

	Friedman test	<i>p</i> -value	Iman-Davenport test	p-value	Outcome
_	χ^2_F	p	F_{ID}	p	
Hyperarea	171.73	0	7.99	0	Reject Ho
Non-dominated solutions	1030.87	0	80.41	0	neject II()

6.3.3 Buffer allocation problem: five machines

Tables B.28 and B.29 present the hyperareas and number of non-dominated solutions and their ranks for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.12 provide a graphical



Figure 6.12: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A2.3.1–A2.3.24.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are given in Table B.30 and presented in Figures B.20 and B.24 for A2.3.1–A2.3.24.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.17. The Friedman and Iman-Davenport test statistics both indicate significance for both the hyperarea and non-dominated solutions ($\chi_F^2 = \{90.47, 274.21\} > \chi_{23}^2$, $p < \alpha$ and $F_{ID} = \{4.05, 13.4\} > F_{23,2277}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

The Nemenyi *post hoc* test is presented in Tables B.31 and B.32 and contains the adjusted p-values of the multiple pairwise comparisons for number of non-dominated solutions. The red

	Friedman test	<i>p</i> -value	Iman-Davenport test	<i>p</i> -value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	90.47	0	4.05	0	Reject Ho
Non-dominated solutions	274.21	0	13.4	0	iteject 110

Table 6.17: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A2.3.1–A2.3.24.

p-values indicate statistical significance, for example there are statistically significant differences between A2.2.1 and A2.2.13, A2.2.15, A2.2.18 and A2.2.19 in terms of hyperarea.

6.3.4 Buffer allocation problem: 10 machines

Tables B.33 and B.34 present the hyperareas and number of non-dominated solutions and their ranks for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.13 provide a graphical summary of the results for hyperparameter combinations A2.4.1–A2.4.24, as presented in Tables B.33 and B.34. The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are given in Table B.35 and presented in Figures B.25 and B.29 for A2.4.1–A2.4.24. The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.17. The Friedman and Iman-Davenport test statistics both indicate significance for both the hyperarea and non-dominated solutions $(\chi_F^2 = \{268.26, 103.06\} > \chi_{23}^2, p < \alpha$ and $F_{ID} = \{13.07, 4.64\} > F_{23,2277}, p < \alpha$). Consequently, H_0 is rejected and the Nemenyi *post hoc* test is conducted.

Table 6.18: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A2.4.1–A2.4.24.

	Friedman test	<i>p</i> -value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	268.26	0	13.07	0	Reject Ho
Non-dominated solutions	103.06	0	4.64	0	neject 110

The Nemenyi *post hoc* test is presented in Tables B.36 and B.37 and contains the adjusted *p*-values of the multiple pairwise comparisons for number of non-dominated solutions. The red *p*-values indicate statistical significance, for example there are statistically significant differences between A2.4.1 and A2.4.8–A2.4.9, A2.4.11–A2.4.12, A2.4.21 and A2.4.24 in terms of hyperarea.

6.3.5 Non-linear buffer allocation problem: 16 machines

Tables B.38 and B.39 present the hyperareas and number of non-dominated solutions and their ranks for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.14 provide a graphical summary of the results for hyperparameter combinations A2.5.1–A2.5.24, as presented in Tables B.38 and B.39. The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are given in Table B.30 and presented in Figures B.30 and B.34 for A2.5.1–A2.5.24.



Figure 6.13: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A2.4.1–A2.4.24.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.19. The Friedman and Iman-Davenport test statistics both indicate significance for both the hyperarea and non-dominated solutions ($\chi_F^2 = \{115.15, 73.33\} > \chi_{23}^2$, $p < \alpha$ and $F_{ID} = \{5.22, 3.26\} > F_{23,2277}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

Table 6.19: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A2.5.1–A2.5.24.

	Friedman test	<i>p</i> -value	Iman-Davenport test	<i>p</i> -value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	115.15	0	5.22	0	Reject Ho
Non-dominated solutions	73.33	0	3.26	0	neject II()



Figure 6.14: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A2.5.1–A2.5.24.

The Nemenyi *post hoc* test is presented in Tables B.41 and B.42 and contains the adjusted *p*-values of the multiple pairwise comparisons for number of non-dominated solutions. The red *p*-values indicate statistical significance, for example there are statistically significant differences between A2.5.1 and A2.5.9 and A2.5.12 in terms of hyperarea.

This concludes the individual (or problem-specific) analysis of the hyperparameter combinations for the NSGA-II. Finally, the single best hyperparameter combination can be determined for the NSGA-II, across all five simulation problems. Again, note that only the hyperareas are considered, since the hyperarea is the better indicator of the quality of an approximation front. Figure 6.15 represents the average hyperareas obtained for each respective hyperparameter combination (collectively referred to as A2.1–A2.24) for the simulation problems. The average number of non-dominated solutions obtained for hyperparameter combinations A2.1–A2.24 is presented in Figure B.35 for each simulation problem.



Figure 6.15: The average hyperareas obtained for each hyperparameter combinations A2.1–A2.24 for the respective simulation optimisation problems.

Figure 6.16 is used to explain the process followed to determine the single best hyperparameter combination (specifically for BAP16) for the NSGA-II. From the *p*-values given in Table B.41 and included in the figure, it can be seen that A2.5.12 is statistically significantly different to A2.5.1, A2.5.4, A2.5.13–A2.5.23 which is illustrated by the red rectangle in Figure 6.16. Because A2.5.1, A2.5.4, A2.5.13–A2.5.23 have smaller average hyperareas they are eliminated from consideration.

Accordingly, solutions A2.5.3, A2.5.7, A2.5.9, A2.5.11, A2.5.12 and A2.5.24 are not statistically significantly different and are therefore considered. A similar process was followed for each problem and the outcome is presented in Table 6.20, where it is clear that hyperparameter combination A2.3 is the common denominator across all five simulation problems. Consequently, hyperparameter combination A2.3 (or rank selection, polynomial mutation, $p_c = 0.9$ and $p_m = 0.3$) is employed in BOCEGAH.

Interestingly, the proposed range for p_m is typically between [0.001, 0.01], since large probabilities could disrupt the search [93], however the hyperparameter study conducted for the NSGA-II



Figure 6.16: An example illustrating the hyperparameter selection process followed for BAP16 based on the Friedman and Iman-Davenport extension hypothesis test and Nemenyi post hoc test conducted for the hyperareas obtained for hyperparameter combinations A2.5.1–A2.5.24.

on the five simulation problems concluded that $p_m = 0.3$ is the common best hyperparameter combination.

Table 6.20: The hyperparameters combinations for the NSGA-II that are considered for implementationin BOCEGAH.

	OMP	IP	BAP5	BAP10	BAP16
A2.1	\checkmark		\checkmark		
A2.2	\checkmark	\checkmark	\checkmark		
A2.3	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
A2.4	\checkmark		\checkmark		
A2.5	\checkmark		\checkmark		
A2.6		\checkmark	\checkmark	\checkmark	
A2.7	\checkmark				\checkmark
A2.8	\checkmark		\checkmark	\checkmark	
A2.9	\checkmark		\checkmark	\checkmark	\checkmark
A2.10	\checkmark				
A2.11	\checkmark		\checkmark	\checkmark	\checkmark
A2.12	\checkmark		\checkmark	\checkmark	\checkmark
A2.13		\checkmark			
A2.14		\checkmark	\checkmark		
A2.15		\checkmark	\checkmark	\checkmark	
A2.16		\checkmark			
A2.17		\checkmark	\checkmark		
A2.18		\checkmark	\checkmark		
A2.19		\checkmark			
A2.20		\checkmark	\checkmark		
A2.21			\checkmark	\checkmark	
A2.22		\checkmark			
A2.23		\checkmark	\checkmark	\checkmark	
A2.24			\checkmark	\checkmark	\checkmark

This concludes the empirical study determining the hyperparameter combination to be employed for the NSGA-II in the BOCEGAH. The results can be used a guideline for choosing hyperparameters for the NSGA-II, when deciding on suitable hyperparameter combinations to employ, or is at least a good place to start.

The next section documents the parametric study conducted for the move operators proposed for DBMOSA.

6.4 Determining the DBMOSA algorithm hyperparameters

The aim in this section is to determine which three move operators for the DBMOSA resulted in the best algorithmic performance, across all five simulation problems. Ultimately, the results of the Friedman and Iman-Davenport extension tests are combined and used to decide which three move operators to implement for the DBMOSA to employ in the BOSAH. First, it is necessary to define which parameters are held constant and are summarised in Table 6.21.

Table 6.21: The hyperparameters that are held constant during the hyperparameter study for DBMOSA.

Parameter	Value
Initial temperature	10
Maximum iterations	1000
Annealing schedule	Geometric
Cooling parameter	0.75
Heating parameter	1.2
Maximum accepts	10
Maximum attempts	5

It is important to know that for DBMOSA each run was run with a different *initial solution* and RNS, an example of which is depicted in Table 6.22 for the IP. For $Run \ 1$ the initial solution is s = 1 and S = 1 with RNS = 1, $Run \ 2$ the initial solution is s = 5 and S = 5 with RNS = 2, until $Run \ 100$ with the initial solution is s = 397 and S = 397 with RNS = 100, applied to all the hyperparameter combinations being evaluated. A similar approach is followed for the other simulation optimisation problems. Moreover, the initial solution for each run remains the same for the different hyperparameter combinations.

	Initial solution	RNS
Run 1	$\{1,1\}$	1
$\operatorname{Run} 2$	$\{5,5\}$	2
÷	:	÷
$\operatorname{Run}100$	$\{397, 397\}$	100

 Table 6.22:
 DBMOSA setup information.

The DBMOSA parameters that are considered for the hyperparameter study are summarised in Table 6.23. A similar naming convention is adopted as before, where A3.1.1 refers to the hyperparameter combination represents *move operator* 1, for the OMP specifically. A search space of four combinations per simulation problem is explored, *i.e.* the number of groups being compared is k = 4. Again, each hyperparameter combination (or group) is run for 100 individual (simulation) runs (unless stated otherwise), *i.e.* n = 100 observations per group resulting in 100 individual hyperareas and number of non-dominated solutions. Also, each run consists of 1 000 solution evaluations (as mentioned previously) to facilitate fair comparison.

The hypothesis tests conducted are first discussed for the OMP, then the IP, BAP5, BAP10 and finally for BAP16. Thereafter, the results of the respective tests are combined and the three best move operators are chosen.

Table 6.23: The hyperparameter search space and corresponding naming convention adopted refer to the specific hyperparameter combinations for the specific simulation problem, resulting in the entire hyperparameter search space considered for the DBMOSA.

Move operator	OMP	IP	BAP5	BAP10	BAP16
1	A3.1.1	A3.2.1	A3.3.1	A3.4.1	A3.5.1
2	A3.1.2	A3.2.2	A3.3.1	A3.4.2	A3.5.2
3	A3.1.3	A3.2.3	A3.3.3	A3.4.3	A3.5.3
4	A3.1.4	A3.2.4	A3.3.1	A3.4.4	A3.5.4

Open mine problem

Recall that for the OMP, only 40 simulation runs per hyperparameter combination were executed, *i.e.* n = 40 observations. Accordingly, the critical values for k = 4 and n = 40 for the Chi-square and F-distributions are $\chi_3^2 = 7.82$ and $F_{3,117} = 2.68$, respectively. Tables B.43 and B.44 present the adjusted hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 39–40 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.17 provide a graphical summary of the results for hyperparameter combinations A3.1.1–A3.1.4, as presented in Tables B.43 and B.44.



Figure 6.17: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A3.1.1–A3.1.4.

The DBMOSA was unable to find the true Pareto set for any of the 40 simulation runs for the four different move operators used. However, this does not in-validate the DBMOSA since the hyperparameter search space is very small. The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.3 and depicted in Figure B.36 for A3.1.1–A3.1.4.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.24. The Friedman and Iman-Davenport test statistics for both the adjusted hyperareas (and the number of non-dominated solutions) do not indicate significance ($\chi_F^2 = \{7.41, 5.29\} < \chi_3^2, p > \alpha$ and $F_{ID} = \{2.57, 1.8\} < F_{3,117}, p > \alpha$). Therefore, the tests fails to reject H_0 , which means that there is no statistically significant difference between any of the groups, *i.e.* the move operators are equally desirable. Consequently, no *post hoc* analysis is required.

Table 6.24: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A3.1.1–A3.1.4.

	Friedman test	p-value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	7.41	0.06	2.57	0.058	Fail to reject H_0
Non-dominated solutions	5.29	0.15	1.8	0.15	Fail to reject 110

(s, S) Inventory problem

Recall that for the IP, BAP5, BAP10 and BAP16, 100 simulation runs per hyperparameter combination were executed, *i.e.* n = 100 observations. Accordingly, the critical values for k = 4and n = 100 for the Chi-square and F-distributions are $\chi_3^2 = 7.82$ and $F_{3,297} = 2.64$, respectively. Tables B.46 and B.47 present the hyperareas and number of non-dominated solutions and their ranks for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.18 provide a graphical summary of the results for hyperparameter combinations A3.2.1–A3.2.4, as presented in Tables B.46 and B.47.



Figure 6.18: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A3.2.1–A3.2.4.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.48 and depicted in Figure B.37 for A3.2.1–A3.2.4.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.25. The Friedman and Iman-Davenport test statistics indicate significance for both the hyperareas and number of non-dominated solutions found ($\chi_F^2 = \{128.68, 28.82\} > \chi_3^2$, $p < \alpha$ and $F_{ID} = \{74.36, 10.52\} > F_{3,297}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

The Nemenyi *post hoc* test is presented in Tables 6.26a and 6.26b and contains the adjusted *p*-values of the multiple pairwise comparisons for both the adjusted hyperareas and number of non-dominated solutions, respectively. The red *p*-values indicate statistical significance. For example, there are statistically significant differences between A3.2.1 and A3.2.2–A3.2.4 for the hyperareas and number of non-dominated solutions.

Table 6.25:	The Fi	riedman	and I	man	Davenport	test	results	based	on	the	ranked	hyperareas	and	non-
dominated so	olutions	obtained	l for	each l	nyperparar	neter	combin	nation	A3.	2.1-	A3.2.4.			

	Friedman test	<i>p</i> -value	Iman-Davenport test	<i>p</i> -value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	128.68	0	74.36	0	Reject Ho
Non-dominated solutions	28.82	0	10.52	0	10,000 110

Table 6.26: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons for (a) hyperarea and (b) number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A3.2.1-A3.2.4.

(a) Hyperarea					(b) Non-dominated solutions				
	A3.2.2	A3.2.3	A3.2.4			A3.2.2	A3.2.3	A3.2.4	
A3.2.1	0	0	0		A3.2.1	1	0	1	
A3.2.2		0	0.01		A3.2.2		0	1	
A3.2.3			0.33		A3.2.3			0	

Buffer allocation problem: five machines

Tables B.49 and B.50 present the hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.19 provide a graphical summary of the results for hyperparameter combinations A3.3.1–A3.3.4, as presented in Tables B.49 and B.50.



Figure 6.19: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A3.3.1–A3.3.4.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.51 and depicted in Figure B.38 for A3.3.1–A3.3.4.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.27. The Friedman and Iman-Davenport test statistics indicate strong significance for both the hyperareas and number of non-dominated solutions found ($\chi_F^2 = \{75.16, 148.04\} > \chi_3^2$, $p < \alpha$ and $F_{ID} = \{33.09, 96.44\} > F_{3,297}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

The Nemenyi post hoc test is presented in Tables 6.28a and 6.28b and contains the adjusted
	Friedman test	p-value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	75.16	0	33.09	0	Reject Ho
Non-dominated solutions	148.04	0	96.44	0	reject 110

Table 6.27: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A3.3.1–A3.3.4.

p-values of the multiple pairwise comparisons for both the adjusted hyperareas and number of non-dominated solutions, respectively. The red *p*-values indicate statistical significance. For example, there are statistically significant differences between A3.3.1 and A3.3.2–A3.3.4 for the hyperareas and number of non-dominated solutions.

Table 6.28: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons for (a) hyperarea and (b) number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A3.3.1–A3.3.4.

	(a) <i>Hy</i>	perarea		(b) N	on-domi	nated sol	utions
	A3.3.2	A3.3.3	A3.3.4		A3.3.2	A3.3.3	A3.3.4
A3.3.1	0	0	0	A3.3.1	0	0	0
A3.3.2		0	0.29	A3.3.2		0	0.64
A3.3.3			0.26	A3.3.3			0

Buffer allocation problem: 10 machines

Tables B.52 and B.53 present the hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.20 provide a graphical summary of the results for hyperparameter combinations A3.4.1–A3.4.4, as presented in Tables B.52 and B.53.



Figure 6.20: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A3.4.1–A3.4.4.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.54 and depicted in Figure B.39 for A3.4.1–A3.4.4.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.29. The Friedman and Iman-Davenport test statistics indicate strong significance for both the hyperareas and number of non-dominated solutions found ($\chi_F^2 = \{105.28, 101.96\} > \chi_3^2, p < \alpha$ and $F_{ID} = \{53.52, 50.97\} > F_{3,297}, p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

Table 6.29: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A3.4.1–A3.4.4.

	Friedman test	p-value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	105.28	0	53.52	0	Reject H_0
Non-dominated solutions	101.96	0	50.97	0	iteject 110

The Nemenyi *post hoc* test is presented in Tables 6.30a and 6.30b and contains the adjusted *p*-values of the multiple pairwise comparisons for both the hyperarea and number of non-dominated solutions. For example, there are statistically significant differences between A3.4.1 and A3.4.2–A3.4.4 for both hyperarea and number of non-dominated solutions.

Table 6.30: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons for (a) hyperarea and (b) number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A3.4.1-A3.4.4.

	(a) Hyperarea				(b) Non-dominated solutions					
	A3.4.2	A3.4.3	A3.4.4			A3.4.2	A3.4.3	A3.4.4		
A3.4.1	0	0	0		A3.4.1	0	0	0		
A3.4.2		0.18	0.01		A3.4.2		0	0.13		
A3.4.3			1		A3.4.3			1		

Non-linear buffer allocation problem: 16 machines

Tables B.55 and B.56 present the hyperareas and number of non-dominated solutions and their ranks (respectively) for run 1–10 and 99–100 as well as the sum of the ranks which is used in the Friedman and Iman-Davenport extension hypothesis tests. The boxplots in Figure 6.21 provide a graphical summary of the results for hyperparameter combinations A3.5.1–A3.5.4, as presented in Tables B.55 and B.56.

The runs that correspond to the worst and best approximation fronts obtained (in terms of hyperarea and number of non-dominated solutions) are presented in Table B.57 and depicted in Figure B.40 for A3.5.1–A3.5.4.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 6.31. The Friedman and Iman-Davenport test statistics indicate strong statistical significance for both the hyperareas and number of non-dominated solutions found ($\chi_F^2 = \{86.96, 66.52\} > \chi_3^2$, $p < \alpha$ and $F_{ID} = \{40.41, 28.21\} > F_{3,297}$, $p < \alpha$). Consequently, H_0 is rejected and the Nemenyi post hoc test is conducted.

The Nemenyi *post hoc* tests are presented in Tables 6.32a and 6.32b and contains the adjusted p-values of the multiple pairwise comparisons for both the hyperareas and number of non-dominated solutions, respectively. For example, there was statistically significant differences between A3.5.1 and A3.5.2–A3.5.4 for both hyperarea and number of non-dominated solutions.

146



Figure 6.21: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for hyperparameter combinations A3.5.1–A3.5.4.

Table 6.31: The Friedman and Iman-Davenport test results based on the ranked hyperareas and nondominated solutions obtained for each hyperparameter combination A3.5.1–A3.5.4.

	Friedman test	p-value	Iman-Davenport test	p-value	Outcome
	χ^2_F	p	F_{ID}	p	
Hyperarea	86.96	0	40.41	0	Reject Ho
Non-dominated solutions	66.52	0	28.21	0	neject m

Table 6.32: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons for (a) hyperarea and (b) number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A3.5.1–A3.5.4.

	(a) <i>Hy</i>	perarea		(b) N	Ion-domi	nated sol	utions
	A3.5.2	A3.5.3	A3.5.4		A3.5.2	A3.5.3	A3.5.4
A3.5.1	0	0	0	A3.5.1	0	0	0
A3.5.2		0.28	0.09	A3.5.2		0.77	0.6
A3.5.3			1	A3.5.3			1

This concludes the individual (or problem-specific) analysis of the four move operators proposed for the DBMOSA. Finally, the three best move operators can be determined for the DBMOSA, across all five simulation problems. Recall that only the hyperareas are considered, since the hyperarea is the better indicator of the quality of an approximation front. Figure 6.22 represents the average hyperareas obtained for each respective hyperparameter combination (collectively referred to as A3.1–A3.4) for each simulation problem. The average number of non-dominated solutions obtained for hyperparameter combinations A3.1–A3.4 is presented in Figure B.41 for each simulation problem.

Figure 6.23 is used to explain the process followed to determine the three best move operators (specifically for BAP16) for the DBMOSA. From the *p*-values given in Table 6.32a and included in the figure, it can be seen that hyperparameter combination A3.5.1 (or move operator 1) is statistically significantly different to A3.5.2–A3.5.4 which is illustrated by the red rectangles in Figure 6.23. Consequently, move operator 1 is considered. Even though move operator 1 is statistically significantly better than move operator 2-4, the hyperheuristic requires three LLHs.

The second and third best move operators need to be determined and because A3.5.2–A3.5.4 are not statistically significantly different $(p > \alpha)$, the two move operators with the largest average



Figure 6.22: The average hyperareas obtained for hyperparameter combinations A3.1–A3.4 for the respective simulation problems.

hyperarea are chosen. Accordingly, move operators 1, 3 and 4 are considered for the BAP16. A similar process was followed for each problem and the outcome is presented in Table 6.33, where it is clear that hyperparameter combination A3.1, A3.3 and A3.4 are the common denominators across four of the five simulation problems. Interestingly, move operator 1 (or A3.1) constantly outperforms the other (proposed) move operators.



Figure 6.23: An example illustrating the hyperparameter selection process followed for BAP16 based on the Friedman and Iman-Davenport extension hypothesis test and Nemenyi post hoc test conducted for the hyperareas obtained for hyperparameter combinations A3.5.1–A3.5.4.

Table 6.33: The move operators that are considered for implementation in BOSAH.

	A3.1	A3.2	A3.3	A3.4
OMP	\checkmark		\checkmark	\checkmark
IP	\checkmark	\checkmark		\checkmark
BAP5	\checkmark		\checkmark	\checkmark
BAP10	\checkmark		\checkmark	\checkmark
BAP16	\checkmark		\checkmark	\checkmark

This concludes the empirical study determining the three hyperparameter combinations to be employed for the DBMOSA in the BOSAH.

6.5 Conclusion: Chapter 6

The purpose of this chapter was to determine the best hyperparameter combinations, respectively for the MOOCEM, the NSGA-II and the DBMOSA, respectively. Ultimately, the aim was to determine common hyperparameter combinations suitable for the subset of problems considered in the study. The final outcome of all the hyperparameter studies for the BOCEGAH and the BOSAH are summarised in Table 6.34.

 Table 6.34:
 The final hyperparameter combinations.

	Hyperparameter combinations
MOOCEM	A1.9
NSGA	A2.3
DBMOSA	A3.1, A3.3, A3.4

The MOOCEM will be employed in the BOCEGAH with hyperparameter combination A1.9, *i.e.* $\alpha = 0.75$ and $p_h = 0.4$. The NSGA-II will also be employed in the BOCEGAH, however with hyperparameter combination A2.3, *i.e.* rank selection, polynomial mutation, $p_c = 0.9$ and $p_m = 0.3$. Finally, the DBMOSA will be implemented in the BOSAH using move operator 1, 3 and 4. The next chapter documents the extensive algorithmic comparative study conducted for the BOCEGAH and the BOSAH.

CHAPTER 7

Algorithm Performance Assessment and Comparison

7.1	BOCEGAH versus MOOCEM and NSGA-II	151
7.2	BOSAH versus DBMOSAs move operators	154
7.3	BOCEGAH versus BOSAH	156
7.4	Conclusion: Chapter 7	158

The previous chapter documented the empirical study that was conducted to determine the best hyperparameter combination for each algorithm, *i.e.* the MOOCEM, the NSGA-II and the DBMOSA.

This chapter serves to fulfil Objective 7, as stated in Chapter 1, evaluating the quality of the BOCEGAH and the BOSAH using the five discrete-event dynamic stochastic simulation optimisation problems, as discussed in Chapter 4. The performances are documented for each hyperheuristic and each simulation problem. Thereafter, the hyperheuristics are compared to their individual LLHs to draw a conclusion on the generalisation capabilities of the proposed hyperheuristics. Finally, the BOCEGAH and the BOSAH are compared to determine whether or not population-based search or single-solution based search performs better, particularly in this study. The chapter closes with a conclusion and summary of its contents.

The Friedman and Iman-Davenport tests are used to compare hyperheuristics with their constituent LLHs. The hypothesis test is first described for the BOCEGAH and its constituent LLHs, *i.e.* the MOOCEM employing hyperparameter combination A1.9 (simply denoted by A1.9) and the NSGA-II employing hyperparameter combination A2.3 (simply denoted by A2.3). Thereafter, a similar process is followed for the BOSAH and the DBMOSA employing move operators A3.1, A3.3 and A3.4. Finally, the tests are used to compare the population-based and the single-solution based search hyperheuristics, the BOCEGAH and the BOSAH.

7.1 BOCEGAH versus MOOCEM and NSGA-II

The critical values applicable for the tests conducted in this section, for the χ^2 and *F*-distributions, are presented in Table 7.1, for k = 3 and $\alpha = 0.05$. Again, recall that n = 40 for the OMP and n = 100 for the IP, BAP5, BAP10 and BAP16. The boxplots in Figure 7.1 provide a graphical summary of the results for A1.9, A2.3 and the BOCEGAH.

Table 7.1: The critical values for the χ^2 and F-distributions at k = 3, $\alpha = 0.05$ but for different n.

		Friedman	Iman-Davenport
k	n	$\chi^2_{k-1,\alpha}$	$F_{k-1,(k-1)(n-1),\alpha}$
3	40	5.99	3.11
3	100	5.99	3.04

CHAPTER 7. ALGORITHM PERFORMANCE ASSESSMENT AND COMPARISON



Figure 7.1: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for each simulation problem comparing A1.9, A2.3 and the BOCEGAH.

The Friedman and Iman-Davenport extension hypothesis test results are summarised in Table 7.2, for each simulation problem. The outcome of the hypothesis tests, to reject H_0 is indicated by a red checkmark. For example, from Tables 7.1 and 7.2 it can be seen that (for BAP5) the Friedman and Iman-Davenport test statistics indicated statistical significance for the hyperarea $(\chi_F^2 = 0.62 < 5.99, \ p > 0.05 \text{ and } F_{ID} = 0.31 < 3.04, \ p > 0.05), \ i.e.$ the tests failed to reject H_0 (denoted by 'x'). However, the tests did not indicate statistical significance for the number of non-dominated solutions $(\chi_F^2 = 31.34 > 5.99, \ p < 0.05 \text{ and } F_{ID} = 18.39 > 3.04, \ p < 0.05), \ i.e.$ the outcome is to reject H_0 (denoted by ' \checkmark ').

Table 7.2: A summary of the Friedman and Iman-Davenport test results (for each simulation problem) based on the ranked hyperareas and non-dominated solutions obtained by comparing A1.9, A2.3 and the BOCEGAH.

		Friedman test	<i>p</i> -value	Iman-Davenport test	<i>p</i> -value	Reject H_0
		χ^2_F	p	F_{ID}	p	
OMP	Hyperarea	50.55	0	66.94	0	(
OMI	Non-dominated solutions	44.04	0	47.76	0	v
ID	Hyperarea	42.98	0	27.10	0	(
11	Non-dominated solutions	143.96	0	254.29	0	v
BAP5	Hyperarea	0.62	0.73	0.31	0.74	×
DAI 5	Non-dominated solutions	31.34	0	18.39	0	\checkmark
BAP10	Hyperarea	137.58	0	218.21	0	.(
DAI 10	Non-dominated solutions	66.47	0	49.28	0	•
BAD16	Hyperarea	155.12	0	342.18	0	(
DAF 10	Non-dominated solutions	112.19	0	126.47	0	V

Recall that §6.2 is used as a template to succinctly present the results obtained for the hypothesis tests and *post hoc* test. Accordingly, the Nemenyi *post hoc* test is conducted for the OMP, IP, BAP10 and BAP16 for both the hyperarea and number of non-dominated solutions.

However, the *post hoc* test for BAP5 is only conducted for the number of non-dominated solutions. The results are presented (where applicable) by the *p*-value tables in Figure 7.2, along with the averages of the n hyperareas and number of non-dominated solutions, for each simulation problem comparing A1.9, A2.3 and the BOCEGAH.



Figure 7.2: The *p*-values obtained by the Nemenyi post hoc test and the averages of the *n* hyperareas and number of non-dominated solutions for each simulation problem comparing A1.9, A2.3 and the BOCEGAH.

By combining A1.9 and A3.2 to create the BOCEGAH, the hope was that the ensemble algorithm would enable each LLH to compensate, to some degree, for the weaknesses of other by method of *algorithmic cooperation*. The result, as reported in [142], is a more (or at least as) effective search method that leads to better (or similar) overall performances when compared with its individual LLHs [36]. From Table 7.3 it can be seen that the performances of A1.9 for the OMP and IP are statistically significantly better to that of the BOCEGAH (in terms of hyperarea).

Table 7.3: A summary indicating the statistically significant differences in terms of hyperareas and number of non-dominated solutions for A1.9, A2.3 and the BOCEGAH for each simulation problem.

Hyperarea					Non	dominate	ed solution	s		
Problem	OMP	IP	BAP5	BAP10	BAP16	OMP	IP	BAP5	BAP10	BAP16
A1.9	\checkmark									
A2.3		\checkmark	\checkmark							
BOCEGAH			\checkmark							

A1.9 (or MOOCEM) and the BOCEGAH outperform A3.2 (or NSGA-II) in terms of hyperarea for the OMP, BAP10 and BAP16, and in terms of the number of non-dominated solutions they outperform NSGA-II for all five simulation problems. By combining MOOCEM and NSGA-II

CHAPTER 7. ALGORITHM PERFORMANCE ASSESSMENT AND COMPARISON

to create the BOCEGAH, the hope was that the ensemble algorithm would enable each LLH to compensate, to some degree, for the weaknesses of other by method of algorithmic cooperation.

However, the hypothesis test and post hoc analysis concluded that (next) MOOCEM with hyperparameter combination A1.9 outperforms the BOCEGAH for the OMP and IP. Therefore, the statement that claims hyperheuristics perform better (or at least similar) to its LLHs cannot be claimed for the population-based hyperheuristic the BOCEGAH. This concludes the performance comparisons between BOCEGAH and its LLHs, and concluded that overall the MOOCEM performed better.

7.2 BOSAH versus DBMOSAs move operators

The critical values applicable for the tests conducted in this section, for the χ^2 and *F*-distributions, are presented in Table 7.4, for k = 4 and $\alpha = 0.05$. The boxplots in Figure 7.3 provide a graphical summary of the results for A3.1, A3.3–A3.4 and the BOSAH.

Table 7.4: The critical values for the χ^2 and F-distributions at k = 4, $\alpha = 0.05$ but for different n.



Figure 7.3: Box plots illustrating the spread of hyperareas and number of non-dominated solutions for each simulation problem comparing A3.1, A3.3–A3.4 and the BOSAH.

The Friedman and Iman-Davenport hypothesis test results are summarised in Table 7.5, for each simulation problem. From Tables 7.4 and 7.5 it can be seen that (for the OMP) the Friedman and Iman-Davenport test statistics do not indicate statistical significance for either the hyperarea or number of non-dominated solutions ($\{\chi_F^2 = \{7.35, 6.01\} < 7.82, p > 0.05\}$) and $\{F_{ID} = \{2.55, 2.06\} < 2.68, p > 0.05\}$), *i.e.* the tests failed to reject H_0 (x). However, for the IP, BAP5, BAP10 and BAP16 the tests indicate strong statistical significance, *i.e.* \checkmark .

		Friedman test	p-value	Iman-Davenport test	p-value	Reject H_0
		χ^2_F	p	F_{ID}	p	
OMP	Hyperarea	7.35	0.06	2.55	0.06	×
0	Non-dominated solutions	6.01	0.11	2.06	0.11	~
IP	Hyperarea	174.84	0	138.3	0	
11	Non-dominated solutions	25.78	0	9.31	0	v
DADE	Hyperarea	83.86	0	38.41	0	.(
DAI 5	Non-dominated solutions	128.09	0	73.76	0	v
BAP10	Hyperarea	84.23	0	38.65	0	
	Non-dominated solutions	82.97	0	37.85	0	•
DAD16	Hyperarea	96.14	0	46.69	0	(
DAT 10	Non-dominated solutions	70.67	0	30.51	0	V

Table 7.5: A summary of the Friedman and Iman-Davenport test results (for each simulation problem) based on the ranked hyperareas and non-dominated solutions obtained comparing A3.1, A3.3–A3.4 and the BOSAH.

Accordingly, the Nemenyi *post hoc* test is conducted for the IP, BAP5, BAP10 and BAP16 for both the hyperarea and number of non-dominated solutions. The results are presented (where applicable) by the *p*-value tables in Figure 7.4, along with the averages of the *n* hyperareas and number of non-dominated solutions, for each simulation problem comparing A3.1, A3.3–A3.4 and the BOSAH.

From Table 7.6 it can be seen that the performances of A3.1 and BOSAH are not statistically significantly different, however they are statistically significantly different to A3.3 and A3.4 (for the IP, BAP5, BAP10 and BAP16). Therefore, it can be said that A3.1 and BOSAH outperform A3.3 and A3.4 (in terms of hyperarea).

Interestingly, A3.1 outperforms BOSAH in terms of the number of non-dominated solution, however, not for the hyperarea. For this reason, only the hyperarea is considered, since it provides a better indication of the quality of an approximation front. Consequently, it can be said that BOSAH performs similar to A3.1 and outperforms A3.3 and A3.4 on four of the five simulation problems.

Again, by combining DBMOSA with three move operators to create the BOSAH, the hope was that the ensemble algorithm would enable each LLH to compensate, to some degree, for the weaknesses of the other and consequently, the statement that claims hyperheuristics perform better or at least similar to its LLHs holds true for the single-solution based hyperheuristic the BOSAH. The result is a more (or at least as) effective search method that leads to better (or similar) overall performances when compared with its individual LLHs.

Table 7.6: A summary indicating the statistically significant differences in terms of hyperareas and number of non-dominated solutions for A3.1, A3.3–A3.4 and the BOSAH for each simulation problem.

			Hyper	area			Non	-dominate	ed solution	s
Problem	OMP	IP	BAP5	BAP10	BAP16	OMP	IP	BAP5	BAP10	BAP16
A3.1	\checkmark									
A3.3	\checkmark					\checkmark				
A3.4	\checkmark					\checkmark	\checkmark			
BOSAH	\checkmark		\checkmark	\checkmark						



Figure 7.4: The *p*-values obtained by the Nemenyi post hoc test and the averages of the *n* hyperareas and number of non-dominated solutions for each simulation problem comparing A3.1, A3.3–A3.4 and the BOSAH.

This concludes the performance comparisons between BOSAH and its LLHs, and concluded that overall BOSAH performed similar to or better than its LLHs, but never worse.

7.3 BOCEGAH versus BOSAH

Finally, the population-based and single-solution based hyperheuristics BOCEGAH and BOSAH are compared. The critical values applicable for the tests conducted in this section, for the χ^2 and *F*-distributions, are presented in Table 7.7, for k = 2 and $\alpha = 0.05$.

Table 7.7: The critical values for the χ^2 and F-distributions at k = 2, $\alpha = 0.05$ but for different n.

		Friedman	Iman-Davenport
k	n	$\chi^2_{k-1,\alpha}$	$F_{k-1,(k-1)(n-1),\alpha}$
2	40	3.84	4.09
2	100	3.84	3.94

The boxplots in Figure 7.5 provide a graphical summary of the results for the BOSAH and the BOCEGAH and provides some insight in terms of which hyperheuristic performed better. To further demonstrate the differences between BOSAH and BOCEGAH, the worst and best approximation sets in terms of hyperarea are presented in Figure E.1. The approximation fronts correspond to the approximation sets obtained for the runs listed in Tables E.2 and E.1. In addition, the worst and best approximation sets in terms of the number of non-dominated



solutions, also specified in Tables E.2 and E.1, are presented in Figure E.2.

Figure 7.5: Box plots illustrating the spread of hyperareas and number of non-dominated solutions found for the BOCEGAH and the BOSAH.

The Friedman and Iman-Davenport hypothesis test results are summarised in Table 7.5, for each simulation problem. Note that the Iman-Davenport test statistic could not be determined, for the number of non-dominated solutions, for the IP, BAP10 and BAP16. This is because $\chi_F^2 = 100$ and n = 100 and accordingly (6.2) results in a division of zero. From Tables 7.7 and 7.8 it can be seen that the Friedman and Iman-Davenport test statistics indicate significance for both hyperarea and number of non-dominated solutions for all five simulation problems, *i.e.* H_0 is rejected (\checkmark).

Since only two groups are being compared (*i.e.* k = 2), it is deemed unnecessary to perform the Nemenyi *post hoc* tests. The averages of the *n* hyperareas and number of non-dominated solutions, for each simulation problem comparing the BOSAH and the BOCEGAH are presented in Figure 7.6.



Figure 7.6: The averages of the *n* hyperareas and number of non-dominated solutions for the BOSAH and the BOCEGAH.

From table 7.9 it can be concluded that the BOSAH outperforms the BOCEGAH on the IP and BAP5 simulation problems, whereas, BOCEGAH outperforms the BOSAH on the OMP, BAP10 and BAP16. In addition, the average of the n hyperareas is presented, along with the

		Friedman test	p-value	Iman-Davenport test	p-value	Reject H_0
		χ^2_F	p	F_{ID}	p	
OMP	Hyperarea	36.1	0	361	0	
Own	Non-dominated solutions	36.1	0	361	0	•
ID	Hyperarea	9	0	9.79	0	
	Non-dominated solutions	100	0	_	_	•
DADE	Hyperarea	14.44	0	16.71	0	.(
DITIO	Non-dominated solutions	73.96	0	281.18	0	•
BAP10	Hyperarea	11.56	0	12.94	0	
	Non-dominated solutions	100	0	_	_	•
BAP16	Hyperarea	36	0	55.69	0	(
	Non-dominated solutions	100	0	_	_	v

Table 7.8: A summary of the Friedman and Iman-Davenport test results (for each simulation problem) based on the ranked hyperareas and non-dominated solutions obtained comparing the BOSAH and the BOCEGAH.

so-called hyperarea ratio.

Even though their performances are statistically significantly different, their hyperarea ratios are relatively similar (next) except for the OMP. It seems the BOSAH struggled to find the true Pareto front which could be due to the fact that the Pareto set was discontinuous, and that single-solution based search is considered exploitation oriented whereas population-based search is considered exploration oriented as they are powerful in the approximation of the whole Pareto set.

Table 7.9: A comparison between the BOSAH and the BOCEGAH.

Problem	OMP	IP	BAP5	BAP10	BAP16
BOSAH		\checkmark	\checkmark		
BOCEGAH	\checkmark			\checkmark	\checkmark
Hyperarea (BOSAH)	7968.49	113 000.49	33.48	21.5	188.98
Hyperarea (BOCEGAH)	13408.65	112654.36	33.16	21.78	200.54
True hyperarea	14485.04	114442.61	33.84	22.66	214.33
Hyperarea ratio (BOSAH)	0.5501	0.9874	0.9894	0.9488	0.8817
Hyperarea ratio (BOCEGAH)	0.9257	0.9844	0.9799	0.9612	0.9357

7.4 Conclusion: Chapter 7

This chapter opened with an extensive algorithmic comparative study focussed on comparing the algorithmic performance of the BOCEGAH with its constituent LLHs (*i.e.* the MOOCEM and the NSGA-II) and the BOSAH with its constituent LLHs (*i.e.* the A3.1, A3.3 and A3.4 for DBMOSA). Later, the BOCEGAH and the BOSAH were compared. Upon considering the evaluation results for the BOCEGAH and the BOSAH, in §7.1 and §7.2, it can be concluded that both hyperheuristics failed to exhibit superior performance and did not indicate favourable performance improvements relative to *all* the individual applications of the LLHs.

CHAPTER 8

Conclusion

8.1	Thesis summary	159
8.2	Appraisal of thesis contributions	160
8.3	Suggestions for future work	161

This final chapter comprises three section. In §8.1 a chapter-by-chapter overview of the research documented in this study is provided. This is followed in §8.2 by an appraisal of the contributions made in this study and §8.3, which contains suggestions for seven avenues of further investigation as possible follow-up work, to fulfil Objective 7 as stated in Chapter 1.

8.1 Thesis summary

This study opened with the introductory, Chapter 1, in §1.1 with a general background in which the need for a more generalise purpose optimisation tool, by means of hyperheuristic approaches (*i.e.* BOCEGAH and BOSAH), especially in the context of simulation optimisation, was motivated. Furthermore, the potential utility of an ANN as metamodel was established, motivating the pilot study that was conducted to determine its feasibility. Thereafter, the problem considered in the study was formally described in \$1.2. Following this was a delimitation of the study's scope to bi-objective discrete-event, dynamic and stochastic simulation optimisation problems and FNNs in a supervised learning paradigm, with specific focus on regression problems, presented in §1.3. In addition, the optimisation approaches considered for inclusion in the proposed hyperheuristic framework were limited to the MOOCEM, the NSGA-II and the DBMOSA. Accordingly, algorithmic comparisons were limited to the BOCEGAH (and its constituent sub-algorithms) and the BOSAH (and its constituent algorithms), for the five biobjective discrete-event, dynamic and stochastic simulation optimisation problems. Next, the objectives pursued in the study were outlined in §1.4, followed by the proposed research methodology in \$1.5. The first chapter closed in \$1.6 with a detailed description of how the remainder of the study was organised into the respective chapters and corresponding appendices.

Apart from the above-mentioned introductory chapter, this study comprised a further seven chapters, a bibliography, and five appendices. Chapter 2 was a literature review in fulfilment of Objective 1 in §1.4, and consisted of six sections. The first section, §2.1, emphasised the need for simulation as a research field and, accordingly, the need for better optimisation approaches to solve complex (combinatorial) simulation optimisation problems. Thereafter, in §2.2, the important notions pertaining to MOO was discussed, which included Pareto dominance, Pareto rank, the quality of MOO algorithms, the notion of archiving and finally MOO performance assessment. This was followed in §2.3 by a discussion of the criteria used to classify metaheuristics and listed the main classes of metaheuristics. Next, in §2.4, the motivations supporting the research on hyperheuristics were discussed and a classification of the hyperheuristic approaches were presented. §2.5, discussed the potential utility of an ANN as metamodel and contained a review of the pertinent literature related to ANNs. The review included the fundamentals of ANNs and the terminology found in the ANN literature, where FNNs were selected to form the

basis of discussion. Furthermore, prominent training algorithms and activation functions were discussed.

In Chapter 3, a pilot study was presented to determine the feasibility of an ANN as metamodel in fulfilment of Objective 2 in §1.4, and consisted of five sections. The first section, §3.1, briefly discussed the regression models used to compare to the ANN, which included MLR, PLR, SVR, DTR, RFR and XGBoost. This was followed in §3.2 by the performance measures used to assess the regression models as to facilitate the comparison. Next, in §3.3, Bayesian optimisation was discussed as the proposed method for finding the best hyperparameters for the FNN with one hidden layer. Thereafter, in §3.4, the training, validation and test results for the respective regression models were documented including the results obtained for the FNN employing the hyperparameters found during Bayesian optimisation.

Chapter 4 discussed the simulation models studied in fulfilment of Objective 3 in §1.4, and consisted of three sections. The first section, §4.1, introduced the statistical prerequisites required for stochastic output analysis. Thereafter, §4.2 presented the respective simulation models, discussing each model in terms of their inputs, outputs as well as the sensible upper bounds that were determined by method of statistical inference. In addition, the hypothesis tests conducted to determine whether or not 100 observations per solution were sufficient was documented for each simulation model and concluded in §4.3.

Chapter 5 documented both hyperheuristics and their constituent sub-algorithm in fulfilment of Objective 4 and 5 in §1.4, and consisted of seven sections. The first section, §5.1, introduced some of the main concepts of P- and S-metaheuristics, which included initial solutions, solution representation and stopping conditions. Following that, in §5.2–§5.4, was a discussion for the MOOCEM, the NSGA-II and the DBMOSA in the context of simulation optimisation and its integration in Tecnomatix. Thereafter, §5.5 and §5.6 discussed the hyperheuristics proposed in the study, also in the context of simulation optimisation and its integration in Tecnomatix.

Chapter 6 documented both hyperheuristics and their constituent sub-algorithms in fulfilment of Objective 6 in §1.4, and consisted of five sections. The first section, §6.1, introduced the non-parametric inferential statistical testing employed in this chapter, *i.e.* the Friedman and Iman-Davenport extension with the Nemenyi *post hoc* procedure. These statistical procedures facilitated the algorithmic parameter evaluation and the algorithmic performance comparisons conducted in Chapter 7. These statistical procedures were documented in §6.2–§6.4 for the MOOCEM, the NSGA-II and the DBMOSA, and concluded the common hyperparameter combination for each algorithm in §6.5.

Chapter 7 documented algorithmic performance comparisons in fulfilment of Objective 7 in §1.4, and consisted of four sections. The first section, §7.1, documented the statistical procedure that facilitated the comparison between the BOCEGAH, the MOOCEM and the NSGA-II. Thereafter, §7.2, documented the statistical procedure that facilitated the comparison between the BOSAH and the DBMOSA employing move operators A3.1, A3.3 and A3.4. Finally, §7.3, documented the statistical procedure that facilitated the comparison between the BOCEGAH and the BOCEGAH and the SOCEGAH and the SOCEGAH and the BOCEGAH and concluded in §7.4.

8.2 Appraisal of thesis contributions

The main contribution of this study is five-fold. This section contains a documentation and appraisal of these contributions.

Contribution 1 The establishment of a formal pilot study, providing an in-depth investigation

8.3. Suggestions for future work

into the workings of an ANN.

The novel pilot study proposed some avenues for further experimentation, some of which are listed in §3.5. The study provided valuable work pertaining to the complex interactions within an ANN and from the Python scripts included in Listing A, it is possible to further the study and ascertain more definitive results pertaining to the feasibility of an ANN as metamodel.

Contribution 2 The addition of five new optimisation tools to solve bi-objective simulation optimisation problems within Tecnomatix.

The addition of two population-based metaheuristics, the MOOCEM and the NSGA-II and the corresponding hyperheuristic tool BOCEGAH, which combined MOOCEM and the NSGA-II. The addition of a single-solution based metaheuristic, the DBMOSA employing four (proposed) move operators and the hyperheuristic tool BOSAH which combined three of the move operators, thereby enhancing the bi-objective simulation optimisation capabilities of Tecnomatix. This may be of particular interest to industries using Tecnomatix.

Contribution 3 The modification and furtherance of the AMALGAM hyperheuristic within the context of single-solution based search.

The application of AMALGAM towards single solutions-based search has been attempts in the literature, see [128], however a new heuristic selection mechanism was proposed and outlined in §5.6.

Contribution 4 An evaluation of the algorithmic parameters for the MOOCEM and the NSGA-II.

The novelty of the hyperparameter search space explored is the valuable insights that can be inferred from the hyperparameter combinations when considering the five discrete-event simulation optimisation problems with varying sizes, approximation front forms and complexities. Moreover, the results can be used as a guideline for choosing hyperparameters for the MOOCEM and the NSGA-II, refer to §6.2 and §6.3.

Contribution 5 The realisation of the inconsistency with the hyperarea calculation for discontinuous approximation fronts.

A linear adjustment was proposed for the hyperarea calculation, taking into account the discontinuity of the approximation front by considering the number of non-dominated solutions in the front *versus* that of the true front, as documented in §6.2.1.

8.3 Suggestions for future work

Suggestions for further investigation is documented in this section. In each case, the suggestion is stated formally and then elaborated upon.

Suggestion 1 Consider the evaluation of the optimisation approaches for different problem contexts.

The problem contexts considered in this study were all (coincidentally) *min-max* problems and therefore *min-min* and *max-max* problems should also be considered.

Suggestion 2 Design more move operators for single-solution based search for discrete vector representation.

The parameter evaluations of DBMOSA's move operators concluded that move operator 1 was statistically significantly better than the other proposed move operators, as documented in §6.4.

- 1. Creating perturbations of x from the Laplacian distribution. The Laplacian distribution is suggested due to its tails that decay relatively slow, ensuring that regions distant from the current solutions are explored [232].
- 2. Use feature importance techniques to determine which inputs have the most influence on the outputs and possibly use that to guide the search.
- **Suggestion 3** Conduct an empirical study for the algorithmic parameters for the DBMOSA algorithm.

The study should consider different annealing schedules (some of which are listed in §5.4.2), initial temperature (see §5.4.3 for some guidelines) and the effect of different initial solutions.

Suggestion 4 Design a complexity measure that encapsulates the complexity of a simulation problem.

Currently, complexity is measured in terms of the number of objectives, number of decision variables, the size of the search space and number of stochastic elements present in the simulation problem. Some have proposed complexity measures [207], however, it is too complex as it takes into account the complexity of the software used, for example the number of objects modeled.

In this study, BAP16 is considered the most complex simulation problem, however no consideration was made for the actual time required to optimise the model. In that case the OMP would have to be considered as it took the longest amount of time to evaluate due to the simulation of the waiting times of the trucks moving to the trains *etc.* The complexity term should also consider the computer used to run the simulations since it influences the time required to optimise a model, *i.e.* consider the central and graphical processing units. Moreover, the time it takes to optimise a model is also influenced by the number of models optimised simultaneously and should therefore also be considered.

Suggestion 5 Design an improved hyperarea calculation that takes into account the true number of non-dominated solutions found.

If the true Pareto set is available, then the hyperarea calculation should only consider the area underneath the non-dominated solutions that corresponds to the true non-dominated solutions found in the true Pareto set.

Suggestion 6 Consolidate the method used to calculate the hyperarea of an approximation front.

In the literature there are various methods proposed for calculating the hyperarea of an approximation set, namely by method of summing the rectangles under the front (for a min-max approximation front), or a more accurate method which includes summing the rectangles and trapezoids (as employed in this study) under the front. Also, some literature suggests that the area under each non-dominated solution with reference to the chosen reference point be summed, however, it does not translate to the dominated area.

Suggestion 7 Further the work done to determine the feasibility of an ANN.

As mentioned, the Python scripts included in Listing A can be used as a starting point to further the study and ascertain more definitive results pertaining to the feasibility of an ANN as metamodel. Also, some suggestions for further experimentation is listed in §3.5.

References

- [1] ABDEL-BASSET M, MANOGARAN G, RASHAD H & ZAIED ANH, 2018, A comprehensive review of quadratic assignment problem: variants, hybrids and applications, Journal of Ambient Intelligence and Humanized Computing, pp. 1–24.
- [2] AIKEN LS, WEST SG & PITTS SC, 2003, Multiple linear regression, Handbook of Psychology, pp. 481–507.
- [3] AMARAN S, SAHINIDIS NV, SHARDA B & BURY SJ, 2016, Simulation optimization: a review of algorithms and applications, Annals of Operations Research, 240(1), pp. 351– 380.
- [4] ANDRADÓTTIR S, 2006, An overview of simulation optimization via random search, Handbooks in Operations Research and Management Science, **13(1)**, pp. 617–631.
- [5] APRIL J, GLOVER FW, KELLY JP & LAGUNA M, 2003, Simulation-based optimization: Practical introduction to simulation optimization, Proceedings of the 35th Winter Simulation Conference: Driving Innovation, New Orleans (NO), pp. 71–78.
- [6] ASTA S, ÖZCAN E, PARKES AJ & ETANER-UYAR AŞ, 2013, Generalizing hyper-heuristics via apprenticeship learning, Proceedings of the European Conference on Evolutionary Computation in Combinatorial Optimization, Berlin, Heidelberg, pp. 169–178.
- [7] ATASHPAZ-GARGARI E & LUCAS C, 2007, Imperialist competitive algorithm: An algorithm for optimization inspired by imperialistic competition, Proceedings of the IEEE Congress on Evolutionary Computation, pp. 4661–4667.
- [8] AWAD M & KHANNA R, "Support vector regression", in: *Efficient learning machines*, Springer, 2015, pp. 67–80.
- [9] AYOB M & KENDALL G, 2003, A monte carlo hyper-heuristic to optimise component placement sequencing for multi head placement machine, Proceedings of the International Conference on Intelligent Technologies, pp. 132–141.
- [10] BÄCK T, 1996, Evolutionary Algorithms in Theory and Practice: Evolution Strategies, Evolutionary Programming, Genetic Algorithms, Oxford University Press, Oxford, UK.
- [11] BÄCK T, FOGEL DB & MICHALEWICZ Z, 1997, Handbook of Evolutionary Computation, New York (NY), Oxford University Press/IOP Publishing, 97(1).
- [12] BAI R, BLAZEWICZ J, BURKE EK, KENDALL G & MCCOLLUM B, 2012, A simulated annealing hyper-heuristic methodology for flexible decision support, A Quarterly Journal of Operations Research (4OR-Q J Oper Res), 10(1), pp. 43–66.
- [13] BAI R & KENDALL G, "An investigation of automated planograms using a simulated annealing based hyper-heuristic", in: *Metaheuristics: Progress as Real Problem Solvers*, Springer, 2005, pp. 87–108.
- [14] BAKER JE, 1985, Adaptive selection methods for genetic algorithms, Proceedings of the International Conference on Genetic Algorithms and their Applications, Pittsburg (PA).
- [15] BAMPORIKI T, BEKKER J & YOON M, 2019, Using a discrete-event, simulation optimisation optimiser to solve a stochastic multi-objective NP-hard problem, Proceedings of the International Conference on Competitive Manufacturing, Stellenbosch.
- [16] BANGSOW S, 2020, Tecnomatix Plant Simulation, 1st Edition, Springer.

[17]	BANZHAF W, NORDIN P, KELLER RE & FRANCONE FD, 1998, Genetic Programming:
	An Introduction, Morgan Kaufmann Publishers, San Francisco (SF).

- [18] BARTON RR & MECKESHEIMER M, 2006, *Metamodel-based Simulation Optimization*, Handbooks in Operations Research and Management Science, **13(1)**, pp. 535–574.
- [19] BASHYAM S & FU M, 1998, Optimization of (s, S) inventory systems with random lead times and a service level constraint, Management Science, 44(1), pp. 243–256.
- [20] BEKKER JF, Introduction to discrete-event simulation, Stellenbosch, July 2019.
- [21] BEKKER J, 2012, Applying the cross-entropy method in multi-objective optimisation of dynamic stochastic systems, PhD thesis, Stellenbosch University, Stellenbosch.
- [22] BEKKER J & ALDRICH C, 2011, The cross-entropy method in multi-objective optimisation: An assessment, European Journal of Operational Research, **211(1)**, pp. 112–121.
- [23] BENLIC U, EPITROPAKIS MG & BURKE EK, 2017, A hybrid breakout local search and reinforcement learning approach to the vertex separator problem, European Journal of Operational Research, 261(3), pp. 803–818.
- [24] BERGMANN B & HOMMEL G, "Improvements of general multiple test procedures for redundant systems of hypotheses", in: *Multiple Hypotheses Testing*, Springer, 1988, pp. 100– 115.
- [25] BERGSTRA J & BENGIO Y, 2012, Random search for hyper-parameter optimization., Journal of machine learning research, 13(2), February, pp. 281–305.
- [26] BERGSTRA J, YAMINS D & COX D, 2013, Making a Science of Model Search: Hyperparameter Optimization in Hundreds of Dimensions for Vision Architectures, Proceedings of the 30th International Conference on Machine Learning (ICML), Atlanta, Georgia (GA), pp. 115–123.
- BETTONVIL B, 1989, A formal description of discrete event dynamic systems including infinitesimal perturbation analysis, European Journal of Operational Research, 42(2), pp. 213–222.
- [28] BETTONVIL B, DEL CASTILLO E & KLEIJNEN JP, 2009, Statistical testing of optimality conditions in multiresponse simulation-based optimization, European Journal of Operational Research, 199(2), pp. 448–458.
- [29] BILGIN B, ÖZCAN E & KORKMAZ EE, 2007, An Experimental Study on Hyper-heuristics and Exam Timetabling, Proceedings of the International Conference on the Practice and Theory of Automated Timetabling, Berlin, Heidelberg, pp. 394–412.
- [30] BISHOP CM, 1995, Neural Networks for Pattern Recognition, Oxford University Press, Cambridge, UK.
- [31] BJORCK J, GOMES C, SELMAN B & WEINBERGER KQ, 2018, Understanding Batch Normalization, Proceedings of the.
- [32] BLICKLE T & THIELE L, 1996, A comparison of selection schemes used in evolutionary algorithms, Evolutionary Computation, 4(4), pp. 361–394.
- [33] BOUTE RN, GIJSBRECHTS J, VAN JAARSVELD W & VANVUCHELEN N, 2022, Deep reinforcement learning for inventory control: A roadmap, European Journal of Operational Research, 298(2), April, pp. 401–412.
- [34] BROWNLEE J, 2017, What is the difference between test and validation datasets, URL: https://machinelearningmastery.com/difference-test-validation-datasets/#: ~:text=That%20the%20%E2%80%9Cvalidation%20dataset%E2%80%9D%20is,it%20to% 20other%20final%20models.

- [35] BURKE EK & BYKOV Y, 2008, A Late Acceptance Strategy in Hill-climbing for Exam Timetabling Problems, Proceedings of the Conference on the Practice and Theory of Automated Timetabling (PATAT), Montreal, Canada.
- [36] BURKE EK, GENDREAU M, HYDE M, KENDALL G, OCHOA G, OZCAN E & QU R, 2013, Hyper-heuristics: A survey of the state of the art, Journal of the Operational Research Society, 64(12), pp. 1695–1724.
- [37] BURKE EK, HYDE M, KENDALL G, OCHOA G, OZCAN E & QU R, 2009, A survey of hyper-heuristics, (Unpublished) Technical Report, School of Computer Science and Information Technology, University of Nottingham, Nottingham, UK.
- [38] BURKE EK, HYDE M, KENDALL G, OCHOA G, ÖZCAN E & WOODWARD JR, 2010, A Classification of Hyper-heuristic Approaches, pp. 449–468 in GENDREAU M & POTVIN JY (EDS), Handbook of Metaheuristics, Springer US, Boston (MA).
- [39] BURKE EK, HYDE MR, KENDALL G, OCHOA G, ÖZCAN E & WOODWARD JR, 2019, A Classification of Hyper-heuristic Approaches: Revisited, pp. 453–477 in GENDREAU M & POTVIN JY (EDS), Handbook of Metaheuristics, Springer, Cham.
- [40] BURKE EK, KENDALL G & SOUBEIGA E, 2003, A tabu-search hyperheuristic for timetabling and rostering, Journal of Heuristics, **9(6)**, pp. 451–470.
- [41] CAN B & HEAVEY C, 2012, A comparison of genetic programming and artificial neural networks in metamodeling of discrete-event simulation models, Computers & Operations Research, 39(2), February, pp. 424–436.
- [42] CHAKHLEVITCH K & COWLING P, 2008, Hyperheuristics: Recent Developments, pp. 3–29 in COTTA C, SEVAUX M & SÖRENSEN K (EDS), Adaptive and Multilevel Metaheuristics, Springer, Berlin, Heidelberg.
- [43] CHEN T & GUESTRIN C, 2016, Xgboost: A scalable tree boosting system, Proceedings of the 22th International Conference on Knowledge Discovery and Data Mining, New York (NY), pp. 785–794.
- [44] CLARKE SM, GRIEBSCH JH & SIMPSON TW, 2005, Analysis of support vector regression for approximation of complex engineering analyses, Journal of Mechanical Design, 127(6), pp. 1077–1087.
- [45] CLEVERT DA, UNTERTHINER T & HOCHREITER S, 2015, Fast and accurate deep network learning by exponential linear units (elus).
- [46] COELLO COELLO CA, 2009, Evolutionary multi-objective optimization: Some current research trends and topics that remain to be explored, Frontiers of Computer Science in China, 3(1), pp. 18–30.
- [47] COELLO COELLO CA, LAMONT GB & VAN VELDHUIZEN DA, 2007, Evolutionary Algorithms for Solving Multi-Objective Problems, 2nd Edition, Springer, New York (NY).
- [48] CONOVER WJ, 1999, Practical nonparametric statistics, 3rd Edition, John Wiley & Sons.
- [49] CORTES C & VAPNIK V, 1995, Support-vector networks, Machine learning, 20(3), pp. 273– 297.
- [50] COWLING P, KENDALL G & SOUBEIGA E, 2001, A Hyperheuristic Approach to Scheduling a Sales Summit, Proceedings of the Practice and Theory of Automated Timetabling III, Berlin, Heidelberg, pp. 176–190.
- [51] COWLING P, KENDALL G & SOUBEIGA E, 2002, Hyperheuristics: A Tool for Rapid Prototyping in Scheduling and Optimisation, Proceedings of the Workshops on Applications of Evolutionary Computation, Berlin, Heidelberg, pp. 1–10.

- [52] CRAWFORD B, SOTO R, ASTORGA G, GARCIÉA J, CASTRO C & PAREDES F, 2017, *Putting continuous metaheuristics to work in binary search spaces*, Machine Learning Applied to Metaheuristics and Combinatorial Problems, **2017(2)**, pp. 1–19.
- [53] CRUZ F, VAN WOENSEL T & SMITH JM, 2010, Buffer and throughput trade-offs in M/G/1/K queueing networks: A bi-criteria approach, International Journal of Production Economics, 125(2), June, pp. 224–234.
- [54] CRUZ FR, DUARTE AR & VAN WOENSEL T, 2008, Buffer allocation in general singleserver queueing networks, Computers & Operations Research, **35(11)**, pp. 3581–3598.
- [55] CUTLER A, CUTLER DR & STEVENS JR, 2012, Random forests, pp. 157–175 in ZHANG C & MA Y (EDS), Ensemble Machine Learning: Methods and Applications, Springer US, Boston (MA).
- [56] CYBENKO G, 1989, Approximation by superpositions of a sigmoidal function, Mathematics of Control, Signals and Systems, 2(4), pp. 303–314.
- [57] DASGUPTA D & MICHALEWICZ Z, 1997, Evolutionary algorithms—an overview, pp. 3–28 in DASGUPTA D & MICHALEWICZ Z (EDS), Springer, Berlin, Heidelberg.
- [58] DE BOER PT, KROESE DP & RUBINSTEIN RY, 2004, A fast cross-entropy method for estimating buffer overflows in queueing networks, Management Science, 50(7), pp. 883– 895.
- [59] DEB K, 2011, Multi-objective optimisation using evolutionary algorithms: An introduction, pp. 3–34 in WANG L, NG AHC & DEB K (EDS), Multi-objective Evolutionary Optimisation for Product Design and Manufacturing, Springer, London (UK).
- [60] DEB K, 2001, Multi-Objective Optimization Using Evolutionary, John Wiley & Sons.
- [61] DEB K, 2010, Recent Developments in Evolutionary Multi-Objective Optimization, pp. 339–368 in Ehrgott M, Figueira JR & Greco S (EDS), Springer, Boston (MA).
- [62] DEB K & AGRAWAL RB, 1995, Simulated binary crossover for continuous search space, Complex Systems, 9(2), pp. 115–148.
- [63] DEB K & AGRAWAL S, 1999, A Niched-Penalty Approach for Constraint Handling in Genetic Algorithms, Proceedings of the Artificial Neural Nets and Genetic Algorithms, Vienna, pp. 235–243.
- [64] DEB K & BEYER HG, 2001, Self-Adaptive Genetic Algorithms with Simulated Binary Crossover, Evolutionary Computation, 9(2), June, pp. 197–221.
- [65] DEB K & DEB D, 2014, Analysing mutation schemes for real-parameter genetic algorithms, International Journal of Artificial Intelligence and Soft Computing, 4(1), February, pp. 1–28.
- [66] DEB K & KUMAR A, 1995, Real-coded genetic algorithms with simulated binary crossover: Studies on multimodal and multiobjective problems, Complex Systems, **9(6)**, pp. 431–454.
- [67] DEB K, PRATAP A, AGARWAL S & MEYARIVAN T, 2002, A fast and elitist multiobjective genetic algorithm: NSGA-II, IEEE Transactions on Evolutionary Computation, 6(2), April, pp. 182–197.
- [68] DEB K & SUNDAR J, 2006, Reference Point Based Multi-Objective Optimization Using Evolutionary Algorithms, Proceedings of the 8th Annual Conference on Genetic and Evolutionary Computation, Seattle, Washington (WA), pp. 635–642.
- [69] DEEP K & THAKUR M, 2007, A new crossover operator for real coded genetic algorithms, Applied Mathematics and Computation, **188(1)**, pp. 895–911.

168

- [70] DEEP K & THAKUR M, 2007, A new mutation operator for real coded genetic algorithms, Applied Mathematics and Computation, **193(1)**, October, pp. 211–230.
- [71] DERRAC J, GARCIÉA S, MOLINA D & HERRERA F, 2011, A practical tutorial on the use of nonparametric statistical tests as a methodology for comparing evolutionary and swarm intelligence algorithms, Swarm and Evolutionary Computation, 1(1), March, pp. 3–18.
- [72] DI PIERRO F, KHU ST & SAVIC DA, 2007, An investigation on preference order ranking scheme for multiobjective evolutionary optimization, IEEE Transactions on Evolutionary Computation, 11(1), pp. 17–45.
- [73] DORIGO M & BLUM C, 2005, Ant colony optimization theory: A survey, Theoretical Computer Science, **344(3)**, November, pp. 243–278.
- [74] DOWSLAND KA, SOUBEIGA E & BURKE E, 2007, A simulated annealing based hyperheuristic for determining shipper sizes for storage and transportation, European Journal of Operational Research, 179(3), June, pp. 759–774.
- [75] DOZAT T, 2016, Incorporating nesterov momentum into adam, URL: https://openreview. net/forum?id=OMOjvwB8jIp57ZJjtNEZ.
- [76] DRAKE JH, KHEIRI A, ÖZCAN E & BURKE EK, 2020, *Recent advances in selection hyperheuristics*, European Journal of Operational Research, **285(2)**, September, pp. 405–428.
- [77] DRÉO J, PÉTROWSKI A, SIARRY P & TAILLARD E, 2006, Metaheuristics for Hard Optimization, 1st Edition, Springer, Berlin, Heidelberg.
- [78] DUCHI J, HAZAN E & SINGER Y, 2011, Adaptive subgradient methods for online learning and stochastic optimization, Journal of Machine Learning Research, 12(7), July, pp. 2121–2159.
- [79] DUECK G, 1993, New Optimization Heuristics: The Great Deluge Algorithm and the Record-to-Record Travel, Journal of Computational Physics, 104(1), January, pp. 86– 92.
- [80] EDMUND K.BURKE GK (ED), 2010, Search Methodologies, 1st Edition, Springer, New York (NY).
- [81] EREMENKO K & DE PONTEVES H, 2019, Machine Learning AZ: Hands-On Python & R In Data Science.
- [82] FANG H, WANG Q, TU YC & HORSTEMEYER MF, 2008, An efficient non-dominated sorting method for evolutionary algorithms, Evolutionary Computation, 16(3), September, pp. 355–384.
- [83] FAUSETT LV, 1993, Fundamentals of Neural Networks: Architectures, Algorithms and Applications, 1st Edition, Pearson.
- [84] FAUSETT LV, 1994, Neural Networks Based on Competition, Proceedings of the Fundamentals of Neural Networks: Architectures, Algorithms and Applications. Englewood Cliffs: Prentice-Hall, pp. 156–217.
- [85] FLEISCHER M, 2003, The Measure of Pareto Optima Applications to Multi-objective Metaheuristics, Proceedings of the International Conference on Evolutionary Multi-criterion Optimization, Berlin, Heidelberg, pp. 519–533.
- [86] FONSECA DJ & NAVARESSE D, 2002, Artificial neural networks for job shop simulation, Advanced Engineering Informatics, **16(4)**, October, pp. 241–246.
- [87] FONSECA DJ, NAVARESSE DO & MOYNIHAN GP, 2003, Simulation metamodeling through artificial neural networks, Engineering Applications of Artificial Intelligence, 16(3), April, pp. 177–183.

170	REFERENCES
[88]	FRIEDMAN JH, 1991, Multivariate adaptive regression splines, The Annals of Statistics, 19(1) , March, pp. 1–67.
[89]	FRIEDMAN M, 1937, The use of ranks to avoid the assumption of normality implicit in the analysis of variance, Journal of the American Statistical Association, 32(200) , May, pp. 675–701.
[90]	Fu MC, 1994, <i>Optimization via simulation: A review</i> , Annals of Operations Research, 53(1) , December, pp. 199–247.
[91]	GEEM ZW, KIM JH & LOGANATHAN GV, 2001, A New Heuristic Optimization Algo- rithm: Harmony Search, Simulation: Transactions of The Society for Modeling and Sim- ulation International, 76(2) , February, pp. 60–68.
[92]	GEMAN S & GEMAN D, 1984, Stochastic Relaxation, Gibbs Distributions, and the Bayesian Restoration of Images, IEEE Transactions on Pattern Analysis and Machine Intelligence (PAMI), 6(6), November, pp. 721–741.
[93]	EL-GHAZALI T, 2009, Metaheuristics from Design to Implementation, John Wiley & Sons.
[94]	GLASSERMAN P & HO YC, 1991, Gradient Estimation via Perturbation Analysis, 1 st Edition, Springer Science & Business Media, New York (NY).
[95]	GLOROT X & BENGIO Y, 2010, Understanding the Difficulty of Training Deep Feed- forward Neural Networks, Proceedings of the Thirteenth International Conference on Artificial Intelligence and Statistics (PMLR), Sandinia, Italy, pp. 249–256.
[96]	GLOROT X, BORDES A & BENGIO Y, 2011, <i>Deep Sparse Rectifier Neural Networks</i> , Proceedings of the Fourteenth International Conference on Artificial Intelligence and Statistics, Fort Lauderdale (FL), pp. 315–323.
[97]	GLOVER F, 1986, Future paths for integer programming and links to artificial intelligence, Computers & Operations Research, 13(5) , pp. 533–549.
[98]	GLOVER F, 1977, <i>Heuristics for integer programming using surrogate constraints</i> , Decision Sciences, 8(1) , January, pp. 156–166.
[99]	GLOVER F, KELLY JP & LAGUNA M, 1996, New Advances and Applications of Combining Simulation and Optimization, Proceedings of the 28 th Conference on Winter Simulation, California (CA), pp. 144–152.
[100]	GLOVER F & LAGUNA M, 1997, <i>Tabu Search Principles</i> , 1 st Edition, Springer, Boston (MA).
[101]	GLOVER F, LAGUNA M & MARTIÉ R, 2015, <i>Scatter Search</i> , pp. 519–537 in GHOSH A & TSUTSUI S (EDS), <i>Advances in Evolutionary Computing</i> , Springer, Berlin, Heidelberg.
[102]	GLYNN PW, 1987, <i>Likelilood ratio gradient estimation: an overview</i> , Proceedings of the 19th Conference on Winter Simulation, Georgia (GA), pp. 366–375.
[103]	GOLDBERG DE & DEB K, 1991, A Comparative Analysis of Selection Schemes Used in Genetic Algorithms, Foundations of Genetic Algorithms, 1, pp. 69–93.
[104]	GOLDBERG DE, KORB B & DEB K, 1989, Messy genetic algorithms: Motivation, analysis, and first results, Complex Systems, 3(5) , pp. 493–530.
[105]	GOLDSMAN D & NELSON BL, 1994, Ranking, selection and multiple comparisons in com- puter simulation, Proceedings of the Winter Simulation Conference, Lake Buena Vista (FL), pp. 192–199.

- [106] GOODFELLOW I, BENGIO Y, COURVILLE A & BENGIO Y, 2016, *Deep learning*, MIT press, Cambridge (UK).
- [107] GREFENSTETTE JJ, 1992, Genetic algorithms for changing environments.
- [108] GUIZZO G, VERGILIO SR, POZO AT & FRITSCHE GM, 2017, A Multi-objective and Evolutionary Hyper-heuristic Applied to the Integration and Test Order Problem, Applied Soft Computing, 56, March, pp. 331–344.
- [109] HACHICHA W, 2011, A simulation metamodelling based neural networks for lot-sizing problem in MTO sector, International Journal of Simulation Modelling, 10(4), pp. 191– 203.
- [110] HAJEK B & SASAKI G, 1989, Simulated annealing—to cool or not, Systems & Control Letters, 12(5), June, pp. 443–447.
- [111] HANSEN P & MLADENOVIĆ N, 2001, Variable neighborhood search: Principles and applications, European Journal of Operational Research, **130(3)**, pp. 449–467.
- [112] HE K, ZHANG X, REN S & SUN J, 2015, Delving deep into rectifiers: Surpassing humanlevel performance on imagenet classification, Proceedings of the IEEE International Conference on Computer Vision (ICCV), pp. 1026–1034.
- [113] HEIBERGER RM & NEUWIRTH E, 1998, Polynomial regression, pp. 235–268 in RAWLINGS JO, PANTULA SG & DICKEY DA (EDS), Applied Regression Analysis: A Research Tool, Springer, New York (NY).
- [114] HERTZ A & DE WERRA D, 1990, The tabu search metaheuristic: How we used it, Annals of Mathematics and Artificial Intelligence, 1(1), September, pp. 111–121.
- [115] HILL NJ & PARKS GT, 2015, Pressurized water reactor in-core nuclear fuel management by tabu search, Annals of Nuclear Energy, 75, January, pp. 64–71.
- [116] HINTON G, SRIVASTAVA N & SWERSKY K, 2012, Neural Networks for Machine Learning, Lecture 6a: Overview of mini-batch gradient descent.
- [117] HO YC & CAO X.-R, 1991, Introduction to Discrete Event Dynamic Systems, pp. 1–13 in, Perturbation Analysis of Discrete Event Dynamic Systems, Springer, Boston (MA).
- [118] HOCHBERG Y & TAMHANE AC, 1987, Multiple Comparison Procedures, John Wiley & Sons.
- [119] HOLLAND JH, 1992, Adaptation in Natural and Artificial Systems: An Introductory Analysis with Applications to Biology, Control, and Artificial Intelligence.
- [120] HOLLAND JH, 1992, Adaptation in Natural and Artificial Systems: An Introductory Analysis with Applications to Biology, Control, and Artificial Intelligence, MIT press.
- [121] HOLLANDER M, WOLFE DA & CHICKEN E, 2015, Nonparametric Statistical Methods, 3rd Edition, John Wiley & Sons.
- [122] HOLM S, 1979, A Simple Sequentially Rejective Multiple Test Procedure, Scandinavian Journal of Statistics, 6(2), pp. 65–70.
- [123] HOOS HH & STÜTZLE T, 2004, Stochastic Local Search: Foundations and Applications, Elsevier, San Francisco (CA).
- [124] HORNIK K, 1991, Approximation capabilities of multilayer feedforward networks, Neural Networks, 4(2), pp. 251–257.
- [125] HORNIK K, STINCHCOMBE M & WHITE H, 1989, Multilayer feedforward networks are universal approximators, Neural Networks, 2(5), pp. 359–366.

172	REFERENCES

- [126] HSIAO PK, JIANG S.-J, SAHAYAM A & HSIAO TC, 2012, Determination of trace elements in silicon powders by inductively coupled plasma quadrupole mass spectrometry with a membrane desolvation sample introduction system, Atomic Spectroscopy, 33(1), January, pp. 1–8.
- [127] HUANG M, 2004, An efficient general cooling schedule for simulated annealing, Proceedings of the Computational Science and Its Applications, Berlin, Heidelberg, pp. 396–404.
- [128] HÜSSELMANN G, 2022, Bus route design and frequency setting for public transit systems, PhD thesis, Stellenbosch University, Stellenbosch.
- [129] IMAN RL & DAVENPORT JM, 1980, Approximations of the critical region of the friedman statistic, Communications in Statistics-Theory and Methods, 9(6), pp. 571–595.
- [130] JAMES G, WITTEN D, HASTIE T & TIBSHIRANI R, 2013, An Introduction to Statistical Learning, 1st Edition, Springer, New York (NY).
- [131] JOHNSON DS, PAPADIMITRIOU CH & YANNAKAKIS M, 1988, *How easy is local search?*, Journal of Computer and System Sciences, **37(1)**, September, pp. 79–100.
- [132] KAELBLING LP, LITTMAN ML & MOORE AW, 1996, *Reinforcement learning: A survey*, Journal of Artificial Intelligence Research, 4, May, pp. 237–285.
- [133] KARABOGA D, 2005, An idea based on honey bee swarm for numerical optimization, (Unpublished) Technical Report, Erciyes University.
- [134] KARIMI-MAMAGHAN M, MOHAMMADI M, MEYER P, KARIMI-MAMAGHAN AM & TALBI E.-G, 2022, Machine Learning at the service of Meta-heuristics for solving Combinatorial Optimization Problems: A state-of-the-art, European Journal of Operational Research, 296(2), January, pp. 393–422.
- [135] KARPATHY A, 2016, CS231n: Convolutional Neural Networks for Visual Recognition, Stanford University.
- [136] KELTON WD, 1997, Statistical analysis of simulation output, Proceedings of the 29th Conference on Winter Simulation, Cincinnati, Ohio (OH), pp. 23–30.
- [137] KENDALL G & HUSSIN NM, 2005, An investigation of a tabu-search-based hyper-heuristic for examination timetabling, Proceedings of the Multidisciplinary Scheduling: Theory and Applications, Boston (MA), pp. 309–328.
- [138] KENDALL G & MOHAMAD M, 2004, Channel assignment in cellular communication using a great deluge hyper-heuristic, Proceedings of the 12th IEEE International Conference on Networks (ICON), pp. 769–773.
- [139] KENNEDY J, 2006, Swarm intelligence, pp. 187–219 in ZOMAYA AY (ED), Handbook of Nature-Inspired and Innovative Computing, Springer.
- [140] KHAN W, SALHI A, ASIF M, ADEEB R & SULAIMAN M, 2015, Enhanced version of multialgorithm genetically adaptive for multiobjective optimization, International Journal of Advanced Computer Science and Applications, 6(12), pp. 279–287.
- [141] KINGMA DP & BA J, 2015, Adam: A method for stochastic optimization, Proceedings of the 3th International Conference for Learning Representations, San Diego (CA).
- [142] KIRAZ B, ETANER-UYAR AŞ & ÖZCAN E, 2013, Selection hyper-heuristics in dynamic environments, Journal of the Operational Research Society, **64(12)**, pp. 1753–1769.
- [143] KIRKPATRICK S, GELATT CD & VECCHI MP, 1983, *Optimization by simulated annealing*, American Association for the Advancement of Science, **220(4598)**, pp. 671–680.

- [144] KLEIJNEN JP, 2009, *Kriging metamodeling in simulation: A review*, European Journal of Operational Research, **192(3)**, February, pp. 707–716.
- [145] KLEIJNEN JP & VAN BEERS WC, 2004, Application-driven sequential designs for simulation experiments: Kriging metamodelling, Journal of the Operational Research Society, 55(8), pp. 876–883.
- [146] KLEIJNEN JP & VAN BEERS WC, 2005, Robustness of Kriging when interpolating in random simulation with heterogeneous variances: Some experiments, European Journal of Operational Research, 165(3), September, pp. 826–834.
- [147] KNOWLES JD, THIELE L & ZITZLER E, 2005, A tutorial on the performance assessment of stochastic multiobjective optimizers, 3rd International Conference on Evolutionary Multi-Criterion Optimization (EMO), 214, February.
- [148] KOÇ Ç, BEKTAŞ T, JABALI O & LAPORTE G, 2016, *Thirty years of heterogeneous vehicle routing*, European Journal of Operational Research, **249(1)**, February, pp. 1–21.
- [149] KONAK A, COIT DW & SMITH AE, 2006, Multi-objective optimization using genetic algorithms: A tutorial, Reliability Engineering & System Safety, 91(9), September, pp. 992– 1007.
- [150] KORTE BH, VYGEN J, KORTE B & VYGEN J, 2012, Combinatorial Optimization, Springer, Berlin, Heidelberg.
- [151] KOUTSOUKAS A, MONAGHAN KJ, LI X & HUAN J, 2017, Deep-learning: Investigating deep neural networks hyper-parameters and comparison of performance to shallow methods for modeling bioactivity data, Journal of Cheminformatics, 9(42), pp. 1–13.
- [152] KROGH A & HERTZ JA, 1991, A simple weight decay can improve generalization, Proceedings of the Advances in Neural Information Processing Systems (NIPS), pp. 950– 957.
- [153] KUBAT M, 1998, Neural networks: A comprehensive foundation by Simon Haykin, Macmillan, 1994, ISBN 0-02-352781-7., The Knowledge Engineering Review, 13(4), pp. 409–412.
- [154] KUHN M & JOHNSON K, 2013, Applied Predictive Modeling, 1st Edition, Springer, New York (NY).
- [155] KUO Y, YANG T, PETERS BA & CHANG I, 2007, Simulation metamodel development using uniform design and neural networks for automated material handling systems in semiconductor wafer fabrication, Simulation Modelling Practice and Theory, 15(8), September, pp. 1002–1015.
- [156] LAGUNA M, *OptQuest*, Opttek Systems, Inc., 2011.
- [157] LAW AM, KELTON WD & KELTON WD, 2007, Simulation Modeling and Analysis, 5th Edition, Mc Graw Hill Education, New York (NY).
- [158] LE QV, 2015, A tutorial on deep learning part 1: Nonlinear classifiers and the backpropagation algorithm, Google Inc., Mountain View, CA, December.
- [159] LECUN YA, BOTTOU L, ORR GB & MÜLLER K.-R, 2012, Efficient backprop, pp. 9–48 in MONTAVON G, ORR GB & MÜLLER K.-R (EDS), Neural Networks: Tricks of the Trade, Springer, Berlin, Heidelberg.
- [160] LEE LH, CHEW EP, TENG S & GOLDSMAN D, 2010, Finding the non-dominated Pareto set for multi-objective simulation models, IIE Transactions, **42(9)**, pp. 656–674.

174 REFERENCES

- [161] LEHRBAUM A & MUSLIU N, 2012, A New Hyperheuristic Algorithm for Cross-domain Search Problems, Proceedings of the International Conference on Learning and Intelligent Optimization (LION), Berlin, Heidelberg, pp. 437–442.
- [162] LI M & YAO X, 2020, Quality evaluation of solution sets in multiobjective optimisation: A survey, ACM Computing Surveys (CSUR), 52(26), March, pp. 1–38.
- [163] LI W, ÖZCAN E & JOHN R, 2017, A learning automata-based multiobjective hyperheuristic, IEEE Transactions on Evolutionary Computation, 23(1), pp. 59–73.
- [164] LI W, OZCAN E & JOHN R, 2017, Multi-objective evolutionary algorithms and hyperheuristics for wind farm layout optimisation, Renewable Energy, 105, May, pp. 473– 482.
- [165] LI YF, NG SH, XIE M & GOH T, 2010, A systematic comparison of metamodeling techniques for simulation optimization in decision support systems, Applied Soft Computing, 10(4), September, pp. 1257–1273.
- [166] LIAGKOURAS K & METAXIOTIS K, 2013, An elitist polynomial mutation operator for improved performance of MOEAs in computer networks, Proceedings of the 22nd International Conference on Computer Communication and Networks (ICCCN), Nassau, Bahamas, pp. 1–5.
- [167] LIN C, YANG JI, LIN KJ & WANG ZD, 1998, Pressurized water reactor loading pattern design using the simple tabu search, Nuclear Science and Engineering, 129(1), pp. 61–71.
- [168] LIU Y, WANG Y & ZHANG J, 2012, New machine learning algorithm: Random forest, Proceedings of the International Conference on Information Computing and Applications (ICICA), Berlin, Heidelberg, pp. 246–252.
- [169] LOTTER DP, 2017, Design of a weapon assignment subsystem within a ground-based air defence environment, PhD thesis, Stellenbosch University, Stellenbosch.
- [170] LOURENÇO HR, MARTIN OC & STÜTZLE T, "Iterated Local Search", in: Handbook of Metaheuristics, ed. by Fred Glover and Gary A. Kochenberger, Boston (MA): Springer, 2003, pp. 320–353.
- [171] LOZANO M, HERRERA F & CANO JR, 2008, Replacement strategies to preserve useful diversity in steady-state genetic algorithms, Information Sciences, 178(23), pp. 4421– 4433.
- [172] MAAS AL, HANNUN AY & NG AY, 2013, Rectifier nonlinearities improve neural network acoustic models, Proceedings of the, (CA).
- [173] MAASHI M, ÖZCAN E & KENDALL G, 2014, A multi-objective hyper-heuristic based on choice function, Expert Systems with Applications, **41(9)**, July, pp. 4475–4493.
- [174] MACGREGOR SMITH J & CRUZ FR, 2005, The buffer allocation problem for general finite buffer queueing networks, IIE Transactions, **37(4)**, pp. 343–365.
- [175] MAHFOUD SW, 1995, Niching methods for genetic algorithms, PhD thesis, University of Illinois, Urbana, Illinois (IL).
- [176] MATTERA D, PALMIERI F & HAYKIN S, 1999, An explicit algorithm for training support vector machines, IEEE Signal Processing Letters, **6(9)**, September, pp. 243–245.
- [177] MAZUR M, A step by step backpropagation example, 2015.
- [178] MEGHABGHAB G & KANDEL A, 2004, Stochastic simulations of web search engines: RBF versus second-order regression models, Information Sciences, 159(1-2), January, pp. 1– 28.

- [179] MEHROTRA K, MOHAN CK & RANKA S, 1996, *Elements of Artificial Neural Networks*, MIT press.
- [180] METROPOLIS N, ROSENBLUTH AW, ROSENBLUTH MN, TELLER AH & TELLER E, 1953, Equation of state calculations by fast computing machines, The Journal of Chemical Physics, 21(6), pp. 1087–1092.
- [181] MIETTINEN K, 1998, Nonlinear Multiobjective Optimization, 1st Edition, Springer Science & Business Media, New York (NY).
- [182] MISIR M, VANCROONENBURG W, VERBEECK K & BERGHE GV, 2011, A selection hyperheuristic for scheduling deliveries of ready-mixed concrete, Proceedings of the 9th Metaheuristics International Conference (MIC 2011), Udine, Italy, pp. 289–298.
- [183] MISIR M, VERBEECK K, DE CAUSMAECKER P & BERGHE GV, 2010, Hyper-heuristics with a dynamic heuristic set for the home care scheduling problem, Proceedings of the Congress on Evolutionary Computation, pp. 1–8.
- [184] MITCHELL TM, 1997, Machine Learning, McGraw-Hill, New York (NY).
- [185] MLADENOVIĆ N & HANSEN P, 1997, Variable neighborhood search, Computers & Operations Research, 24(11), November, pp. 1097–1100.
- [186] MORNATI F, 2013, Pareto Optimality in the work of Pareto, European Journal of Social Sciences, (51-2), pp. 65–82.
- [187] MOSCATO P, 2000, On evolution, search, optimization, genetic algorithms and martial arts: Towards memetic algorithms, Caltech Concurrent Computation Program, October.
- [188] MUELLER JP & MASSARON L, 2019, Deep Learning, For Dummies.
- [189] NAIR V & HINTON GE, 2010, Rectified linear units improve restricted boltzmann machines, Proceedings of the International Conference on Machine Learning (ICML), pp. 807– 814.
- [190] NAKAYAMA H, ARAKAWA M & SASAKI R, 2002, Simulation-based optimization using computational intelligence, Optimization and Engineering, **3(2)**, June, pp. 201–214.
- [191] NEGAHBAN A & SMITH JS, 2014, Simulation for manufacturing system design and operation: Literature review and analysis, Journal of Manufacturing Systems, 33(2), pp. 241– 261.
- [192] NEL GS, 2021, A hyperheuristic approach towards the training of artificial neural networks, PhD thesis, Stellenbosch University, Stellenbosch.
- [193] NESTEROV Y, 1983, A method for unconstrained convex minimization problem with the rate of convergence, Mathematics.
- [194] NIELSEN M, 2015, *How the backpropagation algorithm works*, Proceedings of the Neural Networks and Deep Learning.
- [195] NIELSEN MA, 2019, Neural Networks and Deep Learning, San Francisco (CA).
- [196] OSMAN IH & KELLY JP, 1997, Meta-heuristics theory and applications, Journal of the Operational Research Society, 48(6), pp. 657–657.
- [197] OSMAN IH & KELLY JP, 1996, Meta-heuristics: An overview, pp. 1–21 in OSMAN IH & KELLY JP (EDS), Springer, Boston (MA).
- [198] OSMAN IH & LAPORTE G, 1996, *Metaheuristics: A bibliography*, Annals of Operations Research, **63**, pp. 511–623.
- [199] ÖZCAN E, BILGIN B & KORKMAZ EE, 2008, A comprehensive analysis of hyper-heuristics, Intelligent Data Analysis, **12(1)**, pp. 3–23.

176	REFERENCES
[200]	ÖZCAN E, BYKOV Y, BIRBEN M & BURKE EK, 2009, <i>Examination timetabling using late acceptance hyper-heuristics</i> , Proceedings of the Congress on Evolutionary Computation, pp. 997–1004.
[201]	ÖZCAN E, MISIR M, OCHOA G & BURKE EK, 2012, A reinforcement learning: Great- deluge hyper-heuristic for examination timetabling, pp. 34–55 in , Modeling, Analysis, and Applications in Metaheuristic Computing: Advancements and Trends, IGI Global.
[202]	PAPADOPOULOS HT & HEAVEY C, 1996, Queueing theory in manufacturing systems analysis and design: A classification of models for production and transfer lines, European Journal of Operational Research, 92(1) , July, pp. 1–27.
[203]	PARKINSON AR, BALLING R & HEDENGREN JD, 2013, Optimization Methods for Engi- neering Design, Brigham Young University, 5(11).
[204]	PEREIRA I, MADUREIRA A, DE MOURA OLIVEIRA PB & ABRAHAM A, "Tuning meta- heuristics using multi-agent learning in a scheduling system", in: <i>Transactions on Compu- tational Science XXI</i> , ed. by Marina L. Gavrilova, C.J.Kenneth Tan, and Ajith Abraham, vol. 8160, Lecture Notes in Computer Science book series, Berlin, Heidelberg: Springer, 2013, pp. 190–210.
[205]	PODGORELEC V & ZORMAN M, 2016, Decision tree learning, pp. 1–28 in , Encyclopedia of Complexity and Systems Science, Springer, Berlin, Heidelberg.
[206]	POHLERT T, The pairwise multiple comparison of mean ranks package (PMCMR), 2019, R package, 2014, p. 9.
[207]	POPOVICS G & MONOSTORI L, 2016, An approach to Determine Simulation Model Com- plexity, The Sixth International Conference on Changeable, Agile, Reconfigurable and Virtual Production (CARV), (257–261).
[208]	QIAN N, 1999, On the momentum term in gradient descent learning algorithms, Neural Networks, 12(1) , January, pp. 145–151.
[209]	RAZALI NM & GERAGHTY J, 2011, Genetic algorithm performance with different selection strategies in solving TSP, Proceedings of the 1 th World Congress on Engineering, London (UK), pp. 1–6.
[210]	RIQUELME N, VON LÜCKEN C & BARAN B, 2015, <i>Performance metricsi n multi-objective optimization</i> , Proceedings of the Latin American Computing Conference (CLEI), Arequipa, Peru, pp. 1–11.
[211]	ROJAS R, 2013, <i>Neural networks: A systematic introduction</i> , 1 st Edition, Springer Science & Business Media, Berlin, Heidelberg.
[212]	ROKACH L & MAIMON O, 2005, <i>Decision trees</i> , pp. 165–192 in MAIMON O & ROKACH L (EDS), <i>Data Mining and Knowledge Discovery Handbook</i> , Springer, Boston (MA).
[213]	ROSS P, 2005, Hyper-heuristics, pp. 529–556 in BURKE EK & KENDALL G (EDS), Search Methodologies: Introductory Tutorials in Optimization and Decision Support Techniques, Springer.
[214]	ROSTAMI S, 2019, <i>Hypervolume Indicator</i> , Accessed: 2022-09-27, URL: https://datacrayon.com/posts/search%20and%20optimisation/practical%20evolutionary%20algorithms/hypervolume%20indicator.
[215]	RUBINSTEIN R, 1990, How to optimize discrete-event systems from a single sample path by the score function method, Annals of Operations Research, 27(1) , pp. 175–212.
[216]	RUBINSTEIN RY & SHAPIRO A, 1993, Discrete Event Systems: Sensitivity Analysis and Stochastic Optimization by the Score Function Method, 1 st Edition, Wiley.

- [217] RUBINSTEIN RY & KROESE DP, 2004, The Cross-Entropy Method: A Unified Approach to Combinatorial Optimization, Monte-Carlo Simulation, and Machine Learning, Springer.
- [218] RUSSELL S & NORVIG P, 2009, Artificial Intelligence: A Modern Approach, 3rd Edition, Pearson.
- [219] SABUNCUOGLU I & TOUHAMI S, 2002, Simulation metamodelling with neural networks: An experimental investigation, International Journal of Production Research, 40(11), pp. 2483–2505.
- [220] SACKS J, WELCH WJ, MITCHELL TJ & WYNN HP, 1989, Design and analysis of computer experiments, Statistical Science, 4(4), pp. 409–423.
- [221] SALMAN A, AHMAD I & AL-MADANI S, 2002, Particle swarm optimization for task assignment problem, Microprocessors and Microsystems, 26(8), November, pp. 363–371.
- [222] SCHLUNZ EB, 2016, Multiobjective in-core fuel management optimisation for nuclear research reactors, PhD thesis, Stellenbosch University, Stellenbosch.
- [223] SCHÖLKOPF B, SMOLA AJ & BACH F, 2002, Learning with Kernels: Support Vector Machines, Regularization, Optimization, and Beyond, MIT press.
- [224] SCHRIBER TJ, BRUNNER DT & SMITH JS, 2013, Inside discrete-event simulation software: How it works and why it matters, Proceedings of the 2013 Winter Simulations Conference (WSC), pp. 424–438.
- [225] SCHWEFEL HP & RUDOLPH G, 1995, pp. 7–45 in , *Contemporary Evolution Strategies*, Springer, Berlin, Heidelberg.
- [226] SHAFFER JP, 1986, Modified sequentially rejective multiple test procedures, Journal of the American Statistical Association, **81(395)**, pp. 826–831.
- [227] SHAH T, 2020, About Train, Validation and Test Sets in Machine Learning.
- [228] SHAW P, 1998, Using Constraint Programming and Local Search Methods to Solve Vehicle Routing Problems, Proceedings of the International Conference on Principles and Practice of Constraint Programming, pp. 417–431.
- [229] SHESKIN DJ, 2011, Handbook of Parametric and Nonparametric Statistical Procedures, Chapman & Hall.
- [230] SHIN M, SARGENT RG & GOEL AL, 2002, Gaussian radial basis functions for simulation metamodeling, Proceedings of the Proceedings of the Winter Simulation Conference, San Diego (CA), pp. 483–488.
- [231] SIDNEY S, 1957, Nonparametric statistics for the behavioral sciences, 125(3).
- [232] SMITH KI, EVERSON RM & FIELDSEND JE, 2004, Dominance measures for multi-objective simulated annealing, Proceedings of the Proceedings of the 2004 congress on evolutionary computation (IEEE Cat. No. 04TH8753), pp. 23–30.
- [233] SMITH KI, EVERSON RM, FIELDSEND JE, MURPHY C & MISRA R, 2008, Dominancebased multiobjective simulated annealing, IEEE Transactions on Evolutionary computation, 12(3), pp. 323–342.
- [234] SMOLA AJ & SCHÖLKOPF B, 2004, A tutorial on support vector regression, Statistics and Computing, 14(3), pp. 199–222.
- [235] SNOEK J, LAROCHELLE H & ADAMS RP, 2012, Practical bayesian optimization of machine learning algorithms, Advances in Neural Information Processing Systems (NIPS), 25.

178	REFERENCES
[236]	SOLIÉS JF, FRAIRE HJ, SOTO-MONTERRUBIO JC & PAZOS-RANGEL R, "Multi-Objective Simulated Annealing Algorithms for General Problems", in: <i>Handbook of Research on Military, Aeronautical, and Maritime Logistics and Operations</i> , IGI Global, 2016, pp. 280–292.
[237]	SØRNGÅRD B, 2014, Information Theory for Analyzing Neural Networks, Proceedings of the.
[238]	SRINIVAS N & DEB K, 1994, Muiltiobjective optimization using nondominated sorting in genetic algorithms, Evolutionary Computation, 2(3) , pp. 221–248.
[239]	STEURER M, HILL RJ & PFEIFER N, 2021, Metrics for evaluating the performance of ma- chine learning based automated valuation models, Journal of Property Research, 38(2) , pp. 1–31.
[240]	STORN R & PRICE K, 1997, Differential evolution-a simple and efficient heuristic for global optimization over continuous spaces, Journal of Global Optimization, 11(4) , pp. 341 359.
[241]	SUTSKEVER I, MARTENS J, DAHL G & HINTON G, 2013, On the importance of ini- tialization and momentum in deep learning, Proceedings of the 3 th 30th International Conference on Machine Learning (ICML), Atlanta, Georgia (GA), pp. 1139–1147.
[242]	SUTTON RS & BARTO AG, 2018, <i>Reinforcement Learning: An Introduction</i> , 2 nd Edition, MIT press, Cambridge (MA).
[243]	SVENSÉN M & BISHOP CM, 2007, Pattern Recognition and Machine Learning, Springer.
[244]	SVOZIL D, KVASNICKA V & POSPICHAL J, 1997, Introduction to multi-layer feed-forward neural networks, omputer Science Chemometrics and Intelligent Laboratory Systems, 39(1) , November, pp. 43–62.
[245]	TIELEMAN T & HINTON G, 2012, <i>Neural Networks for Machine Learning</i> , (Unpublished) Technical Report, University of Toronto, Toronto.
[246]	TSANG E & VOUDOURIS C, 1997, Fast local search and guided local search and their appli- cation to British Telecom's workforce scheduling problem, Operations Research Letters, 20(3) , March, pp. 119–127.
[247]	TSUTSUI S, YAMAMURA M & HIGUCHI T, 1999, <i>Multi-parent recombination with simplex crossover in real coded genetic algorithms</i> , Proceedings of the 1st Annual Conference on Genetic and Evolutionary Computation, pp. 657–664.
[248]	VAN GELDER L, DAS P, JANSSEN H & ROELS S, 2014, Comparative study of metamod- elling techniques in building energy simulation: Guidelines for practitioners, Simulation Modelling Practice and Theory, 49(1), December, pp. 245–257.
[249]	VAN LAARHOVEN PJ & AARTS EH, "Simulated Annealing", in: Simulated Annealing: Theory and Applications, Springer, June 1987, pp. 7–15.
[250]	VAN LUONG T, MELAB N & TALBI EG, 2011, <i>GPU computing for parallel local search metaheuristic algorithms</i> , IEEE Transactions on Computers, 62(1) , January, pp. 173–185.
[251]	VAPNIK V & CHERVONENKIS AY, 1974, The method of ordered risk minimization, I, Stochastic Systems, 8, pp. 21–30.
[252]	VENTER G & SOBIESZCZANSKI SOBIESKI J, 2004, Multidisciplinary optimization of a transport aircraft wing using particle swarm optimization, Structural and Multidisciplinary Optimization, 26(1) , January, pp. 121–131.

- [253] VOUDOURIS C & TSANG E, 1999, Guided local search and its application to the traveling salesman problem, European Journal of Operational Research, 113(2), March, pp. 469– 499.
- [254] VRUGT JA & ROBINSON BA, 2007, Improved evolutionary optimization from genetically adaptive multimethod search, Proceedings of the National Academy of Sciences, 104(3), pp. 708–711.
- [255] WHITLEY LD, 1989, The GENITOR algorithm and selection pressure: Why rank-based allocation of reproductive trials is best., Proceedings of the Junth Icga, Fairfax (VA), pp. 116–123.
- [256] WINSTON WL, 2004, Operations Research: Applications and Algorithms, 4th Edition, Duxbury Press.
- [257] WOLPERT DH & MACREADY WG, 1997, No free lunch theorems for optimization, IEEE Transactions on Evolutionary Computation, 1(1), pp. 67–82.
- [258] XU J, HUANG E, CHEN C.-H & LEE LH, 2015, Simulation optimization: A review and exploration in the new era of cloud computing and big data, Asia-Pacific Journal of Operational Research, 32(03).
- [259] YANG S, WU C & HU SJ, 2000, Modeling and analysis of multi-stage transfer lines with unreliable machines and finite buffers, Annals of Operations Research, 93(1), pp. 405– 421.
- [260] YOON M & BEKKER J, 2017, Single-and multi-objective ranking and selection procedures in simulation: A historical review, South African Journal of Industrial Engineering, 28(2), pp. 37–45.
- [261] YOSHIDA H, KAWATA K, FUKUYAMA Y, TAKAYAMA S & NAKANISHI Y, 2000, A particle swarm optimization for reactive power and voltage control considering voltage security assessment, IEEE Transactions on Power Systems, **15(4)**, pp. 1232–1239.
- [262] ZAREMBA W, SUTSKEVER I & VINYALS O, 2014, *Recurrent neural network regularization*, Neural and Evolutionary Computing.
- [263] ZEILER MD, 2012, Adadelta: An adaptive learning rate method, Machine Learning.
- [264] ZHENG YJ, 2015, Water wave optimization: A new nature-inspired metaheuristic, Computers & Operations Research, 55(1), March, pp. 1–11.
- [265] ZHOU Y, HAO J.-K & DUVAL B, 2016, Reinforcement learning based local search for grouping problems: A case study on graph coloring, Expert Systems with Applications, 64, December, pp. 412–422.
- [266] ZITZLER E, 1999, Evolutionary algorithms for multiobjective optimization: Methods and applications, PhD thesis, Swiss Federal Institute of Technology, Zurich.
- [267] ZITZLER E, KNOWLES J & THIELE L, 2008, Quality Assessment of Pareto Set Approximations, pp. 373–404 in BRANKE J, DEB K, MIETTINEN K & SŁOWIŃSKI R (EDS), Multiobjective Optimization: Interactive and Evolutionary Approaches, Springer, Berlin, Heidelberg.
- [268] ZITZLER E & KÜNZLI S, 2004, Indicator-Based Selection in Multiobjective Search, Proceedings of the Parallel Problem Solving from Nature, Berlin, Heidelberg, pp. 832–842.
- [269] ZITZLER E & THIELE L, 1999, Multiobjective evolutionary algorithms: A comparative case study and the strength Pareto approach, IEEE Transactions on Evolutionary Computation, 3(4), November, pp. 257–271.

[270] ZITZLER E, THIELE L, LAUMANNS M, FONSECA CM & DA FONSECA VG, 2003, Performance assessment of multiobjective optimizers: An analysis and review, IEEE Transactions on Evolutionary Computation, 7(2), April, pp. 117–132.

180
APPENDIX A

Metamodel Pilot Study Results

This appendix contains additional results obtained during the pilot study conducted in Chapter 3 to determine the feasibility of an ANN metamodel. The results obtained for the 30 trails conducted for each regression model is presented in Tables A.1–A.5 and the Python script used for MLR is given in Listing A.2. Thereafter, the results for the 30 trials conducted for hyperparameter combinations A–B is given in Tables A.6–A.9 and the Python script for hyperparameter combinations C is given in Listing A.4. The Python script illustrating the process followed for Bayesian optimisation is given in Listing A.3.

Listing A.1: The Python script used for the data preprocessing phase.

```
1
      #Build regressors:
2
      # 1. Multiple Linear Regression
3
4
      from sklearn.linear_model import LinearRegression
      mlr_regressor = LinearRegression()
5
6
      # 2. Support Vector Machine Regression
7
      from sklearn.multioutput import MultiOutputRegressor
8
      from sklearn.svm import SVR
9
      SVM_regressor = SVR(kernel = 'rbf')
10
11
      svm_regressor = MultiOutputRegressor(SVM_regressor)
12
      # 3. Decision Tree Regression
      from sklearn.tree import DecisionTreeRegressor
14
      dt_regressor = DecisionTreeRegressor()
15
16
      # 4. Random Forest Regression
17
      from sklearn.ensemble import RandomForestRegressor
18
      rfr_regressor = RandomForestRegressor(n_estimators = 15, random_state = 0)
19
20
      # 5. K-Nearest Neighbour Regression
21
      from sklearn.neighbors import KNeighborsRegressor
22
23
      knn_regressor = KNeighborsRegressor()
```

Listing A.2: The Python script used to run 30 independent trials for the MLR model.

```
# Multiple Linear Regression
1
      for i in range(1, 31):
2
3
        dataset = pd.read_csv('ssIP_dataset.csv')
        X = dataset.iloc[: , 0:2].values # Independent variables or inputs
4
5
        y = dataset.iloc[: , 2:4].values # Dependent variables or outputs
6
        X_train, X_test, y_train, y_test = train_test_split(X, y,
7
8
                                                    test size = 0.2
9
                                                    random_state = i)
        sc_X = StandardScaler()
10
11
        sc_y = StandardScaler()
        X_train = sc_X.fit_transform(X_train) # Normalise X_train
12
        X_test = sc_X.transform(X_test)  # Normalise X_test
13
        y_train = sc_y.fit_transform(y_train) # Normalise y_train
14
15
        y_test = sc_y.transform(y_test)
                                               # Normalise y_test
16
17
        mlr_regressor = LinearRegression()
                                               # Build regressor
        mlr_regressor.fit(X_train, y_train)
18
19
        t_y_pred = mlr_regressor.predict(X_train)
```

24

```
21
         v_mses = []
        v_maes = []
22
23
        test_maes = []
        test_mses = []
24
        t_mse = mean_squared_error(y_train,t_y_pred)
25
        t_mae = mean_absolute_error(y_train,t_y_pred)
26
27
        # 10-Fold Cross Validation
28
         v_mse = -np.round(cross_val_score(mlr_regressor, X_train,
29
30
                                          y_train,
                                           scoring='neg_mean_squared_error',
31
                                          cv = 10), 4)
32
        v_mses.append(v_mse)
33
34
         v_mse_avg = np.round(v_mse.mean(), 4)
35
         v_mae = -np.round(cross_val_score(mlr_regressor, X_train,
36
                                          y_train,
37
                                          scoring='neg_mean_absolute_error',
                                          cv = 10), 4)
38
39
        v_maes.append(v_mae)
40
         v_mae_avg = np.round(v_mae.mean(), 4)
41
        # Evaluate regressor on Test set
42
43
         test_mse = np.round(mean_squared_error(y_test, y_pred), 4)
        test_mses.append(test_mse)
44
         test_mse_avg = np.round(test_mse.mean(), 4)
45
46
        test_mae = np.round(mean_absolute_error(y_test, y_pred), 4)
47
         test_maes.append(test_mae)
48
        test_mae_avg = np.round(test_mae.mean(), 4)
49
50
         # Populate dataframe
51
         trial_performance_row = [t_mse, t_mae,
                                  v_mse_avg, v_mae_avg,
53
                                  test_mse_avg, test_mae_avg]
        MLR_performance_df.loc['Run_' + str(i)] = trial_performance_row
54
```

Listing A.3: The Python script used to run the Bayesian optimisation and determine suitable hyperparameters for the FNN.

```
def model(hp):
1
2
     # Hyperparameter search space
3
    hp_optimizer = hp.Choice("optimizer", values = ["sgd","adam", "nadam"])
hp_learning_rate = hp.Choice("learning_rate", values = [1e-2, 1e-3, 1e-4])
4
5
    hp_regularizer = hp.Choice("kernel_regularizer", values = ["11", "12", "11_12"])
6
    hp_activation = hp.Choice("activation", values = ["sigmoid", "relu", "elu"])
7
8
     hp_initializer = hp.Choice("kernel_initializer",
     values=["HeUniform", "HeNormal", "GlorotUniform", "GlorotNormal"])
9
    hp_units = hp.Int("units", min_value = 2, max_value = 20, step = 1)
10
11
    # Build ANN
12
    model = Sequential()
13
14
    # Hidden layer
15
     model.add(Dense(units = hp_units,
16
                    kernel_regularizer = hp_regularizer,
17
                     activation = hp_activation,
18
19
                    kernel_initializer = hp_initializer))
20
21
     model.add(BatchNormalization()) # Comment out if not applicable
22
23
     # Output layer
    model.add(tf.keras.layers.Dense(units = 2, activation = hp_activation))
24
25
26
    model.compile(optimizer = hp_optimizer,
27
                    loss = "mean_squared_error",
                    metrics = ["mean_squared_error"]
28
29
    )
30
31
    return model
```

Listing A.4: The Python script used to run 30 independent trails for the hyperparameter combination C found during Bayesian optimisation for the ANN.

```
# Import dataset
    dataset = pd.read_csv('ssIP_dataset.csv')
    X = dataset.iloc[: , 0:2].values # Independent variables, i.e. inputs
    y = dataset.iloc[: , 2:4].values # Dependent variables, i.e. outputs
    for i in range(1, 31):
    # Split dataset using random state i
    X_train, X_test, y_train, y_test = train_test_split(X, y,
                                               test_size = 0.2,
                                               random_state = i)
    # Normalise
    sc_X = StandardScaler()
    sc_y = StandardScaler()
15
    X_train = sc_X.fit_transform(X_train)
16
    X_test = sc_X.transform(X_test)
    y_train = sc_y.fit_transform(y_train)
    y_test = sc_y.transform(y_test)
    # Build ANN using best hyperparameters
    bo_model = Sequential()
    # Hidden layer
    bo_model.add(Dense(units=14,
                      kernel_regularizer = "12",
                      kernel_initializer = "HeNormal",
                      activation='elu'))
    bo_model.add(BatchNormalization()) # Comment out if not applicable
    # Output layer
    bo_model.add(Dense(units=2, activation='elu'))
    bo_optimizer = tf.keras.optimizers.Nadam(learning_rate = 0.001)
    bo_model.compile(optimizer = bo_optimizer,
                      loss = "mean_squared_error",
                      metrics=["mean_squared_error"])
    history_bo = bo_model.fit(X_train,
                             y_train,
                             batch_size = batch_size,
                             epochs = max_epochs,
                             validation_split = 0.25,
                             callbacks=[early_stopping])
    # Evaluate model on Test set
   eval_result = bo_model.evaluate(X_test, y_test)
48
```

1

2

3

4 5

6

7 8

9 10

11

12

13

14

17 18

19

20 21

22

23 24

25

26 27

2829

30 31

32 33 34

35

36 37

38

39

40

41 42

43

44 45

46

Table A.1: Summary of the performance of MLR in terms of MSE and MAE for the 30 trails on the IP dataset.

	Traini	ing set	Validat	tion set	Tes	t set
	MSE	MAE	MSE	MAE	MSE	MAE
Trial 1	0.1479	0.3109	0.1479	0.3109	0.1480	0.3111
Trial 2	0.1481	0.3110	0.1481	0.3110	0.1474	0.3103
Trial 3	0.1479	0.3109	0.1479	0.3109	0.1470	0.3100
Trial 4	0.1482	0.3112	0.1482	0.3112	0.1463	0.3092
Trial 5	0.1480	0.3108	0.1480	0.3108	0.1487	0.3118
Trial 6	0.1480	0.3109	0.1480	0.3109	0.1476	0.3106
Trial 7	0.1481	0.3111	0.1481	0.3111	0.1476	0.3105
Trial 8	0.1480	0.3109	0.1480	0.3109	0.1486	0.3118
Trial 9	0.1482	0.3111	0.1482	0.3111	0.1486	0.3117
Trial 10	0.1479	0.3108	0.1479	0.3108	0.1483	0.3113
Trial 11	0.1481	0.3110	0.1481	0.3110	0.1481	0.3109
Trial 12	0.1481	0.3111	0.1481	0.3111	0.1477	0.3105
Trial 13	0.1477	0.3107	0.1478	0.3107	0.1486	0.3114
Trial 14	0.1478	0.3107	0.1478	0.3107	0.1485	0.3116
Trial 15	0.1480	0.3111	0.1480	0.3111	0.1483	0.3108
Trial 16	0.1479	0.3109	0.1479	0.3109	0.1484	0.3111
Trial 17	0.1480	0.3108	0.1480	0.3109	0.1482	0.3114
Trial 18	0.1478	0.3107	0.1478	0.3107	0.1475	0.3107
Trial 19	0.1480	0.3109	0.1480	0.3109	0.1491	0.3120
Trial 20	0.1481	0.3110	0.1481	0.3110	0.1473	0.3107
Trial 21	0.1480	0.3110	0.1480	0.3110	0.1484	0.3111
Trial 22	0.1481	0.3110	0.1481	0.3110	0.1485	0.3115
Trial 23	0.1480	0.3108	0.1480	0.3108	0.1497	0.3128
Trial 24	0.1479	0.3109	0.1480	0.3110	0.1473	0.3103
Trial 25	0.1478	0.3106	0.1478	0.3106	0.1489	0.3120
Trial 26	0.1476	0.3106	0.1476	0.3106	0.1486	0.3116
Trial 27	0.1480	0.3108	0.1480	0.3108	0.1480	0.3113
Trial 28	0.1479	0.3109	0.1479	0.3109	0.1487	0.3117
Trial 29	0.1479	0.3109	0.1479	0.3110	0.1476	0.3103
Trial 30	0.1481	0.3111	0.1481	0.3111	0.1472	0.3101
Average	0.1480	0.3109	0.1480	0.3109	0.1481	0.3111

Table A.2: Summary of the performance of SVM in terms of MSE and MAE for the 30 trails on the IP dataset.

	Traini	ing set	Validat	tion set	Test	Test set	
	MSE	MAE	MSE	MAE	MSE	MAE	
Trial 1	0.0026	0.0438	0.0026	0.0439	0.0026	0.0438	
Trial 2	0.0026	0.0437	0.0026	0.0438	0.0026	0.0437	
Trial 3	0.0026	0.0439	0.0026	0.0439	0.0026	0.0439	
Trial 4	0.0026	0.0439	0.0026	0.0439	0.0026	0.0439	
Trial 5	0.0026	0.0439	0.0026	0.0439	0.0026	0.0439	
Trial 6	0.0026	0.0440	0.0026	0.0439	0.0026	0.0440	
Trial 7	0.0026	0.0439	0.0026	0.0439	0.0026	0.0439	
Trial 8	0.0026	0.0438	0.0026	0.0438	0.0026	0.0438	
Trial 9	0.0026	0.0439	0.0026	0.0439	0.0026	0.0439	
Trial 10	0.0026	0.0439	0.0026	0.0438	0.0026	0.0439	
Trial 11	0.0026	0.0438	0.0026	0.0439	0.0026	0.0438	
Trial 12	0.0026	0.0437	0.0026	0.0437	0.0026	0.0437	
Trial 13	0.0026	0.0438	0.0026	0.0437	0.0026	0.0438	
Trial 14	0.0026	0.0438	0.0026	0.0440	0.0026	0.0438	
Trial 15	0.0026	0.0439	0.0026	0.0439	0.0026	0.0439	
Trial 16	0.0026	0.0437	0.0026	0.0438	0.0026	0.0437	
Trial 17	0.0026	0.0437	0.0026	0.0436	0.0026	0.0437	
Trial 18	0.0026	0.0437	0.0026	0.0438	0.0026	0.0437	
Trial 19	0.0026	0.0436	0.0026	0.0438	0.0026	0.0436	
Trial 20	0.0026	0.0438	0.0026	0.0438	0.0026	0.0438	
Trial 21	0.0026	0.0440	0.0026	0.0438	0.0026	0.0440	
Trial 22	0.0026	0.0440	0.0026	0.0439	0.0026	0.0440	
Trial 23	0.0026	0.0438	0.0026	0.0439	0.0026	0.0438	
Trial 24	0.0026	0.0440	0.0026	0.0439	0.0026	0.0440	
Trial 25	0.0026	0.0436	0.0026	0.0438	0.0026	0.0436	
Trial 26	0.0026	0.0437	0.0026	0.0438	0.0026	0.0437	
Trial 27	0.0026	0.0438	0.0026	0.0438	0.0026	0.0438	
Trial 28	0.0026	0.0438	0.0026	0.0439	0.0026	0.0438	
Trial 29	0.0026	0.0436	0.0026	0.0436	0.0026	0.0436	
Trial 30	0.0026	0.0438	0.0026	0.0439	0.0026	0.0438	
Average	0.0026	0.0438	0.0026	0.0438	0.0026	0.0438	

Table A.3: Summary of the performance of DTR in terms of MSE and MAE for the 30 trails on the IP dataset.

	Train	ing set	Valida	ation set	Te	st set
	MSE	MAE	MSE	MAE	MSE	MAE
Trial 1	0	0	0	0.0025	0	0.0024
Trial 2	0	0	0	0.0025	0	0.0024
Trial 3	0	0	0	0.0025	0	0.0024
Trial 4	0	0	0	0.0025	0	0.0024
Trial 5	0	0	0	0.0025	0	0.0024
Trial 6	0	0	0	0.0025	0	0.0024
Trial 7	0	0	0	0.0025	0	0.0024
Trial 8	0	0	0	0.0025	0	0.0024
Trial 9	0	0	0	0.0025	0	0.0024
Trial 10	0	0	0	0.0025	0	0.0024
Trial 11	0	0	0	0.0025	0	0.0024
Trial 12	0	0	0	0.0025	0	0.0024
Trial 13	0	0	0	0.0025	0	0.0024
Trial 14	0	0	0	0.0025	0	0.0024
Trial 15	0	0	0	0.0025	0	0.0024
Trial 16	0	0	0	0.0025	0	0.0024
Trial 17	0	0	0	0.0025	0	0.0024
Trial 18	0	0	0	0.0025	0	0.0024
Trial 19	0	0	0	0.0025	0	0.0024
Trial 20	0	0	0	0.0025	0	0.0024
Trial 21	0	0	0	0.0025	0	0.0024
Trial 22	0	0	0	0.0025	0	0.0024
Trial 23	0	0	0	0.0025	0	0.0024
Trial 24	0	0	0	0.0025	0	0.0024
Trial 25	0	0	0	0.0025	0	0.0024
Trial 26	0	0	0	0.0025	0	0.0024
Trial 27	0	0	0	0.0025	0	0.0024
Trial 28	0	0	0	0.0025	0	0.0024
Trial 29	0	0	0	0.0025	0	0.0024
Trial 30	0	0	0	0.0025	0	0.0024
Average	0	0	0	0.0025	0	0.0024

Table A.4: Summary of the performance of RFR in terms of MSE and MAE for the 30 trails on the IP dataset.

	Training set		Valida	ation set	Te	Test set	
	MSE	MAE	MSE	MAE	MSE	MAE	
Trial 1	0	0.00151	0	0.0015	0	0.0015	
Trial 2	0	0.00149	0	0.0015	0	0.0015	
Trial 3	0	0.00150	0	0.0015	0	0.0015	
Trial 4	0	0.00151	0	0.0015	0	0.0015	
Trial 5	0	0.00149	0	0.0016	0	0.0015	
Trial 6	0	0.00149	0	0.0016	0	0.0015	
Trial 7	0	0.00150	0	0.0015	0	0.0015	
Trial 8	0	0.00149	0	0.0016	0	0.0015	
Trial 9	0	0.00151	0	0.0016	0	0.0015	
Trial 10	0	0.00151	0	0.0015	0	0.0015	
Trial 11	0	0.00148	0	0.0015	0	0.0015	
Trial 12	0	0.00149	0	0.0015	0	0.0015	
Trial 13	0	0.00150	0	0.0015	0	0.0015	
Trial 14	0	0.00149	0	0.0016	0	0.0015	
Trial 15	0	0.00150	0	0.0015	0	0.0015	
Trial 16	0	0.00150	0	0.0016	0	0.0015	
Trial 17	0	0.00151	0	0.0016	0	0.0015	
Trial 18	0	0.00149	0	0.0015	0	0.0015	
Trial 19	0	0.00151	0	0.0015	0	0.0015	
Trial 20	0	0.00151	0	0.0015	0	0.0015	
Trial 21	0	0.00149	0	0.0015	0	0.0015	
Trial 22	0	0.00151	0	0.0015	0	0.0015	
Trial 23	0	0.00151	0	0.0015	0	0.0015	
Trial 24	0	0.00149	0	0.0015	0	0.0015	
Trial 25	0	0.00150	0	0.0016	0	0.0015	
Trial 26	0	0.00149	0	0.0015	0	0.0015	
Trial 27	0	0.00149	0	0.0016	0	0.0015	
Trial 28	0	0.00149	0	0.0016	0	0.0015	
Trial 29	0	0.00149	0	0.0015	0	0.0015	
Trial 30	0	0.00150	0	0.0015	0	0.0015	
Average	0	0.00150	0	0.0015	0	0.0015	

Table A.5: Summary of the performance of XGBoost in terms of MSE and MAE for the 30 trails on the IP dataset.

	Train	ing set	Valida	tion set	Tes	Test set	
	MSE	MAE	MSE	MAE	MSE	MAE	
Trial 1	0.0001	0.0072	0.0001	0.0071	0.0001	0.0072	
Trial 2	0.0001	0.0070	0.0001	0.0071	0.0001	0.0070	
Trial 3	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 4	0.0001	0.0072	0.0001	0.0071	0.0001	0.0072	
Trial 5	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 6	0.0001	0.0070	0.0001	0.0070	0.0001	0.0070	
Trial 7	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 8	0.0001	0.0070	0.0001	0.0070	0.0001	0.0070	
Trial 9	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 10	0.0001	0.0072	0.0001	0.0071	0.0001	0.0072	
Trial 11	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 12	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 13	0.0001	0.0070	0.0001	0.0072	0.0001	0.0070	
Trial 14	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 15	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 16	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 17	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 18	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 19	0.0001	0.0072	0.0001	0.0071	0.0001	0.0072	
Trial 20	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 21	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 22	0.0001	0.0072	0.0001	0.0071	0.0001	0.0072	
Trial 23	0.0001	0.0070	0.0001	0.0071	0.0001	0.0070	
Trial 24	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 25	0.0001	0.0071	0.0001	0.0072	0.0001	0.0071	
Trial 26	0.0001	0.0070	0.0001	0.0070	0.0001	0.0070	
Trial 27	0.0001	0.0070	0.0001	0.0071	0.0001	0.0070	
Trial 28	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	
Trial 29	0.0001	0.0070	0.0001	0.0071	0.0001	0.0070	
Trial 30	0.0001	0.0070	0.0001	0.0071	0.0001	0.0070	
Average	0.0001	0.0071	0.0001	0.0071	0.0001	0.0071	

	Traini	ng set	Valida	tion set	Test	t set
	MSE	MAE	MSE	MAE	MSE	MAE
Trial 1	0.052	0.146	0.052	0.146	0.053	0.145
Trial 2	0.052	0.145	0.052	0.145	0.076	0.211
Trial 3	0.054	0.149	0.054	0.149	0.054	0.149
Trial 4	0.054	0.150	0.054	0.150	0.056	0.156
Trial 5	0.052	0.144	0.052	0.144	0.152	0.340
Trial 6	0.052	0.146	0.052	0.146	0.058	0.168
Trial 7	0.053	0.146	0.053	0.146	0.062	0.176
Trial 8	0.054	0.148	0.054	0.148	0.054	0.147
Trial 9	0.053	0.146	0.053	0.146	0.062	0.175
Trial 10	0.055	0.151	0.055	0.151	0.054	0.143
Trial 11	0.055	0.150	0.055	0.150	0.058	0.162
Trial 12	0.055	0.151	0.055	0.151	0.209	0.396
Trial 13	0.054	0.150	0.054	0.150	0.054	0.145
Trial 14	0.055	0.151	0.055	0.151	0.053	0.144
Trial 15	0.054	0.147	0.054	0.147	0.059	0.161
Trial 16	0.054	0.148	0.054	0.148	0.055	0.155
Trial 17	0.054	0.149	0.054	0.149	0.061	0.172
Trial 18	0.052	0.146	0.052	0.146	0.058	0.169
Trial 19	0.053	0.146	0.053	0.146	0.056	0.151
Trial 20	0.052	0.145	0.052	0.145	0.050	0.138
Trial 21	0.053	0.143	0.053	0.143	0.051	0.135
Trial 22	0.054	0.148	0.054	0.148	0.055	0.155
Trial 23	0.053	0.147	0.053	0.147	0.054	0.149
Trial 24	0.054	0.148	0.054	0.148	0.060	0.173
Trial 25	0.054	0.148	0.054	0.148	0.057	0.160
Trial 26	0.052	0.145	0.052	0.145	0.078	0.218
Trial 27	0.049	0.137	0.049	0.137	0.079	0.213
Trial 28	0.052	0.144	0.052	0.144	0.074	0.206
Trial 29	0.051	0.140	0.051	0.140	0.052	0.147
Trial 30	0.053	0.147	0.053	0.147	0.055	0.155
Average	0.053	0.147	0.053	0.147	0.067	0.177

Table A.6: Summary of the performance of the ANN with hyperparameter combination A in terms of MSE and MAE for the 30 trails on the IP dataset.

	Traini	ing set	Validat	tion set	Test	t set
	MSE	MAE	MSE	MAE	MSE	MAE
Trial 1	0.049	0.130	0.049	0.130	0.049	0.129
Trial 2	0.049	0.131	0.049	0.131	0.050	0.131
Trial 3	0.049	0.133	0.049	0.133	0.049	0.133
Trial 4	0.049	0.133	0.049	0.133	0.050	0.134
Trial 5	0.049	0.132	0.049	0.132	0.050	0.132
Trial 6	0.050	0.133	0.050	0.133	0.051	0.133
Trial 7	0.050	0.134	0.050	0.134	0.049	0.134
Trial 8	0.050	0.134	0.050	0.134	0.051	0.135
Trial 9	0.049	0.133	0.049	0.133	0.050	0.133
Trial 10	0.049	0.131	0.049	0.131	0.050	0.133
Trial 11	0.049	0.132	0.049	0.132	0.049	0.132
Trial 12	0.050	0.135	0.050	0.135	0.050	0.135
Trial 13	0.049	0.131	0.049	0.131	0.049	0.130
Trial 14	0.049	0.131	0.049	0.131	0.049	0.131
Trial 15	0.051	0.136	0.051	0.136	0.051	0.136
Trial 16	0.048	0.129	0.048	0.129	0.049	0.130
Trial 17	0.049	0.131	0.049	0.131	0.049	0.132
Trial 18	0.049	0.130	0.049	0.130	0.049	0.131
Trial 19	0.049	0.133	0.049	0.133	0.049	0.131
Trial 20	0.049	0.133	0.049	0.133	0.050	0.133
Trial 21	0.050	0.135	0.050	0.135	0.050	0.135
Trial 22	0.049	0.132	0.049	0.132	0.049	0.131
Trial 23	0.049	0.132	0.049	0.132	0.049	0.132
Trial 24	0.049	0.132	0.049	0.132	0.049	0.132
Trial 25	0.050	0.135	0.050	0.135	0.050	0.134
Trial 26	0.046	0.122	0.046	0.122	0.047	0.123
Trial 27	0.049	0.131	0.049	0.131	0.049	0.131
Trial 28	0.049	0.132	0.049	0.132	0.050	0.132
Trial 29	0.049	0.132	0.049	0.132	0.049	0.131
Trial 30	0.049	0.131	0.049	0.131	0.050	0.131
Average	0.049	0.132	0.049	0.132	0.049	0.132

Table A.7: Summary of the performance of the ANN with hyperparameter combination B in terms of MSE and MAE for the 30 trails on the IP dataset.

	Traini	ing set	Validat	tion set	Test	Test set	
	MSE	MAE	MSE	MAE	MSE	MAE	
Trial 1	0.054	0.148	0.054	0.148	0.053	0.143	
Trial 2	0.051	0.139	0.051	0.139	0.050	0.138	
Trial 3	0.050	0.137	0.050	0.137	0.049	0.136	
Trial 4	0.050	0.134	0.050	0.134	0.050	0.134	
Trial 5	0.049	0.133	0.049	0.133	0.049	0.133	
Trial 6	0.051	0.138	0.051	0.138	0.052	0.137	
Trial 7	0.050	0.137	0.050	0.137	0.049	0.132	
Trial 8	0.052	0.142	0.052	0.142	0.053	0.140	
Trial 9	0.051	0.139	0.051	0.139	0.053	0.140	
Trial 10	0.050	0.135	0.050	0.135	0.049	0.134	
Trial 11	0.051	0.140	0.051	0.140	0.053	0.148	
Trial 12	0.049	0.134	0.049	0.134	0.049	0.129	
Trial 13	0.052	0.142	0.052	0.142	0.051	0.138	
Trial 14	0.051	0.139	0.051	0.139	0.050	0.136	
Trial 15	0.051	0.140	0.051	0.140	0.050	0.138	
Trial 16	0.054	0.148	0.054	0.148	0.055	0.150	
Trial 17	0.050	0.136	0.050	0.136	0.050	0.138	
Trial 18	0.051	0.137	0.051	0.137	0.051	0.135	
Trial 19	0.050	0.135	0.050	0.135	0.049	0.132	
Trial 20	0.053	0.143	0.053	0.143	0.053	0.140	
Trial 21	0.055	0.150	0.055	0.150	0.054	0.147	
Trial 22	0.051	0.139	0.051	0.139	0.050	0.136	
Trial 23	0.050	0.137	0.050	0.137	0.051	0.139	
Trial 24	0.052	0.141	0.052	0.141	0.050	0.138	
Trial 25	0.051	0.138	0.051	0.138	0.050	0.134	
Trial 26	0.053	0.145	0.053	0.145	0.054	0.146	
Trial 27	0.050	0.135	0.050	0.135	0.049	0.131	
Trial 28	0.051	0.139	0.051	0.139	0.050	0.137	
Trial 29	0.050	0.136	0.050	0.136	0.048	0.130	
Trial 30	0.050	0.134	0.050	0.134	0.051	0.139	
Average	0.051	0.139	0.051	0.139	0.051	0.138	

Table A.8: Summary of the performance of the ANN with hyperparameter combination C in terms of MSE and MAE for the 30 trails on the IP dataset.

	Traini	ing set	Valida	tion set	Test	t set
	MSE	MAE	MSE	MAE	MSE	MAE
Trial 1	0.060	0.150	0.060	0.150	0.060	0.150
Trial 2	0.060	0.154	0.060	0.154	0.061	0.155
Trial 3	0.060	0.151	0.060	0.151	0.060	0.151
Trial 4	0.060	0.153	0.060	0.153	0.060	0.153
Trial 5	0.061	0.154	0.061	0.154	0.062	0.154
Trial 6	0.059	0.146	0.059	0.146	0.060	0.147
Trial 7	0.062	0.153	0.062	0.153	0.061	0.153
Trial 8	0.061	0.153	0.061	0.153	0.062	0.153
Trial 9	0.059	0.148	0.059	0.148	0.059	0.149
Trial 10	0.062	0.155	0.062	0.155	0.063	0.155
Trial 11	0.060	0.152	0.060	0.152	0.060	0.152
Trial 12	0.057	0.146	0.057	0.146	0.058	0.146
Trial 13	0.059	0.150	0.059	0.150	0.059	0.150
Trial 14	0.061	0.154	0.061	0.154	0.061	0.154
Trial 15	0.060	0.150	0.060	0.150	0.060	0.150
Trial 16	0.059	0.151	0.059	0.151	0.060	0.151
Trial 17	0.060	0.151	0.060	0.151	0.061	0.151
Trial 18	0.058	0.145	0.058	0.145	0.058	0.146
Trial 19	0.058	0.145	0.058	0.145	0.057	0.145
Trial 20	0.059	0.148	0.059	0.148	0.059	0.148
Trial 21	0.060	0.152	0.060	0.152	0.059	0.151
Trial 22	0.058	0.146	0.058	0.146	0.058	0.146
Trial 23	0.062	0.155	0.062	0.155	0.062	0.155
Trial 24	0.059	0.150	0.059	0.150	0.058	0.149
Trial 25	0.060	0.151	0.060	0.151	0.060	0.150
Trial 26	0.060	0.151	0.060	0.151	0.061	0.152
Trial 27	0.062	0.154	0.062	0.154	0.061	0.153
Trial 28	0.060	0.152	0.060	0.152	0.061	0.153
Trial 29	0.060	0.149	0.060	0.149	0.059	0.149
Trial 30	0.060	0.150	0.060	0.150	0.060	0.151
Average	0.060	0.151	0.060	0.151	0.060	0.151

Table A.9: Summary of the performance of the ANN with hyperparameter combination D in terms of MSE and MAE for the 30 trails on the IP dataset.

APPENDIX B

Algorithmic Parameter Evaluation Results

This appendix contains additional results obtained during the hyperparameter search space evaluation for the respective algorithms as described in Chapter 6 which focussed on determining good parameter values for the BOCEGAH and BOSAH.

The results documented in this appendix were omitted from the respective sections (in Chapter 6) so as to enhance the understanding of the main text. After determining good parameter values for the respective sub-algorithms (or LLHs) they are employed in the BOCEGAH and the BOSAH, respectively.

B.1 MOOCEM

This section presents the results for the hyperparameter study conducted for the MOOCEM.

B.1.1 Open mine problem

Tables B.1 and B.2 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–40 for hyperparameter combinations A1.1.1–A1.1.9. Figures B.1 and B.2 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A1.1.1–A1.1.9 that correspond to the runs given in Table B.3.

APPENDIX B. ALGORITHMIC PARAMETER EVALUATION RESULTS



Figure B.1: The best and worst best approximation fronts for hyperparameter combinations A1.1.1–A1.1.5.



Figure B.2: The best and worst best approximation fronts for hyperparameter combinations A1.1.6–A1.1.9.

B.1.2 (s, S) Inventory problem

Tables B.6 and B.7 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A1.2.1–A1.2.9. Figures B.3 and B.4 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A1.2.1–A1.2.9 that correspond to the runs given in Table B.8.





Figure B.3: The best and worst best approximation fronts for hyperparameter combinations A1.2.1–A1.2.5.



Figure B.4: The best and worst best approximation fronts for hyperparameter combinations A1.2.6–A1.2.9.

B.1.3 Buffer allocation problem: five machines

Tables B.9 and B.10 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A1.3.1–A1.3.9. Figures B.5 and B.6 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A1.3.1–A1.3.9 that correspond to the runs given in Table B.11.





Figure B.5: The best and worst best approximation fronts for hyperparameter combinations A1.3.1–A1.3.5.



Figure B.6: The best and worst best approximation fronts for hyperparameter combinations A1.3.6–A1.3.9.

B.1.4 Buffer allocation problem: 10 machines

Tables B.12 and B.13 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A1.4.1– A1.4.9. Figures B.7 and B.8 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A1.4.1–A1.4.9 that correspond to the runs given in Table B.14.

APPENDIX B. Algorithmic Parameter Evaluation Results



Figure B.7: The best and worst best approximation fronts for hyperparameter combinations A1.4.1–A1.4.5.



Figure B.8: The best and worst best approximation fronts for hyperparameter combinations A1.4.6–A1.4.9.

B.1.5 Non-linear buffer allocation problem: 16 machines

Tables B.15 and B.16 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A1.5.1– A1.5.9. Figures B.9 and B.10 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A1.5.1–A1.5.9 that correspond to the runs given in Table B.17. In Figure B.11 the average number of non-dominated solutions obtained for hyperparameter combinations A1.1–A1.9 is illustrated for the respective simulation problems.





Figure B.9: The best and worst best approximation fronts for hyperparameter combinations A1.5.1–A1.5.5.



Figure B.10: The best and worst best approximation fronts for hyperparameter combinations A1.5.6–A1.5.9.

e ranks	
Th_{0}	
1.1.9.	
I^{-I}	
A1.1	
ons	
inati	
omb	
ter c	ests.
amei	ssis t
rpar.	othe
iype	hyp
for 1	port
[-40]	aver
run [an-D
for	l Im
ined	1 anc
obta	lmaı
lks	Fried
g raı	the
ndin	n of
odsa	tatic
corre	ndm
and	le co
eas (in th
erar	ised
hyp	are u
f the	and
ет о	SSES 2
orevi	enth
A_{I}	part
B.1:	i the
ble]	en in
Tal	givt

APPENDIX B. ALGORITHMIC PARAMETER EVALUATION RESULTS

are

Table B.3: The simulation runs that correspond to the best and worst approximation fronts (obtained by the MOOCEM) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A1.1.1–A1.1.9.

	Wo	orst	В	est
	HA	NDS	HA	NDS
A1.1.1	Run 37	Run 24	Run 3	Run 2
A1.1.2	Run 18	Run 18	$\operatorname{Run} 2$	Run 38
A1.1.3	Run 30	Run 33	$\operatorname{Run} 7$	$\operatorname{Run}25$
A1.1.4	$\operatorname{Run}34$	Run 33	$\operatorname{Run} 2$	$\operatorname{Run} 2$
A1.1.5	$\operatorname{Run}34$	Run 33	$\operatorname{Run} 1$	Run 19
A1.1.6	Run 30	Run 33	$\operatorname{Run} 2$	Run 19
A1.1.7	$\operatorname{Run}32$	Run 33	Run 2	Run 2
A1.1.8	Run 1	Run 9	Run 4	Run 4
A1.1.9	Run 1	Run 9	$\operatorname{Run}4$	$\operatorname{Run}20$

Table B.4: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the adjusted hyperareas of the approximation fronts for the hyperparameter combinations A1.1.1–A1.1.9.

	A1.1.2	A1.1.3	A1.1.4	A1.1.5	A1.1.6	A1.1.7	A1.1.8	A1.1.9	
A1.1.1	1	0.3	1	1	1	1	1	0.94	
A1.1.2		0.54	1	1	1	1	1	1	
A1.1.3			0.25	0.94	1	0.06	1	1	
A1.1.4				1	1	1	1	0.8	
A1.1.5					1	1	1	1	
A1.1.6						1	1	1	
A1.1.7							1	0.24	
A1.1.8								1	

Table B.5: The adjusted *p*-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A1.1.1-A1.1.9.

	A1.1.2	A1.1.3	A1.1.4	A1.1.5	A1.1.6	A1.1.7	A1.1.8	A1.1.9
A1.1.1	1	0.34	1	1	0.2	1	0.17	0.08
A1.1.2		1	1	1	1	1	1	0.8
A1.1.3			1	1	1	1	1	1
A1.1.4				1	1	1	1	1
A1.1.5					1	1	1	1
A1.1.6						1	1	1
A1.1.7							0.99	0.52
A1.1.8								1

are	
ranks	
The	
1.2.9.	
I-A	
A1.2	
tions	
ibinai	
r com	ts.
metei	is tes
para.	othes
hypeı	hype
) for	nport
1 - 100	Dave
. run	man-
ed for	and I
btain	man
ks) o	Fried
g ran	f the .
ndin	ion oi
respo	outat
id coi	coml
as (ar	n the
erare	ised i
e hyp	are u
of the	s and
view	heses
$A \ pre$	narent
3.6: 2	the p
ble E	en in
Tal	give

		A1.2.2	A1.2.3	A1.2.4	A1.2.5	A1.2.6	A1.2.7	A1.2.8	A1.2.9
1	$112\ 268.32\ (2)$	112268.32(2)	$112\ 268.32\ (2)$	$112\ 225.68\ (5)$	$112\ 225.68\ (5)$	$112\ 225.68\ (5)$	111 814.83 (8)	111 814.83 (8)	111 814.83 (8)
2	113556.22 (8)	113556.22 (8)	113556.22 (8)	113692.95(2)	113692.95(2)	113692.95(2)	113581.61 (5)	$113\ 581.61\ (5)$	113581.61 (5)
3	$111086.70\ (6.5)$	$111086.70\ (6.5)$	$113\ 532.90\ (3)$	111599.52 (4)	111559.62(5)	113535.94 (2)	110724.25(8.5)	$110\ 724.25\ (8.5)$	113724.64(1)
4	$112335.51\ (8.5)$	$112335.51\ (8.5)$	$112\ 698.88\ (2)$	112375.57(6.5)	112375.57(6.5)	112613.20 (3)	112395.19(4.5)	$112\ 395.19\ (4.5)$	$113696.57\ (1)$
5 C	112583.02~(8)	112583.02 (8)	112583.02 (8)	112650.31(5.5)	112650.51 (4)	$112650.31\ (5.5)$	$112\ 659.39\ (2)$	$112\ 659.39\ (2)$	112659.39(2)
9	$112\ 076.32\ (1)$	$111646.63\ (6.5)$	$111\ 646.63\ (6.5)$	111894.18(3)	$111663.01\ (4.5)$	$111663.01\ (4.5)$	111996.09(2)	111558.46(8.5)	111558.46(8.5)
7	112408.44 (8)	112408.44 (8)	112408.44 (8)	112460.96(5)	112460.96 (5)	112460.96(5)	112463.91 (2)	$112\ 463.91\ (2)$	112463.91 (2)
×	113000.86(1.5)	113000.86(1.5)	$112\ 934.65\ (6)$	112974.55(3.5)	112974.55(3.5)	112962.09(5)	112898.83 (8.5)	$112\ 898.83\ (8.5)$	112914.42 (7)
6	113055.46(2)	112894.53(5.5)	$112\ 894.53\ (5.5)$	112689.93 (7)	112569.16 (9)	112571.87 (8)	113191.16(1)	$113\ 034.35\ (3.5)$	$113034.35\ (3.5)$
10	112461.86 (5)	112461.86 (5)	$112\ 461.86\ (5)$	113178.06(1.5)	113084.96 (3)	113178.06(1.5)	112395.88 (8)	$112\ 395.88\ (8)$	112395.88 (8)
	•	•	•						
								•••	
99 100	$112\ 256.71\ (2)$	$112\ 256.71\ (2)$	$112\ 256.71\ (2)$	112 082.21 (8) 111 808 64 (5)	112 082.21 (8) 111 808 64 (5)	112 082.21 (8) 111 808 64 (5)	$112\ 153.73\ (5)$	$112\ 153.73\ (5)$	$112\ 153.73\ (5)$
TUU	(7) 00.626 111	(7) 00.626 111	(7) 00.628 111	(c) 70.000 TTT	(c) 1 0.000 111	(c) 1 0.000 111	(0) 01.000 111	(0) 01.000 111	(0) 01.000 111
$\sum_{i} R_{Ci}$	522.50	525.50	515.00	515.50	481.00	491.50	516.50	481.50	451.00
$\sum R_{Ci}^2$	273006.25	276150.25	265225.00	265740.25	231361.00	241572.25	266772.25	231842.25	203401.00

Run	A1.2.1	A1.2.2	A1.2.3	A1.2.4	A1.2.5	A1.2.6	A1.2.7	A1.2.8	A1.2.9
-	208(5)	208(5)	208(5)	206(8)	206(8)	206(8)	214(2)	214(2)	214(2)
2	254(3.5)	254(3.5)	254(3.5)	241 (8)	241 (8)	241(8)	254(3.5)	254(3.5)	254(3.5)
3	233(5.5)	233(5.5)	242(3)	223(9)	231(7)	225(8)	242 (3)	242(3)	250(1)
4	256(4.5)	256(4.5)	235(9)	272 (1.5)	272 (1.5)	269(3)	248 (7.5)	248 (7.5)	255(6)
5 L	183(5)	183(5)	183(5)	190(1.5)	189(3)	190(1.5)	181 (8)	181(8)	181(8)
9	249(3)	199(4.5)	199(4.5)	251(1)	198(6.5)	198(6.5)	250(2)	190(8.5)	190(8.5)
7	233(8)	233(8)	233(8)	240(2)	240(2)	240(2)	238(5)	238(5)	238(5)
x	250(1.5)	250(1.5)	209(3)	204(5.5)	204(5.5)	207(4)	195(8)	195(8)	195(8)
6	221(3)	191(6)	191(6)	230(1.5)	191(6)	201(4)	230(1.5)	179(8.5)	179(8.5)
10	224(2)	224(2)	224(2)	210(7.5)	193(9)	210 (7.5)	219(5)	219(5)	219(5)
. 66	273 (5)	273 (5)	273 (5)	262 (8)	. 262 (8)	. 262 (8)	287 (2)	287 (2)	287 (2)
100	219(3.5)	219(3.5)	219(3.5)	219(3.5)	219(3.5)	219(3.5)	213 (8)	213 (8)	213(8)
$\sum R_{Ci}$	381.5	430	516.5	471.5	548	626	440	509	577.5
$\sum R_{Ci}^2$	145542.25	184900	266772.25	222312.25	300304	391876	193600	259081	333506.25

Appendix B. Algorithmic Parameter Evaluation Results

yperparameter co.	mbinatio	ns A1.2.1	-A1.2.9.			
		We	orst	В	est	
		НА	NDS	HA	NDS	
	A1.2.1	Run 94	Run 89	Run 93	Run 79	

Run 62

 $\operatorname{Run}\,62$

 $\operatorname{Run}\,23$

Run 23

 $\operatorname{Run}\,23$

Run 21

Run 89

Run 89

Run 93

 $\operatorname{Run}\,36$

Run 48

 $\operatorname{Run}\,50$

Run 69

Run 69

Run 69

Run 36

Run 79

 $\operatorname{Run}\,79$

 $\operatorname{Run}\,79$

Run 79

Run 79

Run 79

Run 79

Run 79

A1.2.2

A1.2.3

A1.2.4

A1.2.5

A1.2.6

A1.2.7

A1.2.8

A1.2.9

Run 94

 $\operatorname{Run}\,94$

 $\operatorname{Run}\,30$

Run 72

Run 72

Run 72

Run 3

Run 3

Table B.8: The simulation runs that correspond to the best and worst approximation fronts (obtained by the MOOCEM) for the hyperarea performance indicator and the number of non-dominated solutions found for hy



Figure B.11: The average number of non-dominated solutions obtained for hyperparameter combinations A1.1–A1.9 for the respective simulation problems.

Table B.9: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A1.3.1–A1.3.9. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A1.3.1	A1.3.2	A1.3.3	A1.3.4	A1.3.5	A1.3.6	A1.3.7	A1.3.8	A1.3.9
1	31.78(9)	32.13(7)	32.18(6)	32.30(3)	32.51(1)	32.25(4)	32.04 (8)	32.39(2)	32.24(5)
2	33.43(7)	33.72(2.5)	33.72(2.5)	32.80(9)	33.76(1)	33.64(6)	33.38(8)	33.70(4.5)	33.70(4.5)
3	32.37(3)	32.22(6)	32.29(4)	32.21(8)	33.59(1)	32.71(2)	32.22(7)	32.15(9)	32.28(5)
4	33.50(2.5)	33.50(2.5)	33.26(9)	33.42(6)	33.60(1)	33.27(8)	33.47(4.5)	33.47(4.5)	33.29(7)
5	33.42(7.5)	33.42(7.5)	33.41(9)	33.50(1)	33.45(5)	33.45(6)	33.45(2.5)	33.45(2.5)	33.45(4)
6	33.25(4)	33.34(2)	33.09(6)	33.09(7)	33.33(3)	32.59(9)	33.20(5)	33.37(1)	32.93(8)
7	32.22(5)	32.22(5)	32.22(5)	32.40(2.5)	33.50(1)	32.40(2.5)	32.12(8)	32.12(8)	32.12(8)
8	33.57(6)	33.57(6)	33.57(6)	33.56(8.5)	33.59(4)	33.56(8.5)	33.62(2)	33.62(2)	33.62(2)
9	33.01(6)	33.19(3)	33.20(2)	33.02(5)	33.34(1)	32.76(9)	32.92(8)	32.99(7)	33.03(4)
10	33.65(2)	33.58(5)	33.62(3)	33.57(6)	33.53(9)	33.67(1)	33.54(8)	33.56(7)	33.62(4)
÷	÷	÷	÷	÷	÷	÷	÷	÷	:
99	33.02(3.5)	33.02(3.5)	32.58(6)	32.55(7)	33.46(1)	33.06(2)	32.52(8.5)	32.52(8.5)	32.99(5)
100	33.22~(6.5)	33.22~(6.5)	33.29(5)	33.18(9)	33.33(3)	33.30(4)	$33.51\ (1.5)$	$33.51\ (1.5)$	33.18(8)
$\frac{\sum R_{Ci}}{\sum R_{Ci}^2}$	536.50 287 832.25	$525 \\ 275625$	508.50 258572.25	555 308 025	479 229 441	504.00 254016	485 235 225	$\begin{array}{c} 465.50 \\ 216690.25 \end{array}$	441.50 194922.25

Table B.10: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A1.3.1–A1.3.9. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A1.3.1	A1.3.2	A1.3.3	A1.3.4	A1.3.5	A1.3.6	A1.3.7	A1.3.8	A1.3.9
1	37(9)	55(4)	52(5)	46 (7)	59(2)	49 (6)	41 (8)	58(3)	65(1)
2	73(8)	105(5.5)	105(5.5)	63(9)	112(3)	107(4)	76(7)	117(1.5)	117(1.5)
3	78(3)	69(8)	76(4.5)	71(7)	111(1)	64(9)	81(2)	73(6)	76(4.5)
4	113(5.5)	113(5.5)	86(9)	115(4)	128(1)	88(8)	117(2.5)	117(2.5)	94 (7)
5	83(8.5)	83 (8.5)	105(1)	95(3.5)	94(5)	98(2)	87(6.5)	87(6.5)	95(3.5)
6	71(3)	53(9)	57(7.5)	72(2)	69(4)	57(7.5)	79(1)	65(5)	62(6)
7	68(5)	68(5)	68(5)	74(2.5)	92(1)	74(2.5)	63(8)	63(8)	63(8)
8	126(2)	126(2)	126(2)	108(7.5)	79(9)	108(7.5)	109(5)	109(5)	109(5)
9	64(3)	51(9)	62(4)	55(8)	87 (1)	60(5)	59(6)	56(7)	71(2)
10	107(5)	106~(6)	94(9)	111(3)	120(1)	104(7)	113(2)	108(4)	99(8)
÷	÷	÷	:	÷	:	:	÷	÷	:
99	98(1.5)	98(1.5)	97(3)	87 (7)	96(4)	94 (6)	86(8.5)	86(8.5)	95(5)
100	92 (4.5)	92 (4.5)	71 (8)	106 (1)	70 (9)	80 (6)	93 (2.5)	93 (2.5)	77 (7)
$\frac{\sum R_{Ci}}{\sum R_{Ci}^2}$	$508 \\ 258064$	$566 \\ 320356$	568.5 323192.25	459 210 681	515.5 265740	548 300 304	$404.5 \\ 163620$	$450 \\ 202500$	480.5 230 880.25

Table B.11: The simulation runs that correspond to the best and worst approximation fronts (obtained by the MOOCEM) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A1.3.1–A1.3.9.

	We	orst	Be	est
	HA	NDS	HA	NDS
A1.3.1	Run 75	Run 1	Run 50	Run 50
A1.3.2	Run 75	Run 75	Run 2	Run 18
A1.3.3	Run 75	Run 75	Run 2	Run 18
A1.3.4	Run 75	$\operatorname{Run}27$	Run 92	Run 50
A1.3.5	Run 79	Run 98	Run 2	Run 58
A1.3.6	Run 75	Run 75	Run 10	Run 18
A1.3.7	$\operatorname{Run} 27$	Run 1	Run 60	Run 18
A1.3.8	Run 75	Run 75	Run 60	Run 18
A1.3.9	Run 75	Run 75	Run 2	Run 18

Table B.12: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A1.4.1–A1.4.9. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A1.4.1	A1.4.2	A1.4.3	A1.4.4	A1.4.5	A1.4.6	A1.4.7	A1.4.8	A1.4.9
1	22.10 (2)	21.85(6)	21.92(4)	20.24 (8)	20.24 (8)	20.24 (8)	21.86(5)	22.16 (1)	22.03(3)
2	21.96(3)	21.96(4)	22.08 (1)	21.33(8)	21.33(8)	21.33(8)	21.99 (2)	21.93(5)	21.91(6)
3	21.29(5)	21.62(3)	21.72(2)	21.13(7)	21.13(7)	21.13(7)	21.05(9)	21.42(4)	22.07(1)
4	21.98(3)	21.96(4)	21.65(5)	20.68(8)	20.68(8)	20.68(8)	22.10 (1)	22.07(2)	21.60(6)
5	22.14(2)	21.78(6)	22.15(1)	21.20(8)	21.20(8)	21.20 (8)	22.05(3)	21.82(5)	22.03(4)
6	21.36(6)	21.98(4)	22.08(2)	21.09(8)	21.09(8)	21.09(8)	21.53(5)	22.17(1)	22.07(3)
7	21.71(4)	21.73(2)	21.66(5)	20.94(8)	20.94(8)	20.94(8)	21.57(6)	21.88(1)	21.71(3)
8	21.92(4)	21.91(5)	21.93(3)	21.22(8)	21.22(8)	21.22(8)	21.86(6)	22.05(2)	22.14(1)
9	22.05(1)	21.99(3)	21.89(5)	20.64(8)	20.64(8)	20.64(8)	21.99(2)	21.89(4)	21.83(6)
10	21.72(6)	22.06(1)	21.87(5)	21.03(8)	21.03(8)	21.03(8)	21.95(2)	21.93(4)	21.93(3)
÷	÷	÷	÷	÷	÷	÷	÷	÷	÷
99	21.89(3)	21.73(4)	22.03(2)	20.89(8.5)	20.89(8.5)	21.37(7)	21.72(5)	21.71 (6)	22.07(1)
100	22.03(4)	22.15(1)	21.90(6)	20.83(7.5)	20.83(7.5)	20.80(9)	22.12(3)	22.15(2)	22.00(5)
$\sum R_{Ci}$	371	359	347	799.5	789	799.5	350	373	312
$\sum R_{Ci}^2$	137641	128881	120409	639200.25	622521	639200.25	122500	139129	97344

Table B.13: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A1.4.1–A1.4.9. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A1.4.1	A1.4.2	A1.4.3	A1.4.4	A1.4.5	A1.4.6	A1.4.7	A1.4.8	A1.4.9
1	61 (4)	50(6)	57(5)	42 (8)	42 (8)	42 (8)	63(3)	69(2)	70 (1)
2	64(2.5)	60(5.5)	63(4)	46 (8)	46 (8)	46 (8)	64(2.5)	65(1)	60(5.5)
3	60(1)	59(2.5)	59(2.5)	51(7)	51(7)	51(7)	58(4)	56(5)	47 (9)
4	66(3)	71(1)	63(4)	53(7.5)	53(7.5)	53(7.5)	67(2)	53(7.5)	57(5)
5	59(6)	69(1)	61(4)	49 (8)	49 (8)	49 (8)	68(2.5)	60(5)	68(2.5)
6	64(4.5)	82 (1)	74(2)	40 (8)	40 (8)	40 (8)	52(6)	64(4.5)	71(3)
7	50(4)	55(3)	49(6.5)	49(6.5)	49(6.5)	49(6.5)	60(1)	48 (9)	56(2)
8	80(2)	68(6)	84 (1)	67(8)	67(8)	67(8)	75 (4)	75(4)	75(4)
9	64(3)	65(1.5)	54 (6)	39 (8)	39(8)	39 (8)	62(4)	65(1.5)	59(5)
10	63(4.5)	66(3)	56(6)	47 (8)	47 (8)	47 (8)	68(1)	67(2)	63(4.5)
:	:	:	:	:	:	:	:	:	:
99	61 (4)	58 (5)		57 (7)	57 (7)	57 (7)	69 (3)	56 (9)	. 73 (1)
100	56 (6)	58(5)	65(2)	46 (85)	46 (8 5)	50(7)	60(35)	60(35)	67(1)
100	30 (0)	38 (3)	05 (2)	40 (8.3)	40 (0.0)	50 (7)	00 (3.5)	00 (3.3)	07 (1)
$\sum R_{Ci}$	368	384.5	393.5	744.5	735	762.5	386	365	361
$\sum R_{Ci}^2$	135424	147840.25	154842.25	554280.25	540225	581406.25	148996	133225	130321

Table B.14: The simulation runs that correspond to the best and worst approximation fronts (obtained by the MOOCEM) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A1.4.1–A1.4.9.

	We	orst	Be	est
	HA	NDS	HA	NDS
A1.4.1	Run 75	Run 55	Run 37	Run 92
A1.4.2	Run 75	Run 76	Run 37	Run 49
A1.4.3	$\operatorname{Run}22$	Run 91	$\operatorname{Run}35$	Run 8
A1.4.4	Run 89	$\operatorname{Run}35$	Run 75	Run 8
A1.4.5	$\operatorname{Run}65$	Run 44	Run 75	Run 8
A1.4.6	Run 88	Run 44	Run 60	Run 8
A1.4.7	Run 75	$\operatorname{Run} 82$	Run 88	Run 37
A1.4.8	Run 32	$\operatorname{Run}58$	$\operatorname{Run}62$	Run 40
A1.4.9	Run 47	Run 58	Run 95	Run 90

Table B.15: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A1.5.1–A1.5.9. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A1.5.1	A1.5.2	A1.5.3	A1.5.4	A1.5.5	A1.5.6	A1.5.7	A1.5.8	A1.5.9
1	203.81 (4)	206.29(3)	206.33(2)	206.68(1)	202.53(7)	203.7(5)	202.38 (8)	203.29(6)	201.28 (9)
2	202.7(5)	204.25(1)	203.57(3)	203.63(2)	201.16(7)	200.09(9)	203.38(4)	201.25(6)	200.59(8)
3	202.19(6)	203.39(4)	201.42(8)	202.55(5)	200.46(9)	204.71(1)	201.71(7)	204.07(3)	204.47(2)
4	200.24(8)	201.65(4)	195.51(9)	201.5(6)	204.48(1)	200.68(7)	201.52(5)	201.65(3)	204.3(2)
5	194.67(9)	201.67(4)	200.73(5)	195.99(8)	202.68(3)	203.63(2)	199.85(7)	200.47(6)	204.46(1)
6	200.89(7)	202.62(1)	201.86(2)	196.34(9)	201.84(3)	201.78(4)	197.98(8)	201.08(6)	201.31(5)
7	203.06(3)	205.08(1)	202.76(4)	201.71(5)	201.08(9)	201.69(6)	204.74(2)	201.14(7)	201.09(8)
8	199.21(9)	201.94(6)	201.72(7)	204.56(1)	201.37(8)	203.18(2)	202.27(4)	202.98(3)	202.1(5)
9	196.32(7)	199.6(4)	202.13(1)	193.71(8)	201.15(2)	199.49(5)	191.98(9)	199.06(6)	200.25(3)
10	203.4(8)	204.09(4)	204.1(3)	204.44(2)	202.61 (9)	203.96(6)	205.27(1)	204.06(5)	203.87(7)
÷	÷	÷	÷	÷	÷	÷	÷	÷	:
99	200.86(9)	202.16(7)	203.53(2)	202.4~(6)	202.65(4)	201.5(8)	202.52(5)	205.8(1)	203.28(3)
100	202.61~(6)	202.26(9)	203.17(3)	202.77(5)	202.47(7)	202.94(4)	202.32(8)	203.8(2)	205.31(1)
$\sum R_{Ci}$	585	505	426	563	527	415	533	518	428
$\sum R_{Ci}^2$	342225	255025	181476	316969	277729	172225	284089	268324	183184

Table B.16: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A1.5.1–A1.5.9. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A1.5.1	A1.5.2	A1.5.3	A1.5.4	A1.5.5	A1.5.6	A1.5.7	A1.5.8	A1.5.9
1	63(3)	58(4)	50(8)	44 (9)	64(2)	70 (1)	53(7)	54(6)	56(5)
2	65(1)	61(4)	63(2)	50(9)	54(7)	57(6)	62(3)	58(5)	53(8)
3	58(8)	60(6.5)	63(2.5)	54(9)	63(2.5)	68(1)	60(6.5)	62(4.5)	62(4.5)
4	50(8)	57(3.5)	54(6)	49(9)	52(7)	55(5)	69(1)	64(2)	57(3.5)
5	61(4)	66(1)	48(9)	64(2)	51(6)	62(3)	49 (8)	60(5)	50(7)
6	55(4.5)	62(1)	51(7.5)	43(9)	61(2)	55(4.5)	54(6)	59(3)	51(7.5)
7	46(9)	60(2)	48 (7)	52(4)	49(6)	51(5)	54(3)	47 (8)	65(1)
8	51(7)	66(1)	63(2)	59(4)	48(8)	56(5)	46(9)	52(6)	61(3)
9	56(5.5)	58(3)	63(1)	56(5.5)	54(8.5)	60(2)	57(4)	55(7)	54(8.5)
10	55(8)	60~(6)	62 (4.5)	63(3)	66(1.5)	66(1.5)	55(8)	62 (4.5)	55(8)
÷	:	÷	÷	÷	÷	:	÷	÷	:
99	64(2)	57(6.5)	47 (9)	61(4)	57(6.5)	66(1)	62(3)	50(8)	59(5)
100	59(4.5)	61 (3)	57 (6.5)	52 (9)	57 (6.5)	62(2)	66(1)	54 (8)	59(4.5)
$\frac{\sum R_{Ci}}{\sum R_{Ci}^2}$	511.5 261 632	479 229 441	467 218 089	555.5 308 580	510 260100	460 211 600	532.5 283 556.25	499 249 001	485.5 235 710.25

Table B.17: The simulation runs that correspond to the best and worst approximation fronts (obtained by the MOOCEM) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A1.5.1–A1.5.9.

	We	orst	Best		
	HA	NDS	HA	NDS	
A1.5.1	Run 75	Run 24	Run 19	Run 89	
A1.5.2	Run 75	Run 83	Run 19	Run 29	
A1.5.3	Run 63	Run 60	$\operatorname{Run}43$	Run 57	
A1.5.4	Run 71	$\operatorname{Run}56$	Run 1	Run 50	
A1.5.5	Run 83	Run 29	Run 49	Run 39	
A1.5.6	$\operatorname{Run}75$	$\operatorname{Run}75$	Run 68	Run 77	
A1.5.7	Run 9	Run 93	$\operatorname{Run}35$	$\operatorname{Run}4$	
A1.5.8	Run 75	$\operatorname{Run}28$	$\operatorname{Run}32$	Run 77	
A1.5.9	Run 93	$\operatorname{Run}52$	Run 72	Run 35	

B.2 NSGA-II

This section presents the results for the hyperparameter study conducted for the NSGA-II.

B.2.1 Open mine problem

Tables B.18 and B.19 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–40 for hyperparameter combinations A2.1.1– A2.1.24. Figures B.12 and B.14 combines the approximation fronts for which the runs where found to be the same, whereas Figure B.13 illustrate the worst and best approximation fronts found (in terms of hyperareas and number non-dominated solutions), for the respective hyperparameter combinations A2.1.1–A2.1.24, all corresponding to the runs given in Table B.20.



 $Hyperarea\ and\ non-dominated\ solutions$

Figure B.12: The best and worst best approximation fronts for hyperparameter combinations A2.1.1–A2.1.6, A2.1.8, A2.1.12–A2.1.14.

APPENDIX B. ALGORITHMIC PARAMETER EVALUATION RESULTS



Figure B.13: The best and worst best approximation fronts for hyperparameter combinations A2.1.7, A2.1.9–A2.1.11.

B.2.2 (s, S) Inventory problem

Tables B.23 and B.24 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A2.2.1– A2.2.24. Figures B.15–B.19 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A2.2.1–A2.2.24 that correspond to the runs given in Table B.25.


Hyperarea and non-dominated solutions

Figure B.14: The best and worst best approximation fronts for hyperparameter combinations A2.2.16–A2.2.24.





Figure B.15: The best and worst best approximation fronts for hyperparameter combinations A2.2.1–A2.2.5.



Figure B.16: The best and worst best approximation fronts for hyperparameter combinations A2.2.6–A2.2.10.



Figure B.17: The best and worst best approximation fronts for hyperparameter combinations A2.2.11–A2.2.15.



Figure B.18: The best and worst best approximation fronts for hyperparameter combinations A2.2.16–A2.2.20.





Figure B.19: The best and worst best approximation fronts for hyperparameter combinations A2.2.21–A2.2.24.

B.2.3 Buffer allocation problem: five machines

Tables B.28 and B.29 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A2.3.1– A2.3.24. Figures B.20–B.24 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A2.3.1–A2.3.24 that correspond to the runs given in Table B.30.



Figure B.20: The best and worst best approximation fronts for hyperparameter combinations A2.3.1–A2.3.5.





Figure B.21: The best and worst best approximation fronts for hyperparameter combinations A2.3.6–A2.3.10.



Figure B.22: The best and worst best approximation fronts for hyperparameter combinations A2.3.11–A2.3.15.





Figure B.23: The best and worst best approximation fronts for hyperparameter combinations A2.3.16–A2.3.20.



Figure B.24: The best and worst best approximation fronts for hyperparameter combinations A2.3.21–A2.3.24.

B.2.4 Buffer allocation problem: 10 machines

Tables B.33 and B.34 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A2.4.1– A2.4.24. Figures B.25–B.29 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A2.4.1–A2.4.24 that correspond to the runs given in Table B.35.



Figure B.25: The best and worst best approximation fronts for hyperparameter combinations A2.4.1–A2.4.5.



Figure B.26: The best and worst best approximation fronts for hyperparameter combinations A2.4.6–A2.4.10.

APPENDIX B. Algorithmic Parameter Evaluation Results



Figure B.27: The best and worst best approximation fronts for hyperparameter combinations A2.4.11–A2.4.15.



Figure B.28: The best and worst best approximation fronts for hyperparameter combinations A2.4.16–A2.4.20.

APPENDIX B. ALGORITHMIC PARAMETER EVALUATION RESULTS



Figure B.29: The best and worst best approximation fronts for hyperparameter combinations A2.4.21–A2.4.24.

B.2.5 Non-linear buffer allocation problem: 16 machines

Tables B.38 and B.39 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A2.5.1– A2.5.24. Figures B.30–B.34 illustrate the worst and best approximation fronts found, in terms of hyperareas and number non-dominated solutions, for the respective hyperparameter combinations A2.5.1–A2.5.24 that correspond to the runs given in Table B.40. In Figure B.35 the average number of non-dominated solutions obtained for hyperparameter combinations A2.1–A2.24 is illustrated for the respective simulation problems.



Figure B.30: The best and worst best approximation fronts for hyperparameter combinations A2.5.1–A2.5.5.





Figure B.31: The best and worst best approximation fronts for hyperparameter combinations A2.5.6–A2.5.10.



Figure B.32: The best and worst best approximation fronts for hyperparameter combinations A2.5.11–A2.5.15.





Figure B.33: The best and worst best approximation fronts for hyperparameter combinations A2.5.16–A2.5.20.



Figure B.34: The best and worst best approximation fronts for hyperparameter combinations A2.5.21–A2.5.24.

0 0 0 0 0 0 0 0 0			A2.1.2		A2.1.4	A2.1.5		A2.1.6	A2	.1.19	A2.1.2	0	A2.1.21	A2.1.22	A	2.1.23	A2.1.24
0.3	630.37 (2) $86.77 (16)$	12412.64(12626.15($\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\frac{48}{15}$ (13) .15 (8)	$\frac{10415.26}{11843.21} \left(\begin{array}{c} 0 \\ 0 \end{array} \right)$	(1) 11 188 (1) 8423.7	3.10(8) .	$8861.30\ (173612.05\ (243612.05)$	7) 10 4) 10	892.46(9) 110.91(13	(10.456.)	.37(10) 44(11) 8	$egin{array}{c} 9\ 342.55\ (14)\ 8\ 528.62\ (19) \end{array}$	6603.35 12680.9		$\begin{array}{c} 912.77 \ (16) \\ 2 \ 401.58 \ (7) \end{array}$	$) 6 \; 991.79 \; (2)$ $) 13 \; 181.44 \; ($
).).	15.35 (13)	9 668.67 (1	0) 10182	.95 (6)	9489.43 (12)	10 537	.10 (3)	9522.34 (11	1) 74	70.06 (22)	8 774.4	(3(15))	10466.60 (4)	6 957.33	3 (24) 10	0 922.96 (1)	7 295.40 (2
4 6	485.04(1.5)	72685.30	6) 13260	.07 (10)	13240.51(1	(2) 13683	1.96 (7)	13 658.39 (8	8) 13 7.13	205.74 (13	3) 12 081.	01(22)	12255.68 (20	() 12395.1	8 (19) 1: . (81) 2:	2772.71 (1)	7) 11419.97 (
א מ ס ס	09.04 (11) 88.53 (4)	0 803 30 (3	0 00035	(14) 10 (14) 10	9 000.14 (9) 7 577 59 (18	9 1923	94 (o) 56 (8)	1)/1.0//11 (0)/13//0)	T) 60	02.17 (10)	7 148 8	50 (24) 1 (10)	9	6 700 57	(7 (03) 7 7 (03) 7	200.00 (7) 016 77 (15)	0 0 0 0 0 1 0 1 0 0 0 0 0 0 0 0 0 0 0 0
101	533.44 (15)	11 711.32 (6) 8741.6	(12) 65	11 939.68 (5	12.278	.55 (4)	0.762.40 (1	13) 94	32.93 (18)	10.852	26 (12) 9	9 409.07 (19)	11 603.1	6 (6) 10	0.867.03 (1	1) 8585.68 (2
13	928.72 (7)	13 241.53 (12) 13 195	.36 (13)	14 485.04 (1.	.5) 11 736.	.77 (23)	13 106.97 (1	15) 13.	281.77 (10	1) 12 667.	80 (18)	11499.04 (24) 13661.5	53 (8) 1:	3 133.24 (1	4) 12 450.48 (
11	258.29 (5)	9 970.75 (1	2) 10680	.67 (10)	12 402.05 (2	() 12 069	0.08 (4)	10756.01 (8)	3) 95	60.30 (16)	10 047.	21 (11) 2	8635.27(23)	8 824.23	3 (22) 9	182.61 (20)	9 858.45 (1)
12	878.16(2)	11 752.39 (4) 9448.0	35(16)	11 504.02 (6	i) 10460	0.52(11)	11 331.51 (7	7) 95	47.95(15)	11 122.	.05 (8)	8 391.34 (20)	9604.07	7 (14) 8	017.31 (22)	9 158.01 (1
;		. 110 0 7		. (00) 11											1017 01		
11	350.58(5) 789.72(7)	9 600.27 (2	14) 11 792 11 792 1) 12 083	(51 (5))	$14\ 059.460\ (9)$ 11 128.37 (1)	(1) 13 $(0.43)2) 13 (0.92)$		12778.12(1)	$100 \dots 12$	750.59 (18 87.24 (20)	5) 14485. 9355.4	.04 (2) 7 (22) &	123/4.41 (20 8854.36 (24)) 13 083.0 10 034.0	1 (11) 1 1 (11) 1 1 (11) 1	3257.19(1)) 11 901.04 (
i 289		423	473		303	391		522	595	5	543		658	594	5	17	656
. 83	521	178929	22372	6	91809	15288	31	272484	354	4025	294.84	6	432964	352836	5	66772	430336
Run	A2.1.1	A2.1.2	A2.1.3	A2.1.4	A2.1.5 A	A2.1.6	A2.1.7	A2.1.8	A2.1.	.17 A2.	1.18 A	12.1.19	A2.1.20 A	2.1.21	A2.1.22	A2.1.23	A2.1.24
	32(2)	31(3.5)	24(13.5)	26 (12)	28 (8.5) 2	33 (17.5)	23 (17.5)	21 (21.5)	30 (5	5.5) 23 (17.5) 2	8 (8.5)	27 (10.5) 2	3 (17.5)	21 (21.5)	23 (17.5)	18 (24)
~	27(12.5)	32 (6)	30(9)	30(9)	21 (21.5) 1	7 (24)	34(2)	34(2)	19 (2	33) 22 ((19.5) 2	5(15.5)	29 (11) 2	2(19.5)	32(6)	32(6)	33(4)
~ ·	24(13)	25(10)	26(6.5)	25(10)	27 (3) 2	24(13)	21 (21.5)	26(6.5)	21 (2	21.5) 25 ((10) 2	1(21.5)	23 (16.5) 2	7 (3)	20(24)	28(1)	21(21.5)
বা	36(1.5)	34(7.5)	33(13)	33(13)	34(7.5) 3	34 (7.5)	33(13)	36(1.5)	33 (1	(3) 31 ((20) 3	(13)	31(20) 3	1(20)	31(20)	32(17)	29(23)
<u>د</u> م	24(7.5)	21 (17.5)	23 (12)	23 (12)	24(7.5) 3	0 (I)	29(2)	22 (15)	18 (2	22) 24 ((7.5) 2	(1.17.5)	15(24) 2	3(12)	20(20)	24(7.5)	25(4.5)
10	20 (0) 20 (10)	20(5)	27 (1.5) 09 (01 E)	20 (18.5)	- 24 (9) 2 91 (9 E) 5	(a) (a) (a)	22 (13) 22 (15)	20 (0) 21 (0 E)	1) 22	[3) I9 (E) A0 (1 (0.1Z)	((23.5)	20 (18.5) 2	0 (13.5) 6 (17.5)	17 (23.5) 20 (e E)	0.21 (15.5) 07 (15 E)	24 (9) 99 (94)
- 00	26 (12) 35 (5.5)	33 (12.5)	23 (21.5) 33 (12.5)	36 (2) 36 (2)	30 (22) 3	3 (12.5)	34 (8.5)	32 (17.5)	35 (5	(.5) 20 (.5) 32 ((12) 2 (17.5) 3	$\frac{1}{3}$ (12.5)	32 (17.5) 2	9 (24)	23 (0.0) 34 (8.5)	33 (12.5)	$\frac{22}{31}$ (20)
6	28 (6)	25(14)	27 (8.5)	31(2)	30 (4) 2	7 (8.5)	28 (6)	31 (2)	21 (2	(4) 25 (14) 2	5(14)	25(14) 2	2(23)	23(21)	23 (21)	25(14)
10	32(2)	30(3.5)	24(15.5)	29(6)	26(11.5) 2	(9) (6)	29(6)	30(3.5)	$\dots 21(2$	(1.5) 26	(11.5) 2	4(15.5)	28 (8.5) 2	3(19)	24(15.5)	21(21.5)	23(19)
- 5				· 1) · 10		1 . t . (·, •0	,	ہ ب	. I		í	1		
40 10	36(3) 30(7.5)	33 (15) 24 (22.5)	30(22.5) 31(4.5)	35(7.5) 28(14)	34(11.5)3 35(1)2	(17.5)	36(3) 30(7.5)	36(3) 27(17)	31 (1	(19.5) 30 (4) 31 ((22.5) 3 (4.5) 2	(4.7.6)	36(3) 3 24(22.5) 2	$\begin{array}{c} 1 & (19.5) \\ 4 & (22.5) \end{array}$	$\frac{34}{26} (11.5)$	33 (15) 30 (7.5)	30(22.5) 24(22.5)
		1				1							1	, 1 , 1			
N N N	71 285	446.5 100.369.95	479	313	391 5	531.5 	310.5	321.5	685.5	716	ъ. ъ	78.5	545 6	57.5	580	509.5	634.5

Table B.18: A preview of the hyperareas (and corresponding ranks) obtained for run 1-40 for hyperparameter combinations A2.1.1-A2.1.24. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

238

	We	orst	Be	est
	HA	NDS	HA	NDS
A2.1.1	Run 27	Run 27	Run 4	Run 4
A2.1.2	$\operatorname{Run}5$	$\operatorname{Run}5$	$\operatorname{Run}16$	Run 16
A2.1.3	$\operatorname{Run}21$	$\operatorname{Run}21$	$\mathrm{Run}\ 18$	Run 18
A2.1.4	$\operatorname{Run}31$	$\operatorname{Run}31$	Run 8	Run 8
A2.1.5	$\operatorname{Run}37$	$\operatorname{Run}37$	Run 16	Run 16
A2.1.6	$\operatorname{Run} 2$	$\operatorname{Run} 2$	Run 4	Run 4
A2.1.7	$\operatorname{Run}21$	$\operatorname{Run}21$	$\operatorname{Run}20$	Run 16
A2.1.8	$\mathrm{Run}\ 27$	$\mathrm{Run}\ 27$	Run 4	Run 4
A2.1.9	Run 1	Run 6	Run 19	Run 19
A2.1.10	$\operatorname{Run}21$	$\operatorname{Run}34$	Run 8	Run 8
A2.1.11	$\mathrm{Run}\ 27$	$\mathrm{Run}\ 27$	Run 16	Run 36
A2.1.12	$\operatorname{Run}34$	$\operatorname{Run}34$	Run 19	Run 19
A2.1.13	$\mathrm{Run}\ 27$	$\mathrm{Run}\ 27$	$\operatorname{Run}16$	Run 16
A2.1.14	$\operatorname{Run}5$	$\operatorname{Run}5$	Run 39	Run 39
A2.1.15	$\mathrm{Run}\ 27$	$\mathrm{Run}\ 27$	$\operatorname{Run}15$	Run 15
A2.1.16	$\operatorname{Run}32$	$\operatorname{Run}32$	Run 39	Run 39
A2.1.17	$\operatorname{Run}21$	$\operatorname{Run}21$	$\operatorname{Run}16$	Run 16
A2.1.18	$\operatorname{Run}32$	$\operatorname{Run}32$	$\operatorname{Run}36$	Run 36
A2.1.19	Run 6	Run 6	Run 19	Run 19
A2.1.20	$\operatorname{Run}5$	$\operatorname{Run}5$	Run 16	Run 16
A2.1.21	$\mathrm{Run}\ 27$	$\mathrm{Run}\ 27$	Run 19	Run 19
A2.1.22	$\mathrm{Run}\ 27$	$\mathrm{Run}\ 27$	$\operatorname{Run}25$	$\operatorname{Run}25$
A2.1.23	$\mathrm{Run}\ 27$	$\mathrm{Run}\ 27$	$\operatorname{Run}16$	Run 16
A2.1.24	Run 1	Run 1	Run 16	Run 16

Table B.20: The simulation runs that correspond to the best and worst approximation fronts (obtained by the NSGA-II) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A2.1.1–A2.1.24.

e multiple comparisons based on the hyperareas of the approximation	
for th	
$c \ test$	
st ho	
nyi pc	
Neme	.1.24.
v the	.1–A2
ied by	A2.1
obtain	ations
alues	ombin
p-v	eter c
djuste	aram
The a	yperp
21:	the h_{i}
е В.	s for
Tabl	front

$\begin{array}{cccccccccccccccccccccccccccccccccccc$											OT.T						
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1	1	1	1	1	0	0	0	0	0	0	0	0.02	0	0	0.09	0
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1	1	1	1	1	0.03	0.01	0	0.08	0.01	0	1	1	0.06	1	1	0.06
1 0.15 1 1 1 0.36 0	1	1	0.39	0.46	1	0.52	0.18	0.01	1	0.13	0.07	1	1	0.95	1	1	1
1 1 0.36 0	1	1	1	1	1	0	0	0	0	0	0	0	1	0.15	0.04	0	1
0.36	1	1	1	1	1	0	0	0	0.01	0	0	0.35	1	0.01	0.37	1	0.01
	0.18	1	0.02	0.02	1	1	1	0.17	1	1	1	1	1	1	1	1	1
	1	1	1	1	1	0	0	0	0	0	0	0	0.11	0	0	0.48	0
		1	1	1	1	0	0	0	0	0	0	0.001	0.05	0	0	0.25	0
			1	1	1	0.001	0	0	0	0	0	0.09	1	0	0.1	1	0
				1	1	0	0	0	0	0	0	0	0.01	0	0	0.03	0
					1	0	0	0	0	0	0	0	0.01	0	0	0.04	0
						0	0	0	0	0	0	0.03	0.63	0	0.03	1	0
							1	1	1	1	1	1	1	1	1	1	1
								1	1	1	1	1	1	1	1	1	1
									1	1	1	1	0.54	1	1	0.12	1
										1	1	1	1	1	1	1	1
											1	1	1	1	1	1	1
												1	1	1	1	0.79	1
													1	1	1	1	1
														1	1	1	1
															1	1	1
																1	1
																	1

Stellenbosch University https://scholar.sun.ac.za

Table B.22: The adjusted p-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A2.1.1–A2.1.24.

A	2.1.2 A2.	.1.3 A2.1	1.4 1.4.					AT.T.74		AL.L.AN	A	FT-T-7C	01.1.24	0T.1.2A	11.1.24	01.1.24	A				07.1.70	14.1.40
1.1.1	1 0.5	596 1	1	0.027	1	1	1	1	1	1	0	0	0	0	0	0	0	0.01	0	0	0.11	0
12.1.2		1 1	1	1	1	1	1	1	1	1	0.19	0.07	0	0.25	0.04	0.01	1	1	0.23	1	1	0.82
V2.1.3		1	1	1	1	1	1	0.31	0.54	1	1	0.46	0.02	1	0.30	0.048	1	1	1	1	1	1
12.1.4			1	0.152	1	1	1	1	1	1	0	0	0	0	0	0	0.01	1	0.15	0.07	0	1
12.1.5				1	1	1	1	1	1	1	0.01	0	0	0.01	0	0	0.84	1	0.007	0.77	1	0.03
A2.1.6					0.13	0.25	1	0.01	0.02	1	1	1	0.55	1	1	0.95	1	1	1	1	1	1
12.1.7						1	1	1	1	1	0	0	0	0	0	0	0.006	0.06	0	0.01	0.46	0
12.1.8							1	1	1	1	0	0	0	0	0	0	0.013	0.11	0	0.01	0.82	0
12.1.9								1	1	1	0	0	0	0	0	0	0.35	1	0.002	0.32	1	0.01
2.1.10									1	1	0	0	0	0	0	0	0	0.01	0	0	0.05	0
2.1.11										1	0	0	0	0	0	0	0.001	0.01	0	0	0.09	0
2.1.12											0	0	0	0.001	0	0	0.17	1	0	0.15	1	0
2.1.13												1	1	1	1	1	1	1	1	1	1	1
2.1.14													1	1	1	1	1	1	1	1	1	1
2.1.15														1	1	1	1	1	1	1	0.16	1
2.1.16															1	1	1	1	1	1	1	1
2.1.17																1	1	1	1	1	1	1
2.1.18																	1	1	1	1	0.29	1
2.1.19																		1	1	1	1	1
2.1.20																			1	1	1	1
2.1.21																				1	1	1
2.1.22																					1	1
2.1.23																						1

$\begin{array}{c} 271 \ (8) \\ 251 \ (10.f \\ 221 \ (6) \\ 228 \ (5.5) \\ 228 \ (5.5) \\ 228 \ (10) \\ 2264 \ (14) \\ 2264 \ (14) \\ 2261 \ (5.5) \\ 2281 \ (16.5 \\ 238 \ (16) \ (16)$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{c} 7 \ (14) \\ 8 \ (12.5) \\ 1 \ (6) \\ 3 \ (4) \\ 5 \ (9) \end{array}$

Table B.23: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A2.2.1–A2.2.24. The ranks

	Wo	rst	Be	st
	HA	NDS	HA	NDS
A2.2.1	Run 17	Run 30	Run 7	Run 26
A2.2.2	Run 39	Run 60	Run 92	$\operatorname{Run} 7$
A2.2.3	$\operatorname{Run}72$	$\operatorname{Run}78$	Run 88	$\operatorname{Run}90$
A2.2.4	$\operatorname{Run}57$	$\operatorname{Run}23$	$\operatorname{Run}29$	$\operatorname{Run}76$
A2.2.5	$\operatorname{Run} 17$	$\operatorname{Run}65$	Run 62	$\operatorname{Run}68$
A2.2.6	Run 3	Run 8	Run 14	$\operatorname{Run}85$
A2.2.7	Run 3	Run 11	Run 79	$\operatorname{Run}26$
A2.2.8	Run 12	Run 1	Run 44	$\operatorname{Run}28$
A2.2.9	Run 12	Run 39	$\operatorname{Run}25$	$\mathrm{Run}\ 17$
A2.2.10	Run 57	Run 88	Run 62	Run 79
A2.2.11	$\operatorname{Run}72$	Run 11	Run 79	$\operatorname{Run}48$
A2.2.12	Run 3	$\operatorname{Run}22$	$\mathrm{Run}\ 100$	$\operatorname{Run}37$
A2.2.13	Run 3	$\operatorname{Run}44$	Run 9	$\operatorname{Run}16$
A2.2.14	Run 39	Run 44	Run 7	Run 79
A2.2.15	Run 83	$\operatorname{Run}65$	$\operatorname{Run}54$	$\operatorname{Run}43$
A2.2.16	$\operatorname{Run}57$	Run 60	$\operatorname{Run}15$	$\operatorname{Run}26$
A2.2.17	Run 3	$\operatorname{Run}44$	$\operatorname{Run}21$	$\operatorname{Run}94$
A2.2.18	$\mathrm{Run}\ 100$	$\operatorname{Run}44$	$\operatorname{Run}70$	$\operatorname{Run}71$
A2.2.19	Run 94	$\operatorname{Run}35$	$\operatorname{Run}74$	$\operatorname{Run}79$
A2.2.20	$\mathrm{Run}\ 12$	Run 38	Run 11	Run 71
A2.2.21	$\operatorname{Run} 17$	$\operatorname{Run}43$	Run 73	$\operatorname{Run}45$
A2.2.22	$\operatorname{Run}57$	Run 61	Run 63	$\mathrm{Run}\ 27$
A2.2.23	$\operatorname{Run}57$	$\operatorname{Run}29$	Run 36	$\operatorname{Run}50$
A2.2.24	Run 97	Run 89	Run 79	Run 24

Table B.25: The simulation runs that correspond to the best and worst approximation fronts (obtained by the NSGA-II) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A2.2.1–A2.2.24.

arisons based on the hyperareas of the approximation	
Nemenyi post hoc test for the multiple compa-	.2.24.
able B.26: The adjusted p-values obtained by the	onts for the hyperparameter combinations A2.2.1–A2

	A2.2.4	4 A2.2.5	A2.2.6	A2.2.7	42.2.8	A2.2.9 A	2.2.10 A	2.2.11 A	2.2.12 A	2.2.13 A	.2.2.14 A	v2.2.15 A	2.2.16 /	A2.2.17	A2.2.18	A2.2.19	42.2.20	A2.2.21	A2.2.22	A2.2.25	A2.2.24
	1	1	1	1	1	1	1	1	1	0.01	0	0.04	1	1	0	0	1	1	1	1	1
	1	1	1	1	1	1	1	1	0.01	1	1	1	1	1	1	1	1	1	1	1	1
	1	1	1	0.31	1	0.045	0.14	0.25	0	1	1	1	1	1	1	1	1	1	1	1	1
		1	1	1	1	1	1	1	1	0.11	0.04	0.41	1	1	0.02	0.03	1	1	1	1	1
			1	1	1	1	1	1	1	0.047	0.02	0.19	1	1	0.01	0.01	1	1	1	1	1
				1	1	1	1	1	0.045	1	0.94	1	1	1	0.44	0.72	1	1	1	1	1
					1	1	1	1	1	0	0	0.01	1	0.37	0	0	1	1	0.37	0.53	1
- 0 - 0 0 4 10 c						1	1	1	1	0.01	0	0.05	1	1	0	0	1	1	1	1	1
0 - 0 6 4 6 9							1	1	1	0	0	0	1	0.05	0	0	1	1	0.05	0.08	1
								1	1	0	0	0	1	0.17	0	0	1	1	0.17	0.26	1
									1	0	0	0.01	1	0.3	0	0	1	1	0.3	0.44	1
										0	0	0	0	0	0	0	0.01	0.67	0	0	0.77
											1	1	1	1	1	1	1	0.18	1	1	0.15
10 50												1	1	1	1	1	1	0.07	1	1	0.06
													1	1	1	1	1	0.63	1	1	0.55
														1	1	1	1	1	1	1	1
~															1	1	1	1		1	1
~																1	1	0.03	1	1	0.02
•																	1	0.049	1	1	0.04
0																		1	1	1	1
1																			1	1	1
5																				1	1
																					1

Stellenbosch University https://scholar.sun.ac.za

Stellenbosch University https://scholar.sun.ac.za

4	42.2.2	A2.2.3	A2.2.4	A2.2.5	A2.2.6 /	12.2.7	12.2.8	12.2.9 4	A2.2.10 /	12.2.11	A2.2.12	A2.2.13	A2.2.14	A2.2.15	A2.2.16	A2.2.17	A2.2.18	8 A2.2.16	A2.2.20) A2.2.21	A2.2.22	A2.2.23	A2.2.2
2.2.1	1	0.03	1	1	0.01	1	1	1	1	1	0.53	0	0	0	0	0	0	0	0	0	0	0	0
2.2.2		1	1	1	1	1	1	1	1	1	1	0	0	0	0	0	0	0	0	0	0	0	0
2.2.3			1	1	1	0.03	0	1	0	1	1	0	0	0	0	0	0	0	0	0	0	0	0
2.2.4				1	0.52	1	1	1	1	1	1	0	0	0	0	0	0	0	1	0.52	0	0	0
2.2.5					1	1	0.18	1	0.04	1	1	0	0	0	0	0	0	0	0	0	0	0	0
2.2.6						0.01	0	1	0	1	1	0	0	0	0	0	0	0.01	0	0	0	0	0
2.2.7							1	1	1	1	0.55	0	0	0	0	0	0	0	0	0	0	0	0
2.2.8								1	1	1	0.07	0	0	0	0	0	0	0	0	0	0	0	0
2.2.9									1	1	1	0	0	0	0	0	0	0	0	0	0	0	0
2.2.10										0.81	0.02	0	0	0	0	0	0	0	0	0	0	0	0
2.2.11											1	0	0	0	0	0	0	0	0	0	0	0	0
2.2.12												0	0	0	0	0	0	0	0	0	0	0	0
2.2.13													1	1	1	1	1	1	1	1	1	1	1
2.2.14														1	1	1	1	1	1	1	1	1	1
2.2.15															1	1	1	0.75	1	1	1	1	1
2.2.16																1	1	1	1	1	1	1	1
2.2.17																	1	0.66	1	1	1	1	1
2.2.18																		0.01	0.04	1	0.04	1	1
2.2.19																			1	1	1	1	0.13
2.2.20																				1	1	1	0.48
2.2.21																					1	1	1
2.2.22																						1	0.5
00.00																							

	Run	A2.2.1	A2.2.2	A2.2.3	A2.2.4	A2.2.	5 A2.2	2.6 A2.	.2.7 A2	2.2.8	A2.2.	17 A2.2	.18 A2.2	.19 A2	.2.20	A2.2.21	A2.2.22	A2.2.23	A2.2.24	1
	1 2 0	33.57 (13) 33.01 (13) 33.01 (13)) 33.63 (10)) 33.31 (8) 22 72 (15)	$\begin{array}{c} 33.54 \ (1) \\ 33.45 \ (12) \\ 32.45 \ (12) \\ 32.45 \ (14) \end{array}$	33.64 (8 32.97 (5	33.62 33.24 33.24	(17) 33.6 (19) 32.2 (18) 32.2	7 (14) 33. 6 (3) 32.9 7 (5) 32.9	67 (6) 33 95 (14) 32.	.63 (7) .25 (10) .11 (11)	32.79 33.3 ($\begin{pmatrix} (22) & 22 & (5) \\ (9) & 33.35 \\ (21) & 23 & 75 \\ (21)$	33.54) 33.55 3 (6) 32.23 7 (12) 22 20	$\begin{pmatrix} (12) & 33 \\ (22) & 32 \\ (0) & 32 \\ 32 & 32 \\ 33 & 33$	$ \begin{array}{c} 54 & (16) \\ 95 & (16) \\ 32 & (32) \\ 33 & (32) \\ 34 & (32) \\ 35 & (32)$	32.82 (21 33.54 (1) 33.72 (16	$ \begin{array}{c} 32.75 \\ 32.75 \\ 32.75 \\ 32.77 $	4) 33.26 (20 8) 33.46 (2) 22 20 (8)	$ \begin{array}{c} 33.38 & (19) \\ 33.37 & (7) \\ 22 & (7) \\ 22 & (7) \\ \end{array} $	
	o 4	32.10 (24) 33.61 (5)	33.65 (3)	33.65(14)	33.47 (2	.2) 33.65 33) 33.65	(12) 33.4	4 (0) 52. 8 (2) 33.	5 (17) 33.	.54(11)	33.46	(18) 33.55	5 (9) 33.4	(22) 33 (22) 33	53 (11)	32.42 (10 33.44 (20) 33.44 (1) 33.44 (1)	9) 33.64 (4)	33.52 (1)	
	ഗയ	32.3(21) 32.03(24)	32.21 (24) 33 (10)	33.42(3)	33.3 (8) 32.14 (2	32.97 1) 33.31	(22) 33.2 (12) 33.4	(7 (6) 33 32.	$39\ (13)\ 33$ $74\ (23)\ 33$	(23 (2) + (13))	33.31	(12) 32.6. (9) 32.45	l (20) 33.45 3 (17) 32.44	8 (5) 32 L (16) 32	.69(19)	33.42 (7) 32.33 (19	33.34 (1 32.18 (2	1) 33.35 (10 2) 32.4 (18)) 33.66 (1) 33.4 (4)	
		33.45 (5)	32.56 (20)	32.26 (22)	33.38 (2	(1) 33.32	(18) 33.4	7 (24) 33.	34 (8) 33.	.37 (7)	33.48	(3) 33.01	7 (14) 33.04	(15) 33	01 (16)	32.7 (19)	32.95 (1)	7) 33.56 (1)	33.49 (2)	
	¢ 6	33.37 (23, 32.94 (16)) 33.08 (1) 33.02 (13)	33.42 (24) 33.23 (1)	32.26 (2)	(2) 33.25	(12) 33.5	2(21)33.	01 (11) 33 53 (23) 32.	(0z) / c (6) 69:	33.34	(12) 33.3((18) 32.62	3 (10) 33.04 3 (20) 32.14	(4) 33 (24) 33	232 (13)	33.33 (10 33.33 (6)) 33.04 (9 32.97 (1) 33.40 (18 5) 33.36 (5)) 33.43 (3) 33.43 (3)	
	10	33.6(2)	33.44 (7)	33.36(1)	33.44(1	2) 32.22	(17) 33.3	4 (15) 33	$36(8)^{32}$.73 (22)	33.41	(11) 33.4 $($	$3(9)^{33.32}$	(19) 33	.53(5)	33.42 (10) 32.4 (23	$)^{\circ}$ 33.37 $(14$) 33.56 (4)	
											•••									
	$\frac{99}{100}$	$32.37 \\ 33.33 \\ (11)$	$\stackrel{.}{1}33.13\overset{.}{10}$	$\begin{array}{c} 33.04 \\ 32.24 \\ (3) \end{array}$	32.71 (1 32.12 (1) 33.43 4) 33.08	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c} .3 (14) 33 \\ (22) 33 \end{array} $	$ \begin{array}{c} 33 \\ 27 \\ (24) \\ 33. \end{array} $.07(9) .37(16)	32.21 33.59	. (24) 33.36 (1) 33.05	$\begin{pmatrix} & & & \\ & & & \\ & & & \\ & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & $	(4) (4) (20) 33	$ \stackrel{.}{.}84 (15) $ $ \stackrel{.}{.}32 (12) $	$33.05 \\ 32.82 \\ (19$	$) \begin{array}{c} 32.52 \\ 33.38 \\ 9 \end{array}$	$ \begin{array}{c} 9) & 33.38 \\ 0 & 33.45 \\ 33.45 \\ (4) \end{array} $	$32.49 \\ 33.44 \ (5)$	-
	$\sum_{\substack{R_{Ci}^2\\R_{Ci}^2}} R_{Ci}^2$	1372 1882384	$1\ 132$ $1\ 281\ 424$	$\frac{1\ 265}{1\ 600\ 225}$	1319 1739761	1229 1 1510	$\begin{array}{cccc} 1 & 1 \\ 1 & 1 \\ 4 & 1 \\ 1 & 1 \\ 1 & 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1$	8 133 1344 178	37 11 37 569 12	109	$\dots 1252$ $\dots 1567$	1 1 0 7 5 0 4 1 2 2 5	1 508 1 508 449 2 274	064 16	69 10 361	$\frac{1}{1} \frac{042}{085} 764$	1421 2019241	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	$\begin{array}{c} 1 \ 075 \\ 1 \ 155 \ 625 \end{array}$	
Table	B.29:	A previ	iew of th	ie numb	er of 1	non-do	minate	d soluti	ions (an	id corr	espona	ling ran	ks) obtā	ined f	or run	1-100	for hy	perparan	neter con	- nbinatio
12.3.1-	-A2.3.24	4. The I	anks are	e given i.	n the p	oarentl	heses ar	nd are ı	ised in a	the coi	mputat	ion of t	he [¯] Fried	man a	nd Im	an-Dav	enport	hypothes	is tests.	
Run	A2.3.1	A2.3.2	A2.3.5	3 A2.2.	4 A	2.2.5	42.2.6	A2.2.7	A2.2.	8 A2	.2.9	A2.2.17	A2.2.1	8 A2.	2.19 A.	2.2.20	A2.2.21	A2.2.22	A2.2.23	A2.2.24
	160(4.5)	152 (9)	143 (1	[4] 118 (j	17) 10	3 (18) 1	128 (16)	164 (2)	162 (;	3) 144	$\begin{array}{c} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $	88 (22)	83 (24	() 88 ()	22) 88	(22)	39 (20)	153 (8)	157 (7)	144 (12.5)
n w	$92 (4) \\ 98 (4.5)$	02 (8.5	(2) 39 (2) (1) 106 (1	1.5) 89 (1: 1.5) 89 (15	9) 02 3) 93	6 (7) 6	30 (11) 38 (4.5)	82 (0) 82 (22)	$^{64}(1)$ 92 (8.	u) 106 .5) 106	3 (1.5)	88 (15) 88 (15)	(9) CS 07 (6)	47	$ \frac{14}{24} 84 $	(3) (19.5) 8	39 (13) 89 (13)	(1 (10) 85 (17.5)	80 (23) 80 (23)	(c.01) 2 / (10) 01 (10)
4 1	123(5)	121 (7)	110 (1	17.5) 111 (:	15.5) 11	(4 (13) 1	(117(9))	144(1)	122 (6) 12E	5 (4)	127 (2.1	5) 111 (1)	5.5) 120	(8) 11	5(10.5)	99 (24)	127(2.5)	106(19)	102(22)
ი 0	72(12.5)	(1) 78 (7)	71 (10 84 (4)	0) /0 (1. 76 (8.	5) 04 5) 74	i (10) 5	74 (0) 54 (23)	57(21)	92 (2 76 (8.	, 5) 81 .5) 81	(1) (5.5)	50 (0) 72 (12.5	i) 53 (24	000 (c. ((16.5) 64	(0.11)	$^{(4)}_{71}$	62 (19)	$^{(4)}_{73}(11)$	$ \begin{array}{c} 91 \\ 81 \\ (5.5) \end{array} $
6 0	100 (4)	81 (17.	5) 79 (15 67 (01	9) 81 (1	7.5) 10	(2) (2)	(6) (6) (6)	102 (3)	105 (1) 87	(12.5)	65 (24)	77 (20	.5) 76 (22.5) 96	(9)	37 (12.5)	95(7)	89 (10)	83 (15.5)
0 0	56 (17)	66 (9) 66 (9) 66 (9)	87 (2)	1) 151 (1 55 (19	2) 11 3) 56	6 (0.0) c (17) 6	38 (8) 38 (8)	53(21)	63 (1.	(1) 10 10 10 10 10 10 10 10 10 10 10 10 10	. : (T)	54(20)	75 (4.1	5) 37 ()	23) 57	(14.5)	76 (3)	33(24)	61 (12.5)	75 (4.5)
10	76 (20)	66(23)	89 (5)) 88 (6.	5) 84	1 (13.5) 8	86 (11)	82 (15.5	6) 87 (8.	.5) 79	(17)	73 (22)	87 (8.	5) 65 (24) 88	(6.5)	34(13.5)	74(21)	82 (15.5)	100(3)
$99 \\ 100$	$\begin{array}{c} 87 \ (18.5) \\ 75 \ (16.5) \end{array}$	$ \begin{array}{ccc} 103 & (6) \\ 0 & (0) & (23) \end{array} $	96 (10 70 (20	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	2) 81 5) 92	$\begin{bmatrix} (23) & 1\\ (2.5) & 7\\ \end{bmatrix}$	$\begin{array}{ccc} 106 & (5) \\ 76 & (15) \end{array}$	$\begin{array}{c} 91 & (15) \\ 72 & (19) \end{array}$	$\begin{array}{c} 91 & (1) \\ 82 & (9) \end{array}$	5) 77 .5) 75	$(24) (16.5) \dots$	85(21.6) 92(2.5)	5) 100 (7 77 (14	97 ()	$\begin{array}{ccc} 9) & 11 \\ 24) & 84 \end{array}$	2 (1) (6)	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 85 & (21.5) \\ 70 & (20.5) \end{array}$	$\begin{array}{c} 98 & (8) \\ 90 & (4) \end{array}$	$\begin{array}{c} 89 & (17) \\ 79 & (12) \end{array}$
$\sum_{\substack{R \\ P \\ R^2}} R_{Ci}^2$	1 313.50	1 144.50 25 1 309 85	0 1050 3025 11025	1 276. 500 1 6297	50 11 452.95.1.9	131 1	$1\ 190.50$	1 086.50	006.5	0 875 378 01)	1423.50	0 1447	160	4 15 0 016 16	374 876 876	159.50	1 563.50	1 224	1 132.50 1 300 556

Table B.28: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A2.3.1–A2.3.24. The ranks

	We	orst	В	est
	HA	NDS	HA	NDS
A2.3.1	Run 6	Run 44	Run 17	Run 1
A2.3.2	$\operatorname{Run}34$	$\operatorname{Run}71$	Run 89	$\operatorname{Run}48$
A2.3.3	$\mathrm{Run}\ 27$	$\operatorname{Run}97$	Run 1	Run 39
A2.3.4	$\operatorname{Run}44$	$\operatorname{Run}44$	$\operatorname{Run}95$	Run 39
A2.3.5	$\mathrm{Run}\ 27$	Run 82	Run 73	Run 39
A2.3.6	$\operatorname{Run}34$	$\operatorname{Run}34$	Run 96	Run 89
A2.3.7	$\operatorname{Run}59$	Run 9	$\operatorname{Run}50$	Run 89
A2.3.8	Run 21	Run 9	Run 76	Run 1
A2.3.9	Run 44	Run 12	Run 16	Run 39
A2.3.10	Run 34	$\operatorname{Run}34$	Run 89	Run 1
A2.3.11	Run 34	Run 98	Run 13	Run 1
A2.3.12	Run 66	Run 2	Run 89	Run 1
A2.3.13	Run 44	Run 44	Run 1	Run 15
A2.3.14	Run 39	Run 44	Run 7	Run 79
A2.3.14	Run 12	Run 60	Run 84	Run 1
A2.3.15	Run 28	Run 2	Run 63	Run 1
A2.3.16	Run 44	Run 44	Run 89	Run 15
A2.3.17	Run 32	Run 32	Run 16	Run 48
A2.3.18	Run 45	Run 6	$\operatorname{Run} 25$	Run 65
A2.3.19	Run 21	Run 9	Run 78	Run 39
A2.3.20	Run 71	Run 12	$\operatorname{Run} 25$	Run 18
A2.3.21	$\operatorname{Run}87$	Run 12	Run 94	Run 39
A2.3.22	$\operatorname{Run}97$	Run 9	Run 96	Run 15
A2.3.23	Run 44	Run 1	Run 49	Run 39
A2.3.24	Run 21	Run 97	Run 74	Run 15

Table B.30: The simulation runs that correspond to the best and worst approximation fronts (obtained by the NSGA-II) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A2.3.1–A2.3.24.

oximation		
e appr		
s of th		
erarea		
he hyp		
d on t		
s base		
oarison		
e com		
nultipl		
ir the i		
test fc		
st hoc		
nyi po		
Neme	2.3.24.	
by the	.3.1-A	
tained	ons A2	
ues ob	$_{1binati}$	
d p-val	ter con	
djustee	arame	
The a	hyperp	
B.31:	or the.	
Table	fronts f	

Ā	V2.3.2 A2	2.3.3 A2	1.0.4 MAN	.0 A2.3.	6 AZ.3.	1 A2.3.	8 A2.3.	9 A2.3.10	A2.3.11	A2.3.12	A2.3.13	A2.3.14	A2.3.15	A2.3.16	1T.C.ZA	01.0.2A	LL2.0.10		14.0.40	11.0.041	01-0-041	A2.0.2A
12.3.1	1	1	1 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	0.27	1	1	0.82
12.3.2		1	1 1	1	1	1	1	0.38	1	1	0.26	1	1	0.53	1	1	0.047	1	1	1	1	1
12.3.3			1 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
12.3.4			1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
12.3.5				1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
12.3.6					1	1	1	1	1	1	1	1	1	1	1	1	0.38	1	1	1	1	1
12.3.7						1	1	1	1	1	1	1	1	1	1	1	1	1	0.88	1	1	1
12.3.8							1	0.17	1	1	0.11	1	1	0.24	1	1	0.02	1	1	0.5	1	1
12.3.9								0.14	1	1	0.09	1	1	0.2	1	1	0.02	1	1	0.42	1	1
2.3.10									1	0.13	1	1	1	1	1	0.16	1	1	0.01	1	0.18	0.045
2.3.11										1	1	1	1	1	1	1	0.998	1	1	1	1	1
2.3.12											0.09	1	1	0.19	1	1	0.01	1	1	0.39	1	1
2.3.13												1	1	1	1	0.10	1	1	0.01	1	0.12	0.03
2.3.14													1	1	1	1	1	1	1	1	1	1
2.3.15														1	1	1	0.22	1	1	1	1	1
.2.3.16															1	0.22	1	1	0.02	1	0.26	0.07
.2.3.17																1	1	1	1	1	1	1
2.3.18																	0.02	1	1	0.47	1	1
2.3.19																		1	0	1	0.02	0
2.3.20																			1	1	1	1
2.3.21																				0.04	1	1
2.3.22																					0.53	0.15
2.3.23																						1

Stellenbosch University https://scholar.sun.ac.za

Table B.32: The adjusted p-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the number of non-dominated solutions found for the approximation fronts for the hyperparameter combinations A2.3.1–A2.3.24.

4	v2.3.2 A	2.3.3 A2.	3.4 A2.3.	5 A2.3.	5 A2.3.7	A2.3.8	A2.3.9	A2.3.10	A2.3.11	A2.3.12	A2.3.13	A2.3.14	A2.3.15	A2.3.16	A2.3.17	A2.3.18	A2.3.19	A2.3.20	A2.3.21	A2.3.22	A2.3.23	A2.3.24
A2.3.1	1	1 1		1	1	0.013	0	1	0.004	0.048	0.01	1	1	0.21	1	1	1	1	1	1	1	1
A2.3.2		1 1		1	1	1	1	1	1	1	0	0.7	1	0	1	0.7	0.001	1	1	0.01	1	1
A2.3.3		1		1	1	1	1	1	1	1	0	0.02	0.98	0	0.05	0.02	0	0.33	1	0	1	1
A2.3.4			1	1	1	0.06	0.02	1	0.02	0.2	0	1	1	0.05	1	1	0.29	1	1	1	1	1
A2.3.5				1	1	1	1	1	1	1	0	0.44	1	0	0.95	0.44	0	1	1	0	1	1
A2.3.6					1	1	0.51	1	0.54	1	0	1	1	0	1	1	0.01	1	1	0.05	1	1
A2.3.7						1	1	1	1	1	0	0.09	1	0	0.21	0.09	0	1	1	0	1	1
A2.3.8							1	1	1	1	0	0	0	0	0	0	0	0	1	0	0.41	1
A2.3.9								1	1	1	0	0	0	0	0	0	0	0	1	0	0.16	1
A2.3.10									1	1	0	0.231	1	0	0.57	0.23	0	1	1	0.002	1	1
A2.3.11										1	0	0	0	0	0	0	0	0	1	0	0.17	1
12.3.12											0	0	0.02	0	0	0	0	0	1	0	1	1
A2.3.13												1	0.04	1	0.72	1	1	0.13	0	1	0	0
A2.3.14													1	1	1	1	1	1	1	1	1	0.47
12.3.15														0.56	1	1	1	1	1	1	1	1
12.3.16															1	1	1	1	0	1	0.01	0
12.3.17																1	1	1	1	1	1	1
A2.3.18																	1	1	1	1	1	0.458
12.3.19																		1	0	1	0.04	0
12.3.20																			1	1	1	1
12.3.21																				0.02	1	1
12.3.22																					0.19	0.01
12.3.23																						1

- 	$\begin{array}{cccccccccccccccccccccccccccccccccccc$		ombin; 3. 3. A2.4.	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
	21.03 21.29 0 21.29 0 0 0 21.14 0 0 0 21.14 0 0 0 21.23 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 0 21.14 0 0 <	21.4 (8) 993 986 049	eter co s tests A2.4.2	$\begin{array}{c} 38 & (24) \\ 552 & (5) \\ 556 & (13) \\ 555 & (71) \\ 662 & (7) \\ 662 & (7) \\ 662 & (7) \\ 682 & (7) \\ 682 & (7) \\ 682 & (7) \\ 682 & (7) \\ 682 & (7) \\ 682 & (7) \\ 682 & (7) \\ 682 & (7) \\ 682 & (16) $
	$\begin{array}{c} 1.01 \ (16) \\ 0.9 \ (16) \\ 1.32 \ (5) \\ 1.1 \ (23) \\ 1.1 \ (23) \\ 1.02 \ (20) \\ 1.46 \ (2) \\ 0.93 \ (22) \\ 1.6 \ (7) \\ 1.46 \ (2) \\ 1.46 \ (2) \\ 1.46 \ (2) \\ 1.46 \ (2) \\ 1.46 \ (3) \\ 1.45 \ (6) \ (6$	$\frac{1.09\ (15)}{146}$ 313 316	param othesi A2.4.22	$\begin{array}{c} 48 \ (13) \\ 39 \ (20) \\ 55 \ (4) \\ 55 \ (4) \\ 47 \ (12) \\ 52 \ (7) \\ 39 \ (22) \\ 39 \ (22) \\ 39 \ (22) \\ 39 \ (22) \\ 52 \ (3.5) \\ 52 \ (3.5) \\ 52 \ (3.5) \\ 52 \ (3.5) \\ 52 \ (3.5) \\ 52 \ (3.5) \\ 53 \ (776) \ (776) \$
	$ \begin{array}{c} 62 & (23) \\ 95 & (22) \\ 97 & (24) \\ 25 & (11) \\ 25 & (11) \\ 25 & (11) \\ 25 & (11) \\ 22 & (11) \\ 22 & (15) \\ 22 & (15) \\ 22 & (15) \\ 22 & (15) \\ 22 & (15) \\ 22 & (15) \\ 22 & (15) \\ 21 & (21) $	87 (23) 2 58 1 27 364 1	hyper ort hyr A2.4.21	$\begin{array}{c} 49 & (11.5) \\ 51 & (13) \\ 51 & (13) \\ 61 & (3) \\ 52 & (3) \\ 33 & (24) \\ 33 & (23) \\ 33 & (23) \\ 33 & (23) \\ 33 & (23) \\ 41 & (18.5) \\ 41 & (120) \\ 51 & (9) \\ 50 & (9) \\ 1218 \\ 128524 \\ 128524 \end{array}$
	$ \begin{array}{c} (4) \\ (21) \\ (2$	$\frac{(21)}{150}$ $\frac{150}{242}$ $\frac{1}{2}$	avenp	7) 11) 234) 244) 224) 2355 2355 11.5 11.5 2006.255 2006.
1 01 96	$\begin{array}{c} 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 $	3) 20.95 969 4 93896	un 1–1 man-L ^{19 A2.4}	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
	21.17 (8 21.28 (8) 21.28 (8) 21.28 (11 19.72 (12) 19.72 (2) 21.6 (1) 21.6 (1) 21.5 (2) 21.5 ($\frac{21.15(1)}{1248}$ 155750	l for r 1 and I A2.4.	50 9 42 11 42 11 57 12 57 11 50 12 51 12 52 12 53 11 53 12 54 11 57 11 58 11 53 11 54 11 55 11 57 11 57 11
(11)	$\begin{array}{c} 0.888 \\ 0.888 \\ 1.1.25 \\ 0.59 \\ 0.57 \\ 1.98 \\ 0.57 \\ 1.98 \\ 1.1.08 \\ 1.1.08 \\ 1.1.08 \\ 1.1.08 \\ 1.1.23 \\ 1.1.29 \\$	20.96 (18) 539 368 521	btained iedmar A2.4.18	$\begin{array}{c} 47 & (14.5) \\ 54 & (2.5) \\ 57 & (2.5) \\ 57 & (2.5) \\ 57 & (2.5) \\ 62 & (1.5) \\ 62 & (1.5) \\ 47 & (12) \\ 447 & (12) \\ 48 & (8.5) \\ 48 & (8.5) \\ 50 & (9) \\ 50 & (9) \\ 50 & (9) \\ 50 & (2) \\ 50 & (2) \\ 1262 \end{array}$
3 (6) 5	$\begin{array}{c} 1.17\\ 1.17\\ 1.17\\ 1.17\\ 1.17\\ 1.17\\ 1.17\\ 1.17\\ 1.12\\$	$\begin{array}{c c} .1 & (14) & 2 \\ 01 & 1 \\ 42 & 401 & 2 \end{array}$	anks) o the Fr ^{A2.4.17}	$\begin{array}{c} 43 \ (21) \\ 47 \ (11) \\ 59 \ (4) \\ 48 \ (7.5) \\ 48 \ (7.5) \\ 44 \ (14) \\ 47 \ (17) \\ 47 \ (17) \\ 45 \ (12) \\ 45 \ (12) \\ 41 \ (20) \ (20) \\ 41 \ (20) \$
19) 21	$\begin{array}{c} \begin{array}{c} \begin{array}{c} & & \\ & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \end{array} \\ \end{array} \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ \end{array} \end{array} \\ \end{array} \\ \end{array} \end{array} \\ \begin{array}{c} & & \\ \end{array} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \end{array} \\ \end{array} \end{array} \\ \end{array} \\ \end{array} \end{array} \\ \end{array} \\ \end{array} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \end{array} \\ \\ \end{array} \\ \end{array} \\ \end{array} \\ \\ \end{array} \\ \end{array} \\ \\ \end{array} \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \\ \end{array} $ \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\ \\ \end{array} \\	5 (20) 21 3 3 3 12 4896 14 4	ding ra tion of 4.16	$\begin{array}{c} 18.5 \\ 221 \\ 5.2 \\ 2.2 \\ 19.5 \\ 19.5 \\ 119.5 \\ 148.5 \\ 148.5 \\ 148.5 \\ 148.5 \\ 149 \\ 140 $
21 ($\dots 20.9$ $\dots 1336$ $\dots 178^{2}$	respon mputa	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
1 07 (13)	$\begin{array}{c} 1.68 \\ 1.68 \\ 1.26 \\ 1.3 \\ 1.3 \\ 1.3 \\ 1.37 \\ 1.37 \\ 1.37 \\ 1.48 \\ 1.68 \\ 1.48 \\ 1.6 \\ 1.48 \\ 1.6 \\ 1.48 \\ 1.6 \\ 1.48 \\ 1.6 \\ 1.137 \\ 1.33 \\ 1$	$\frac{1.64(1)}{008}$ 016064	nd cor the co A2.4.9	$\begin{array}{c} 50 & (9.5) \\ 42 & (17.5) \\ 57 & (18.5) \\ 57 & (18.5) \\ 56 & (6) \\ 54 & (10) \\ 54 & (10) \\ 54 & (10) \\ 57 & (13) \\ 57 & (13) \\ 57 & (13) \\ 53 & (2) \\ 1078 \\ \end{array}$
0 (01) 1	$\begin{array}{c} \begin{array}{c} & & & & \\ & & & & \\ & & & & \\ & & & & $	$ \begin{array}{c cccccccccccccccccccccccccccccccccc$	ions (ɛ used in A2.4.8	$\begin{array}{c} 43 \\ 47 \\ 111 \\ 52 \\ 11$
10 (1)	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	8 (16) 21. 881 19.	d solut 1d are A2.4.7	$ \begin{array}{c} 61 & (1) \\ 54 & (9) \\ 54 & (10) \\ 54 & (10) \\ 54 & (10) \\ 54 & (10) \\ 35 & (24) \\ 81 & (23) \\ 61 & (11) \\ 51 & (11) \\ $
	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{rrr} 17 & 21.08 \\ 1 & 1 & 109 \\ 14 & 1 & 229 \end{array}$	ninate eses al ^{12.4.6}	$\begin{array}{c} 44 \ (5.5) \\ 44 \ (2.5) \\ 68 \ (2.1) \\ 68 \ (2.1) \\ 68 \ (2.1) \\ 68 \ (2.5) \\ 61 \ (2.5) \ (2.5) \\ 61 \ (2.5) $
7 21 12 (21.15 (1) 21.15 (8) 21.15 (8) 21.15 (8) 21.15 (17) 21.17 (17) 21.17 (17) 21.15 (17) 21.15 (17) 21.17 (17)	$\frac{21.05}{1} \left(\frac{1152}{1327} \right)$	101-doi 0arenth 2.4.5	$ \begin{array}{c} 2 & (8) \\ 3 & (15.5) \\ (15.5) \\ (15.5) \\ (15.5) \\ (15.5) \\ (15.5) \\ (15.5) \\ (11.6) \\ (11.6) \\ (11.6) \\ (12.5) \\$
	$\begin{array}{c} 21.01 \left(17\\ 220.85 \left(18\\ 220.85 \left(18\\ 221.43 \left(16\\ 220.92 \left(21\\ 220.99 \left(26\\ 220.99 \left(26\\ 221.33 \left(5\right) \right) \right) \right) \right) \\ 221.33 \left(5\right) \\ 221.33 \left($	$\frac{21.29}{1476}(9)$ $\frac{1476}{2178576}$	n the p	$\begin{array}{c} 3 \\ 3 \\ 1 \\ 3 \\ 3 \\ 3 \\ 5 \\ 5 \\ 5 \\ 5 \\ 5 \\ 5 \\ 5$
0.91 (20)	$\begin{array}{c} \begin{array}{c} 1.33 \\ 1.33 \\ 1.23 \\ 1.23 \\ 1.58 \\ 1.46 \\ 9) \\ 1.37 \\ 5) \\ 1.37 \\ 5) \\ 1.37 \\ 5) \\ 1.31 \\ 21 \\ 0.044 \\ (24) \\ 0.094 \\ (20) \\ 0.094 \\ (20) \end{array}$	$\frac{1.63}{120}$ 254 400	e numl given i A2.	5) 56 55 55 55 55 55 55 50 55 51 55 51 55 51 55 52 53 53 54 61 1 54 55 55 54 55 55 55 55 55 55 55 55 55 55
0 (10) - 0	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	v of th lks are A2.4.3	$\begin{array}{c} 45 \ (16. \\ 334 \ (24) \\ 551 \ (25) \\ 552 \ (25) \\ 553 \ (12) \\ 553 \ (12) \\ 553 \ (12) \\ 513 \ (12) \\ 42 \ (16) \\ 42 \ (17) \\ 64 \ (17) \\ 43 \ (17) \\ 43 \ (17) \\ 43 \ (17) \\ 43 \ (17) \\ 44 \ (17)$
	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccc} 1 & (22) & 21. \\ & & 1 & 37 \\ 849 & 1 & 89 \\ \end{array}$	previev The ran ^{A2.4.2}	$\begin{array}{c} 49 & (11.5) \\ 55 & (5) \\ 58 & (16) \\ 58 & (16) \\ 58 & (16) \\ 58 & (16) \\ 58 & (12) \\ 57 & (17) \\ 65 & (3.5) \\ 55 & (3.5) \\ 55 & (5) \\ 55 & (5) \\ 56 & (12) \\ 55 & (5) \\ 1182 \\ 1182 \\ 1182 \\ 1182 \\ 1305 \\ 427 \\ 116.5 \\ 1305 \\ 422 \\ 427 \\ 16.5 \\ 1305 \\ 422 \\ 427 \\ 16.5 \\ 1305 \\ 422 \\ 427 \\ 16.5 \\ 1305 \\ 422 \\ 427 \\ 16.5 \\ 1305 \\ 427 \\ 427 \\ 16.5 \\ 1305 \\ 427 \\$
21.14	21.45 20.4 21.15 21.17 21.175 21.175 21.175 21.51 21.51 21.51 21.51 21.51	$\begin{array}{c} 20.94\\ 20.94\\ C_i & 1457\\ C_{i} & 2122\\ \end{array}$	4: A j 4.24. 7	$ \begin{array}{c} (2) \\ (13.5) \\ (15.5) \\ (15.5) \\ (20) \\ (20) \\ (20) \\ (20) \\ (25) \\ (20) \\ (25) \\ (20) \\ (25) \\ (20) \\ (25) \\ (20$
-	0.04 - 00 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0	\sum_{R}^{100}	e B.3 <i>1</i> – <i>A</i> 2	$\begin{array}{c} 57 \\ 45 \\ 53 \\ 53 \\ 53 \\ 51 \\ 51 \\ 51 \\ 51 \\ 5$

Table B.33: A preview of the hyperareas (and corresponding ranks) obtained for run 1-100 for hyperparameter combinations A2.4.1–A2.4.24. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

250

	We	orst	В	est
	HA	NDS	НА	NDS
A2.4.1	Run 66	Run 66	Run 6	Run 39
A2.4.2	$\operatorname{Run}79$	$\operatorname{Run}26$	$\operatorname{Run}25$	Run 16
A2.4.3	$\operatorname{Run}83$	$\operatorname{Run}58$	Run 98	Run 15
A2.4.4	$\operatorname{Run}44$	$\operatorname{Run}42$	$\operatorname{Run}77$	$\operatorname{Run}50$
A2.4.5	$\operatorname{Run}32$	$\operatorname{Run} 12$	$\operatorname{Run}5$	Run 96
A2.4.6	$\operatorname{Run}59$	$\operatorname{Run}21$	$\operatorname{Run}15$	Run 84
A2.4.7	$\operatorname{Run}79$	$\operatorname{Run}82$	$\operatorname{Run}24$	$\operatorname{Run} 18$
A2.4.8	$\operatorname{Run}82$	Run 68	$\operatorname{Run}38$	Run 15
A2.4.9	$\operatorname{Run}83$	$\operatorname{Run}58$	$\operatorname{Run}76$	Run 4
A2.4.10	$\mathrm{Run}\ 12$	$\operatorname{Run}64$	Run 63	Run 15
A2.4.11	$\mathrm{Run}\ 28$	Run 9	$\operatorname{Run}76$	$\operatorname{Run}73$
A2.5.12	$\operatorname{Run}71$	Run 60	$\operatorname{Run}70$	Run 19
A2.5.13	$\mathrm{Run}\ 68$	$\mathrm{Run}\ 27$	$\operatorname{Run}15$	$\operatorname{Run} 18$
A2.5.14	$\operatorname{Run} 2$	$\operatorname{Run} 2$	Run 4	Run 36
A2.5.15	$\operatorname{Run}74$	Run 82	$\operatorname{Run}47$	Run 18
A2.5.16	$\operatorname{Run}26$	$\operatorname{Run}26$	Run 46	Run 16
A2.5.17	$\operatorname{Run}58$	Run 26	Run 4	Run 39
A2.5.18	$\operatorname{Run}21$	$\operatorname{Run}71$	Run 81	Run 65
A2.4.19	Run 13	Run 13	$\operatorname{Run}50$	Run 39
A2.4.20	Run 6	Run 44	Run 63	Run 15
A2.4.21	$\operatorname{Run}59$	$\operatorname{Run}82$	Run 96	Run 15
A2.5.22	$\operatorname{Run}74$	$\mathrm{Run}\;12$	$\operatorname{Run}75$	$\operatorname{Run}73$
A2.5.23	$\operatorname{Run}74$	$\operatorname{Run}55$	Run 33	Run 96
A2.5.24	Run 44	Run 44	Run 3	Run 15

Table B.35: The simulation runs that correspond to the best and worst approximation fronts (obtained by the NSGA-II) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A2.4.1–A2.4.24.

4.1 1	0.207	7 1	0.63	0.138	-	0.002	0	1	0.003	0	0.88	1	0.52	1	1	1	1	1	0	1	0.52	0		
1.2	1	1	1	1	1	0.06	0	1	0.10	0	0.047	1	1	1	1	1	1	1	0.01	1	1	0.035		
.3		0.102	1	1	1	1	1	0.193	1	0.7	0	1	1	0.01	1	1	0.01	1	1	0.003	1	1		
.4			0.33	0.067	1	0	0	1	0	0	1	1	0.27	1	1	1	1	0.33	0.07	1	0	1		
5				1	1	1	1	0.591	1	0.23	0	1	1	0.04	1	1	0.03	1	1	0.01	1	1		
.6					1	1	1	0.128	1	0.998	0	1	1	0.01	1	1	0.005	1	1	0.002	1	1		
.7						0.03	0	1	0.049	0	0.1	1	1	1	1	1	1	1	0.01	1	1	0.016		
8.							1	0	1	1	0	0.97	1	0	0.29	1	0	1	1	0	1	1		
6.								0	1	1	0	0.01	1	0	0	0.47	0	0.09	1	0	1	1		
.10									0.003	0	0.94	1	0.48	1	1	1	1	1	0	1	0.48	0.001		
.11										1	0	1	1	0	0.44	1	0	1	1	0	1	1		
12											0	0	0.29	0	0	0.04	0	0.01	1	0	0.29	1		
.13												0.002	0	1	0.009	0	1	0	0	1	0	0		
.14													1	1	1	1	1	1	0.26	1	1	0.59		
15														0.03	1	1	0.02	1	1	0.01		1		
16															1	0.23	1	1	0	1	0.028	0		
17																1	1	1	0.07	1	1	0.17		
.18																	0.2	1	1	0.1	1	1		
.19																		0.998	0	1	0.02	0		
20																			1	0.53	1	1		
21																				0	1	1		
.22																					0.01	0		
.23																						Г		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	4															0	11.5.70	01.1.711						
--	----	---	---	---	---	---	---	------	---	-----	---	---	------	------	------	------	---------	----------	------	-----	---	------	------	------
2 1		1	1	1	1	1	1	0.65	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	5		1	1	1	1	1	1	1	1	1	1	0.78	1	1	1	1	1	1	1	1	0.09	1	1
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ę.			1	1	1	1	0.85	1	1	1	1	1	1	1	1	1		1	1	1	1	1	1
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	4.				1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	ы.					1	1	1	1	1	1	1	0.1	1	1	0.14	1	1	0.25	1	1	0.01	1	г
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	9.						1	0.84	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
.8 .9 .9 .0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.01 0 0.06 0.	٢.							1	1	1	1	1	0.02	1	0.71	0.03	0.8	1	0.05	1	1	0	1	1
9 1 1 1 1 0.02 1 0.59 0.02 0.66 10 1	×.								1	0.8	1	1	0	0.15	0.01	0	0.01	0.85	0	0.5	1	0	1	1
10 1 1 1 1 1 1 11 1 1 1 1 1 1 1 12 1<	6									1	1	1	0.02	1	0.59	0.02	0.66	1	0.04	1	1	0	1	-
11 1 0.01 1 0.5 0.02 0.56 13 1 1 1 1 1 1 14 1 1 1 1 1 1 15 1 1 1 1 1 1 16 1 1 1 1 1 1 16 1 1 1 1 1 1 17 1 1 1 1 1 1 16 1 1 1 1 1 1 17 1 1 1 1 1 1 17 1 1 1 1 1 1 17 1 1 1 1 1 1 18 1 1 1 1 1 1 18 1 1 1 1 1 1 18 1 1 1 1 1 1 19 1 1 1 1 1 1 10 1 1 1 1 1 1 10 1 1 1 1 1	10										1	1	1	1	1	1	1	1	1	1	1	1	1	1
12 0 1 0.14 0 0.14 13 1 <td< td=""><td>11</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>1</td><td>0.01</td><td>1</td><td>0.5</td><td>0.02</td><td>0.56</td><td>1</td><td>0.03</td><td>1</td><td>1</td><td>0</td><td>1</td><td>1</td></td<>	11											1	0.01	1	0.5	0.02	0.56	1	0.03	1	1	0	1	1
13 1 1 1 1 14 1 1 1 1 1 15 1 1 1 1 1 1 1 16 1 <td< td=""><td>12</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>0</td><td>1</td><td>0.14</td><td>0</td><td>0.16</td><td>1</td><td>0.01</td><td>1</td><td>1</td><td>0</td><td>1</td><td>1</td></td<>	12												0	1	0.14	0	0.16	1	0.01	1	1	0	1	1
14 1 15 1 16 1 17 1 17 1 18 19 19 10 10 11 12 13 14 15 16 17 18 19 10 10 11 12 13 14 15 16 17 18 19 10 10 11 11 12 13 14 15 15 16 17 18 19 10 11 11 12 13 14 15 16 17 18 19 10 11 11 12 13 14 15 15 16 1	13													1	1	1	1	1	1	1	1	1	1	0.70
15 1 16 1 17 1 18 1 19 1 10 1 21 1 21 1 22 1	14														1	1	1	1	1	1	1	1	1	1
16 17 18 19 19 20 21 22	15															1	1	1	1	1	1	1	1	1
17 18 19 20 21 22	16																1	1	1	1	1	1	1	0.89
18 19 20 21 22 22	17																	1	1	1	1	1	1	1
19 20 21 22 22 22	8																		1	1	1	1	1	г
20 21 22	61																			1	1	1	1	1
21 22	50																				1	1	1	г
22	21																					0.33	1	1
	22																						0.22	0.08
07	23																							1

Table B.37: The adjusted p-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the number of non-dominated solutions

found for the approximation fronts for the hyperparameter combinations A2.4.1–A2.4.24.

$\begin{array}{c} 4, \\ 220, 176\\ 13, 171\\ 18, 162\\ 18, 162\\ 7, 187\\ 9, 179\\ 9, 177\\ 137\\ 137\\ 137\\ 137\\ 137\\ 137\\ 137\\$		f the e g	$\begin{array}{c} 1 & 184.17 \\ 1 & 184.17 \\ 1 & 184.5 \\ 1 & 183.14 \\ 1 & 187.28 \\ 1 & 177.2 \\ 2 & 2 \\ 1 & 177.2 \\ 1 & 274.641 \\ 1 & 274.641 \\ 1 & 274.641 \\ 1 & 1 & 274.641 \\ 1 & 2 & 1 & 1 \\ 1 & 2 & 1 &$	$\begin{array}{c} 33) & 180.4 (1) \\ 1 & 173.22 (1) \\ 1 & 184.58 (1) \\ 2 & 184.58 (2) \\ 2 & 184.58 (2) \\ 2 & 169.37 (1) \\ 3 & 183.97 (1) \\ 1 & 183.97 (1) \\ 1 & 183.97 (1) \\ 1 & 1990 921 \\ 1 & 1990 9$	$\begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \end{array} \\ \begin{array}{c} \end{array} \\ \end{array} $	$\begin{array}{c} (22) 177 \\ (21) 188 \\ (11) 188 \\ (11) 188 \\ (11) 188 \\ (17) 179 \\ (17) 179 \\ (17) 179 \\ (17) 188 \\ (21)$	$\begin{array}{c} \begin{array}{c} 3.34 & (19) \\ 3.34 & (19) \\ 5.58 & (8) \\ 5.58 & (8) \\ 1 \\ 7.2 & (12) \\ 1 \\ 7.2 & (12) \\ 1 \\ 1 \\ 7.2 & (12) \\ 1 \\ 1 \\ 1 \\ 1 \\ 2 \\ 7 \\ 6 \\ 1 \\ 1 \\ 1 \\ 2 \\ 7 \\ 6 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1$	$\begin{array}{c} \begin{array}{c} (88).94 & (2)\\ (88).94 & (2)\\ (177.9 & (2)\\ (177.7) & (2)\\ (179.9 & (2)\\ (179.9 & (2)\\ (177.9) & (2)\\ (177.9 & (2)\\ (179.9 & (2)\\ (173.8 & (2)\\ (113.3 & (2)\\ (2) & (2)\\ (2) & (2) & (2)\\ (2) & (2) & (2)\\ (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2) & (2)\\ (2) & (2) & (2) & (2) & (2) & (2) & (2)\\ (2) & $	(1,1,1,1)	822.2 (6) 834.76 (12) 84.79 (12) 81.45 (8) 81.45 (8) 81.45 (8) 81.45 (8) 716.73 (20) 721344 721344 721344 rtation	$\begin{array}{c} \begin{array}{c} 1000000\\ 1000000000\\ 1000000000\\ 100000000$	$\begin{array}{c} \begin{array}{c} 33 \\ 222 \\ 153 \\ 151 \\ 151 \\ 177.8 \\ 151 \\ 177.8 \\ 151$	$\begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \begin{array}{c} \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \begin{array}{c} \end{array} \\ \begin{array}{c} \begin{array}{c} \end{array} \\ \end{array} $	$\begin{array}{c} 77.83 \\ 78.49 \\ 17.83 \\ 17.83 \\ 76.01 \\ 17.63 \\ 11.1 \\ 11.1 \\ 12.1 \\ 11.1 \\ 1466521 \\ 11.1 \\ 1466521 \\ 11.1$	$\begin{array}{c} \begin{array}{c} 1 & 172.33 \\ 1 & 171.64 \\ 1 & 171.64 \\ 1 & 182.38 \\ 1 & 182.38 \\ 1 & 182.38 \\ 1 & 192.99 \\ 1 & 173.97 \\ 1 & 173.97 \\ 1 & 167.84 \\ 1 & 1386 \\ 1 & 1386 \\ 1 & 19209 \\ 1 & 19209 \end{array}$	$\begin{array}{c} (21) & 170 \\ (19) & 181 \\ (19) & 188 \\ (16) & 188 \\ (11) & 182 \\ (11) & 182 \\ (11) & 182 \\ (11) & 182 \\ (11) & 182 \\ (24) & 183 \\ (24) & 183 \\ (24) & 183 \\ (24) & 183 \\ (26) & 178 \\ (26) & 188 \\ (26) & 188 \\ (26) & 188 \\ (26) & 188 $	$\begin{array}{c} \begin{array}{c} 9.07 \\ 1.1.48 \\ 8.73 \\ 1.1.48 \\ 8.73 \\ 1.1.48 \\ 8.73 \\ 1.1.28 \\$	$\begin{array}{c} \begin{array}{c} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $	$\begin{array}{c} \begin{array}{c} \begin{array}{c} & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ \end{array} \end{array} $
9) 17 (9) 17 13		$\begin{array}{c} \vdots \\ 9.42 (10) \\ 9.04 (18) \\ \hline 23 969 \\ \hline 23 969 \\ \hline bf the \\ are g \end{array}$)) 185.52 ()) 182.57 (1129 1274641 1274641	: 10) 180.54 (1) 10) 180.54 (1) 1411 1990 921 1990 921 the parent	$\begin{array}{c} & \begin{array}{c} & & \\ & & $	(1) 184 2 (1) 184 2 (12) 180 225 152 152 152 152 153 and ar	$\begin{array}{c} \vdots \\ 4.55 (6) \\ 1.97 (13) \\ 1 \\ 36 \\ 27 696 \\ 1 \\ 27 696 \\ 1 \\ 1 \\ 27 696 \\ 1 \\ 1 \\ 1 \\ 27 696 \\ 1 \\ 1 \\ 27 696 \\ 1 \\ 1 \\ 27 696 \\ 1 \\ 1 \\ 27 696 \\ 1 \\ 1 \\ 27 696 \\ 1 \\ 1 \\ 27 \\ 27 \\ 6 \\ 1 \\ 1 \\ 27 \\ 27 \\ 1 \\ 1 \\ 27 \\ 27 \\ 1 \\ 1 \\ 27 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ $	$\frac{1}{100}$: ranks) of the l	14) 184.4 22) 180.3 1 229 0 1 510 0 1 510 <i>Obtaii</i> <i>Friedm</i>	$\begin{array}{c} \vdots \\ 45 (7) 1 \\ 35 (15) 1 \\ 1 \\ 1 \\ 1441 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $; 71.85 (18 90.09 (3) 211 .466 521 .466 521 .466 521 .4m	3) 173.97 167.84 1386 19209 1-100 f	$\frac{(17) 177}{(24) 183}$	$\frac{1}{2}$	3) 169.7 176.0 1117 1247 1247 thesis	: 71 (21) 06 (21) 7 689 ter cc ter cc tests
-		of the grand	number iven in	¹⁴¹¹ 1990 921 • of non-c the paren A2.5.4	1235 15252 15252 10minat hominat hominat A2.5.5 43 (14)	12: 225 15: 225 15: 15: 15: A2:5.6	$\begin{array}{c c} 36 & 1\\ 27696 & 1\\ utions (\\ e used i\\ A2.5.7 \end{array}$	1133 1283689 and cc n the c	orrespe	0nding	^{1 400} 1 960 000 : ranks) of the 1	0 1 5 10 0 1 5 10 0 btair Friedm	1 1 1441 1 100 for 1an and	211 466521 r run l d Iman	1386 19209 1-100 f	138 178 for hyp nport h	³³⁷ 787 569 perpa. hypot	1117 1247 tramet thesis	7 689 ter cc tests
iev an		с и С	A9 E 2	1.0.70	43 (14)	0.0.70	1.0.44	64	о и	6 V	л 10 /	A 9 E 10	н С V И	1 C V 0 C	10.2	00 X 0 V	0.4	1 203	40 E 0
		2.5.2 (16.5)	A2.5.3	38 (10)		46 (6)	47 (A E	AZ.	(16 E)	A2.	10 F) A	A2.5.19	AZ.5. AF (7	20 A2.5	12.0	AZ.5.22 33 (94)	A2.	(1 E) 1	AZ.5.24
	$^{+}_{+}^{+}$	(2000)	$\begin{array}{c} 45 & (5) \\ 37 & (22.5) \end{array}$	29(24) 39(21)	32 (21.5 43 (13.5	5) 39 (12. 5) 40 (18.	(5) 34 $(20)(5) 40 (18)$	5) 48 42	(15)		(8) (12) - 4	$ \begin{array}{c} 36 (17) \\ 41 (16) \\ \end{array} $	30(2)	(3) 42 (3) 42 (42) (42) (42) (42) (42) (42) (42)	9.5) 9.5)	$\begin{array}{c} 32 & (21.5) \\ 32 & (21.5) \\ 40 & (18.5) \end{array}$) 37	(14.5) 4 (9) 4	44 (6.5) 45 (9)
	46	(12)	38 (22) 36 (10 E)	44(15)	50 (6) 43 (10)	40(19)) 36 (24)	47	(10)	52 (39 (20.5) 25 (21)	43 (1 51 (1	(1) 50 (30 (6) 1 <i>6</i> E) [1	52(2)	, 46	(12) (12)	50 (6) 50 (6)
	46 %	(e.e.1) (9) (9)	37 (18.5)	40 (13) 39 (14)	$\frac{42}{51}$ (1.5)	$\frac{42}{10}$ 51 (1.5)	$\begin{pmatrix} 42 & 10 \\ 5 & 47 & 5 \end{pmatrix}$	34	(23)	40 ((11) 3 (11) 3	38 (16)	37 (1 37 (1) 30 (8.5)37 (18.5)	35(21.5)) 40 ((14)	30 (2.3 44 (7)
	36	(22)	41 (13.5)	39(18)	44 (8.5)	(16) 40 (16)	(45(7))	42	(12)	35	(23)	40(16)	44 (8	$(5)^{41}$	13.5)	47(3.5)	43	(10.5) 4	46(5.5)
~ ~	36	(11)	$\begin{array}{c} 43 \ (16) \\ 38 \ (12.5) \end{array}$	38 (21) 32 (20.5)	$\begin{array}{c} 47 & (10) \\ 44 & (5) \end{array}$	49(6.5 43(6.5	$\begin{array}{cccccccccccccccccccccccccccccccccccc$) 43) 42	(16) (8)	··· 50 ((5) 4 (6.5) 4	$\begin{array}{c} 43 & (16) \\ 41 & (9.5) \end{array}$	$^{47}(1)$	$\begin{array}{c} 0 \\ 2 \\ 2 \\ 46 \\ 6 \end{array}$	18.5) 4) ::	$44 (13.5) \\ 36 (15)$) $52 \\ 41$	(3) (9.5) 4	$47 (10) \\ 47 (3)$
	40	8 (8)	40(20)	48 (8)	45 (12.5	5) 43 (17)) 50 (4.5	() 49	(9)	44 ((15.5) 4	48 (8)	58 (1) 38 (22.5)	46(10)	44	(15.5) 5	51(3)
-	43	: : (7.5)	48 (2)	· · · · 6	(E)	42 (11)		42			(20.5) 3		45 (4) 35 (43 (7.5)	32	: : (20.5) 3	31 (22.
~	41	1		37 (16)	(1) 70		47 (35) 40	(13.5)	29 ((23.5) 3	38 (17)	44 (7) 49 (1)	33 (21.5)) 42	(10) 3	37 (18.

Table B.38: A preview of the hyperareas (and corresponding ranks) obtained for run 1-100 for hyperparameter combinations A2.5.1-A2.5.24. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

254

Stellenbosch University https://scholar.sun.ac.za

	We	prst	Be	est
	HA	NDS	HA	NDS
A2.5.1	Run 13	Run 13	Run 84	Run 63
A2.5.2	Run 9	$\operatorname{Run}34$	$\operatorname{Run}73$	$\operatorname{Run}25$
A2.5.3	$\operatorname{Run}44$	$\operatorname{Run}79$	$\mathrm{Run}\ 47$	Run 38
A2.5.4	$\operatorname{Run}75$	$\operatorname{Run}57$	$\operatorname{Run}81$	Run 73
A2.5.5	$\operatorname{Run}28$	Run 29	Run 18	Run 99
A2.5.6	$\operatorname{Run}75$	Run 16	Run 39	Run 35
A2.5.7	Run 9	$\operatorname{Run}57$	Run 16	Run 47
A2.5.8	Run 6	$\operatorname{Run}74$	$\operatorname{Run}47$	Run 18
A2.5.9	Run 68	Run 32	$\operatorname{Run}42$	Run 78
A2.5.10	$\operatorname{Run}27$	Run 99	Run 84	Run 74
A2.5.11	Run 91	Run 98	Run 93	Run 16
A2.5.12	$\operatorname{Run}34$	Run 87	Run 10	$\operatorname{Run} 82$
A2.5.13	$\operatorname{Run}75$	Run 98	$\operatorname{Run}20$	$\operatorname{Run}52$
A2.5.14	Run 75	Run 9	Run 73	Run 93
A2.5.15	$\operatorname{Run}27$	Run 92	Run 39	Run 96
A2.5.16	Run 75	$\operatorname{Run}28$	Run 73	Run 19
A2.5.17	Run 99	Run 60	$\operatorname{Run}56$	Run 86
A2.5.18	$\operatorname{Run}27$	Run 75	$\operatorname{Run}47$	Run 92
A2.5.19	Run 75	$\operatorname{Run}72$	Run 86	Run 36
A2.5.20	Run 6	Run 2	Run 10	Run 48
A2.5.21	$\operatorname{Run}75$	Run 46	Run 39	Run 81
A2.5.22	$\operatorname{Run}32$	$\operatorname{Run} 2$	$\operatorname{Run}36$	Run 96
A2.5.23	$\operatorname{Run}75$	$\operatorname{Run}97$	Run 39	$\operatorname{Run}92$
A2.5.24	Run 75	Run 26	Run 62	Run 33

Table B.40: The simulation runs that correspond to the best and worst approximation fronts (obtained by the NSGA-II) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A2.5.1–A2.5.24.

	2 A4.U.	10.14			0.74.0		.0 A2.0.	9 AZ-0.1(11.0.24	A4.5.14	01.0.2 H	174.U.14	01.0.74	01.0.28	11.0.24	AZ-0-10	01.0.40	04.0.40		44.0.44		
5.1 1	1	1	1	1	1	1	0	1	1	0.02	1	1	1	1	1	1	1	1	1	1	1	1
5.2	1	1	1	1	1	1	0	1	1	0.17	1	1	1	1	1	1	1	1	1	1	1	1
5.3		1	1	1	1	1	0.09	1	1	1	1	1	1	0.12	1	1	1	1	1	1	1	1
5.4			1	1	1	1	0	1	0.72	0	1	1	1	1	1	1	1	1	1	1	1	1
5.5				1	1	1	0	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
5.6					1	1	0	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
5.7						1	0.08	1	1	1	1	1	1	0.14	1	1	1	1	1	1	1	1
5.8							0	1	1	1	1	1	1	1	1	1	1		1	1	1	1
5.9								0.002	0.18	1	0	0	0	0	0	0	0	0	0	0	0	0.14
5.10									1	1	1	1	1	1	1	1	1	1	1	1	1	1
5.11										1	1	1	1	0.06	1	1	1	1	1	1	1	1
5.12											0.02	0.01	0.44	0	0.02	0.18	0	1	1	0.01	0.07	1
5.13												1	1	1	1	1	1	1	1	1	1	1
5.14													1	1	1	1	1	1	1		1	1
5.15														1	1	1	1	1	1	1	1	1
5.16															1	1	1	1	1	1	1	0.08
5.17																1	1	1	1	1	1	1
5.18																	1	1	1	1	1	-
5.19																		1	1	1	1	1
5.20																			1	1	1	г
5.21																				1	1	1
5.22																					1	1
5.23																						1

Stellenbosch University https://scholar.sun.ac.za

Table B.41: The adjusted p-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the hyperareas of the approximation fronts for the hyperparameter combinations A2.5.1-A2.5.24. Table B.42: The adjusted p-values obtained by the Nemenyi post hoc test for the multiple comparisons based on the number of non-dominated solutions

A2.	5.2 A2	.5.3 A2.	5.4 A2.	.5.5 A2	2.5.6 A2.	5.7 A2.	5.8 A2.5	.9 A2.5.1	10 A2.5.3	11 A2.5.1	12 A2.5.1	.3 A2.5.1	4 A2.5.1	5 A2.5.16	A2.5.17	A2.5.18	A2.5.19	A2.5.20	A2.5.21	A2.5.22	A2.5.25	A2.5.2
A2.5.1 1		1 1		_	1	. 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
A2.5.2		1 1		_	1	. 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
A2.5.3		1		_	1	. 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
A2.5.4			-	_	1 0.8	38 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
A2.5.5					1	. 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
A2.5.6					-		1	1	1	1	1	1	1	0.63	1	0.78	1	1	1	1	1	1
A2.5.7						1	1	1	1	1	1	1	1	0.01	0.37	0.02	0.07	1	1	0.049	1	1
A2.5.8							1	1	1	1	1	1	1	0.03	0.65	0.04	0.13	1	1	0.09	1	1
A2.5.9								1	1	1	1	1	1	0.04	0.81	0.046	0.17	1	1	0.12	1	1
A2.5.10									1	1	1	1	1	1	1	1	1	1	1	1	1	1
A2.5.11										1	1	1	1	0.23	1	0.29	0.92	1	1	0.69	1	1
A2.5.12											1	1	1	1	1	1	1	1	1	1	1	1
A2.5.13												1	1	1	1	1	1	1	1	1	1	1
A2.5.14													1	1	1	1	1	1	1	1	1	1
A2.5.15														1	1	1	1	1	1	1	1	1
A2.5.16															1	1	1	0.02	1	1	1	0.82
A2.5.17																1	1	0.43	1	1	1	1
A2.5.18																	1	0.02	1	1	1	1
A2.5.19																		0.08	1	1	1	1
A2.5.20																			1	0.06	1	1
A2.5.21																				1	1	1
A2.5.22																					1	1
19 5 93																						



Figure B.35: The average number of non-dominated solutions obtained for hyperparameter combinations A2.1–A2.24 for the respective simulation problems.

This section presents the results for the hyperparameter study conducted for the DBMOSA.

B.3.1 Open mine problem

Tables B.43 and B.44 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–40 for hyperparameter combinations A3.1.1– A3.1.4. In Figure B.36 the worst and best approximation fronts are plotted for the respective hyperparameter combinations (or move operators) A3.1.1–A3.1.4, for the respective simulation problems as summarised in Table B.45.

B.3.2 (s, S) Inventory problem

Tables B.46 and B.47 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A3.2.1– A3.2.4. In Figure B.37 the worst and best approximation fronts are plotted for the respective hyperparameter combinations (or move operators) A3.2.1–A3.2.4, for the respective simulation problems as summarised in Table B.48.

B.3.3 Buffer allocation problem: five machines

Tables B.49 and B.50 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A3.3.1– A3.3.4. In Figure B.38 the worst and best approximation fronts are plotted for the respective hyperparameter combinations (or move operators) A3.3.1–A3.3.4, for the respective simulation problems as summarised in Table B.51.

B.3.4 Buffer allocation problem: 10 machines

Tables B.52 and B.53 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A3.4.1– A3.4.4. In Figure B.39 the worst and best approximation fronts are plotted for the respective hyperparameter combinations (or move operators) A3.4.1–A3.4.4, for the respective simulation problems as summarised in Table B.54.

B.3.5 Non-linear buffer allocation problem: 16 machines

Tables B.55 and B.56 present a preview of the hyperareas and number on non-dominated solutions and their corresponding ranks for run 1–100 for hyperparameter combinations A3.5.1– A3.5.4. In Figure B.40 the worst and best approximation fronts are plotted for the respective hyperparameter combinations (or move operators) A3.5.1–A3.5.4, for the respective simulation problems as summarised in Table B.57.

In Figure B.41 the average number of non-dominated solutions obtained for hyperparameter combinations A3.1–A3.4 is illustrated for the respective simulation problems.

Table B.43: A preview of the hyperareas (and corresponding ranks) obtained for run 1–40 for hyperparameter combinations A3.1.1–A3.1.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.1.1	A3.1.2	A3.1.3	A3.1.4
1	9 199.23 (2)	5 384.99 (4)	8898.58 (3)	9678.68 (1)
2	9117.74~(1)	6114.05~(3)	7637.54~(2)	5675.39~(4)
3	$10800.56\ (1)$	7399.36(3)	7021.79~(4)	9473.27(2)
4	7334.52~(3)	8640.86(1)	7532.07~(2)	6736.11(4)
5	7811.52~(3)	8018.12(1)	7969.69~(2)	6519.96~(4)
6	6701.76(3)	4 420.00 (4)	8274.31(1)	7573.69 (2)
7	8122.81 (2)	5582.25~(4)	7408.69~(3)	9303.05~(1)
8	6454.90 (2)	5577.79 (3)	5243.40 (4)	7952.99~(1)
9	6447.06~(3)	5103.89~(4)	8112.39(1)	7099.57~(2)
10	12070.97 (1)	4620.59~(4)	10656.92~(2)	8936.71 (3)
:	•	•	:	÷
39	6518.99(3)	7925.88(1)	7124.13~(2)	6054.66~(4)
40	$6\ 500.90\ (4)$	$6\ 909.59\ (3)$	10594.92~(1)	8848.14 (2)
$\sum R_{Ci}$	90	109	88	113
$\sum R_{Ci}^2$	8 100	11881	7744	12769

Table B.44: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–40 for hyperparameter combinations A3.1.1–A3.1.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

A3.1.1	A3.1.2	A3.1.3	A3.1.4
24(2)	14(4)	23(3)	25(1)
24(1)	16(3)	20(2)	15(4)
27(1)	20(3)	19(4)	24(2)
19(3)	22(1)	21(2)	17(4)
20(2.5)	21(1)	20(2.5)	18 (4)
18(3)	12 (4)	21 (1)	20(2)
21 (2)	16(4)	19(3)	25(1)
18(2)	15(3)	14 (4)	20 (1)
17(3)	14 (4)	21 (1)	19(2)
31(1)	14 (4)	27(2)	24(3)
:	:	:	:
·	00 (1)	·	•
17(3)	20(1)	19(2)	16(4)
18(3.5)	18(3.5)	27(1)	22(2)
93	109	88.5	109.5
8 6 4 9	11 881	7832.25	11990.25
	A3.1.1 24 (2) 24 (1) 27 (1) 19 (3) 20 (2.5) 18 (3) 21 (2) 18 (2) 17 (3) 31 (1)	A3.1.1 A3.1.2 24 (2) 14 (4) 24 (1) 16 (3) 27 (1) 20 (3) 19 (3) 22 (1) 20 (2.5) 21 (1) 18 (3) 12 (4) 21 (2) 16 (4) 18 (2) 15 (3) 17 (3) 14 (4) 31 (1) 14 (4) 17 (3) 20 (1) 18 (3.5) 18 (3.5)	A3.1.1 A3.1.2 A3.1.3 24 (2) 14 (4) 23 (3) 24 (1) 16 (3) 20 (2) 27 (1) 20 (3) 19 (4) 19 (3) 22 (1) 21 (2) 20 (2.5) 21 (1) 20 (2.5) 18 (3) 12 (4) 21 (1) 21 (2) 16 (4) 19 (3) 18 (2) 15 (3) 14 (4) 17 (3) 14 (4) 27 (2) \vdots \vdots \vdots 17 (3) 20 (1) 19 (2) 18 (3.5) 18 (3.5) 27 (1) 93 109 88.5 8 649 11 881 7 832.25

Table B.45: The simulation runs that correspond to the best and worst approximation fronts (obtained by the DBMOSA) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A3.1.1–A3.1.4.



Figure B.36: The best and worst best approximation fronts for hyperparameter combinations (or move operators) A3.1.1–A3.1.4.

Table B.46: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.2.1–A3.2.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.2.1	A3.2.2	A3.2.3	A3.2.4
1	112804.07(1)	111 381.09 (3)	108 706.16 (4)	111633.64 (2)
2	112712.34(1)	111546.09 (3)	111315.14 (4)	112078.12 (2)
3	$113925.8\ (1)$	$113110.8\ (2)$	105669.79(4)	110381.22 (3)
4	112902.02 (2)	113541.5(1)	111165.58~(4)	112155.7 (3)
5	$112666.71\ (1)$	111847.99 (3)	108289.54~(4)	112666.35(2)
6	113200.88(1)	111054.7 (3)	110461.12~(4)	112589.25~(2)
7	$113016.04\ (1)$	111289.27~(2)	100622.78~(4)	$111211.16\ (3)$
8	113544.92~(2)	113597.58(1)	111551.79 (3)	111162.96~(4)
9	112723.49(1)	110520.39(3)	109914.84~(4)	111870.97~(2)
10	113462.11 (1)	111493.86~(4)	112956.93~(3)	$113026.35\ (2)$
:	÷	:	÷	
99	112944.81 (2)	113485.79(1)	112324.02 (3)	112230.69 (4)
100	113225.38(1)	111076.44 (3)	109549.16~(4)	111697.29 (2)
$\sum R_{Ci}$	138	235	331	296
$\sum R_{Ci}^2$	19044	55225	109561	87616

Table B.47: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.2.1–A3.2.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.2.1	A3.2.2	A3.2.3	A3.2.4
1	198 (2)	313(1)	23(4)	145(3)
2	183(2)	259(1)	44 (4)	118(3)
3	156(3)	135(4)	208(2)	219(1)
4	169(1)	149(2)	118(4)	129(3)
5	214(1)	112(3)	102(4)	177(2)
6	167(4)	321(1)	184(3)	204(2)
7	162(3)	414 (1)	80 (4)	165(2)
8	156(3)	163(2)	108(4)	177(1)
9	186(2)	318(1)	147(3)	124(4)
10	137(3)	198(1)	42(4)	189(2)
:	÷	÷	÷	÷
99	160(3)	178(2)	58(4)	214(1)
100	188 (1)	163(2)	66 (4)	117 (3)
$\sum R_{Ci}$	240.5	223.5	309	227
$\sum R_{Ci}^2$	57840.25	49952.25	95481	51529

Table B.48: The simulation runs that correspond to the best and worst approximation fronts (obtained by the DBMOSA) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A3.2.1–A3.2.4.



Figure B.37: The best and worst best approximation fronts for hyperparameter combinations (or move operators) A3.2.1–A3.2.4.

Table B.49: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.3.1–A3.3.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.3.1	A3.3.2	A3.3.3	A3.3.4
1	33.67(2)	33.65(3)	33.72(1)	33.54 (4)
2	33.70(1)	33.42(3)	33.61(2)	33.20(4)
3	33.73(1)	33.69(2)	33.49(3)	33.38(4)
4	33.67(1)	33.53(4)	33.63(3)	33.65(2)
5	33.63(2)	33.46(3)	33.65(1)	33.42(4)
6	33.64(3)	33.71(1)	33.70(2)	33.22(4)
7	33.68(2)	33.54(4)	33.72(1)	33.65(3)
8	33.65(1)	33.19(4)	33.46(2)	33.44(3)
9	33.70(1)	33.45(3)	33.45(2)	33.19(4)
10	33.75(1)	33.43(4)	33.59(3)	33.69(2)
÷		:	÷	÷
99	33.52(3)	33.59(2)	33.07(4)	33.67(1)
100	32.20(4)	33.24(3)	33.48(2)	33.50(1)
$\sum R_{Ci}$	164	315	242	279
$\sum R_{Ci}^2$	26896	99225	58564	77841

Table B.50: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.3.1–A3.3.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.3.1	A3.3.2	A3.3.3	A3.3.4
1	62(1)	55(2)	54(3)	46 (4)
2	60(1)	49(4)	52(2.5)	52(2.5)
3	76(1)	39(4)	48(2)	41 (3)
4	55(1)	43(4)	54(2)	46(3)
5	60(1)	31(4)	55(2)	40(3)
6	64(2)	50(4)	65(1)	53(3)
7	52(2)	47 (4)	63(1)	49(3)
8	56(1)	46(3)	39(4)	52(2)
9	63(1)	46(2)	45(3)	43(4)
10	77 (1)	39(4)	56(2)	43(3)
:	:	÷	:	•
99	67(1)	56(3)	62(2)	44 (4)
100	41 (2)	38 (3)	59(1)	37(4)
$\sum R_{Ci}$	139.5	338	214	308.5
$\sum R_{C_i}^2$	19460.25	114244	45796	95172.25

Table B.51: The simulation runs that correspond to the best and worst approximation fronts (obtained by the DBMOSA) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A3.3.1–A3.3.4.



Figure B.38: The best and worst best approximation fronts for hyperparameter combinations (or move operators) A3.3.1–A3.3.4.

Table B.52: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.4.1–A3.4.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.4.1	A3.4.2	A3.4.3	A3.4.4
1	22.1(1)	21.86(4)	21.98(2)	21.97(3)
2	21.36(1)	20.87(3)	21.16(2)	20.78(4)
3	22.02(1)	21.29(2)	20.98(3)	19.57 (4)
4	21.76(1)	21.37(3)	20.78(4)	21.59(2)
5	21.8(1)	21.32(2)	21.18(3)	21.07(4)
6	21.05(2)	21.11(1)	19.81(3)	18.81(4)
7	21.45(2)	20.18(4)	20.94(3)	21.45(1)
8	21.65(1)	20.53(4)	20.64(3)	21(2)
9	21.36(1)	20.25(3)	20.46(2)	19.95~(4)
10	21.97(1)	19.37(4)	20.32(3)	20.7(2)
:	:	:	:	:
99	21.23(2)	19.89(4)	21.04(3)	21.64(1)
100	21.12 (3)	19.56(4)	21.14 (2)	21.25(1)
$\sum R_{Ci}$	145	327	274	254
$\sum R_{Ci}^2$	21025	106929	75076	64516

Table B.53: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.4.1–A3.4.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.4.1	A3.4.2	A3.4.3	A3.4.4
1	35(3)	31 (4)	39(1)	38 (2)
2	40 (1)	27(4)	29(3)	33(2)
3	46(1)	28(3)	33(2)	26(4)
4	39(2)	31(3.5)	31(3.5)	41 (1)
5	40 (1)	26(4)	33(3)	35(2)
6	34(1)	29(2)	27(3)	19(4)
7	36(2)	29(4)	35(3)	42 (1)
8	40 (1)	30(3.5)	30(3.5)	36(2)
9	33(1.5)	24(4)	33(1.5)	28(3)
10	47 (1)	29(3.5)	34(2)	29(3.5)
:	•	•	•	•
99	43 (1)	28(4)	34(2)	32(3)
100	32 (2)	29 (4)	41 (1)	31 (3)
$\sum R_{Ci}$	148.5	328.5	250	273
$\sum R_{C_i}^2$	22052.25	107912.25	62500	74529

Table B.54: The simulation runs that correspond to the best and worst approximation fronts (obtained by the DBMOSA) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A3.4.1–A3.4.4.



Figure B.39: The best and worst best approximation fronts for hyperparameter combinations (or move operators) A3.4.1–A3.4.4.

Table B.55: A preview of the hyperareas (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.5.1–A3.5.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.5.1	A3.5.2	A3.5.3	A3.5.4
1	205.74(2)	178.94(4)	203.61 (3)	205.85(1)
2	183.55(3)	182.99(4)	191.91(2)	200.81(1)
3	205.97(1)	184.54(2)	179.41(3)	178.97(4)
4	180.93(1)	174.15(4)	174.74(3)	179.45(2)
5	183.24(1)	172.86(3)	164.59(4)	177.03(2)
6	174.51(3)	179.59(2)	170.54(4)	187.56(1)
7	191.64(1)	173.07(4)	179.66(3)	182.57(2)
8	188.19(1)	173.72(4)	179.59(2)	179.59(3)
9	188.33(1)	164.58(4)	181.09(2)	171.32(3)
10	202.83(1)	179.86(2)	173.66(4)	177.67(3)
:	•	•	:	:
99	178.91(3)	184.86(2)	188.48 (1)	166.98(4)
100	182.53(2)	177.85(3)	172 (4)	182.68 (1)
$\sum R_{Ci}$	153	318	269	260
$\sum R_{Ci}^2$	23409	101124	72361	67600

Table B.56: A preview of the number of non-dominated solutions (and corresponding ranks) obtained for run 1–100 for hyperparameter combinations A3.5.1–A3.5.4. The ranks are given in the parentheses and are used in the computation of the Friedman and Iman-Davenport hypothesis tests.

Run	A3.5.1	A3.5.2	A3.5.3	A3.5.4
1	42 (1)	31(4)	37(2)	33(3)
2	31(4)	33(2)	37(1)	32(3)
3	53(1)	31(3)	27(4)	36(2)
4	32(2.5)	32(2.5)	35(1)	30(4)
5	35(1)	29(3)	25(4)	33(2)
6	33(2)	28(4)	34(1)	30 (3)
7	31(3)	26(4)	35(1.5)	35(1.5)
8	32(2)	31(3)	36(1)	28(4)
9	33(1)	26(3)	32(2)	25(4)
10	38(1)	28(2.5)	25(4)	28(2.5)
:	:	:	:	:
99	.34(3)	.40(2)	49 (1)	28 (4)
100	32 (1)	26 (4)	27 (3)	31 (2)
$\sum R_{Ci}$	163.5	306	266.5	264
$\sum R_{Ci}^2$	26732.25	93636	71022.25	69 696

Table B.57: The simulation runs that correspond to the best and worst approximation fronts (obtained by the DBMOSA) for the hyperarea performance indicator and the number of non-dominated solutions found for hyperparameter combinations A3.5.1–A3.5.4.



Figure B.40: The best and worst best approximation fronts for hyperparameter combinations (or move operators) A3.5.1–A3.5.4.



Figure B.41: The average number of non-dominated solutions obtained for hyperparameter combinations A3.1–A3.4 for the respective simulation problems.

APPENDIX C

The Backpropagation Training Algorithm

This appendix contains details concerning the backpropagation training algorithm for ANNs training using gradient methods, namely, gradient descent and stochastic gradient descent. The derivation of the backpropagation of errors is presented in the context of a single layered perceptron. This derivation is a modified reproduction of the derivatives presented in [30, 158, 177, 194, 159].

Furthermore, the gradient-based optimisation technique employed in the training algorithm is also elaborated upon. This appendix should be read in conjunction with Chapter 2, as the terminologies and notations employed in this appendix are introduced there.

The purpose of the backpropagation algorithm is to find suitable values for the network weights between the input and hidden layer as well as between the hidden and the output layer. This is achieved by minimising the error function, denoted by E. The individual errors $\{E_1, \ldots, E_k, \ldots, E_m\}$, where m is the number of outputs, may be defined separately, where E is the summation of the individual errors, *i.e.* $\sum_{k=1}^{m} E_i$. Assume that the individual errors can be expressed as a differentiable function of the network outputs.

The backpropagation procedure obtains the derivatives of the error function, where the error function is influenced by the network weights, biases and activations. Note that only the weights and biases may be changed to minimise E. The weights and biases are updated in proportion to their contribution to E.

C.1 Backpropagation



Figure C.1 is used to help elucidate the process followed to perform backpropagation.

FIGURE C.1: Backpropagation visual representation of the chain rule.

Assume that input training vector \boldsymbol{x} has been propagated forward through the network, therefore the activations of the hidden and output neurons have been calculated using (2.9)–(2.12) and are rewritten in the context of Figure C.1 to simplify the explanation, such that

$$z^{L} = w^{L}a^{L-1} + b^{L}, (2.9), (C.1)$$

$$a^L = \sigma(z^L), \tag{C.2}$$

$$z^{L-1} = w^{L-1}a^{L-2} + b^{L-1},$$
 (2.11), (C.3)

$$a^{L-1} = \sigma(z^{L-1}),$$
 (2.12). (C.4)

The output layer (L):

The partial derivative of the error function given the weight in layer L is,

$$\frac{\partial E}{\partial w^L} = \frac{\partial E}{\partial a^L} \frac{\partial a^L}{\partial z^L} \frac{\partial z^L}{\partial w^L},\tag{C.5}$$

where the partial derivatives are defined as follows. Given $E = \frac{1}{2}(a^L - y)^2$, the derivative is

$$\frac{\partial E}{\partial a^L} = (a^L - y). \tag{C.6}$$

Given $a^L = \sigma(z^L)$, the derivative is

$$\frac{\partial a^L}{\partial z^L} = \sigma'(z^L). \tag{C.7}$$

Given $z^L = w^L a^{L-1} + b^L$, the derivative is

$$\frac{\partial z^L}{\partial w^L} = a^{L-1}.$$
 (C.8)

Substituting (C.6)-(C.8) into (C.5) yields,

$$\frac{\partial E}{\partial w^L} = (a^L - y) \ \sigma'(z^L) \ a^{L-1}.$$
(C.9)

The hidden layer (L-1):

The partial derivative of the error function given the weight in layer L-1 is,

$$\frac{\partial E}{\partial w^{L-1}} = \frac{\partial E}{\partial a^{L-1}} \frac{\partial a^{L-1}}{\partial z^{L-1}} \frac{\partial z^{L-1}}{\partial w^{L-1}} \tag{C.10}$$

where the partial derivatives are defined as follows

$$\frac{\partial E}{\partial a^{L-1}} = \sum_{k=1}^{m} \frac{\partial E_k}{\partial a^{L-1}}, \tag{C.11}$$

$$= \sum_{k=1}^{m} \frac{\partial E_k}{\partial z_k^L} \frac{\partial z_k^L}{\partial a_k^{L-1}}, \qquad (C.12)$$

$$= \sum_{k=1}^{m} \frac{\partial E_k}{\partial a_k^L} \frac{\partial a_k^L}{\partial z_k^L} \frac{\partial z_k^L}{\partial a_k^{L-1}}, \qquad (C.13)$$

C.1. Backpropagation

and since backpropagation is performed on a MFNN with one hidden layer (C.13), can simplify to

$$\frac{\partial E}{\partial L} = \frac{\partial E}{\partial L} \frac{\partial a^L}{\partial L} \frac{\partial z^L}{\partial L}, \qquad (C.14)$$

$$\frac{\partial L}{\partial a^L} = \frac{\partial L}{\partial a^L} \frac{\partial L}{\partial z^L} \frac{\partial L}{\partial a^{L-1}},$$

$$= (a^L - u) \sigma'(z^L) w^L.$$
(C.14)
(C.15)

$$= (u - g) o (z) w$$
. (0.13)

The partial derivative of the activation of layer L-1 with respect to the weighted sum of layer L-1 is given by

$$\frac{\partial a^{L-1}}{\partial z^{L-1}} = \sigma'(z^{L-1}). \tag{C.16}$$

The partial derivative of the activation of layer L-1 with respect to the weighted sum of layer L-1 is given by

$$\frac{\partial z^{L-1}}{\partial w^{L-1}} = a^{L-2}. \tag{C.17}$$

Substituting (C.14)–(C.17) into (C.10) yields

$$\frac{\partial E}{\partial w^{L-1}} = (a^L - y) \ \sigma'(z^L) \ w^L \ \sigma'(z^{L-1}) \ a^{L-2}.$$
(C.18)

The weights are updated as follows:

$$w^{L^+} = w^L - \eta \ \frac{\partial E}{\partial w^L},\tag{C.19}$$

$$w^{L-1^+} = w^{L-1} - \eta \, \frac{\partial E}{\partial w^{L-1}}, \qquad (C.20)$$

where w^{L^+} represents the updated weight for layer L, w^{L-1^+} the updated weight in layer L-1 and η is the learning rate.

Appendix C. The Backpropagation Training Algorithm

APPENDIX D

Discrete-event Simulation Optimisation Problem Definitions

This appendix defines each of the bi-objective simulation optimisation problems studied. First, the (s, S) inventory problem (IP) is presented. Next, a well-researched problem, the bufferallocation problem (BAP) is studied as well as instances thereof and lastly, the Open mine problem (OMP) is presented. The experiments used to determine a sensible upper bound and the sufficient number of observations per solution are also defined, where applicable, for the respective problems.

D.1 The (s, S) inventory problem

[The content of this section is an extract from unpublished notes, from the University of Stellenbosch (US), and are included here to define the problem.]

The (s, S) IP used in this study is dynamic and stochastic in nature as it deals with uncertain demand, where s refers to the *re-order level* and S to the *re-order quantity*. For a detailed description of the problem, the reader is referred to [19].

A company sells a single commodity and has to determine the level of inventory necessary for the following n months in order to satisfy the demand. The inventory level can be calculated as stock on hand minus backorders minus demand and the inventory position as inventory level plus outstanding orders. Customer inter-arrival times follow an exponential distribution with a mean of 3 minutes. Assume that the demand follows a distribution of [Weibull(1, 8)] for each customer i and order lead time is distributed TRI(12,14,20) hours. In order to know what quantity should be ordered, the company assesses the inventory level after each sale. Once the inventory level drops below the reorder point s after the sale, a quantity of S items are ordered. Due to the order lead time and the continuity of demand, there exists a possibility that the inventory level will reach zero. If this is the case, and replenishment stock has not yet arrived, a stock-out period is experienced and customer demand cannot be fulfilled, which leads to lost sales.

When a customer requests a certain number of units x, the inventory level I will drop by x:

$$I_{t+1} = I_t - x.$$
 (D.1)

Once the inventory level drops below the re-order level s, a supplier is notified. The supplier will dispatch a quantity equal to the re-order quantity S. There is a delay (in hours), distributed TRI(1,2,4), between the notification and the delivery of S. It is assumed that the supplier is reliable and will always deliver the number S units, but it is delayed as stated earlier. Ignoring possible capacity constraints for the moment, a delivery has the following effect on inventory:

$$I_{t+1} = I_t + S. \tag{D.2}$$

276 Appendix D. Discrete-event Simulation Optimisation Problem Definitions

If the inventory level drops too low to be able to meet customer demand, customer satisfaction is affected negatively. Assume that a customer whose exact demand cannot be met, will not be satisfied with an alternative amount and will leave without having drawn cash.. The seller therefore has to maintain an inventory level high enough to meet most customers' demand. Refer to the percentage of demand met as service level and it is defined in (D.3).

Service level =
$$\frac{\text{Number of customers serviced}}{\text{Number of customers requiring service}}$$
. (D.3)

There is an inventory cost C_I associated with inventory. Inventory being stored has to be paid for, and each delivery also has a cost associated with it. Delivery cost is denoted by C_D . The break down of the total cost C_T incurred is shown in (D.4), (D.5) and (D.6).

$$C_T = C_D + C_I,\tag{D.4}$$

where

$$C_D = d \times c_d,\tag{D.5}$$

and

$$C_I = \sum_{i=1}^{n} 0.06 \times u_i,$$
 (D.6)

where u_i is the number of units in store at the end of day *i* and 0.06 is the holding cost per unit per day. A single delivery *d* costs R120 (this is denoted by c_d) and *n* denotes the total number of days. The goal is to minimise *total inventory cost* while maximising *service level*. The typical inventory consumption and replenishment process is shown in Figure D.1.



Figure D.1: Some characteristics of the generalised (s, S) inventory problem

Figure D.2 shows the simulation model as it is built in Tecnomatix and represents the IP.



Figure D.2: A screenshot of the IP model in Tecnomatix.

Observations per solution

Table D.1 shows the experiments used to determine whether or not 100 observations per solution are sufficient for the IP and is summarised in Table D.2.

Table D.1: Experiments used to determine a sufficient number of observations per solution for the IP

			Tota	l inventory	$\cos t$	_	S	ervice lev	vel
	s	S	А	В	С		А	В	\mathbf{C}
Exp 1	50	50	744.08	750.95	744.58	2	8.98	29.22	28.66
Exp 2	100	100	816.61	814.90	809.19	4	0.05	40.43	39.92
Exp 3	150	150	914.15	909.34	906.91	5	1.20	51.69	51.07
Exp 4	200	200	1049.33	1044.23	1042.70	6	2.54	63.37	62.32
Exp 5	250	250	1209.69	1200.14	1194.99	7	2.55	72.83	72.15
Exp 6	300	300	1361.72	1382.71	1380.13	8	31.08	81.08	80.47
Exp 7	350	350	1594.72	1604.73	1599.25	8	37.96	87.72	86.84
Exp 8	400	400	1863.48	1847.64	1840.77	g	1.62	91.70	91.06
Exp 9	450	450	2118.69	2122.16	2121.81	g	3.54	94.04	93.22
$\mathrm{Exp}~10$	500	500	2434.06	2438.60	2417.66	ç	3.77	94.94	94.25

Table D.2: Summary of the experiments used to determine the sufficient number of observations per solution for the IP.

		Tota	al inventory	cost		Service level				
Groups	Count	Sum	Average	Variance	Sum	Average	Variance			
Exp 1	3	2239.61	746.54	14.66	86.86	28.95	0.08			
Exp 2	3	2440.69	813.56	15.07	120.40	40.13	0.07			
Exp 3	3	2730.41	910.14	13.60	153.96	51.32	0.11			
Exp 4	3	3136.26	1045.42	12.04	188.23	62.74	0.31			
Exp 5	3	3604.82	1201.61	55.64	217.54	72.51	0.12			
Exp 6	3	4124.56	1374.85	131.02	242.63	80.88	0.12			
Exp 7	3	4798.70	1599.57	25.13	262.52	87.51	0.34			
Exp 8	3	5551.89	1850.63	135.72	274.38	91.46	0.12			
Exp 9	3	6362.66	2120.89	3.64	280.80	93.60	0.17			
Exp 10	3	7290.32	2430.11	121.29	282.96	94.32	0.34			

278 Appendix D. Discrete-event Simulation Optimisation Problem Definitions

D.2 The buffer-allocation problems

[The content of this section is an extract from [21] and is included here with permission from the author.]

Finite queuing networks are associated with many practical systems through which discrete or continuous flow occurs, such as manufacturing systems and telecommunication networks. These networks often exhibit flow variation or asynchronous part movement; hence the need for buffer space in the network. One of the network design priorities is to maximise the network throughput or throughput rate, which increases with more buffer space [53].

However, buffer space may be costly for several reasons in commercial projects and costs must be minimised. This gives rise to the BAP which is usually formulated as a stochastic, non-linear, integer mathematical programming problem and which is computationally hard to solve [54]. The problem has several variations, including problems with reliable or unreliable servers (for example, manufacturing machines), the type of distribution assigned to service time, repair times and time-to-failure, synchronous or asynchronous part movement, and the network topology.

Papadopoulos & Heavey [202] classified queuing network models for production and transfer lines, while [54] identified four traditional methodological approaches to the BAP: simulation methods, metaheuristics, dynamic programming and search methods. In some studies the processing times were assumed to be deterministic, while time-to-failure and repair times were exponentially distributed. Researchers also adopt different performance measures (objectives) and consider single or multi-objective problem variants, while decision variables include buffer sizes and server processing rates.

A production line consists of a series of m machines with m-1 buffers (or niches). Discrete parts are processed by each of these machines in sequence, and each discrete part takes up one buffer space or occupies a machine. There are n such spaces available, while the spaces in front of the first and beyond the last machine are considered infinite.

Generally, the machines have exponentially distributed processing and repair times with mean rates μ_i and r_i , respectively. The machine failures of the models in this study are operation dependent failures (ODF), which are more realistic than time-based failures [259]. A machine thus fails after a number of operations has been completed, and these operation counts are dictated by Poisson distributions with rates β_i .

Note that the ODF approach results in longer times between failures when finite buffers are required. Suppose the time between failures for Machine 1 is exponentially distributed with rate β_1 and the repair rate is r_1 . Then the expected number of failures for this machine for a simulation run length of T, when infinite buffers are used, is

$$\frac{T}{\frac{1}{\beta_1} + \frac{1}{r_1}}.$$

When the buffer sizes are limited, this number decreases and the time between failures increases. An upstream machine can become blocked when its successor has failed, while a downstream machine can eventually become starved if its predecessor has failed. The basic performance measures of such a system are the throughput rate and the *work-in-progress* (WIP).

The objective is to determine the best buffer allocations, given a certain size of n. For example if n = 10, there are 10 buffers slots that can each hold a part. All 10 of these slots can be assigned to one machine and zero to the other three machines. Two buffers slots can be assigned, each

to three machines and four buffers slots to the remaining machine. The goal is to minimise *work-in-progress* while maximising the *throughput*.

The three instances of the BAP are discussed in the next sections.

D.2.1 The buffer-allocation problem: five machines

The linear buffer allocation problem with m = 5 (BAP5) has exponential processing and repair times, schematically presented in Figure D.3.

$$B_0 = \infty \xrightarrow{B_1} M_2 \xrightarrow{B_2} M_3 \xrightarrow{B_3} M_4 \xrightarrow{B_4} M_5 \xrightarrow{B_5 = \infty}$$

Figure D.3: A series of five machines M_1, \ldots, M_5 with finite buffers B_1, \ldots, B_{5-1} in a queuing network, representative of BAP5

Table D.3 shows the processing times for machines $M_1 - M_5$, respectively, note that the processing times become progressively faster.

Table D.3: Processing times for the machines in BAP5

	M1	M2	M3	M4	M5
Processing time	1:00:00	55:00	50:00	46:00	43:00

Figure D.4 shows the simulation model as it is built in Tecnomatix and represents BAP5.



Figure D.4: A screenshot of the BAP5 model in Tecnomatix.

Upper bounds

Table D.4 shows the experiments used to determine a sensible upper bound for BAP5.

Observations per solution

Table D.5 shows the experiments used to determine whether or not 100 observations per solution are sufficient for BAP5 and is summarised in Table D.6.

280 Appendix D. Discrete-event Simulation Optimisation Problem Definitions

	B1	B2	B3	B4	Work-in-progress	Throughput
Exp 1	10	10	10	10	2.13	91.83
Exp 2	12	12	12	12	2.28	92.14
Exp 3	14	14	14	14	2.38	92.23
Exp 4	16	16	16	16	2.43	92.31
Exp 5	17	17	17	17	2.46	92.28
Exp 6	18	18	18	18	2.48	92.30
Exp 7	19	19	19	19	2.49	92.34
Exp 8	50	50	50	50	2.54	92.34
Exp 9	100	100	100	100	2.54	92.34

Table D.4: The experiments used to determine a suiteable upper bound for BAP5

Table D.5: Experiments used to determine a sufficient number of observations per solution for BAP5

					Work-in-progress			T	hroughp	ut
	B1	B2	B3	Β4	А	В	С	А	В	С
Exp 1	1	1	1	1	0.37	0.38	0.39	69.30	70.15	71.34
Exp 2	2	2	2	2	0.73	0.71	0.72	77.10	77.45	78.37
Exp 3	3	3	3	3	0.99	1	1.02	80.40	82.22	82.94
Exp 4	4	4	4	4	1.26	1.21	1.25	83.60	85.54	86.05
Exp 5	5	5	5	5	1.45	1.43	1.46	87.30	87.73	88.02
Exp 6	6	6	6	6	1.73	1.62	1.65	88.30	88.95	89.52
Exp 7	7	7	7	7	1.92	1.76	1.80	89.80	89.87	90.50
Exp 8	8	8	8	8	2.07	1.89	1.93	88.50	90.22	91.08
Exp 9	9	9	9	9	2.05	2	2.04	90.10	90.52	91.59
$\mathrm{Exp}\ 10$	10	10	10	10	2.14	2.06	2.13	89.20	90.80	91.83

Table D.6: Summary of the experiments used to determine the sufficient number of observations per solution for BAP5.

		v	Work-in-pro	gress		Throughput					
Groups	Count	Sum	Average	Variance	Sum	Average	Variance				
Exp 1	3	1.14	0.38	0	210.79	70.26	1.05				
Exp 2	3	2.16	0.72	0	232.92	77.64	0.43				
Exp 3	3	3.00	1.00	0	245.56	81.85	1.71				
Exp 4	3	3.71	1.24	0	255.19	85.06	1.67				
Exp 5	3	4.34	1.45	0	263.05	87.68	0.13				
Exp 6	3	5.00	1.67	0	266.77	88.92	0.38				
Exp 7	3	5.48	1.83	0.01	270.17	90.06	0.15				
Exp 8	3	5.90	1.97	0.01	269.80	89.93	1.72				
Exp 9	3	6.08	2.03	0	272.21	90.74	0.59				
$Exp \ 10$	3	6.34	2.11	0	271.83	90.61	1.75				

D.2.2 The buffer-allocation problem: 10 machines

The linear buffer allocation problem with m = 10 (BAP10) is similar to BAP5, *i.e.* has exponential processing and repair times. BAP10 is schematically presented in Figure D.5. Table D.7 shows the processing times for machines $M_1 - M_{10}$, respectively. Again, note that the processing times become progressively faster. Figure D.6 shows the simulation model as it is built in Tecnomatix and represents BAP10.

$$B_0 = \infty \underbrace{M_1}_{M_2} \underbrace{B_2}_{M_3} \underbrace{M_3}_{M_4} \underbrace{B_4}_{M_5} \underbrace{B_5}_{M_6} \underbrace{B_6}_{M_7} \underbrace{B_7}_{M_8} \underbrace{B_8}_{M_9} \underbrace{B_9}_{M_{10}} \underbrace{B_{10}}_{M_1} = \infty$$

Figure D.5: A series of ten machines M_1, \ldots, M_10 with finite buffers B_1, \ldots, B_{10-1} in a queuing network, representative of BAP10

Table D.7: Processing times for the machines in BAP10

-		M1	M2	M3	M4	M5	M6	M7	M8	M9	M10	-
_	Processing time	1:00:00	55:00	50:00	46:00	43:00	39:00	35:00	31:00	27:00	23:00	_
S. Models	BAD 10 (75%)											-
iventController	Buffer-Allocation model w	ith 10 machines										
User-interface	Model input: Buff, Capacity-2 Buff, Capacity-2 Buff, Capacity-2 Buff, Capacity-7 Buff, Capacity-7 Buff, Capacity-7 Buff, Capacity-7 Buff, Capacity-7 Buff, Capacity-3 Buff, Capacity-3 Buff, Capacity-3 Buff, Capacity-3 Buff, Capacity-3 Buff, Capacity-3 Buff, Capacity-3 Buff, Capacity-3	Model out WP=0.0027 Throughpu MIIFailureTime	s sputx spots spots spots at = 0 Be LatheFailureTimeBe Monocomment at the failureTimeBe mRivetingFailureTimeBe	ndingFailureTBendi	ngFailureTimeS	TableFile Buff TableFile Buff ThroughputCalc CreateEntity	erPunchMaxOccup 2.18421052631579 erBendingMaxOccu AvgBufferSizes=7.21 SumBuffers TableRow=1	r tion=0 BPun BLatt pation1=0 BBen 480041919657 BRive BBen BBen BBen BBen BBen	<pre>rms ch = 1.1022845566578 ie=1.3712445702869 d = 0.5525682625074c d = 0.6827037471754 d2=0.2667224106298 d3=0.7425484627696 d4=0.2446083612555 d5=0.2018349987312</pre>	37 B1_Count=1 1 B1_ODF=0 36 B2_Count= 12 B3_COUF=0 343 B3_COUF=0 343 B3_COUF=0 352 B4_Count= 375 B4_ODF=0 371 B5_Count=	0 85_00F 86_00 0 86_00 87_00F 88_00F 88_00F 0 88_00F 89_00F	:=17 F=21 int=16 F=18 int=3 F=17 int=15 F=25

Figure D.6: A screenshot of the BAP10 model in Tecnomatix.

Observations per solution

Table D.8 shows the experiments used to determine whether or not 100 observations per solution are sufficient for BAP10 and is summarised in Table D.9.

Table D.8: Experiments used to determine a sufficient number of observations per solution for BAP10

										Work-in-progress		5	Throughp	out	
	B1	B2	B3	B4	B5	B6	B7	B8	B9	А	В	С	А	В	С
Exp 1	1	1	1	1	1	1	1	1	1	0.32	0.34	0.34	62	62.74	62.77
Exp 2	2	2	2	2	2	2	2	2	2	0.54	0.61	0.63	66.5	69.64	69.59
Exp 3	3	3	3	3	3	3	3	3	3	0.81	0.84	0.86	71.4	73.67	73.74
Exp 4	4	4	4	4	4	4	4	4	4	1.00	1.02	1.04	71.2	75.94	76.02
Exp 5	5	5	5	5	5	5	5	5	5	1.19	1.15	1.17	72.6	77.64	77.7
Exp 6	6	6	6	6	6	6	6	6	6	1.26	1.27	1.29	76.3	78.39	78.73
Exp 7	7	7	7	7	7	7	7	7	7	1.45	1.38	1.38	72.7	78.35	79.25
Exp 8	8	8	8	8	8	8	8	8	8	1.42	1.43	1.45	74.8	78.93	79.58
Exp 9	9	9	9	9	9	9	9	9	9	1.48	1.5	1.51	77.7	79.12	79.69
$\mathrm{Exp}\ 10$	10	10	10	10	10	10	10	10	10	1.49	1.53	1.55	76.3	78.69	79.93

D.2.3 The non-linear buffer-allocation problem: 16 machines

The non-linear buffer allocation problem with m = 16 (BAP16), proposed by MacGregor Smith *et al.* [174], has exponential processing and repair times as well as probabilistic routings. BAP16 is schematically presented in Figure D.7.

Table D.10 shows the processing times for machines $M_1 - M_{16}$, respectively. However, note that

		V	Work-in-pro	ogress		Throughput				
Groups	Count	Sum	Average	Variance	Sum	Average	Variance			
Exp 1	3	1.00	0.33	0	187.51	62.50	0.19			
Exp 2	3	1.78	0.59	0	205.73	68.58	3.23			
Exp 3	3	2.51	0.84	0	218.81	72.94	1.77			
Exp 4	3	3.06	1.02	0	223.16	74.39	7.61			
Exp 5	3	3.51	1.17	0	227.94	75.98	8.57			
Exp 6	3	3.82	1.27	0	233.42	77.81	1.73			
Exp 7	3	4.21	1.40	0	230.30	76.77	12.61			
Exp 8	3	4.31	1.44	0	233.31	77.77	6.71			
Exp 9	3	4.48	1.49	0	236.51	78.84	1.05			
$\mathrm{Exp}\ 10$	3	4.57	1.52	0	234.92	78.31	3.41			

Table D.9: Summary of the experiments used to determine the sufficient number of observations per solution for the BAP10.

the processing times do not become progressively faster because of the non-linear topology.

Table D.10: Processing times for the machines in BAP16

	M1	M2	M3	M4	M5	M6	M7	M8	M9	M10	M11	M12	M13	M14	M15	M16
Processing time	6:00	6:00	6:00	12:00	12:00	12:00	12:00	12:00	12:00	12:00	12:00	12:00	12:00	6:00	6:00	3:00



Figure D.7: The non-linear BAP with sixteen machines M_1, \ldots, M_{16} with finite buffers B_1, \ldots, B_{16-1} in a queuing network

Figure D.8 shows the simulation model as it is built in Tecnomatix and represents BAP16.

Observations per solution

Table D.11 shows the experiments used to determine whether or not 100 observations per solution are sufficient for BAP16 and is summarised in Table D.12.





Figure D.8: A screenshot of the BAP16 model in Tecnomatix.

Table D.11: Experiments used to determine a sufficient number of observations per solution for BAP16

												Work	-in-pro	ogress	1	hroughp	out
	B1	B2	B3	B4	B5		B11	B12	B13	B14	B15	А	В	\mathbf{C}	А	В	\mathbf{C}
Exp 1	1	1	1	1	1		1	1	1	1	1	0.24	0.24	0.25	452.4	449.09	453.98
Exp 2	2	2	2	2	2		2	2	2	2	2	0.43	0.42	0.43	473.4	487.57	481.43
Exp 3	3	3	3	3	3		3	3	3	3	3	0.66	0.60	0.59	524.6	502.92	503.47
Exp 4	4	4	4	4	4		4	4	4	4	4	0.75	0.69	0.72	504.5	510.35	516.28
Exp 5	5	5	5	5	5		5	5	5	5	5	0.81	0.81	0.82	525.4	521.95	526.17
Exp 6	6	6	6	6	6	•••	6	6	6	6	6	0.90	0.90	0.92	534.8	533.93	537.18
Exp 7	7	7	7	7	7	•••	7	7	7	7	7	1.02	0.96	1.00	548.2	536.7	546.74
Exp 8	8	8	8	8	8	•••	8	8	8	8	8	0.94	1.05	1.08	552.1	551.52	553.38
Exp 9	9	9	9	9	9	•••	9	9	9	9	9	1.18	1.14	1.12	560.8	554.92	557.29
Exp 10	10	10	10	10	10	•••	10	10	10	10	10	1.32	1.18	1.19	575.9	559.06	561.72

Table D.12: Summary of the experiments used to determine the sufficient number of observations per solution for the BAP16.

		Work-in-progress			Throughput				
Groups	Count	Sum	Average	Variance	Sum	Average	Variance		
Exp 1	3	0.73	0.24	0	1355.47	451.82	6.22		
Exp 2	3	1.28	0.43	0	1442.40	480.80	50.50		
Exp 3	3	1.85	0.62	0	1530.99	510.33	152.79		
Exp 4	3	2.16	0.72	0	1531.13	510.38	34.69		
Exp 5	3	2.44	0.81	0	1573.52	524.51	5.05		
Exp 6	3	2.71	0.90	0	1605.91	535.30	2.83		
Exp 7	3	2.99	1.00	0	1631.64	543.88	39.21		
Exp 8	3	3.07	1.02	0.01	1657.00	552.33	0.90		
Exp 9	3	3.45	1.15	0	1673.01	557.67	8.75		
$Exp \ 10$	3	3.69	1.23	0.01	1696.68	565.56	81.94		

284 Appendix D. Discrete-event Simulation Optimisation Problem Definitions

D.3 The Open mine problem

[The content of this section is an extract from unpublished notes, from the University of Stellenbosch (US), and are included here to define the problem.]

A rail operator has to transport crushed ore to a port for export. There are a number of trains in service which cycles between the open mine and the port. Each train consists of a set of locomotives and trucks. The trains arrive at a buffer yard where one of two loaders in the mine site is assigned to each train for loading.

There are maintenance requirements that need to be met and as a result the loaders must be cycled, *i.e.* when Loader 1 has just been assigned a train, the next train is assigned to Loader 2, regardless of whether Loader 2 is occupied or not. The cycle is thus Loader 1–Loader 2–Loader 1–Loader 2. There are two different physical waiting areas that the trains can be assigned to, one in front of each loader.

The mine operation uses a shovel (see Figure $D.9a^1$) that loads a number of mine trucks (see Figure $D.9b^2$). The shovel loads TRI(140, 160, 200)*1000 kg per truck per loading time as given in Table D.13. The mine trucks then transport the ore to a crusher, which in turn forms a stockpile. The crushing processing time of a load is given in Table D.13.

 Table D.13: Loading and processing times for OMP

	Distribution (minutes)
Shovel loading time	TRI(18,12,28)
Crusher processing time	TRI(9,6,15)

The distance between the shovel and crusher is 2 500m and the trucks travel at 15km/hr (loaded and unloaded). The delivery of a truck load of ore has the following effect on the stockpile (after crushing):

$$StackSize = StackSize + Truck.LoadMass.$$
(D.7)

The loaders retrieve the crushed product from the stockpile to fill the train trucks. The loaded mass is distributed UNI(65,85) metric tons and it takes EXP(3) minutes to load a train truck. A loader can only start to retrieve ore from the stockpile if the amount of ore is available to fill the alternate loader's train (*i.e.* the number of train trucks to be filled multiplied by 80 tons, plus 100 tons safety margin). The 80 ton value is just a guide the mine agreed upon.

If sufficient ore is not available in the stockpile, the loader must wait for the mine trucks to replenish the stockpile. If the stockpile gets depleted while a loader is loading, the loader will cease operation until some ore is available. The other loader must wait until the current loader has completely filled the train it was busy with. The two loaders thus do not compete for crushed ore when the stockpile gets low – one train is filled and dispatched rather than having two trains waiting.

So, suppose Loader 1 has loaded 36 of its train trucks and Loader 2 has loaded seven of its train trucks, when the stockpile becomes depleted. Now the loaders have to wait. When there are at least 200 metric tons of crushed ore available again, the loading process at Loader 1 resumes

¹source: http://www.thompsoncreekmetals.com/i/photos/milligan/milligan9.jpg

²source: http://i.telegraph.co.uk/multimedia/archive/02256/Metal_to_Medal_0_2256835b.jpg



(a) Mine shovel



(b) Mine truck

Figure D.9: Examples of a (a) mine shovel and (b) mine truck in an open mine

and Loader 2 does not load at all. Once Loader 1 has completed loading its 50 train trucks, Loader 2 may resume loading, provided the stockpile has sufficient (more than 200 metric tons) crushed ore. After a train is loaded by one of the loaders, the train is inspected. Inspection time is distributed EXP(10) minutes, after which it starts its 24 hour journey to the port.

Figure D.10 shows the simulation model as it is built in Tecnomatix and represents OMP.

Observations per solution

Table D.14 shows the experiments used to determine whether or not 100 observations per solution are sufficient for the OMP and is summarised in Table D.15.

286 Appendix D. Discrete-event Simulation Optimisation Problem Definitions



Figure D.10: A screenshot of the OMP model in Tecnomatix.

Table D.14: Experiments used to determine a sufficient number of observations per solution for theOMP

					Total cost	Se	erved tra	lins	
	Shovels	Crushers	Trucks	А	В	С	А	В	С
Exp 1	4	4	4	1773.93	1773.82	1 773.81	11.8	11.61	11.55
Exp 2	5	5	5	2217.44	2217.41	2217.37	13.8	13.85	13.84
Exp 3	6	6	6	2660.96	2660.88	2660.89	14	14.16	14.18
Exp 4	7	7	7	3104.66	3104.45	3104.41	14	14.16	14.18
Exp 5	8	8	8	3548.03	3547.93	3547.92	14	14.16	14.18
Exp 6	9	9	9	3991.66	3991.43	3991.38	14	14.16	14.18
Exp 7	10	10	10	4434.92	4434.78	4434.83	14	14.16	14.18
Exp 8	11	11	11	4878.51	4878.30	4878.31	14	14.16	14.18
Exp 9	12	12	12	5321.86	5321.78	5321.77	14	14.16	14.18
Exp 10	13	13	13	5765.36	5765.14	5765.20	14	14.16	14.18

Table D.15: Summary of the experiments used to determine the sufficient number of observations per solution for the OMP.

		Total cost					Served tra	ins
Groups	Count	Sum	Average	Variance		Sum	Average	Variance
Exp 1	3	5321.57	1773.86	0		34.96	11.65	0.02
Exp 2	3	6652.23	2217.41	0		41.49	13.83	0
Exp 3	3	7982.74	2660.91	0		42.34	14.11	0.01
Exp 4	3	9313.51	3104.50	0.02		42.34	14.11	0.01
Exp 5	3	10643.88	3547.96	0		42.34	14.11	0.01
Exp 6	3	11974.47	3991.49	0.02		42.34	14.11	0.01
Exp 7	3	13304.53	4434.84	0.01		42.34	14.11	0.01
Exp 8	3	14635.12	4878.37	0.01		42.34	14.11	0.01
Exp 9	3	15965.41	5321.80	0		42.34	14.11	0.01
Exp 10	3	17295.70	5765.23	0.01		42.34	14.11	0.01

APPENDIX E

Algorithmic Comparison Results

This appendix contains additional results obtained during the algorithmic comparisons for the hyperheuristics, as described in Chapter 7 and were omitted from the respective sections so as to enhance the understanding of the main text.

In Figures E.1 and E.2, the best and worst approximation fronts are plotted for the BOSAH and the BOCEGAH for hyperarea and number of non-dominated solutions, for the respective simulation problems as summarised in Tables E.2 and E.1.

Table E.1: The simulation runs that correspond to the best and worst approximation fronts (obtained by the BOCEAH) for the hyperarea performance indicator and the number of non-dominated solutions found for each simulation problem.

	Wo	orst	Be	est
	HA	NDS	HA	NDS
OMP	Run 1	Run 1	Run 30	Run 5
IP	Run 94	Run 49	Run 19	Run 14
BAP5	Run 60	Run 34	Run 16	Run 89
BAP10	$\operatorname{Run}57$	$\operatorname{Run}27$	Run 35	Run 48
BAP16	$\operatorname{Run}75$	$\operatorname{Run}75$	$\mathrm{Run}~97$	Run 89

Table E.2: The simulation runs that correspond to the best and worst approximation fronts (obtained by the BOSAH) for the hyperarea performance indicator and the number of non-dominated solutions found for each simulation problem.

	Wo	orst	В	est
	HA	NDS	HA	NDS
OMP	Run 39	Run 39	Run 3	Run 3
IP	$\operatorname{Run}97$	$\operatorname{Run}42$	Run 3	Run 16
BAP5	$\operatorname{Run}57$	Run 83	Run 35	$\operatorname{Run}25$
BAP10	$\operatorname{Run}57$	Run 79	$\operatorname{Run}76$	Run 1
BAP16	$\mathrm{Run}\ 27$	Run 19	Run 3	$\operatorname{Run}24$



Figure E.1: The best and worst best approximation fronts obtained for the BOSAH and the BOCEGAH in terms of hyperarea.


Figure E.2: The best and worst best approximation fronts obtained for the BOSAH and the BOCEGAH in terms of the number of non-dominated solutions.